



## Magnetohydrodynamic Equations Around Couette Flow

Zur Erlangung des akademischen Grades eines

## DOKTORS DER NATURWISSENSCHAFTEN

von der KIT-Fakultät für Mathematik des Karlsruher Instituts für Technologie (KIT) genehmigte

## DISSERTATION

von

## NIKLAS KNOBEL

Tag der mündlichen Prüfung: 24.07.20241. Referent: PD Dr. Christian Zillinger2. Referent: JProf. Dr. Xian Liao

## Abstract

This thesis studies the stability of the magnetohydrodynamic equations around an affine shear flow and constant magnetic field. The dynamics of the equations change drastically depending on the fluid viscosity  $\nu \geq 0$  and magnetic resistivity  $\kappa \geq 0$ . The main goal of this thesis is to establish results on (nonlinear) stability and instability for selected dissipation regimes. These stability and instability results are established in Chapters 3-5.

In the first part, we consider the inviscid  $\nu = 0$  and resistive  $\kappa > 0$  case. We linearize around explicit low-frequency solutions of traveling waves to infer the main growth model. Small data in Gevrey 2 spaces are necessary and sufficient for this main growth model to ensure stability.

In the second part, we consider the MHD equations with viscosity and horizontal resistivity  $\nu = \kappa_x > 0$  and  $\kappa_y = 0$ . We show that small initial data in Sobolev spaces ensure stability. Furthermore, we show that for the viscous  $\nu > 0$  and non-resistive  $\kappa = 0$  case, for the linearized MHD equations, there exists initial data such that the magnetic field grows unbounded as  $t \to \infty$ . Thus, global in time stability of the magnetic field in Sobolev spaces cannot hold for the nonlinear system.

In the third part, we consider the case when resistivity is smaller than viscosity  $0 < \kappa \leq \nu$ . We show that small initial data in Sobolev spaces yield stability. Furthermore, the stability properties qualitatively differ for the cases  $\kappa \geq \nu^3$ and  $\kappa \leq \nu^3$ . For the case  $\kappa \leq \nu^3$  we obtain norm inflation of order  $\nu \kappa^{-\frac{1}{3}}$  due to growth in the magnetic field.

## Acknowledgements

First and foremost, I would like to thank my advisor, Christian Zillinger, for supervising me during my PhD. For all the helpful discussions that built my understanding of fluid dynamics, for his guidance during my PhD and for his patience with my writing. I also would like to thank Xian Liao for being a valuable member of our working group and for agreeing to act as a referee for my thesis. Next, I want to thank Diego Alonso-Orán for advising my Master's thesis and for his guidance.

Furthermore, I would like to thank the CRC 1173 for the opportunity to write my thesis in the Project AP6. I also want to thank my working group of nonlinear differential equations for enriching mathematical and non-mathematical conversations. In particular, Wolfgang Reichel, Gerd Herzog, Björn, Sebastian, Jannis and Julia.

Furthermore, I would like to thank my colleagues and friends from the second floor. The coffee break at 4 p.m. was always a good opportunity for a break. I especially want to thank Marvin, Gianmichele, Lucrezia, Alex, Friedrich, and Max for being awesome.

Finally, many thanks to my Family, my brother Moritz and my parents, Achim and Petra, for always supporting me.

# Contents

1	Mixing and Stability of the Magnetohydrodynamic Equations					
2	Ma 2.1 2.2 2.3 2.4	thematical Effects of the MHD Equations Linear Effects of the Navier-Stokes Equations with $b = 0$ and $\alpha = 0$ Linear Effects of the Ideal MHD Equations with $\nu = \kappa = 0$ Linear Effects of the Dissipative MHD Equations with $\nu > 0$ or $\kappa > 0$	7 8 9 12 17			
	2.5	Nonlinear Behavior	18			
3	On Echoes in Magnetohydrodynamics with Magnetic Dissipa-					
	tion 3.1 3.2 3.3 3.4 3.5 3.6	Introduction and Main Results	<ol> <li>22</li> <li>27</li> <li>39</li> <li>45</li> <li>76</li> <li>79</li> </ol>			
<b>4</b>	On the Sobolev Stability Threshold for the 2D MHD Equations					
	wit	h Horizontal Magnetic Dissipation	81			
	4.1		81			
	4.2 4.3	Bootstran Hypotheses and Outline of Proof	00 91			
	4.4	Instability of the Non-Resistive MHD System	.06			
<b>5</b>	Sobolev Stability for the 2D MHD Equations in the Non-Resistive					
	Lim	it 1	13			
	5.1	Introduction	13			
	5.2	Linear Stability and Norm Inflation	.20			
	5.3	Sobolev Stability for the Nonlinear System	.24			
	$5.4 \\ 5.5$	Local Wellposedness	43			

Α	Mathematical Background and Notation	146
в	Derivation of the MHD Equations from the Navier-Stokes and	d
	Maxwell Equations	149
Re	eferences	152

## Chapter 1

# Mixing and Stability of the Magnetohydrodynamic Equations

The magnetohydrodynamic (MHD) equations are a common model used in astrophysics, planetary magnetism and controlled nuclear fusion [Dav16]. They model the interaction of the velocity and magnetic field of a conducting nonmagnetic fluid. The MHD equations are derived from the Navier-Stokes equations coupled with the Maxwell equations assuming vanishing charge density (see Appendix B for a detailed derivation).

In this thesis, we consider the two-dimensional MHD equations in a finite periodic channel,

$$\partial_t V + V \cdot \nabla V + \nabla \Pi = \nu \Delta V + B \cdot \nabla B,$$
  

$$\partial_t B + V \cdot \nabla B = \kappa \Delta B + B \cdot \nabla V,$$
  

$$\nabla \cdot V = \nabla \cdot B = 0,$$
  

$$B\Big|_{t=0} = B_{in}, \quad V\Big|_{t=0} = V_{in},$$
  

$$(t, x, y) \in \mathbb{R}^+ \times \mathbb{T} \times \mathbb{R} =: \Omega.$$
(1.1)

Here  $V: \Omega \to \mathbb{R}^2$  is the velocity of an incompressible conducting fluid interacting with the magnetic field  $B: \Omega \to \mathbb{R}^2$ . The quantity  $\Pi: \Omega \to \mathbb{R}$  denotes the pressure. The dissipation parameters  $\nu, \kappa \geq 0$  represent fluid viscosity and magnetic resistivity.

The MHD equations (1.1) are notoriously challenging. Already in the 2D case, global-in-time wellposedness is open for general initial data. Stability results for vanishing resistivity in the literature are often either local-in-time or limited to small perturbations around special solutions (mostly around a large constant magnetic field). Therefore, weak or vanishing resistivity poses great challenges. For the resistive  $\kappa > 0$  and viscous  $\nu > 0$  case we obtain global-in-time wellposedness [ST83]. Furthermore, a strong magnetic field is stable for

small initial data [WZ17]. Also for the inviscid  $\nu = 0$  and resistive  $\kappa > 0$  MHD equations, one obtains global-in-time wellposedness [Koz89] for initial data in Sobolev spaces. For the viscous  $\nu > 0$  and non-resistive  $\kappa = 0$  MHD equations, one obtains local-in-time wellposedness [CMRR16, FMRR17], but global-in-time wellposedness is remains open. Furthermore, one obtains stability of a large enough constant magnetic field for small initial data [LXZ15, RWXZ14]. For the ideal MHD equations  $\nu = \kappa = 0$ , only local-in-time wellposedness is established [Koz89, Sch88, Sec93, MY06]. In [BSS88], it is shown that the dynamics of small initial perturbations of the ideal MHD equations around a large constant magnetic field behave like the linearized system.

An essential challenge for the MHD equations is the solutions' stability and long-time behavior. Specifically, we aim to understand the stability and instability of specific global in-time solutions that combine the effects of mixing and magnetic (de)stabilization. This dissertation focuses on the combination of an affine shear flow, called Couette flow, and a constant magnetic field

$$V_s(t, x, y) = \begin{pmatrix} y \\ 0 \end{pmatrix},$$
  

$$B_s(t, x, y) = \begin{pmatrix} \alpha \\ 0 \end{pmatrix}, \qquad \alpha \in \mathbb{R}.$$
(1.2)

This stationary solution combines the effects of mixing due to shear and coupling by the magnetic field. The Couette flow mixes perturbations, which leads to damping. The constant magnetic field induces a strong coupling of the magnetic and velocity perturbations. Therefore, perturbations of (1.2) highlight the velocity and magnetic field interaction.

The main goal of this thesis is to contribute to the understanding of stability and instability of perturbations of (1.2). Here, stability is understood in the Lipschitz sense, which we state in the following. We consider the perturbative unknowns

$$v(t, x, y) = V(t, x + yt, y) - V_s(t, x + yt, y),$$
  

$$b(t, x, y) = B(t, x + yt, y) - B_s(t, x + yt, y).$$
(1.3)

The change of variables  $x \mapsto x + yt$  follows the characteristics of the Couette flow. Then the equations (1.1) become

$$\begin{aligned} \partial_t v + v_2 e_1 &= \nu \Delta_t v + \alpha \partial_x b + b \cdot \nabla_t b - v \cdot \nabla_t v - \nabla_t \pi, \\ \partial_t b - b_2 e_1 &= \kappa \Delta_t b + \alpha \partial_x v + b \cdot \nabla_t v - v \cdot \nabla_t b, \\ \nabla_t \cdot v &= \nabla_t \cdot b = 0, \\ b\big|_{t=0} &= b_{in}, \quad v\big|_{t=0} = v_{in}, \\ (t, x, y) \in \mathbb{R}_+ \times \mathbb{T} \times \mathbb{R}. \end{aligned}$$
(1.4)

Here we write  $v = \begin{pmatrix} v_1 \\ v_2 \end{pmatrix}$ ,  $b = \begin{pmatrix} b_1 \\ b_2 \end{pmatrix}$  and  $e_1 = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$ . The spatial derivatives become time-dependent due to the change of variables and change to  $\partial_y^t :=$ 

 $\partial_y - t\partial_x$ ,  $\nabla_t := (\partial_x, \partial_y^t)^T$  and  $\Delta_t := \partial_x^2 + (\partial_y^t)^2$ . For different values of  $\nu$  and  $\kappa$ , we obtain different linear behavior and as a result, the nonlinear terms yield different interactions.

**Problem 1.1.** Let f be a choice of unknowns which describes a solution of (1.4). Let  $X_1$  and  $X_2$  be two Banach spaces, then we aim to find a bound on the initial data  $\varepsilon_0 = \varepsilon_0(f, X_i, \nu, \kappa)$  and a Lipschitz constant  $L = L(f, X_i, \nu, \kappa)$  such that for initial data which satisfies

$$||f_{in}||_{X_1} = \varepsilon \le \varepsilon_0,$$

the corresponding solution is globally bounded in time by

$$\|f(t)\|_{X_2} \le L\varepsilon$$

Furthermore, the stability threshold are  $\gamma_1, \gamma_2 \in \mathbb{R}$ , such that for  $\varepsilon_0 = c_0 \nu^{\gamma_1} \kappa^{\gamma_2}$ with a  $c_0 > 0$  we obtain

$$\begin{split} \|f_{in}\|_{X_1} &\leq c_0 \nu^{\gamma_1} \kappa^{\gamma_2} \to \ stability, \\ \|f_{in}\|_{X_1} &\gg c_0 \nu^{\gamma_1} \kappa^{\gamma_2} \to \ possible \ instability. \end{split}$$

Stability thresholds are a typical approach and are also used for related equations such as the Navier-Stokes, Euler or Boussinesq equations. Our definition generalizes previous definitions [BVW18, BGM19, Lis20] to allow for two parameters  $\gamma_1$  and  $\gamma_2$ .

In the following, we give an overview of the literature. First, we give an overview of the closely connected cases of the Navier-Stokes ( $\nu > 0$ ) equations and the Euler equations ( $\nu = 0$ ) around Couette flow. This corresponds to the MHD equations around Couette flow without a magnetic field ( $b = 0 \in \mathbb{R}^2$  and  $\alpha = 0 \in \mathbb{R}$ ) and has similar properties but behaves qualitatively differently. We refer to [BGM19] for a more comprehensive overview.

For the Euler equations around Couette flow, Gevrey 2 spaces (see Appendix A) are expected to be optimal for stability. In [BM15a], stability is shown for small initial data in Gevrey  $\sigma$  spaces for  $1 \leq \sigma < 2$ . When linearizing around small explicit low-frequency solutions, one obtains stability and norm inflation in Gevrey 2 spaces, depending on the radius of convergence [DZ21]. For the nonlinear system, one obtains nonlinear instability in a space slightly weaker than Gevrey 2 [DM23].

In the case of the Navier-Stokes equations around Couette flow, small initial data in Sobolev spaces with respect to  $\nu$  are sufficient to obtain stability. In [BVW18] it is shown that initial data in Sobolev spaces smaller than  $\nu^{\frac{1}{2}}$  yield stability. Masmoudi and Zhao improved this to  $\nu^{\frac{1}{3}}$  [MZ22], which is expected to be optimal. In [BMV16], it is shown that small initial data in Gevrey  $\sigma$  spaces for  $1 \leq \sigma < 2$  are sufficient to ensure stability independent of  $\nu$ . Furthermore, one can trade smallness in terms of  $\nu$  and Gevrey regularity to obtain stability [LMZ22].

The MHD equations around Couette flow and constant magnetic field recently received substantial attention [Lis20, KZ1, ZZ24, Dol24, KZ2, K, CZ24]. This includes the three publications and preprints which are the basis for this thesis:

- Niklas Knobel and Christian Zillinger, On Echoes in Magnetohydrodynamics with Magnetic Dissipation, Journal of Differential Equations, 367:625–688, 2023, [KZ1]. Main result: Theorem 3.3.
- 2. Niklas Knobel and Christian Zillinger, On the Sobolev Stability Threshold for the 2D MHD Equations with Horizontal Magnetic Dissipation arXiv preprint, arXiv:2309.00496, 2023, [KZ2]. Main results: Proposition 4.2 and Theorem 4.3
- Niklas Knobel, Sobolev Stability for the 2D MHD Equations in the Non-Resistive Limit arXiv preprint, arXiv:2401.12548, 2024, [K]. Main result: Theorem 5.1

For the MHD equations around Couette flow and constant magnetic field, the linear dynamics change depending on the values of the dissipation parameters  $\nu$  and  $\kappa$ . As a consequence, the nonlinear terms exhibit different behaviors. For an intuition on linear and nonlinear effects on the MHD equations, we refer to Chapter 2. Here we give an overview of the various results.

In 2020, Liss proved the first stability result for the MHD equations around Couette flow and constant magnetic field [Lis20]. He considered the fully dissipative regime,  $\kappa = \nu > 0$ , and established stability of the three-dimensional MHD equations for initial data small in Sobolev spaces for a large enough constant magnetic field. For the analogous two-dimensional problem, Chen and Zi considered in [CZ24] the stability of the 2D MHD equations around a shear flow close to Couette for small initial data in Sobolev spaces. Dolce [Dol24] proved stability for small initial data in Sobolev spaces for the more general setting of  $0 < \kappa^3 \leq \nu \leq \kappa$ .

In [KZ1], which is part of this thesis in Chapter 3, Zillinger and the author consider the regime of vanishing viscosity  $\nu = 0$  and resistivity  $\kappa > 0$ . In the linearized dynamics, the equations decouple with respect to frequency and are stable in Sobolev spaces. However, the nonlinear terms yield an interaction between different frequencies, possibly leading to norm inflation. In order to capture this effect, we consider traveling waves, which are explicit low-frequency solutions, from which we deduce the main growth model. There we showed that resonance chains can lead to Gevrey 2 norm inflation. On the one hand, Gevrey 2 spaces with a large enough radius of convergence ensure stability. On the other hand, there are specific initial data in Gevrey 2 spaces with a small radius of convergence such that the solutions grow unbounded in Sobolev spaces. Thus, Gevrey 2 spaces are sufficient and necessary for stability. In a corresponding nonlinear stability result, Zhao and Zi [ZZ24] proved the almost matching nonlinear stability result for small perturbations in Gevrey  $\sigma$  spaces for  $1 \leq \sigma < 2$ .

In [KZ2], which is a part of this thesis in Chapter 4, Zillinger and the author consider the case of horizontal resistivity and full viscosity,  $\nu = \kappa_x > 0$  and  $\kappa_y =$ 

0. We establish stability for small initial data in Sobolev spaces. Furthermore, we prove that for the non-resistive MHD equations,  $\nu > 0$  and  $\kappa = 0$ , the linearised dynamics yield growth of the Sobolev norm by  $\nu t$ . Therefore, stability in Sobolev spaces cannot hold for the magnetic field. In this sense, horizontal resistivity is indeed necessary for stability.

In [K], which is a part of this thesis in Chapter 5, the author considers the case when resistivity is smaller than viscosity  $0 < \kappa \leq \nu$ . The author proves stability for small enough initial data in Sobolev spaces. The behavior qualitatively differs between the cases  $\nu^3 \leq \kappa$  and  $\nu^3 \geq \kappa$ . The relation  $\nu^3 \leq \kappa$ corresponds to the case when resistivity and viscosity are close to each other and thus yield stability and no norm inflation. In the case when  $\nu^3 \geq \kappa$ , the resistivity is very small compared to the viscosity. Thus the velocity field gets damped down on time scales  $\nu^{-1}$  leading to growth in the magnetic field until times scales of order  $\kappa^{-\frac{1}{3}}$ . Therefore, this case yields stability with norm inflation of order  $\nu \kappa^{-\frac{1}{3}}$ .

To summarize the previously mentioned results for the MHD equations around Couette flow, the stability of the cases  $\nu \approx \kappa > 0$  [Lis20, Dol24, CZ24, K] and  $\nu = 0 < \kappa$  [KZ1, ZZ24] are by now well understood, with key contribution being part of this dissertation. The case  $\kappa = 0 < \nu$  is still open due to the growth of the magnetic field and we cannot expect global-in-time stability for the magnetic field [KZ2]. Furthermore, [KZ2, K] contribute to the understanding of the non-resistive limit or the case of partially horizontal, anisotropic resistivity. For the ideal case, no results exist yet. We remark that the non-resistive and ideal MHD equations, which remain open in our setting, are exactly the two regimes where global-in-time wellposedness is still open for the MHD equations in general, as mentioned at the beginning. The main results of this thesis are summarized in the following table:

Paper	Ch	Result	Dissipation	Description
[KZ1]	3	Thm 3.3	$\nu = 0,  \kappa > 0$	(In)stability of traveling waves
[KZ2]	4	Thm 4.3	$\nu = \kappa_x > 0, \ \kappa_y = 0$	Nonlinear stability
[KZ2]	4	Prop 4.2	$\nu > 0, \ \kappa = 0$	Linear growth of magnetic field
[K]	5	Thm 5.1	$\nu \ge \kappa \gtrsim \nu^{\frac{1}{3}} > 0$	Nonlinear stability
[K]	5	Thm 5.1	$\nu^{\frac{1}{3}} \gtrsim \kappa > 0$	Nonlinear stability, norm inflation

The remainder of this dissertation is structured as follows:

- In Chapter 2, we give a short overview of the main mathematical effects of the linear and nonlinear dynamics.
- In Chapter 3, we consider the inviscid  $\nu = 0$  and resistive case  $\kappa > 0$ . While the linearised dynamics is stable in Sobolev spaces, we cannot expect stability for the nonlinear dynamics due to resonances in the nonlinear terms. We model these resonances by deriving a model from explicit low-frequency solutions. For this model, we show that Gevrey 2 spaces are necessary for stability. This chapter is based on a joint publication

with Christian Zillinger [KZ1], published in the Journal of Differential Equations.

- In Chapter 4, we consider the case with viscosity and horizontal resistivity  $\nu = \kappa_x > 0$  and vanishing vertical resistivity  $\kappa_y = 0$ . Smallness of initial data in Sobolev norms is sufficient for nonlinear stability. Furthermore, we show that for the viscous  $\nu > 0$  and non-resistive  $\kappa = 0$  case, for specific initial data, the magnetic field grows linearly in time. This is based on a joint preprint with Christian Zillinger [KZ2].
- In Chapter 5, we consider the case when resistivity is smaller than viscosity  $0 < \kappa \leq \nu$ . We obtain nonlinear stability for initial data small in Sobolev spaces. There are two different behaviors. If resistivity is close to viscosity, we obtain stability without norm inflation. If the resistivity is much smaller than viscosity, we obtain norm inflation depending on the resistivity and viscosity. In particular, stability estimates degenerate in the non-resistive limit. This is based on the author's preprint [K].
- In Appendix A, we give an overview of the necessary mathematical background.
- In Appendix B, we derive the magnetohydrodynamic equations from the Navier-Stokes and Maxwell equations.

## Chapter 2

# Mathematical Effects of the MHD Equations

In this chapter, we give a short overview of the leading mathematical effects of the MHD equations around Couette flow and constant magnetic field (1.4), for the unknowns (1.3),

$$\begin{aligned} \partial_t v + v_2 e_1 &= \nu \Delta_t v + \alpha \partial_x b + b \cdot \nabla_t b - v \cdot \nabla_t v - \nabla_t \pi, \\ \partial_t b - b_2 e_1 &= \kappa \Delta_t b + \alpha \partial_x v + b \cdot \nabla_t v - v \cdot \nabla_t b, \\ \nabla_t \cdot v &= \nabla_t \cdot b = 0, \\ b\Big|_{t=0} &= b_{in}, \quad v\Big|_{t=0} = v_{in}, \\ (t, x, y) \in \mathbb{R}_+ \times \mathbb{T} \times \mathbb{R}. \end{aligned}$$

$$(2.1)$$

Choosing the right unknowns is crucial for equation (2.1). Depending on the parameters  $\nu, \kappa \geq 0$  and  $\alpha \in \mathbb{R}$ , switching to different unknowns adapted to the linearized dynamics is natural. In particular, in this chapter, we will often use the adapted unknowns  $p = \Lambda_t^{-1}(\nabla_t^{\perp} \cdot u, \nabla_t^{\perp} \cdot b)$ , the vorticity  $w = \nabla_t^{\perp} \cdot v$  and the current  $j = \nabla_t^{\perp} \cdot b$ . The fractional Laplacian is defined as  $\Lambda_t := (-\Delta_t)^{\frac{1}{2}}$ . Furthermore, we perform a Fourier transform such that for the variables  $(x, y) \in \mathbb{T} \times \mathbb{R}$  we obtain  $(k, \xi) \in \mathbb{Z} \times \mathbb{R}$  in Fourier space. With slight abuse of notation, we omit writing the hat of the Fourier transform.

We briefly outline the structure of this chapter. In the first three sections, we describe the linearized effects of the MHD equations. Section 2.1 considers the case when the magnetic field vanishes and corresponds to the Navier Stokes equation. Section 2.2 discusses effects in the ideal MHD (i.e.  $\nu = \kappa = 0$ ) equations. In Section 2.3, we consider the dissipative MHD equations and give an overview of the different dissipation regimes. Section 2.4, provides an overview of all the useful unknowns. In Section 2.5 we describe the leading nonlinear effects in terms of toy models.

## 2.1 Linear Effects of the Navier-Stokes Equations with b = 0 and $\alpha = 0$

This section describes the linear effects of the MHD equations without a magnetic field. In particular, these are the linearised Navier-Stokes equations around Couette flow

$$\partial_t v + v_2 e_1 = \nu \Delta_t v - v \cdot \nabla_t v - \nabla_t \pi,$$
  
$$\nabla_t \cdot v = 0.$$

The effects described in this section are well-known (see for example [Orr07, BM15a, BVW18, Zil17]). In vorticity  $w = \nabla_t^{\perp} \cdot v$  formulation, the linearized Navier-Stokes equations read

$$\partial_t w = \nu \Delta_t w,$$
  

$$v = \nabla_t^{\perp} \Delta_t^{-1} w.$$
(2.2)

Assuming vanishing x-average, we obtain two important stabilizing effects in linearized dynamics: inviscid damping and enhanced dissipation. After a Fourier transformation (as announced we omit the hat in the Fourier transform), we obtain

$$\partial_t w = -\nu (k^2 + (\xi - kt)^2)w,$$
  

$$v = -\frac{i}{k^2 + (\xi - kt)^2} \begin{pmatrix} \xi - kt \\ -k \end{pmatrix} w.$$
(2.3)

The equation decouples in frequency space. Due to the assumption of vanishing x-average we obtain  $k \neq 0$ , since  $k \in \mathbb{Z}$  that means  $|k| \geq 1$ .

### Inviscid Damping with $\nu = 0$

The Couette flow induces a mixing of vorticity. This leads to a damping of the velocity field independent of  $\nu$ , often called inviscid damping. For  $\nu = 0$  the solution of (2.2) stays constant

$$w(t,k,\xi) = w_{in}(k,\xi)$$

and thus, for the velocity field, we infer

$$v(t,k,\xi) = \frac{i}{k} \frac{1}{1 + (t - \frac{\xi}{k})^2} \binom{t - \frac{\xi}{k}}{1} w_{in}(k,\xi).$$
(2.4)

For fixed frequencies, the absolute value of the velocity reaches its maximum at time  $t = \frac{\xi}{k}$  and is damped as t increases further. For norm estimates, lower Sobolev norms will decay in time [Orr07]

$$\langle t \rangle \| v_1 \|_{H^{N+1}} + \langle t \rangle^2 \| v_2 \|_{H^N} \lesssim \| w_{in} \|_{H^{N+1}} = \| v_{in} \|_{H^{N+2}}.$$
(2.5)

### Enhanced Dissipation with $0 < \nu \ll 1$

Mixing by the Couette flow leads to faster dissipation rates, often called enhanced dissipation. In particular, at times  $t \approx \nu^{-\frac{1}{3}}$  the dissipation leads to fast decay and compares to  $t \approx \nu^{-1}$  for the heat equation. For  $\nu > 0$ 

$$w(t,k,\xi) = \exp\left(-\nu k^2 \int_0^t (1+(t-\frac{\xi}{k})^2)d\tau\right) w_{in}(k,\xi)$$

solves equation (2.3). We estimate this exponential term by

$$\exp\left(-\nu k^2 \int_0^t (1+(t-\frac{\xi}{k})^2) d\tau\right) \lesssim \exp(-c\nu^{\frac{1}{3}}t)$$

and thus, for  $\nu \ll 1$ , we obtain the enhanced dissipation estimate

$$||w||_{H^{N+1}} \le e^{-c\nu^{\frac{1}{3}}t} ||w_{in}||_{H^{N+1}}.$$

For the velocity field, we obtain both enhanced dissipation and inviscid damping

$$\langle t \rangle \| v_1 \|_{H^{N+1}} + \langle t \rangle^2 \| v_2 \|_{H^N} \lesssim e^{-c\nu^{\frac{1}{3}}t} \| w_{in} \|_{H^{N+2}}.$$

## 2.2 Linear Effects of the Ideal MHD Equations with $\nu = \kappa = 0$

This section describes the linear effects of the ideal MHD equations,  $\nu = \kappa = 0$ . Here, there is competition between the growth of the magnetic field, the decay of the velocity field, and their interaction. We will describe these effects and then show stability for a large constant magnetic field  $\alpha > \frac{1}{2}$ . To show these effects, we consider the adapted unknowns

$$p = \Lambda_t^{-1} (\nabla_t^{\perp} \cdot v, \nabla_t^{\perp} \cdot b).$$

These unknowns are established in [KZ2] (Chapter 4) and similar unknowns are established independently in [Dol24]. Then (2.1) changes to

$$\partial_t p_1 = \partial_x \partial_y^t \Delta_t^{-1} p_1 + \alpha \partial_x p_2,$$
  

$$\partial_t p_2 = -\partial_x \partial_y^t \Delta_t^{-1} p_2 + \alpha \partial_x p_1,$$
  

$$p\big|_{t=0} = p_{in}.$$
(2.6)

We apply a Fourier transformation, only consider modes  $k \neq 0$  and, by slight abuse of notation, we replace  $p_2$  by  $-ip_2$  to obtain the equation

$$\partial_t p_1 = -\frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} p_1 - \alpha k p_2,$$
  

$$\partial_t p_2 = \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} p_2 + \alpha k p_1,$$
  

$$p\Big|_{t=0} = p_{in}.$$
(2.7)

For simplicity of notation, we introduce the new variable  $s = t - \frac{\xi}{k}$  and the initial time  $s_{in} = -\frac{\xi}{k}$ . Then equation (2.7) reads

$$\partial_s p_1 = -\frac{s}{1+s^2} p_1 - \alpha k p_2,$$
  

$$\partial_s p_2 = \frac{s}{1+s^2} p_2 + \alpha k p_1,$$
  

$$p\Big|_{s=s_{in}} = p_{in}.$$
(2.8)

In the following, we will discuss all the different effects appearing in (2.8).

#### Decay and Growth of the Adapted Unknowns with $\alpha = 0$

As the first toy model, we consider

$$\partial_s p_1 = -\frac{s}{1+s^2} p_1,$$
  

$$\partial_s p_2 = \frac{s}{1+s^2} p_2.$$
(2.9)

This corresponds to the case  $\alpha = 0$ , when there is no coupling between  $p_1$  and  $p_2$ . Equation (2.9) has an explicit solution

$$p_1(s) = \frac{\langle s_{in} \rangle}{\langle s \rangle} p_{1,in},$$
  
$$p_2(s) = \frac{\langle s \rangle}{\langle s_{in} \rangle} p_{2.in}.$$

So in this model the  $p_2$  grows by  $\langle s \rangle$  and  $p_1$  decays by  $\langle s \rangle^{-1}$ . The decay in  $p_1$  corresponds to the inviscid damping of the Navier-Stokes equations discussed in (2.4). However, the  $p_2$  unknown grows in time, which is a major challenge for longtime stability. It destabilizes the nonlinear equations in the settings of small magnetic field  $0 \leq \alpha \ll \frac{1}{2}$  or for small values of the resistivity  $0 \leq \kappa \ll \nu$ .

# Constant Magnetic Field and Circular Movement with $\alpha \in \mathbb{R}$ and without Couette Flow

In this subsection, we highlight the effect of the constant magnetic field  $\alpha$  in (2.8). This is very similar to the effect of Alfén waves [Alf42, Dav16]. We consider the toy model

$$\partial_s p_1 = -\alpha k p_2, 
\partial_s p_2 = \alpha k p_1,$$
(2.10)

which corresponds to the second part of (2.8). This is solved by

$$p(s) = \begin{pmatrix} \cos(\alpha k(s - s_{in})) & -\sin(\alpha k(s - s_{in})) \\ \sin(\alpha k(s - s_{in})) & \cos(\alpha k(s - s_{in})) \end{pmatrix} p_{in}.$$

We call this *circular movement*, which leads to an exchange between  $p_1$  and  $p_2$ .

## Stability and Loss of Inviscid Damping $\alpha > \frac{1}{2}$

We consider the linearized ideal MHD equations

$$\partial_{s} p_{1} = -\frac{s}{1+s^{2}} p_{1} - \alpha k p_{2}, 
\partial_{s} p_{2} = \frac{s}{1+s^{2}} p_{2} + \alpha k p_{1}.$$
(2.11)

In particular, this is the combination of (2.9) and (2.10). For a large constant magnetic field  $\alpha > \frac{1}{2}$ , the circular movement averages the growth of the magnetic field and the decay in the velocity field. We define the energy

$$E = |p|^2 + \frac{2}{\alpha k} \frac{s}{1+s^2} p_1 p_2.$$
(2.12)

Using an energy of this form is a common approach used similarly in [BBZD23, Dol24, KZ2]. The energy E is strictly positive definite since  $\alpha > \frac{1}{2}$  and satisfies

$$(1 - \frac{1}{2\alpha|k|})|p|^2 \le E \le (1 + \frac{1}{2\alpha|k|})|p|^2$$
 (2.13)

globally in time. Then we estimate

$$\partial_s E = \frac{2}{|\alpha k|} \frac{1-s^2}{(1+s^2)^2} |p_1 p_2| \le \frac{2}{2\alpha |k|-1} \frac{1}{1+s^2} E$$

and thus we infer the energy estimate

$$\exp(-\frac{2\pi}{2\alpha|k|-1})E(s_{in}) \le E(s) \le \exp(\frac{2\pi}{2\alpha|k|-1})E(s_{in}).$$

Using (2.13), this implies stability

$$|p|(s) \approx |p_{in}|.$$

That means we are stable in the p variables. However, this comes with a cost to the inviscid damping: For the velocity field, we obtain

$$v = \frac{i}{\sqrt{1+s^2}} \begin{pmatrix} s \\ 1 \end{pmatrix} p_1,$$
  
$$b = \frac{i}{\sqrt{1+s^2}} \begin{pmatrix} s \\ 1 \end{pmatrix} p_2.$$
 (2.14)

Hence, using  $t = s + \frac{\xi}{k}$  we infer the estimate

$$\|(v_1, b_1)\|_{H^{N+1}} + \langle t \rangle \|(v_2, b_2)\|_{H^N} \lesssim \|(v_{in}, b_{in})\|_{H^{N+1}}.$$
(2.15)

In particular, compared to the Euler equations (2.5) both components lose decay of order  $\langle t \rangle^{-1}$ . This happens due to the previously mentioned growth in the magnetic field.

## 2.3 Linear Effects of the Dissipative MHD Equations with $\nu > 0$ or $\kappa > 0$

The MHD equations behave very differently depending on the relation of the dissipation parameters. We consider the linearised MHD equations with dissipation

$$\partial_t p_1 = \partial_x \partial_y^t \Delta_t^{-1} p_1 + \alpha \partial_x p_2 + \nu \Delta_t p_1,$$
  

$$\partial_t p_2 = -\partial_x \partial_y^t \Delta_t^{-1} p_2 + \alpha \partial_x p_1 + \kappa \Delta_t p_2,$$
  

$$p\big|_{t=0} = p_{in}.$$
(2.16)

After a Fourier transform, replacing  $p_2$  by  $-ip_2$  and the variable change  $s = t - \frac{\xi}{k}$ and  $s_{in} = -\frac{\xi}{k}$  (similar as in the previous section), we obtain

$$\partial_{s} p_{1} = -\frac{s}{1+s^{2}} p_{1} - \alpha k p_{2} - \nu k^{2} (1+s^{2}) p_{1},$$
  

$$\partial_{s} p_{2} = \frac{s}{1+s^{2}} p_{2} + \alpha k p_{1} - \kappa k^{2} (1+s^{2}) p_{2},$$
  

$$p|_{s=s_{in}} = p_{in},$$
(2.17)

for  $k \neq 0$ . From the stability and asymptotic of (2.17) in the *p* unknowns the stability of (2.16) and (2.1) follows directly, similar as in the previous section. This section focuses on the various effects appearing in (2.17). For a rigorous calculation, we refer to Chapters 3, 4 and 5. We saw in the inviscid case that for  $\alpha > \frac{1}{2}$ , the *p* unknowns are stable. That is not necessarily true for the dissipative regime since the imbalance of the dissipation counteracts the circular movement. In particular, we distinguish between different cases,  $\nu = \kappa$ ,  $\nu \approx \kappa$ , close to resonant s = 0,  $\nu \ll \kappa$ ,  $\kappa \ll \nu$  and ( $\nu = \kappa_x > 0$  and  $\kappa_y = 0$ ). If both dissipation parameters are close to each other  $\nu \approx \kappa$ , a strong constant magnetic field  $\alpha > \frac{1}{2}$  is sufficient to ensure stability. Both *p* get damped by a rate close to each other, ensuring stability. However, if one dissipation is much larger than the other, one component is damped to zero very fast. In contrast, in the other component, we obtain behavior similar to Subsection 2.2 for  $\alpha = 0$  with some additional terms.

#### The Case ' $\nu = \kappa$ '

This case was first considered by Liss in [Lis20]. In the case when  $\kappa = \nu$  and  $\alpha > \frac{1}{2}$ , the MHD equations are stable as in the ideal case with additional enhanced dissipation decay

$$|p|(s) \lesssim \exp(-c\nu^{\frac{1}{3}}(s-s_{in}))|p_{in}|.$$

With the energy (2.12) we obtain

$$\partial_s E + 2\nu k^2 (1+s^2) E \le \frac{2}{\alpha} \frac{1}{1+s^2} p_1 p_2$$

Thus we infer for  ${\cal E}$ 

$$E(s) \le \exp\left(\int_{s_{in}}^{s} \frac{2}{\alpha} \frac{1}{1+s^2} - 2\nu k^2 (1+s^2)\right) E(s_{in})$$
  
$$\lesssim \exp(-2c\nu^{\frac{1}{3}}(s-s_{in})) E(s_{in})$$

for some  $c = c(\alpha) > 0$ . With  $E \approx |p|^2$  for  $\alpha > \frac{1}{2}$  we obtain

$$|p|(s) \lesssim \exp(-c\nu^{\frac{1}{3}}(s-s_{in}))|p_{in}|.$$

### The Case ' $\nu \approx \kappa$ '

The regime  $\nu \leq \kappa \leq \nu^{\frac{1}{3}}$  was first considered by Dolce in [Dol24]. When  $\nu$  and  $\kappa$  are close with  $\nu^3 \leq \kappa \leq \nu^{\frac{1}{3}}$ , for a strong magnetic field  $\alpha > \frac{1}{2}$ , the circular movement ensures linear stability. Our enhanced dissipation changes to  $\exp(-c\min(\nu,\kappa)^{\frac{1}{3}}t)$  with the minimum of the dissipation parameters. We obtain the estimate

$$|p|(s) \lesssim \exp(-c\min(\nu,\kappa)^{\frac{1}{3}}(s-s_{in}))|p_{in}|$$

In the following, we show this for the regime  $\kappa \leq \nu$ , the opposite regime is analog. We derive the energy (2.12)

$$\partial_s E + 2\nu k^2 (1+s^2) |p_1|^2 + 2\kappa k^2 (1+s^2) |p_2|^2$$
  
$$\leq \frac{2}{\alpha k} \frac{1-s^2}{(1+s^2)^2} p_1 p_2 - \frac{2}{\alpha} s k(\nu+\kappa) p_1 p_2.$$

Similarly to the previous cases, the first term will lead to finite growth. The second term appears due to the imbalance of dissipation and is relevant for stability. We estimate and split the second term

$$\begin{aligned} -\frac{2}{\alpha}sk(\nu+\kappa)p_1p_2 &\leq \nu k^2(1+s^2)p_1^2 + \frac{1}{\alpha}\nu p_2^2 \\ &\leq \nu k^2(1+s^2)p_1^2 + \kappa k^2(1+s^2)p_2^2 + (\frac{1}{\alpha}\nu - \kappa k^2(1+s^2))p_2^2. \end{aligned}$$

We absorb the first two terms into the dissipation. Thus we infer

$$\partial_s E + 2c\kappa k^2 (1+s^2) E$$
  

$$\leq \frac{2}{\alpha} \frac{1}{1+s^2} p_1 p_2 + (\frac{1}{\alpha}\nu - \kappa k^2 (1+s^2))_+ p_1 p_2$$
  

$$\lesssim (\frac{1}{1+s^2} + (\frac{1}{\alpha}\nu - \kappa k^2 (1+s^2))_+) E$$

for some  $c = c(\alpha) > 0$ . At this point we require the assumption  $\nu^3 \le \kappa$ , to bound

$$\int (\frac{1}{\alpha}\nu - \kappa k^2 (1+\tau^2))_+ d\tau \le 2 \sup_s (\frac{1}{\alpha} s\nu - \frac{\kappa}{3} k^2 s^3) \le 4\frac{4}{\alpha}.$$

Thus we deduce the estimate

$$E(s) \lesssim \exp(-2c\min(\nu,\kappa)^{\frac{1}{3}}(s-s_{in}))E(s_{in}).$$

Thus, with  $\alpha > \frac{1}{2}$  we infer the estimate

$$|p|(s) \lesssim \exp(-c\min(\nu,\kappa)^{\frac{1}{3}}(s-s_{in}))|p_{in}|$$

### Frequencies Close to Resonant

Before discussing the cases where there is a large imbalance of the dissipation, we consider the case close to resonant which ensures stability in the p variables. For the non-resistive limit of Chapter 5 this is particularly important for stability. For two times  $s, s_0$  ( $s_0$  is any time, possibly different from  $s_{in}$ ) satisfying  $|s - s_0| \leq \min(\nu^{-1}, \kappa^{-1})$ , a large constant magnetic field  $\alpha > \frac{1}{2}$  ensures that the circular movement is stronger than the imbalance of dissipation yielding stability

$$|p|(s) \lesssim |p(s_0)|.$$

The derivative of the energy (2.12) satisfies

$$\begin{split} \partial_s E + 2\nu k^2 (1+s^2) |p_1|^2 + 2\kappa k^2 (1+s^2) |p_2|^2 \\ &= \frac{2}{\alpha k} \frac{1-s^2}{(1+s^2)^2} p_1 p_2 + \frac{2}{\alpha} |\nu + \kappa| sk p_1 p_2 \\ &\lesssim (\frac{1}{1+s^2} + \frac{2}{\alpha} |\nu + \kappa| sk) p_1 p_2. \end{split}$$

We estimate the second term by

$$\frac{2}{\alpha}|\nu+\kappa|skp_1p_2 \le \frac{4}{\alpha^2}\max(\nu,\kappa)|p_1|^2 + \max(\nu,\kappa)k^2(1+s^2)|p_2|^2,\\ \frac{2}{\alpha}|\nu+\kappa|skp_1p_2 \le \frac{4}{\alpha^2}\max(\nu,\kappa)|p_2|^2 + \max(\nu,\kappa)k^2(1+s^2)|p_1|^2,$$

which yields

$$\frac{2}{\alpha}|\nu+\kappa|skp_1p_2 \le \frac{4}{\alpha^2}\max(\nu,\kappa)|p|^2 + \nu k^2(1+s^2)|p_1|^2 + \kappa k^2(1+s^2)|p_2|^2.$$

So we infer the estimate

$$\partial_s E + c \min(\kappa, \nu) k^2 (1 + s^2) E$$
  

$$\lesssim \left(\frac{1}{1+s^2} + \max(\nu, \kappa)\right) p_1 p_2$$
  

$$\lesssim \left(\frac{1}{1+s^2} + \max(\nu, \kappa)\right) E$$
(2.18)

for some  $c = c(\alpha)$ . By Growall's lemma, we obtain

$$E(s) \lesssim \exp(\max(\nu, \kappa)(s - s_0))E(s_{in}) \lesssim E(s_0)$$

and thus with  $\alpha > \frac{1}{2}$  we deduce

$$|p|(s) \lesssim |p(s_0)|.$$

We note that, from (2.18) one obtains an enhanced dissipation rate of  $\exp(-c(\min(\kappa,\nu))^{\frac{1}{3}}(s-s_0))$ . However, this effect only appears if  $|s-s_0| \gg \min(\kappa,\nu)^{-\frac{1}{3}}$ . Therefore, to reach time scales where enhanced dissipation is relevant, we need the assumption  $\nu^3 \le \kappa \le \nu^{\frac{1}{3}}$  of the previous section.

### The Effect of Strong Resistivity with $\nu \ll \kappa$

In this case the viscosity is much smaller than the resistivity, see Chapter 3 for more details. In the first model, we formally set  $p_2 = 0$ . Then we obtain the behavior of the Euler and Navier-Stokes equations of Section 2.1. The assumption  $p_2 = 0$  is too strong for a more precise analysis. We consider the vorticity and current formulation

$$\partial_t w = -\alpha k j - \nu k^2 (1 + s^2) w, 
\partial_t j = (\frac{2s}{1 + s^2} - \kappa k^2 (1 + s^2)) j + \alpha k w.$$
(2.19)

We define the good unknown  $G = j + \frac{\alpha}{\kappa} \frac{1}{1+s^2} w$  and so (2.19) changes to

$$\partial_t w = -\frac{\alpha^2}{\kappa} \frac{1}{1+s^2} w - \nu k^2 (1+s^2) w + \alpha G, \partial_t G = (\frac{2s}{1+s^2} - \kappa k^2 (1+s^2)) G + \frac{\alpha}{\kappa} \frac{s}{(1+s^2)^2} w + \frac{\alpha}{\kappa} \partial_s (\frac{1}{1+s^2} w).$$

The resistivity is large enough for high frequencies k to ensure  $G \approx 0$ . Therefore we obtain for  $\beta = \frac{\kappa}{\alpha^2}$ 

$$\partial_t w = -\frac{1}{\beta} \frac{1}{1+s^2} w - \nu k^2 (1+s^2) u$$

compared to the Navier-Stokes case, the vorticity w obtains additional damping of  $-\frac{1}{\beta}\frac{1}{1+s^2}w$ .

### The Effect of Strong Viscosity with $\kappa \ll \nu$

In this case the resistivity is much smaller than the viscosity, see Chapter 5 for more details. At first glance, one would expect that large values of viscosity lead to better stability. This doesn't hold since for  $\nu \gg \kappa^{\frac{1}{3}}$  we obtain growth in the  $p_2$  unknown which corresponds to growth in the magnetic field. Due to the viscosity, for large times  $s \gtrsim \nu^{-1} \approx s_0$  we obtain  $p_1 \approx 0$  and so  $p_1$  can be neglected (In Chapter 5, Proposition 5.2 and 5.3 we prove the following with the  $p_1$  unknown). For times  $s \gtrsim \nu^{-1} \approx s_0$ , we reduce (2.19) to the toy model

$$\partial_t p_2 = \left(\frac{s}{1+s^2} - \kappa k^2 (1+s^2)\right) p_2. \tag{2.20}$$

We obtain the estimate

$$|p_2|(s) \lesssim \nu \kappa^{-\frac{1}{3}} e^{-c\kappa^{\frac{1}{3}}(s-s_0)} |p_2|(s_0), \qquad (2.21)$$

which is optimal in the sense that for k = -1 we obtain the norm inflation

$$p_2(\kappa^{-\frac{1}{3}}, k = -1) \approx \nu \kappa^{-\frac{1}{3}} p_2(s_0, k = -1).$$
 (2.22)

In particular, the  $p_2$  unknown exhibit growth of the size  $\nu \kappa^{-\frac{1}{3}}$ .

The first term in (2.20) leads to linear growth until the resistivity is strong enough for the second term to take over. This is seen in the explicit solution of (2.20)

$$p_2(s) = \frac{\langle s \rangle}{\langle s_0 \rangle} \exp\left(-\kappa k^2 \int_{s_0}^s 1 + \tau^2 \, d\tau\right) p_2(s_0).$$

Then (2.22) and (2.21) follow directly. The reader may expect that the enhanced dissipation timescale  $\nu^{-\frac{1}{3}}$  would be the relevant timescale, but the combination of circular movement and the viscosity gives enough decay for  $p_2$  such that the linear growth is suppressed until the time  $\nu^{-1}$ .

# The Effect of Horizontal Resistivity with $\kappa_x = \nu > 0$ and $\kappa_y = 0$

We consider the effect of anisotropic resistivity when  $\mu := \kappa_x = \nu > 0$  and  $\kappa_y = 0$  for a strong constant magnetic field  $\alpha > \frac{1}{2}$ , see Chapter 4 for more details. Then we obtain the linearized equations

$$\partial_t p_1 = -\frac{s}{1+s^2} p_1 - \alpha k p_2 - \mu k^2 (1+s^2) p_1, 
\partial_t p_2 = \frac{s}{1+s^2} p_2 + \alpha k p_1 - \mu k^2 p_2.$$
(2.23)

The horizontal resistivity is sufficient to ensure stability

$$|p|(s) \lesssim e^{-c\mu(s-s_{in})}|p_{in}|$$

This is shown with the energy E in (2.12)

$$\partial_s E + 2\mu k^2 (1+s^2) |p_1|^2 + 2\mu k^2 |p_2|^2$$
  
=  $\frac{2}{\alpha k} \frac{1-s^2}{(1+s^2)^2} p_1 p_2 + \frac{2}{\alpha} \mu k s (1+\frac{1}{1+s^2}) p_1 p_2.$ 

The last term is estimated by

$$\frac{2}{\alpha}\mu ksp_1p_2 \le 2\mu k^2 s^2 p_1^2 + \frac{1}{2\alpha^2}\mu k^2 p_2^2,$$
$$\frac{2}{\alpha}\mu ks\frac{1}{1+s^2}p_1p_2 \le \mu kp_1^2 + \frac{4}{1+s^2}p_2^2.$$

This yields

$$\partial_s E + 2c\mu k^2 E \le \frac{8}{1+s^2} p_1 p_2,$$

for some  $c = c(\alpha) > 0$  and so with Gronwall's Lemma, we infer

$$E(s) \lesssim e^{-2c\mu(s-s_{in})} E(s_{in}).$$

Finally, we deduce

$$|p|(s) \lesssim e^{-c\mu(s-s_{in})}|p_{in}|.$$

### 2.4 Tailored Unknowns

As seen in the previous section, we use several notions of unknowns. In this section, we summarize the most important unknowns and their relations. Usually, the different unknowns are used if we consider the part of the equation without x-average. For the x-average, we use the original equations. Then, the different unknowns are equivalent formulations of the MHD equations. In the following, we only consider the part without x-average and omit writing the projection. The unknowns satisfy the following relation

$$\begin{array}{c} p_1 \underbrace{ \overset{\Lambda_t^{-1} \nabla_t^{\perp}}{ - \Lambda_t^{-1} \nabla_t^{\perp} } v, \\ p_2 \underbrace{ \overset{\Lambda_t^{-1} \nabla_t^{\perp}}{ - \Lambda_t^{-1} \nabla_t^{\perp} } b, \end{array} \end{array}$$

and

$$p_1 \xrightarrow{\Lambda_t} w,$$

$$\phi \xrightarrow{-\Lambda_t} p_2 \xrightarrow{\Lambda_t} j.$$

Furthermore, if one dissipation parameter is much stronger, the symmetry of the p unknowns breaks. The dissipation and the linear forcing by the other unknown lead to a balance. Good unknowns are a typical approach to use this balance to obtain better estimates [Zil23, MSHZ22]. For the MHD equations, the relevant good unknowns are

$$G_w = \partial_x w + \frac{\alpha}{\nu} \partial_x^2 \phi,$$
  

$$G_\phi = \partial_x \Delta_t \phi + \frac{\alpha}{\kappa} \partial_x^2 \Delta_t^{-1} w.$$

All these different unknowns lead to different equations. In the following, we summarize the most relevant equations:

Velocity and magnetic field formulation: The standard form of the MHD equations is written in terms of the velocity and magnetic field

$$\partial_t v + v_2 e_1 = \nu \Delta_t v + \alpha \partial_x b + b \cdot \nabla_t b - v \cdot \nabla_t v - \nabla_t \pi, 
\partial_t b - b_2 e_1 = \kappa \Delta_t b + \alpha \partial_x v + b \cdot \nabla_t v - v \cdot \nabla_t b, 
\nabla_t \cdot v = \nabla_t \cdot b = 0.$$
(2.24)

The p unknowns formulation: The p unknowns reduce from the vectorial unknowns (v, b) to two scalar unknowns  $(p_1, p_2)$ . The equations for the p unknown read

$$\partial_t p_1 - \partial_x \partial_y^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \nu \Delta_t p_1 + \Lambda_t^{-1} \nabla_t^{\perp} (b \cdot \nabla_t b - v \cdot \nabla_t v), 
\partial_t p_2 + \partial_x \partial_y^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = \kappa \Delta_t p_2 + \Lambda_t^{-1} \nabla_t^{\perp} (b \cdot \nabla_t v - v \cdot \nabla_t b).$$
(2.25)

This formulation is more useful for calculations and energy estimates. In particular, as seen in the previous sections, the p unknowns allow us to quantify the relation between circular movement, inviscid damping and different dissipation regimes. In Sobolev spaces, they are equivalent to the velocity and magnetic field.

Vorticity and magnetic potential formulation: If there is a strong imbalance between resistivity and viscosity, linear stability will shift from the p unknowns to the vorticity w and magnetic potential  $\phi$ 

$$\partial_t w + v \cdot \nabla w = \nu \Delta_t w + \alpha \partial_x \Delta_t \phi + \nabla^\perp \phi \cdot \nabla \Delta_t \phi, 
\partial_t \phi + v \cdot \nabla \phi = \kappa \Delta_t \phi + \alpha \partial_x \Delta_t^{-1} w.$$
(2.26)

Vorticity and magnetic good-unknown formulation: For the resistive  $\kappa > 0$  and inviscid  $\nu = 0$  case, the resistivity damps the magnetic potential to a balance with linear forcing by the vorticity. The good unknown  $G_{\phi}$  corresponds to this balance

$$\partial_t w = -\frac{\alpha^2}{\kappa} \partial_x^2 \Delta_t^{-1} w + \alpha G_\phi + b \cdot \nabla_t j - v \cdot \nabla_t w, \partial_t G_\phi = \kappa \Delta_t G_\phi + \partial_x \Delta_t (v \cdot \nabla_t \phi) + \frac{\alpha}{\kappa} \partial_x (\Delta_t^{-1} w).$$
(2.27)

Fluid good-unknown and magnetic potential formulation: In the viscous  $\nu > 0$  and non-resistive case  $\kappa = 0$ , the viscosity damps the magnetic potential to a balance with the linear forcing by the magnetic potential. The good unknown  $G_w$  corresponds to this balance

$$\partial_t G_w = \nu \Delta_t G_w + \frac{\alpha}{\nu} \partial_t \partial_x^2 \phi + \partial_x (\nabla^\perp \phi \cdot \nabla \Delta_t \phi - v \cdot \nabla w), \partial_t \phi = -\frac{\alpha^2}{\nu} \partial_x^2 \Delta_t^{-1} \phi + \alpha \Delta_t^{-1} G_w - v \cdot \nabla \phi.$$
(2.28)

### 2.5 Nonlinear Behavior

For the MHD equations, the linearized dynamics decouple in frequency space and we obtain stability for a suitable choice of adapted unknowns. The nonlinear terms differ depending on the choice of unknowns and yield an interaction between different frequencies. Understanding and estimating the nonlinear interaction is the major challenge for establishing nonlinear stability. In the following, we give a simplified model to get an intuition for the main growth mechanism in nonlinear terms. We consider the toy model of an active scalar equation

$$\partial_t f = (a \cdot \nabla_t) f,$$
  

$$a = \nabla_t^{\perp} \Lambda_t^{-m} f.$$
(2.29)

For f we assume vanishing x-average. The m corresponds to different choices of unknowns, for m = 2 this is the nonlinearity of (2.27) and m = 1 is a

simplification of the nonlinearities in (2.25) (i.e.  $\Lambda_t^{-1} \nabla_t^{\perp}(b \nabla_t b)$  replaced by  $b \nabla_t (\Lambda_t^{-1} \nabla_t^{\perp} b) = b \nabla_t p$ ). After a Fourier transform of (2.29) we obtain

$$\partial_t f(k,\xi) = \sum_{k-l,l\neq 0} \int d\eta \frac{\xi l - k\eta}{((\xi - \eta - (k-l)t)^2 + (k-l)^2)^{\frac{m}{2}}} f(k-l,\xi-\eta) f(l,\eta).$$
(2.30)

#### Nonlinear Toy Model: Reaction Term

We apply a paraproduct decomposition to the quadratic nonlinearity to isolate the main resonance mechanism in (2.30). This is a common approach, see for example [BM14, Zil23]. We split the solution into high and low frequencies  $f = f_{lo} + f_{hi}$ . This can be made precise with the help of a Littlewood-Paley decomposition [Gra14]. We thus obtain the system

$$\partial_t f_{hi} = (a_{lo} \nabla_t) f_{hi} + (a_{hi} \nabla_t) f_{lo} + (a_{hi} \nabla_t) f_{hi}, \partial_t f_{lo} = (a_{lo} \nabla_t) f_{lo}.$$

For this splitting, the  $a_{lo}\nabla_t$  term acts as a transport and the  $(a_{hi}\nabla_t)p_{hi}$  term is small in high Sobolev spaces. Our main contribution is then the term

$$\partial_t f_{hi} \approx (a_{hi} \nabla_t) f_{lo}.$$
 (2.31)

We model this effect by replacing

$$f_{lo} \rightarrow -2c\cos(x).$$

These are called a traveling wave solution. The traveling in the name comes from the fact that our coordinates follow the characteristic of the Couette flow, which means we 'travel' in the original coordinates by x - yt. The cos is a solution of frequency k = 1 and yields a nearest-neighbor interaction between the k and  $k \pm 1$  modes. This is a very good model for the largest possible growth in the full nonlinear system. More generally, we could use any periodic and smooth function in x. With this low frequent solution, we obtain

$$\partial_t f = -\nabla^{\perp} \Lambda_t^{-m} f \cdot \nabla 2c \cos(x)$$
$$= 2c \sin(x) \partial_y \Lambda_t^{-m} f,$$

which is our main growth model. After a Fourier transform, we obtain

$$\partial_t f(k,\xi) = c \frac{\xi}{(k+1)^m} \frac{1}{(1+(\frac{\xi}{k+1}-t)^2)^{\frac{m}{2}}} f(k+1,\xi) - c \frac{\xi}{(k-1)^m} \frac{1}{(1+(\frac{\xi}{k-1}-t)^2)^{\frac{m}{2}}} f(k-1,\xi).$$

We simplify this further by looking at the k mode acting on k-1 mode

$$\partial_t f(k-1,\xi) = c \frac{\xi}{k^m} \frac{1}{(1+(\frac{\xi}{k}-t)^2)^{\frac{m}{2}}} f(k,\xi).$$

We distinguish between the two main cases for m.

• Let m = 2, this corresponds to the main model for the resistive MHD equations in Chapter 3

$$\partial_t f(k-1,\xi) = c \frac{\xi}{k^2} \frac{1}{1 + (\frac{\xi}{k} - t)^2} f(k,\xi)$$

The main effect of f(k) acting on f(k-1) will appear close to the time  $\tilde{t}_k = \frac{\xi}{k}$ , which we call resonant time. Close to this time, the f(k) stays approximately constant and thus after integrating in time

$$f(t_{k-1}, k-1, \xi) \approx c\pi \frac{\xi}{k^2} f(t_k, k, \xi)$$

for times  $t_l = \frac{1}{2}(\tilde{t}_l + \tilde{t}_{l+1})$  between the resonant times. If we iterate this growth  $k \to k - 1 \to \cdots \to 1$  for initial data  $f(t_k, l, \xi) = \delta_{kl}$  we obtain

$$f(t_1, 1, \xi) \approx \prod_{l=1}^k c\pi \frac{\xi}{l^2} = \frac{(c\pi\xi)^k}{(k!)^2}.$$

This is maximized by Sterling's approximation at  $k \approx \sqrt{c\xi}$  to

$$f(t_1, 1, \xi) \approx \sqrt{\xi} \exp(\tilde{C}\sqrt{\xi}) f(t_k, k, \xi)$$

which is a loss if Gevrey 2 regularity of  $\tilde{C} = \tilde{C}(c) > 0$  in the radius of convergence. Which corresponds to our result in Chapter 3.

• The case m = 1 corresponds to the main model of the p unknowns in Chapter 5

$$\partial_t f(k-1,\xi) = c_k^{\xi} \frac{1}{(1+(\frac{\xi}{k}-t)^2)^{\frac{1}{2}}} f(k,\xi).$$

With the previous approach, we would need regularity stronger than Gevrey 1, which would impose problems in the nonlinear estimates. Here we perform a different approach, we show under which circumstances we can suppress the nonlinear effect with dissipation. At times  $t \geq \min(\nu, \kappa)^{-\frac{1}{3}}$ , the enhanced dissipation appears and damps the equation down quickly. Thus we consider times  $t \leq \mu^{-\frac{1}{3}}$  with  $\mu = \min(\nu, \kappa)$ . We write

$$\frac{\xi}{k} \frac{1}{(1 + (\frac{\xi}{k} - t)^2)^{\frac{1}{2}}} = 1 + \frac{t}{(1 + (\frac{\xi}{k} - t)^2)^{\frac{1}{2}}}$$

and after integrating in time

$$\int_0^t \frac{\xi}{k} \frac{1}{(1 + (\frac{\xi}{k} - \tau)^2)^{\frac{1}{2}}} d\tau \lesssim t \ln(2\sqrt{1 + t^2}) \le \mu^{-\frac{1}{3}} \ln(4\mu^{-\frac{1}{3}}).$$

If we assume that f(k) is constant for  $t \approx \frac{\xi}{k}$ , we obtain then

$$f(t_{k-1}k - 1, \xi) \le 2c\mu^{-\frac{1}{3}}\ln(4\mu^{-\frac{1}{3}})f(t_k, k, \xi).$$

Thus for  $c < \frac{1}{2}\mu^{\frac{1}{3}}\left(\ln(4\mu^{-\frac{1}{3}})\right)^{-1}$  these resonances get supresed. The idea here corresponds to what is done in the nonlinear estimates if  $\nu \approx \kappa > 0$ . However, the nonlinear estimates are more challenging since the supremum norm in time and the  $L^2$  norm in space do not commute. Therefore, the nonlinear estimates are worse than conjectured here for the toy model.

## Chapter 3

# On Echoes in Magnetohydrodynamics with Magnetic Dissipation

This chapter consists of the paper [KZ1], published in the Journal of Differential Equations and is a joint work with Christian Zillinger. In this Chapter, we often refer to Gevrey spaces in the sense of (A.1) since the important effects are visible there. We note here that the space (3.19) with (3.9) also includes the Gevrey spaces in the sense of (A.2).

NIKLAS KNOBEL AND CHRISTIAN ZILLINGER

Abstract. We study the long time asymptotic behavior of the inviscid magnetohydrodynamic equations with magnetic dissipation near a combination of Couette flow and a constant magnetic field. Here we show that there exist nearby explicit global in time low frequency solutions, which we call waves. Moreover, the linearized problem around these waves exhibits resonances under high frequency perturbations, called echoes, which result in norm inflation Gevrey regularity and infinite time blow-up in Sobolev regularity.

## **3.1** Introduction and Main Results

In this article we consider the two-dimensional magnetohydrodynamic (MHD) equations with magnetic resistivity  $\kappa > 0$  but without viscosity

$$\partial_t V + V \cdot \nabla V + \nabla p = B \cdot \nabla B,$$
  

$$\partial_t B + V \cdot \nabla B = \kappa \Delta B + B \cdot \nabla V,$$
  

$$\operatorname{div}(B) = \operatorname{div}(V) = 0,$$
  

$$(t, x, y) \in \mathbb{R}_+ \times \mathbb{T} \times \mathbb{R},$$
  
(3.1)

near the stationary solution

$$V(t, x, y) = (y, 0), B(t, x, y) = (\alpha, 0).$$
(3.2)

The MHD equations are a common model of the evolution of conducting fluids interacting with (electro-)magnetic fields in regimes where the magnetization of the fluid can be neglected. They describe the evolution of the fluid in terms of the fluid velocity V, pressure p and magnetic field B. The constant mass and charge densities are normalized to 1. Here particular examples of applications range from the modeling of solar dynamics to geomagnetism and the earths molten core to using liquid metals in industrial applications or in fusion applications [Dav16].

A main aim of this article is to analyze the long-time asymptotic behavior of solutions to this coupled system and, in particular, the interaction of instabilities, partial dissipation and the system structure of the equations. Here we note that due to the affine structure of the stationary solution (3.2), the corresponding linearized problem around this solution decouples in Fourier space and can be shown to be stable in arbitrary Sobolev (or even analytic) regularity, as we prove in Section 3.2.

#### **Lemma 3.1.** Let $\alpha \in \mathbb{R}$ be given and consider the linear problem

$$\partial_t V + y \partial_x V + (V_2, 0) = \alpha \partial_x B,$$
  

$$\partial_t B + y \partial_x B - (B_2, 0) = \kappa \Delta B + \alpha \partial_x V,$$
  

$$\operatorname{div}(B) = \operatorname{div}(V) = 0,$$
  

$$(t, x) \in \mathbb{R}_+ \times \mathbb{T} \times \mathbb{R}.$$

Then these equations are stable in  $H^s$  for any  $s \in \mathbb{R}$  in the sense that there exists a constant C > 0 such that for any choice of initial data and all times t > 0 it holds that

$$\begin{aligned} \| (\nabla^{\perp} \cdot V)(t, x - ty, y) \|_{H^s}^2 + \| (\nabla^{\perp} \cdot B)(t, x - ty, y) \|_{H^s}^2 \\ &\leq (1 + \kappa^{-2/3})^2 (\| \nabla^{\perp} \cdot V |_{t=0} \|_{H^s}^2 + \| \nabla^{\perp} \cdot B |_{t=0} \|_{H^s}^2). \end{aligned}$$

Here  $\nabla^{\perp} \cdot V =: W$  is the vorticity of the fluid and  $\nabla^{\perp} \cdot B =: J$  is the (magnetically induced) current.

In contrast to this to this very strong linear stability result, the stability results for the inviscid nonlinear equations are expected to crucially rely on very high, Gevrey regularity (see Section 3.2.2 for a definition). More precisely, similarly to the nonlinear Euler equations [DM23, DZ21, Zil23] or Vlasov-Poisson equations [Bed20, Zil21a, MV11] the nonlinear equations are not a priori expected to not remain close to the linear dynamics due to "resonances" or "echoes" [MWGO68, YOD05], which may lead to unbounded norm inflation of any Sobolev norm. It is the main aim of this article to identify and capture this resonance mechanism for the resistive MHD equations. In particular, we ask to

which extent magnetic dissipation can stabilize the dynamics. As we discuss in Section 3.2.2 the main nonlinear resonance mechanism is expected to be given by the repeated interaction of a high frequency perturbation with an underlying low frequency perturbation of (3.2). In this article we thus explicitly construct such low frequency nonlinear solutions, called *traveling waves* (a combination of an Alfvén waves and shear dynamics; see Section 3.2 and Lemma 3.5 for further discussion).

**Lemma 3.2.** Let  $\kappa > 0$  and  $\alpha \in \mathbb{R}$  and let  $(f_0, g_0) \in \mathbb{R}^2$ . Then there exist smooth global in time solutions of the nonlinear, resistive MHD equations (3.1), which are of the form

$$V(t, x, y) = (y, 0) - \frac{f(t)}{1 + t^2} \nabla^{\perp} \cos(x - ty),$$
  
$$B(t, x, y) = (\alpha, 0) + \frac{g(t)}{1 + t^2} \nabla^{\perp} \sin(x - ty),$$

with  $(f(0), g(0)) = (f_0, g_0)$ . Furthermore, for a suitable choice of  $f_0, g_0$  it holds that

$$f(t) \to 2c,$$
  
$$g(t) \to 0,$$

as  $t \to \infty$ .

In view of the underlying shear dynamics it is natural to change to coordinates

$$(x - ty, y).$$

In these coordinates the corresponding vorticity  $W = \nabla^\perp \cdot V$  and current  $J = \nabla^\perp \cdot B$  read

$$W = -1 + f(t)\cos(x),$$
  
$$J = 0 - g(t)\sin(x).$$

Unlike the stationary solution (3.2) these waves have a non-trivial x-dependence. As we discuss in Section 3.2.2 this x-dependence allows resonances to propagate in frequency and underlies the nonlinear instability of the stationary solution (3.2). More precisely, we show that the (simplified) *linearized* equations around these waves exhibit the above mentioned *nonlinear* resonance mechanism (in terms of both upper and lower bounds on solutions). In particular, we aim to obtain a precise understanding of the dependence of the resonance mechanism on the resistivity  $\kappa > 0$  and the frequency-localization of the initial perturbation. The research on well-posedness and asymptotic behavior of the magneto-hydrodynamic equations is a very active field of research and we in particular mention the recent work [Lis20], which considers a related, fully dissipative setting in 3D, as well as the articles [JW22, ZZ23, WZ21, BLW20, FL19, DYZ19,

HXY18, LCZL18, WZ17]. More precisely, in [Lis20] Liss studied the nonlinear, fully dissipative, three-dimensional MHD equations around the same stationary solution (3.2) in a doubly-periodic three-dimensional channel  $\mathbb{T} \times \mathbb{R} \times \mathbb{T}$  and established bounds on the Sobolev stability threshold as  $\nu = \kappa \downarrow 0$ . In contrast, this article considers the 2D setting with partial dissipation  $\nu = 0, \kappa > 0$  in Gevrey regularity. Similar questions on the stability of systems with partial dissipation in critical spaces are also a subject of active research in other (fluid) systems, such as the Boussinesq equations [CW13, EW15, DWZZ18].

For simplicity of presentation and to simplify the analysis in this article we modify the linearized equations for the vorticity and current perturbations w, j

$$\partial_t w = \alpha \partial_x j - (2c \sin(x) \partial_y \Delta_t^{-1} w)_{\neq}$$
  

$$\partial_t j = \kappa \Delta_t j + \alpha \partial_x w - 2 \partial_x \partial_y^t \Delta_t^{-1} j,$$
  

$$\Delta_t = \partial_x^2 + (\partial_y - t \partial_x)^2,$$
(3.3)

and fix the x-averages of w and j, which also fixes the underlying shear flow. Here, for simplicity we have also replaced f(t), g(t) by 2c and 0, respectively. In analogy to other fluid systems [BBZD23, BM15b], a similar structure of the equations can be achieved by considering the coordinates

$$(x - \int_0^t \int V_1 dx dt, \frac{1}{t} \int_0^t \int V_1 dx dt) =: (X, Y),$$

which however makes estimates of  $\Delta_t^{-1}$  technically more involved and less transparent [Zil17]. In the interest of a clear presentation of the resonance mechanism we hence instead fix Y = y by a small forcing.

**Theorem 3.3.** Let  $0 < \alpha < 10$  and  $0 < \kappa < 1$  with  $\beta := \frac{\kappa}{\alpha^2}$  and  $c \leq \min(10^{-3}\beta^{\frac{16}{3}}, 10^{-4})$  be given. Consider the (simplified) linearized equations (3.3) around the wave of Lemma 3.2.

Then there exists a constant C such that for any initial data  $w_0, j_0$  whose Fourier transform satisfies

$$\sum_{k} \int \exp(C\sqrt{|\xi|}) (|\mathcal{F}w_0(k,\xi)|^2 + |\mathcal{F}j_0(k,\xi)|^2) d\xi < \infty$$

the corresponding solution stays regular for all times up to a loss of constant in the sense that for all t > 0 it holds that

$$\sum_{k} \int \exp(\frac{C}{2}\sqrt{|\xi|}) (|\mathcal{F}w(t,k,\xi)|^2 + |\mathcal{F}j(t,k,\xi)|^2) d\xi < \infty.$$

Moreover, there exists initial data  $w_0, j_0$  and  $0 < C^* < C$  such that

$$\sum_{k} \int \exp(C^* \sqrt{|\xi|}) (|\mathcal{F}w_0(k,\xi)|^2 + |\mathcal{F}j_0(k,\xi)|^2) d\xi < \infty,$$

but so that the corresponding solution w, j grows unbounded in Sobolev regularity as  $t \to \infty$ .

Let us comment on these results:

- As we discuss in Section 3.2.2 the linearized equations around a traveling wave closely resemble the interaction of high and low-frequency perturbations in the nonlinear equations. These equations thus are intended to serve as slightly simplified model of the nonlinear resonance mechanism. We remark that in the full nonlinear problem the x-averages and hence the underlying shear dynamics change with time and the corresponding change of coordinates has to be controlled. For simplicity and clarity the present model instead fixes this change of coordinates.
- The Fourier integrability with a weight  $\exp(C\sqrt{|\xi|})$  corresponds to Gevrey 2 regularity with respect to y. For simplicity of presentation the above results are stated with  $L^2(\mathbb{T})$  regularity in x. All results also extend to more general Fourier-weighted spaces, such as  $H^N$  for any  $N \in \mathbb{N}$  or suitable Gevrey or analytic spaces (see Definitions 3.8 and 3.9).
- The stability and norm inflation in Gevrey 2 regularity matches the regularity classes of the (nonlinear) Euler equations. In particular, the magnetic field and magnetic dissipation are shown to not be strong enough to suppress this growth. We remark that our choice of coupling between the size of the magnetic field and magnetic dissipation is made so that both effects are "of the same magnitude" and hence their interaction plays a more crucial role (see Section 3.2.3 for a discussion).
- These results complement the work of Liss [Lis20] on the Sobolev stability threshold in 3D with full dissipation. Indeed, the above derived upper and lower bounds establish Gevrey 2 as the optimal regularity class of the linearized problem in 2D with partial dissipation. We expect that as for the Euler [BM15a] or Vlasov-Poisson equations [BMM16] nonlinear stability results match the regularity classes of the linearized problem around appropriate traveling waves.

We further point out that the instability result of Theorem 3.3 also implies a norm inflation result for the nonlinear problem around each wave in slightly different spaces (see Corollary 3.26). In particular in any arbitrarily small analytic neighborhood around the stationary solution (3.2) there exist nonlinearly unstable solutions (with respect to lower than Gevrey 2 regularity).

The remainder of the article is structured as follows:

- In Section 3.2 we discuss the linearized problem around the stationary state (3.2) and introduce waves as low-frequency solutions of the nonlinear problem.
- In Section 3.2.2 we discuss the resonance mechanism for a toy model. In particular, we discuss optimal spaces for norm inflation and (in)stability results as well as the time- and frequency-dependence of resonances.

- The main results of this article are contained in Section 3.4, where we establish upper bounds and lower bounds on the norm inflation.
- The 3.5 contains some auxiliary estimates of a growth factor in Section 3.4. In the second 3.6 we prove a nonlinear instability result for the traveling waves.

## 3.2 Linear Stability, Traveling Waves and Echo Chains

In this section we establish the linear stability of the resistive MHD equations (3.1)

$$\partial_t V + (V \cdot \nabla)V + \nabla p = (B \cdot \nabla)B,$$
  

$$\partial_t B + (V \cdot \nabla)B = \kappa \Delta B + (B \cdot \nabla)V,$$
  

$$\nabla \cdot B = \nabla \cdot V = 0,$$
(3.4)

around the stationary solution (3.2) as stated in Lemma 3.1. Furthermore, we sketch the nonlinear resonance mechanism underlying the norm inflation result of Theorem 3.3, which is given by the repeated interaction of high and low frequency perturbations. This mechanism motivates the construction of the traveling wave solutions of Lemma 3.5 and the corresponding (simplified) linearized equations around these waves, which are studied in the remainder of the article.

In order to simplify notation we may restate the MHD equations with respect to other unknowns. That is, since we consider vector fields in two dimensions and V and B are divergence-free, we may introduce the magnetic potential  $\Phi$ , magnetic current J and fluid vorticity W by

$$J = \nabla^{\perp} \cdot B,$$
  
$$\Delta \Phi = J,$$
  
$$W = \nabla^{\perp} \cdot V.$$

Under suitable decay assumptions (or asymptotics) in infinity the equations can then equivalently be expressed as

$$\partial_t W + (V \cdot \nabla) W = (B \cdot \nabla) \Delta \Phi, \partial_t \Phi + (V \cdot \nabla) \Phi = \kappa \Delta \Phi,$$
(3.5)

or in terms of J:

$$\partial_t W + (V \cdot \nabla)W = (B \cdot \nabla)J, \partial_t J + (V \cdot \nabla)J = \kappa \Delta J + (B \cdot \nabla)W - 2(\partial_i V \cdot \nabla)\partial_i \Phi.$$
(3.6)

With these formulations we are now ready to establish the linear stability of the stationary solution (3.2).

Proof of Lemma 3.1. Consider the formulation of the MHD equations as (3.5), then the linearization around  $V = (y, 0), W = -1, B = (\alpha, 0), \Phi = \alpha y$  is given by

$$\partial_t W + y \partial_x W = \alpha \partial_x \Delta \Phi,$$
  
$$\partial_t \Phi + y \partial_x \Phi + V_2 \alpha = \kappa \Delta \Phi.$$

We note that all operators other than  $y\partial_x$  are constant coefficient Fourier multipliers. Hence we apply a change of variables

$$(x,y) \mapsto (x-ty,y)$$

to remove this transport term and obtain

$$\begin{aligned} \partial_t w &= \alpha \partial_x \Delta_t \phi, \\ \partial_t \phi &= -\alpha \partial_x \Delta_t^{-1} w + \kappa \Delta_t \phi, \end{aligned}$$

where  $w, \phi$  denote the unknowns with respect to these variables and  $\Delta_t = \partial_x^2 + (\partial_y - t\partial_x)^2$ . We note that this system decouples in Fourier space and for simplicity of notation express it in terms of the (Fourier transform of the) current  $j = \Delta_t \phi$ :

$$\partial_t w = ik\alpha j,$$
  
$$\partial_t j = \frac{2k(kt-\xi)}{k^2 + (\xi-kt)^2}j - \kappa(k^2 + (\xi-kt)^2)j + ik\alpha w,$$

where  $k \in \mathbb{Z}$  and  $\xi \in \mathbb{R}$  denote the Fourier variables with respect to  $x \in \mathbb{T}$  and  $y \in \mathbb{R}$ , respectively. Here and in the following, with slight abuse of notation, we reuse w and j to refer to the Fourier transforms of the vorticity and current perturbation. For k = 0 these equations are trivial and we hence in the following we may assume without loss of generality that  $k \neq 0$ . Furthermore, we note that the right-hand-side depends on  $\xi$  only in terms of  $\frac{\xi}{k} - t$ . Hence, by shifting time we may further assume that  $\xi = 0$ .

With this reduction we first note that by anti-symmetry for all  $\alpha \in \mathbb{R}$  it holds that

$$\partial_t (|w|^2 + |j|^2)/2 = (\frac{2t}{1+t^2} - \kappa k^2 (1+t^2))|j|^2.$$

We make a few observations:

- If  $\kappa k^2 \ge 1$  the horizontal dissipation is sufficiently strong to absorb growth for all times.
- If  $\kappa k^2 \leq 1$  is small, then for sufficiently large times  $|t| \geq (k^2 \kappa)^{-1/3}$  the right-hand-side is non-positive.
- It thus only remains to estimate the growth on the time interval  $|t| \leq (k^2 \kappa)^{-1/3}$ , where

$$\partial_t (|w|^2 + |j|^2) \le \frac{4(t)_+}{1+t^2} (|w|^2 + |j|^2).$$

The latter case can be bounded by an application of Gronwall's lemma and after shifting back in time it yields

$$|w(t)|^2 + |j(t)|^2 \le (1 + (k^2 \kappa)^{-2/3})^2 (|w(0)|^2 + |j(0)|^2)$$

for all t > 0.

While the ground state is thus linearly stable in arbitrary Sobolev or even analytic regularity, nonlinear stability poses to be a much more subtle question with stronger regularity requirements.

#### 3.2.1 Wave-type Perturbations

In order to investigate the stability of the MHD equations, it is a common approach to consider wave-type perturbation. Here a classical result considers perturbations around a constant magnetic field and a vanishing velocity field.

**Lemma 3.4** ((c.f. [Alf42, Dav16])). Consider the ideal MHD equations (i.e.  $\kappa = 0$ ) in three dimensions linearized around a constant magnetic field  $B = B_0 e_z$  and vanishing velocity field V = 0. Then a particular solution is of the form

$$B = (B_1(t, z), 0, 0), V = (V_1(t, z), 0, 0)$$

where  $B_1$  and  $V_1$  are solutions of the wave equation

$$\partial_t^2 B_1 - B_0^2 \partial_z^2 B_1 = 0,$$
  
$$\partial_t^2 V_1 - B_0^2 \partial_z^2 V_1 = 0,$$

The linearized problem thus admits wave-type solutions propagating in the direction  $e_z$  of the constant magnetic field and pointing into an orthogonal direction. These solutions are known as *Alfvén waves* [Alf42].

Proof of Lemma 3.4. We make the ansatz that B and v only depend on t and z and express the linearized equations in terms of the current  $J = \nabla \times B$  and vorticity  $W = \nabla \times V$ . Then the equations reduce to

$$\partial_t J = B_0 \partial_z W,$$
  
$$\partial_t W = B_0 \partial_z J.$$

These equations are satisfied if both J and W solve a wave equation and are chosen compatibly. More precisely, two linearly independent solutions are given by

$$W = f(z + B_0 t) = J$$

and

$$W = g(z - B_0 t) = -J,$$
where f and g are arbitrary smooth function.

We remark that since B = B(t, z) and V = V(t, z) point into a direction orthogonal to the z-axis, they are divergence-free for all times. Finally, since both functions are independent of x it follows that all nonlinearities  $V \cdot \nabla V, B \cdot$  $\nabla V, V \cdot \nabla B, B \cdot \nabla B$  identically vanish, so these are also nonlinear solutions.  $\Box$ 

In the following we consider the two-dimensional setting and extend this construction to also include an underlying affine shear flow. We call the resulting solutions *traveling waves* in analogy to dispersive equations and related constructions for fluids and plasmas [DZ21, Bed20, Zil21a, Zil23, DM23]. As we sketch in Section 3.2.2 the non-trivial x-dependence of these waves will allow us to capture the main nonlinear norm inflation mechanism in the linearized equations around these waves (as opposed to linearizing around the stationary solutions (3.2)).

**Lemma 3.5.** Let  $\alpha \in \mathbb{R}$  and  $\kappa \geq 0$  be given. Then for any choice of parameters  $(f(0), g(0)) \in \mathbb{R}^2$  there exists a solution of (3.6) of the form

$$W = -1 + f(t)\cos(x - yt) J = -g(t)\sin(x - yt).$$
(3.7)

We call such a solution a traveling wave.

We remark that this construction also allows for general profiles h(t, x - ty)in place of  $\cos(x - ty)$ . This particular choice is made so that for f(0) and g(0)small, such a wave is an initially small, analytic perturbation of the stationary solution (3.2) and localized at low frequency.

*Proof of Lemma 3.5.* For easier reference we note that for this ansatz, we obtain

$$V = (y, 0) + \frac{f(t)}{1+t^2} \sin(x - yt)(t, 1)$$
  

$$W = -1 + f(t) \cos(x - yt)$$
  

$$B = (\alpha, 0) + \frac{g(t)}{1+t^2} \cos(x - yt)(t, 1)$$
  

$$J = -g(t) \sin(x - yt)$$
  

$$\Phi = \alpha y + \frac{g(t)}{1+t^2} \sin(x - yt).$$

Inserting this into the equation (3.6) the nonlinearities vanish due to the onedimensional structure of the waves. Therefore, this ansatz yields a solution if and only if f and g solve the ODE system

$$f'(t) = -\alpha g(t),$$
  

$$g'(t) = -\kappa (1+t^2)g(t) + \alpha f(t) + \frac{2t}{1+t^2}g(t).$$
(3.8)

Thus by classical ODE theory for any choice of initial data there indeed exists a unique traveling wave solution.  $\hfill \Box$ 

Given such a traveling wave we are interested in its behavior, in particular for large times, and how it depends on the choices of  $\kappa$  and  $\alpha$ .

**Lemma 3.6.** Let  $\alpha > 0$  and  $\kappa > 0$  then for any choice of initial data the solutions f(t), g(t) of the ODE system (3.8)

$$f'(t) = -\alpha g(t),$$
  

$$g'(t) = -\kappa (1+t^2)g(t) + \alpha f(t) + \frac{2t}{1+t^2}g(t),$$
(3.9)

satisfy the following estimates:

$$\begin{cases} |f(t)|^2 + |g(t)|^2 \le (1+t^2)^2 (|f(0)|^2 + |g(0)|^2) & \text{if } 0 < t < \kappa^{-1/3} \\ |f(t)|^2 + |g(t)|^2 \le |f(\kappa^{-1/3})|^2 + |g(\kappa^{-1/3})|^2 & \text{if } t > \kappa^{-1/3}. \end{cases}$$
(3.10)

Furthermore, for a specific choice of initial data it holds

$$|f(t) - \epsilon| \le \frac{1}{2}\epsilon,$$

for all  $t \ge 4\beta^{-1}$  and

$$|g(t)| \to 0$$

as  $t \to \infty$ .

*Proof of Lemma 3.6.* We first observe that by anti-symmetry of the coefficients it holds that

$$\partial_t (|f|^2 + |g|^2) = 2|g|^2 \left(-\kappa(1+t^2) + \frac{2t}{1+t^2}\right).$$

In particular, for  $t > \kappa^{-1/3}$  the last factor is negative and hence  $|f|^2 + |g^2|$  is non-increasing. For times smaller than this, we may derive a first rough bound from the estimate

$$\partial_t (|f|^2 + |g|^2) \le (|f|^2 + |g|^2) \frac{4t}{1+t^2},$$

which yields an algebraic lower and upper bound growth bound. We next turn to the case of special data, due to lower and upper norm bounds (3.9) is time reversible. Therefore, we can obtain

$$f(t_0) = 1, \qquad g(t_0) = 0$$

for  $t_0 = 4\beta^{-1}$ . Then we deduce

$$g(t) = \alpha \int_{t_0}^t d\tau \; \exp(-\kappa(t - \tau + \frac{1}{3}(t^3 - \tau^3))) \frac{1 + t^2}{1 + \tau^2} f(\tau)$$

and thus

$$f(t) = 1 - \alpha \int_{t_0}^t d\tau_1 g(\tau_1)$$
  
=  $1 - \frac{\kappa}{\beta} \int_{t_0}^t d\tau_1 (1 + \tau_1^2) \int_{t_0}^{\tau_1} d\tau_2 \exp(-\kappa(\tau_1 - \tau_2 + \frac{1}{3}(\tau_1^3 - \tau_2^3))) \frac{1}{1 + \tau_2^2} f(\tau_2)$   
=  $1 - \frac{1}{\beta} \int_{t_0}^t d\tau_2 \frac{1}{1 + \tau_2^2} f(\tau_2) (1 - \exp(-\kappa(t - \tau_2 + \frac{1}{3}(t^3 - \tau_2^3)))).$ 

This gives the estimate

$$0 \le 1 - f(t) \le \frac{2}{\beta t_0},\tag{3.11}$$

which implies that after time  $t_0$  the value of f satisfies the same bound. Similarly, for g we recall that

$$\partial_s g = \left(\frac{2t}{1+t^2} - \kappa(1+t^2)\right)g(t) + \alpha f(t)$$

and hence for  $t_1 = 2\kappa^{-\frac{1}{3}}$  it holds that

$$g(t_1) \le \alpha \frac{t_1^2}{t_0^2}.$$

Furthermore, this implies that for times  $t \ge t_1$  it holds that

$$g(t) \le \alpha \frac{t_1^2}{t_0^2} \exp(-\frac{\kappa}{3} t^2 (t - t_1)) + \alpha \int_{t_1}^t \exp(-\frac{\kappa}{3} (1 + t^2) (t - \tau)) \\ \le \alpha \frac{t_1^2}{t_0^2} \exp(-\frac{\kappa}{3} t^2 (t - t_1)) + \frac{3\alpha}{\frac{\kappa}{3} (1 + t)^2}.$$

Finally, for times  $t \gg \kappa^{-\frac{1}{3}}$  we may estimate

$$g \le \frac{4\alpha}{\kappa t^2}.\tag{3.12}$$

#### 3.2.2 Paraproducts and an Echo Model

As mentioned in Section 3.1 the main mechanism for nonlinear instability is expected to be given by the repeated interaction of high- and low-frequency perturbations of the stationary solution (3.2). In the following we introduce a model highlighting the role of the traveling waves and discuss what stability and norm inflation estimates can be expected.

For this purpose we note that the nonlinear MHD equations (3.5) for the perturbations  $w, \phi$  of the groundstate (3.2) in coordinates (x - ty, y) can be expressed as

$$\partial_t w + \nabla^{\perp} \Delta_t^{-1} w \cdot \nabla w = \alpha \partial_x \Delta \phi + \nabla^{\perp} \phi \cdot \nabla \Delta_t \phi,$$
  

$$\partial_t \phi + \nabla^{\perp} \Delta_t^{-1} w \cdot \nabla \phi = \alpha \partial_x \Delta^{-1} w + \kappa \Delta_t \phi,$$
  

$$\Delta_t = \partial_x^2 + (\partial_y - t \partial_x)^2,$$
(3.13)

where we used cancellation properties of  $\nabla^{\perp} \cdot \nabla$ . The stability result of Lemma 3.1 considered the linearized problem around the trivial solution (0,0), which removes all effects of the nonlinearities. In order to incorporate these effects into our model we thus consider the nonlinear equations as a coupled system for the low frequency part of the solution  $(w_{low}, \phi_{low})$  (defined as the Littlewood-Payley projection to frequencies  $\langle N/2 \rangle$  for some dyadic scale N) and the high frequency part  $(w_{hi}, \phi_{hi})$ . If we for the moment consider the low frequency part as given then the action of the nonlinearities on the high frequency perturbation of the vorticity can be decomposed as

$$\nabla^{\perp} \Delta_t^{-1} w_{low} \cdot \nabla w_{hi} + \nabla^{\perp} \Delta_t^{-1} w_{hi} \cdot \nabla w_{low} + \nabla^{\perp} \Delta_t^{-1} w_{hi} \cdot \nabla w_{hi}.$$
(3.14)

Here the first term is of transport type and hence unitary in  $L^2$  and we expect  $\nabla^{\perp}\Delta_t^{-1}w_{low}$  to decay sufficiently quickly in time that this term should not yield a large contribution to possible norm inflation. Similarly for the last term we note that both factors are at comparable frequencies and that we by assumption consider a small high frequency perturbation and thus this term is also not expected to have a large impact on the evolution. The main norm inflation mechanism thus is expected to be given by the high frequency velocity perturbation interacting with a non-trivial low frequency vorticity perturbation.

In order to build our toy model we thus focus on this part and formally replace  $w_{low}, \phi_{low}$  by the traveling waves, which are solutions of the nonlinear problem. Furthermore, as a simplification by a similar reasoning as above we also fix the underlying shear flow for our model. Then the equations for the (high frequency part of the) current perturbation  $j = \Delta \phi$  and vorticity perturbation w read

$$\partial_t w = \alpha \partial_x j - (2c \sin(x) \partial_y \Delta_t^{-1} w)_{\neq}$$
  
$$\partial_t j = \kappa \Delta_t j_{\neq} + \alpha \partial_x w - 2\partial_x \partial_y^t \Delta_t^{-1} j,$$
  
(3.15)

where we also simplified to f(t) = 2c, g(t) = 0.

We note that compared to the linearization around the stationary solution these equations break the decoupling in Fourier space. Indeed taking a Fourier transform and relabeling  $j \mapsto -ij$  we arrive at

$$\partial_t w(k) = -\alpha k j(k) - c \frac{\xi}{(k+1)^2} \frac{1}{1 + (\frac{\xi}{k+1} - t)^2} w(k+1) + c \frac{\xi}{(k-1)^2} \frac{1}{1 + (\frac{\xi}{k-1} - t)^2} w(k-1),$$
  

$$\partial_t j(k) = \left(2 \frac{t - \frac{\xi}{k}}{1 + (\frac{\xi}{k} - t)^2} - \kappa k^2 (1 + (\frac{\xi}{k} - t)^2)) j(k) + \alpha k w(k).$$
(3.16)

Furthermore, if  $t \approx \frac{\xi}{k}$  then the additional term is of size  $c\frac{\xi}{(k)^2}$  and hence can possibly lead to a very large change of the dynamics. In reference to the experimental results mentioned in Section 3.1 we can interpret this as the low frequency and high frequency perturbation resulting in an "echo" around the time  $t \approx \frac{\xi}{k}$ . For the following toy model we neglect all modes except those at frequency k and k-1 and only include the action of the resonant mode k on the non-resonant mode k-1. **Lemma 3.7** (Toy model). Let  $c, \kappa, \alpha$  be as in Theorem 3.3 such that  $\beta = \frac{\kappa}{\alpha^2} \ge \pi$ and consider the Fourier variables  $k \ge 2$  and  $\xi \ge 10 \max(\kappa^{-1}, \frac{k^2}{c})$ . Then for

$$t_k := \frac{1}{2}\left(\frac{\xi}{k+1} + \frac{\xi}{k}\right) < t < \frac{1}{2}\left(\frac{\xi}{k} + \frac{\xi}{k-1}\right) =: t_{k-1}$$

we consider the toy model

$$\partial_t w(k) = -\alpha k j(k),$$
  

$$\partial_t j(k) = \left(2 \frac{t - \frac{\xi}{k}}{1 + (\frac{\xi}{k} - t)^2} - \kappa k^2 (1 + (\frac{\xi}{k} - t)^2)) j(k) + \alpha k w(k),$$
  

$$\partial_t w(k-1) = -\alpha (k-1) j(k-1) + c \frac{\xi}{k^2} \frac{1}{1 + (\frac{\xi}{k} - t)^2} w(k),$$
  

$$\partial_t j(k-1) = -\kappa \frac{\xi^2}{k^2} j(k-1) + \alpha (k-1) w(k-1).$$
  
(3.17)

Then for initial data  $w(k, t_k) = 1$  and  $w(k - 1, t_k) = j(k, t_k) = j(k - 1, t_k) = 0$ we estimate

$$(|w(k)| + |w(k-1)| + \alpha k|j(k)| + \alpha (k-1)|j(k-1)|)|_{t=t_{k-1}} \le 2\pi c \frac{\xi}{k^2}.$$

Furthermore, this bound is attained up to a loss of constant in the sense that

$$|w(k-1,t_{k-1})| \ge \frac{\pi}{2}c\frac{\xi}{k^2}.$$

Proof of Lemma 3.7. We perform a shift in time such that  $t = \frac{\xi}{k} + s$  and thus  $s_0 := -\frac{\xi}{2} \frac{1}{k^2+k} \le s \le \frac{\xi}{2} \frac{1}{k^2-k} =: s_1$ . Integrating the equations in time, for our choice of initial data we obtain that

$$j(s,k) = \alpha k \int_{s_0}^t \frac{1+s^2}{1+\tau_2^2} \exp(-\kappa k^2 (s-\tau_2 + \frac{1}{3}(s^3 - \tau_2^3))) w(k,\tau_2) d\tau_2$$

and thus

$$w(k,s) = 1 - \alpha k \int j(\tau_1,k) d\tau_1$$
  
=  $1 - \alpha^2 k^2 \int_{s_0}^s \int_{s_0}^{\tau_1} \frac{1 + \tau_1^2}{1 + \tau_2^2} \exp(-\kappa k^2 (\tau_1 - \tau_2 + \frac{1}{3}(\tau_1^3 - \tau_2^3))) w(k,\tau_2) d\tau_2 d\tau_1.$ 

For the second term we insert  $\alpha^2 = \frac{\kappa}{\beta}$  and deduce that

$$\begin{split} \frac{\kappa}{\beta}k^2 \int_{s_0}^s \int_{s_0}^{\tau_1} \frac{1+\tau_1^2}{1+\tau_2^2} \exp(-\kappa k^2(\tau_1-\tau_2+\frac{1}{3}(\tau_1^3-\tau_2^3)))w(k,\tau_2)d\tau_2d\tau_1 \\ &= \frac{1}{\beta} \int_{s_0}^s \frac{1}{1+\tau_2^2} \int_{\tau_2}^s \kappa k^2(1+\tau_1^2)\exp(-\kappa k^2(\tau_1-\tau_2+\frac{1}{3}(\tau_1^3-\tau_2^3)))w(k,\tau_2)d\tau_1d\tau_2 \\ &= \frac{1}{\beta} \int_{s_0}^s \frac{w(k,\tau_2)}{1+\tau_2^2} \left[\exp(-\kappa k^2(\tau_1-\tau_2+\frac{1}{3}(\tau_1^3-\tau_2^3)))\right]_{\tau_1=\tau_2}^{\tau_1=s} d\tau_2 \\ &= \frac{1}{\beta} \int_{s_0}^s \frac{w(k,\tau_2)}{1+\tau_2^2} (1-\exp(-\kappa k^2(s-\tau_2+\frac{1}{3}(s^3-\tau_2^3))))d\tau_2. \end{split}$$

This further yields that

$$w(k,s) = 1 - \frac{1}{\beta} \int_{s_0}^{s} d\tau_2 \frac{w(k,\tau_2)}{1+\tau_2^2} (1 - \exp(-\kappa k^2 (s - \tau_2 + \frac{1}{3} (s^3 - \tau_2^3)))). \quad (3.18)$$

Therefore, if  $1 \ge w(k, s) \ge 0$  we obtain

$$|w(k,s) - 1| \le \frac{1}{\beta} \int_{s_0}^s \frac{1}{1 + \tau_2^2} d\tau_2 \le \frac{1}{\beta} (\arctan(s) + \frac{\pi}{2}).$$

and by bootstrap this assumption holds for all times if  $\beta \ge \pi$ . For the current j(k) we similarly estimate

$$\begin{split} \int_{s_0}^{s_1} \frac{1+s^2}{1+\tau_2^2} \exp(-\kappa k^2 (s-\tau_2 + \frac{1}{3}(s^3 - \tau_2^3))) w(k,\tau_2) d\tau_2 \\ &\leq (\int_{s_0}^{\frac{\xi}{5k^2}} + \int_{\frac{\xi}{5k^2}}^{s_1}) \exp(-\kappa k^2 (s-\tau_2 + \frac{1}{3}(s^3 - \tau_2^3))) d\tau_2 \\ &\leq \frac{1}{\kappa k^2} (\exp(-\kappa \xi (\frac{1}{3} + 3^{-4}\frac{\xi^2}{k^4})) + \frac{4}{\eta^2}) \leq \frac{c}{\kappa k^2}, \end{split}$$

which yields

$$\alpha k j(s,k) \le (\alpha k)^2 \frac{c}{\kappa k^2} = \frac{c}{\beta}.$$

Concerning the k-1 mode we argue similarly and write

$$j(k-1) = \alpha(k-1) \int \exp(\kappa \frac{\xi^2}{k^2}(s-\tau))w(k-1)d\tau$$

and

$$w(k-1) = c\frac{\xi}{k^2} \int \frac{1}{1+\tau^2} w(k) d\tau - \alpha(k-1) \int j(k-1) d\tau_1$$
  
=  $c\frac{\xi}{k^2} \int \frac{1}{1+\tau^2} w(k) d\tau$   
-  $\alpha^2(k-1)^2 \iint \exp(\kappa \frac{\xi^2}{k^2} (\tau_1 - \tau_2)) w(\tau_2, k-1) d\tau_2 d\tau_1$   
=  $c\frac{\xi}{k^2} \int_{s_0}^{s_1} \frac{1}{1+\tau^2} w(k) d\tau - \frac{\alpha^2 k^2 (k-1)^2}{\kappa \xi^2} \int d\tau_2 w(\tau_2, k-1).$ 

Since

$$\left|\frac{\alpha^2 k^2 (k-1)^2}{\kappa \xi^2} \int d\tau_2\right| \le \frac{1}{\beta \frac{\xi}{k^2}}$$

we deduce by bootstrap that

$$|w(k-1)| \le 2\pi c \frac{\xi}{k^2}$$

and thus

$$\begin{split} |w(k-1) - \pi c_{k^2}^{\xi}| &\leq c_{k^2}^{\xi} \int_{\tau \notin [s_0, s_1]} \frac{1}{1 + \tau^2} d\tau \\ &+ c_{k^2}^{\xi} \frac{1}{\beta} \int_{s_0}^{s_1} \frac{1}{1 + \tau^2} (\arctan(\tau) - \frac{\pi}{2}) d\tau + \frac{1}{\beta \frac{\xi}{k^2}} 2\pi c_{k^2}^{\xi} \\ &\leq \pi c_{k^2}^{\xi} (\frac{2}{\pi} \frac{k^2}{\xi} + \frac{\pi}{2\beta} + \frac{2}{\beta \frac{\xi}{k^2}}) \\ &\leq \frac{\pi}{2} c_{k^2}^{\xi} \end{split}$$

and

$$\begin{aligned} \alpha(k-1)j(k-1) &\leq 2\pi c \frac{\xi}{k^2} (\alpha(k-1))^2 \int \exp(\kappa \frac{\xi^2}{k^2} (s-\tau)) d\tau \\ &= 2\pi c \frac{\xi}{k^2} (\alpha(k-1))^2 \frac{1}{\kappa \frac{\xi^2}{k^2}} \\ &= \pi c \frac{1}{\beta \xi} \ll \pi c \frac{\xi}{k^2}. \end{aligned}$$

	_	_	_	

Based on this model we may thus expect that a repeated interaction or chain of resonances starting at  $k_0$ 

$$k_0 \mapsto k_0 - 1 \mapsto \cdots \mapsto 1$$

results in a possible growth

$$|w(1,t_1)| \ge |w(k_0,t_{k_0})| \prod_{k=1}^{k_0} C'(1+c\frac{\xi}{k^2}),$$

where  $C' = C'(\beta)$ . Choosing  $k_0 \approx \sqrt{C'c\xi}$  to maximize this product and using Stirling's approximation formula we may estimate this growth by an exponential factor:

$$\prod_{k=1}^{k_0} C' c \frac{\xi}{k^2} = \frac{(C' c \xi)^{k_0}}{(k_0!)^2} \approx \exp(\sqrt{C' c \xi})$$

This suggests that stability can only be expected if the initial decays in Fourier space with such a rate, which is naturally expressed in terms of Gevrey spaces.

**Definition 3.8.** Let  $s \ge 1$ , then a function  $u \in L^2(\mathbb{T} \times \mathbb{R})$  belongs to the Gevrey class  $\mathcal{G}_s$  if its Fourier transform satisfies

$$\sum_{k} \int \exp(C|\xi|^{1/s}) |\mathcal{F}(u)(k,\xi)|^2 d\xi < \infty$$

for some constant C > 0.

In view of the more prominent role of the frequency with respect to y and for simplicity of notation this definition only includes  $|\xi|^{1/s}$  as opposed to  $(|k| + |\xi|)^{1/s}$  in the exponent. All results in this article also extend to more general Fourier weighted spaces X (see Definition 3.9) with respect to x with norms

$$\sum_{k} \int \exp(C|\xi|^{1/s}) \lambda_k |\mathcal{F}(u)(k,\xi)|^2 d\xi.$$
(3.19)

We remark that any Gevrey function is also an element of  $H^N$  for any  $N \in \mathbb{N}$  and that Gevrey classes are nested with the strongest constraint, s = 1, corresponding to analytic regularity with respect to y.

As the main result of this article and as summarized in Theorem 3.3 we show that the above heuristic model's prediction is indeed accurate and that the optimal regularity class for the (simplified) linearized MHD equations around a traveling wave are given by Gevrey 2.

# 3.2.3 Magnetic Dissipation, Coupling and the Influence of $\beta$

In the preceding proof we have seen that the interaction of interaction of w(k)and j(k) is determined by the combination of the action of the underlying magnetic field of size  $\alpha$  and magnetic resistivity  $\kappa > 0$  through the parameter

$$\beta = \frac{\kappa}{\alpha^2}.$$

More precisely, we recall that ignoring the influence of neighboring modes w(k) and j(k) are solutions of a coupled system:

$$\partial_t w(k) = -\alpha k j(k), \partial_t j(k) = \left(2 \frac{t - \frac{\xi}{k}}{1 + (\frac{\xi}{k} - t)^2} - \kappa k^2 (1 + (\frac{\xi}{k} - t)^2)) j(k) + \alpha k w(k). \right)$$

Hence starting with data  $w(k, s_0) = 1, j(k, s_0) = 0$  three different mechanisms interact to determine the size of w(k, s):

- The vorticity w(k, s) by means of the constant magnetic field generates a current perturbation j(k, s).
- The current perturbation j(k, s) is damped by the magnetic resistivity.
- The current j(k, s) in turn by means of the constant magnetic field acts on the vorticity and damps it.

In this system several interesting regimes may arise, which are distinguished by the parameter  $\beta$ .

In the limit of infinite dissipation,  $\beta \to \infty$ , the current is rapidly damped and the system hence formally reduces to the Euler equations

$$\partial_t w(k) = 0,$$
  
$$j(k) = 0,$$

where w(k, s) remains constant in time.

As the opposite extremal case, if  $\beta \downarrow 0$  we obtain the inviscid MHD equations and the system

$$\partial_s w(k) = -\alpha k j(k),$$
  
$$\partial_s j(k) = 2 \frac{s}{1+s^2} j(k) + \alpha k w(k).$$

Hence at least for |s| large this suggests that

$$w(k) \approx c_1(1+s)\cos(\alpha ks), \quad j(k) \approx c_1(1+s)\sin(\alpha ks).$$

In particular, in stark contrast to the Euler equations (i.e.  $\alpha = 0$ ) for the inviscid MHD equations with a magnetic field the vorticity w(k) and current perturbations j(k) cannot be expected to remain close to 1 and 0, respectively.

This article considers the regime  $0 < \beta < \infty$ , where the interaction of both extremal phenomena results in behavior which is qualitatively different from both limiting cases. Indeed, recall that by a repeated application of Duhamel's formula w(k, s) satisfies the integral equation (3.18):

$$w(k,s) = 1 - \frac{1}{\beta} \int_{s_0}^s \frac{w(k,\tau_2)}{1+\tau_2^2} (1 - \exp(-\kappa k^2 (s - \tau_2 + \frac{1}{3} (s^3 - \tau_2^3)))) d\tau_2$$

Hence, as a first case which we also discussed in the toy model of Lemma 3.7, if we restrict to  $\beta \ge \pi$  then the integral term is bounded and small

$$\frac{1}{\beta} \int_{s_0}^s d\tau_2 \frac{1}{1 + \tau_2^2} \le 1.$$

Hence, for large  $\beta$  the integral term can be treated as a perturbation and w(k, s) remains comparable to 1 uniformly in s and thus close to the Euler case. However, unlike for the Euler equations the evolution of the current remains non-trivial.

If instead  $0 < \beta < \pi$  we obtain different behaviour depending on the dissipation  $\kappa k^2$ , the size of the magnetic field and the frequencies considered, whose interaction determines the behavior of the solution. More precisely, considering the integrand

$$\frac{1}{1+\tau_2^2}(1-\exp(-\kappa k^2(s-\tau_2+\frac{1}{3}(s^3-\tau_2^3))))),$$

we observe that for  $\kappa k^2 \gg 1$  large the magnetic dissipation is very strong and hence the integrand is well-approximated by  $\frac{1}{1+\tau_2^2}$ . In particular, this suggests that for these s it holds that

$$\begin{split} w(k,s) &\approx 1 - \frac{1}{\beta} \int_{s_0}^s d\tau_2 \frac{w(k,\tau_2)}{1+\tau_2^2}, \\ \Leftrightarrow & \partial_s w(k,s) \approx -\frac{1}{\beta} \frac{1}{1+s^2} w(k,s), \\ \Leftrightarrow & w(k,s) \approx \exp(-\frac{1}{\beta} (\arctan(s) + \frac{\pi}{2})) w(k,s_0), \end{split}$$

and hence w(k, s) might decay by a factor comparable to  $\exp(-\frac{\pi}{\beta})$ .

If instead  $\kappa k^2 \leq 1$  is small, different effects interact and involve the following natural time scales:

- Mixing enhanced magnetic dissipation becomes relevant on time scales  $(\kappa k^2)^{-1/3} \gg 1.$
- The resonant interval  $I^k$  is of size about  $\frac{\xi}{k^2}$ .
- Within this resonant interval most of the  $L^1$  norm of  $\frac{1}{1+\tau_2^2}$  is achieved on a much smaller sub-interval of size about 1.

Hence, for times  $|s| < s^* \ll (\kappa k^2)^{-1/3}$  which are small compared to the disspation time scale the integrand is small and we may therefore expect that

$$w(k,s) \approx 1$$

remains constant. If we instead consider very large times  $|s| \gg (\kappa k^2)^{-1/3} \gg s^*$  in view of the exponential factor and the decay of  $\frac{1}{1+\tau_s^2}$  the size of w(k,s) should largely be determined by the action of the time interval  $(-s^*, s^*)$ , that is

$$w(k,s) \approx 1 - \frac{1}{\beta} \int_{-s^*}^{s^*} \frac{1}{1 + \tau_2^2} d\tau_2$$
$$\approx 1 - \frac{\pi}{\beta},$$

provided such such s exist, that is if the size  $\frac{\xi}{k^2}$  of  $I^k$  is much bigger than the dissipative time scale. In particular, the size of w(k,s) transitions from being close to 1 for  $|s| < s^*$  to being very far from 1 for  $|s| \gg (\kappa k^2)^{-1/3}$  and further needs to be controlled on intermediate time scales. These different regimes all have to be considered in the upper and lower bounds of Section 3.4 and we in particular need to control the size of w(k,s) in order to estimate the resulting norm inflation due to resonances. For this purpose we estimate w(k,s) in terms of a growth factor L such that

$$|w(k,s)| \le Lw(k,s_0),$$

as we discuss in 3.5. For our upper bounds we will require that  $cL \ll 1$  is sufficiently small to control back-coupling estimates.

### 3.3 Stability for Small and Large Times

In this section we establish some general estimates on the (simplified) linearized MHD equations (3.16). We note that these equations decouple with respect to  $\xi$ . In the following we hence treat  $\xi$  as an arbitrary but fixed parameter of the equations and consider (3.16) as an evolution equation for the sequences  $w(\cdot, \xi, t)$  and  $j(\cdot, \xi, t)$ . As mentioned following the statement of Theorem 3.3 in addition to  $\ell^2(\mathbb{Z})$  all proofs in the remainder of the article hold for a rather general family of weighted spaces:

**Definition 3.9.** Consider a weight function  $\lambda_l > 0$  such that

$$\sup_{l} \frac{\lambda_{l\pm 1}}{\lambda_{l}} =: \hat{\lambda} < 10$$

Then we define the Hilbert space X associated to this weight function as the set of all sequences  $u : \mathbb{Z} \to \mathbb{C}$  such that  $(u_l \lambda_l)_l \in \ell^2$ .

This definition for instance includes  $\ell^2$  ( $\lambda_l = 1$ ), (Fourier transforms of) Sobolev spaces  $H^s$  ( $\lambda_l = 1 + C|l|^{2s}$  with C > 0 sufficiently small) or Gevrey regular or analytic functions with a suitable radius of convergence.

As sketched in Section 3.2.2 for a given frequency  $\xi \in \mathbb{R}$  we expect the norm inflation for evolution by (3.16) to be concentrated around times  $t_k \approx \frac{\xi}{k}$  for suitable  $k \in \mathbb{Z}$ . In particular, if the time is too large,  $t > 2\xi$ , there exists no such k and we expect the evolution to be stable. Similarly, if t is small also the size of the resonance predicted by the toy model is small and we again expect the evolution to be stable. The results of this section show that this heuristic is indeed valid and establish stability for "small" and "large" times. The essential difficulty in proving Theorem 3.3 thus lies in control the effects of resonances in the remaining time intervals, which are studied in Section 3.4. In the following we will often write  $L_t^{\infty}$  as the supremum norm till time t.

**Lemma 3.10** (Large time). Consider the equation (3.16) on the time interval  $(2\xi, \infty)$ . Then the possible norm inflation is controlled uniformly in time

$$||w, j||_X(t) \le \frac{1}{1-4c} \frac{1}{1-2c\lambda} ||w, j||_X(2\xi).$$

where  $\hat{\lambda} = \max_{l} \frac{\lambda_{l}}{\lambda_{l\pm 1}}$  is as in Definition 3.9.

*Proof.* Let  $\hat{w}(k) = |(w(k), j(k))|$ . Then we infer

$$\frac{1}{2}\partial_t \hat{w}^2(k) \le (a(k-1)w(k-1) - a(k+1)w(k+1))w(k) + b(k)j(k)^2,$$

where we introduced the short-hand notation a, b for the coefficient functions. Since  $b(t, k) \leq 0$  for  $t \geq 2\xi$ , we further deduce that

$$\frac{1}{2}\partial_t \hat{w}^2(k) \le c \frac{\xi}{1+(t-\xi)^2} (\hat{w}(k+1) + \hat{w}(k-1) + 2\hat{w}(k))\hat{w}(k)$$
  
$$\rightsquigarrow \ \hat{w}^2(k,t) < \hat{w}^2(k,2\xi) + 2c(|\hat{w}^2(k)|_{L^{\infty}_t} + \frac{1}{2}|\hat{w}^2(k+1)|_{L^{\infty}_t} + \frac{1}{2}|\hat{w}^2(k-1)|_{L^{\infty}_t}).$$

Hence by a bootstrap argument we control

$$\hat{w}^2(k,t) \le \frac{1}{1-4c} \sum_l (2c)^{|k-l|} \hat{w}^2(l,2\xi).$$

Summing this estimate with the weight  $\lambda_k$  then concludes the proof:

$$|w, j\|_{X}(t) \leq \frac{1}{1-4c} \sqrt{\sum_{k} \lambda_{k} \sum_{l} (2c)^{|k-l|} \hat{w}^{2}(l, 2\xi)}$$
$$\leq \frac{1}{1-4c} \sqrt{\sum_{l} \lambda_{l} \hat{w}^{2}(l, 2\xi) \sum_{k} (2c\hat{\lambda})^{|k-l|}}$$
$$\leq \frac{1}{1-4c} \frac{1}{1-2c\lambda} \|w, j\|_{X}(2\xi)$$

Thus it suffices to study the evolution for times  $t < 2\xi$ . In view of the estimates of Section 3.2.2 it here is convenient to partition  $(0, 2\xi)$  into intervals where  $t \approx \frac{\xi}{k}$  for some  $k \in \mathbb{Z}$ .

**Definition 3.11.** Let  $\xi > 0$  be given, then for any  $k \in \mathbb{N}$  we define

$$t_k = \frac{1}{2} \left( \frac{\xi}{k+1} + \frac{\xi}{k} \right) \text{ if } k > 0,$$
  
$$t_0 = 2\xi.$$

We further define the time intervals  $I^k = (t_k, t_{k-1})$ , for  $\xi < 0$  we define  $t_k$ analogously for  $-k \in \mathbb{N}$ .

Note that

$$t_k < \frac{\xi}{k} < t_{k-1}$$

and

$$t_{k-1} - t_k = \frac{1}{2} \left( \frac{\xi}{k+1} - \frac{\xi}{k-1} \right) = \frac{\xi}{k^2 - 1}.$$

Hence  $I_k$  is an interval containing the time of resonance  $\frac{\xi}{k}$  and is of size about  $\frac{\xi}{k^2}.$ 

The next lemma provides a very rough energy-based estimate, which will allow us to control the evolution for small times and frequencies. That is, we show that it is easy to obtain a energy estimate with  $\xi t$  in the exponent. If the time t or the frequency  $\xi$  are small this rough estimate is sufficient. However, for Gevrey 2 norm estimates it will be necessary to improve this control to a  $C\sqrt{\xi}$  term in the exponent in subsequent estimates. Furthermore, we remark that also the magnetic part needs to be handled adequately, since it may give an additional growth by  $\exp(\frac{4}{3}\kappa^{-\frac{1}{2}})$ .

**Lemma 3.12** (Rough estimate). Consider a solution of (3.16), then for fixed  $\xi$  and for all times t > 0 it holds that

$$|w, j||_X(t) \le \exp(\frac{4}{3}\kappa^{-\frac{1}{2}})\exp((1+\hat{\lambda})c\xi t)||w, j||_X(0).$$

*Proof.* We define  $\hat{w}(k) = |w, j|(k)$ , then

$$\frac{1}{2}\partial_t \hat{w}^2(k) = (a(k+1)w(k+1) - a(k-1)w(k-1))w(k) + b(k)j(k)^2$$

with  $b(k,t) = (2\frac{t-\frac{\xi}{k}}{1+(\frac{\xi}{k}-t)^2} - \kappa k^2(1+(\frac{\xi}{k}-t)^2))$ . We further define the define the growth factor

$$M(k,t) = \begin{cases} 1 & \text{if } t - \frac{\xi}{k} \le 0 \text{ or } \kappa k^2 \ge 1, \\ \frac{1}{1 + (\frac{\xi}{k} - t)^2} & \text{if } 0 \le t - \frac{\xi}{k} \le (\frac{2}{\kappa k^2})^{\frac{1}{3}} \text{ and } \kappa k^2 \le 1, \\ \frac{1}{1 + (\frac{2}{\kappa k^2})^{\frac{1}{3}}} & \text{if } (\frac{2}{\kappa k^2})^{\frac{1}{3}} \le t - \frac{\xi}{k} \text{ and } \kappa k^2 \le 1. \end{cases}$$

We note that this weight satisfies  $b(k,t) + \frac{M'}{M}(k,t) \leq 0$ . Hence, defining the energy

$$E = (\prod_{l} M(l,t))^2 \sum_{k} \lambda_k \hat{w}(k,t)^2,$$

we deduce that

$$\begin{split} \frac{1}{2}\partial_t E &\leq \left(\prod_l M(l,t)\right)^2 \sum_k \lambda_k \left(a(k+1)\hat{w}(k+1,t) + a(k-1)\hat{w}(k-1,t)\right)\hat{w}(k) \\ &\leq (\prod_l M(l,t))^2 \sum_k (\lambda_k a(k) + \frac{a(k-1)\lambda_{k-1} + a(k+1)\lambda_{k+1}}{2})\hat{w}^2(k,t) \\ &= (1+\hat{\lambda})c\xi E. \end{split}$$

Applying Gronwall's inequality thus yields

$$E(t) \le \exp(2(1+\hat{\lambda})c\xi t)E(0)$$

This in turn leads to the estimate

$$\begin{split} \|w, j\|_X &\leq \exp((1+\hat{\lambda})c\xi t) \prod_l \|M(l,t)\|^{-1} \|w_0, j_0\|_X \\ &\leq \exp((1+\hat{\lambda})c\xi t) \prod_{l=0}^{\kappa^{-\frac{1}{2}}} (1+(\frac{2}{\kappa l^2})^{\frac{1}{3}}) \|w_0, j_0\|_X \end{split}$$

Finally, we can use Stirling's approximation of the factorial, which results in the desired estimate:

$$\|w, j\|_X(t) \le \exp(\frac{4}{3}\kappa^{-\frac{1}{2}})\exp((1+\hat{\lambda})c\xi t)\|w_0, j_0\|_X.$$

In the following we establish upper and lower bounds for small times. Here we use that for modes k such that  $\frac{\xi}{k^2}$  is small nay possible resonance will not produce large enough norm inflation and the evolution can hence be treated perturbatively. More precisely, we consider the evolution on the time interval

$$I = [0, \frac{\xi}{2} (\frac{1}{k_0} + \frac{1}{k_0 - 1})]$$

for fixed  $k_0$  to be determined later. For this purpose we introduce the parameter  $\eta_0 := \frac{\xi}{k_0^2}$  which later will be chosen as  $\eta_0 \approx \frac{1}{10c}$ .

**Lemma 3.13.** Let w, j be a solution of (3.16), define  $d := c^{-1}$  and let  $\xi, k_0$  be such that  $\eta_0 \leq d^2$ . Then for all times  $0 \leq t \leq t_{k_0}$  it holds that

$$||w(t), j(t)||_X^2 \le \exp(2(1+\hat{\lambda})\max(c\eta_0, 1)\sqrt{\xi\eta_0})||w_0, j_0||_X^2$$
  
$$\le \exp(C\sqrt{\xi})||w_0, j_0||_X^2.$$

Furthermore, suppose that  $k_0 \ge \kappa^{-\frac{1}{2}}$  and  $10d \le \frac{\xi}{k_0^2} \le \frac{1}{100c^2}$ , then for the initial data  $w(k,0) = \delta_{k_0,k}$  and j(k,0) = 0 we obtain that

$$w(k_0, t_{k_0}) \ge \frac{1}{2} \max(1, w(k, t_{k_0}), j(k, t_{k_0}))$$
  
$$j(k_0, t_{k_0}) \le \frac{1}{\alpha_{k_0} \xi \eta_0}.$$

*Proof.* Computing the time derivative, we obtain

$$\frac{1}{2}\partial_t \|w, j\|_X^2 = \sum_l (a(l+1)w(l+1) + a(l-1)w(l-1))\lambda_l w(l) + b(l)\lambda_l j(l)^2,$$

where the coefficient functions satisfy

$$a(l) \leq \begin{cases} c\eta_0 & l \geq k_0\\ 4c\frac{1}{1+\eta_0} & l \leq k_0\\ \leq \max(c\eta_0, 4c),\\ b(l) \leq 1. \end{cases}$$

Therefore, we conclude that

$$\partial_t \|w, j\|_X^2 \le 2(1+\hat{\lambda}) \max(c\eta_0, 1) \|w, j\|_X^2, \|w, j\|_X^2(t) \le \exp(2(1+\hat{\lambda}) \max(c\eta_0, 1)t) \|w_0, j_0\|_X^2, \|w, j\|_X^2(t_{k_0}) \le \exp(2(1+\hat{\lambda}) \max(c\eta_0, 1)\sqrt{\xi\eta_0}) \|w_0, j_0\|_X^2.$$

To prove lower bounds on the norm inflation we further need to show that for  $w(k,0) = \delta_{k_0,k}$  and j(k,0) = 0, the mode  $w(k_0,t_{k_0})$  will stay the largest mode. Therefore, we introduce the short-hand notation

$$\hat{w}(k,t) = |w,j|(k,t)$$

and have to estimate the growth of  $\hat{w}(\cdot, t)$ . Since on the interval  $[0, t_{k_0}]$  it holds that  $b(k) \leq 0$  as  $k_0 \geq \kappa^{-\frac{1}{2}}$ , we obtain the system

$$\partial_t \hat{w}(k) \le a(k+1)\hat{w}(k+1) + a(k-1)\hat{w}(k-1)$$
  
 $\hat{w}(k,0) = \delta_{k,k_0}.$ 

Let  $\sqrt{\xi\pi} = l_0 \ge l \ge k_0$  to be fixed later. We want to prove by induction that

$$\hat{w}(m, t_{l-1}) \leq 6\pi c \eta_0 (2c)^{|m-k_0|} 
\hat{w}(l, t_{l-1}) \leq 4(2c)^{|l-k_0|} 
\hat{w}(n, t_{l-1}) \leq 2(2c)^{|m-k_0|}$$
(3.20)

for all m > l > n. Induction start: We integrate a in time to estimate

$$\int_{0}^{t_{l_{0}-1}} a(k) = c \frac{\xi}{k^{2}} \int_{0}^{t_{l_{0}-1}} \frac{1}{1 + (\frac{\xi}{k} - t)^{2}} \\ \leq \begin{cases} \pi c \frac{\xi}{k^{2}} & k > l_{0} \\ c & k \le l_{0} \\ \le c. \end{cases}$$

Thus we obtain that

$$\hat{w}(k, t_{l_0-1}) < \delta_{k_0, k} + c(|\hat{w}(k+1)|_{L_t^{\infty}} + |\hat{w}(k-1)|_{L_t^{\infty}})$$

which by a bootstrap argument yields that

$$\hat{w}(k, t_{l_0-1}) \le \frac{1}{1-2c} (2c)^{|k_0-k|}$$

for all k which satisfy (3.20).

**Induction step:** We fix l and we assume that (3.20) holds for all  $\tilde{l}$  with  $l_0 \geq \tilde{l} \geq l+1 \geq k_0+1$  and then prove that it holds also for l. We here argue by bootstrap. That is, we show that the estimate (3.20) at least holds up until a time  $t^*$  with  $t_l \leq t^* \leq t_{l-1}$  and that the maximal time with this property is given by  $t^* = t_{l-1}$ . For n < l we estimate

$$\hat{w}(n, t_{l-1}) \leq \delta_{n,k_0} + \int_0^{t_{l-1}} a(n \pm 1, \tau) w(n \pm 1, \tau)$$
  
$$\leq \delta_{n,k_0} + c(4(2c)^{|n+1-k_0|} + 2(2c)^{|n-1-k_0|})$$
  
$$< 2(2c)^{|n-k_0|}.$$

To estimate the l mode we estimate the integral between  $t_l$  and  $t_{l-1}$  to deduce

$$\hat{w}(l, t_{l-1}) \leq \hat{w}(l, t_l) + \int_{t_l}^{t_{l-1}} a(l \pm 1, \tau) w(l \pm 1, \tau)$$
  
$$\leq 2(2c)^{l-k_0} + 6\pi c \eta_0 (2c)^{l+1-k_0} + 2c(2c)^{|l-1-k_0|}$$
  
$$\leq 4(2c)^{l-k_0}.$$

For m > l we split the integrals as

$$\hat{w}(m, t_{l-1}) \leq \hat{w}(m, t_{m-1}) + \int_{t_{m-1}}^{t_l} a(m+1, t) \hat{w}(m+1, \tau) + \int_{t_{m-1}}^{t_{m-2}} a(m-1, t) \hat{w}(m-1, \tau) + \int_{t_{m-2}}^{t_l} a(m-1, t) \hat{w}(m-1, \tau) \leq 4(2c)^{m-k_0} + 12\pi c^2 \eta_0 (2c)^{|m-k_0|} + 4\pi \eta_0 (2c)^{m-k_0} + 6\pi c \eta_0 (2c)^{m-k_0} \leq 6\pi \eta_0 (2c)^{m-k_0}.$$

So we finally deduce that

$$\hat{w}(k, t_{k_0}) \le (2c)^{|k-k_0|} \begin{cases} 4 & k \le k_0 + 1\\ 6\pi\eta_0 & k > k_0 + 1 \end{cases}$$

Thus we established an upper bound for all modes, the next step is to show that for w indeed the  $k_0$  mode is one of the largest modes. Therefore, we estimate  $j(k_0)$  by

$$j(k_0,t) = \alpha k_0 \int_0^t d\tau \; \frac{1 + (\frac{\xi}{k_0} - t)^2}{1 + (\frac{\xi}{k_0} - \tau)^2} \exp\left(-\kappa k_0^2 (t - \tau + \frac{1}{3}((\frac{\xi}{k_0} - t)^3 - (\frac{\xi}{k_0} - \tau)^3)\right) w(k_0,\tau)$$

and hence obtain that

$$j(t) \le \alpha k_0 \int \exp(-\kappa k_0^2 \eta^2 (t-\tau))$$
$$\le \frac{1}{\sqrt{\beta \kappa \xi \eta_0^3}} = \frac{1}{\alpha_{k_0} \xi \eta_0}$$

and

$$\begin{aligned} \alpha k_0 \int j(k_0, \tau_2) d\tau_1 &= \frac{\kappa k_0^2}{\beta} \int_0^t d\tau_1 \int_0^{\tau_1} d\tau_2 \ \frac{1 + (\frac{\xi}{k_0} - \tau_1)^2}{1 + (\frac{\xi}{k_0} - \tau_2)^2} \\ &\times \exp\left(-\kappa k_0^2 (\tau_1 - \tau_2 + \frac{1}{3}((\frac{\xi}{k_0} - \tau_1)^3 - (\frac{\xi}{k_0} - \tau_2)^3)\right) \\ &\leq \frac{1}{\beta} \int d\tau_1 \frac{1}{1 + (\frac{\xi}{k_0} - \tau_1)^2} \\ &\leq \frac{4}{\beta \eta_0}. \end{aligned}$$

With this we conclude that

$$|w(k_0, t_{k_0}) - 1| \le \int a(k_0 + 1)w(k_0 + 1) + a(k_0 - 1)w(k_0 - 1) + \alpha k_0 j(t)$$
  
$$\le 16\pi\eta_0 c^2 + 8c^2 + \frac{4}{\beta\eta_0} \le \frac{1}{2},$$

which in turn yields

$$|w(k_0, t_{k_0})| \ge \frac{1}{2} \ge \max_{k \ne k_0} (\hat{w}(k, t_{k_0}), |j(k_0, t_{k_0})|).$$

# 3.4 Resonances and Norm Inflation

Having discussed the evolution for small times and large times in Section 3.3 it remains to discuss the evolution on the interval

$$(t_{k_0}, 2\xi) = \bigcup_{1 \le k \le k_0} I^k$$

with  $I^k$  as in Definition 3.11.

Based on the heuristics of the toy model of Section 3.2.2 our aim here is to establish both upper lower and upper bounds on the norm inflation on each resonant interval  $I^k$ , where the resonant mode w(k) can possibly lead to a large growth of its neighboring modes  $w(k \pm 1)$ . In order to simplify notation we introduce the growth factor

$$L = L(\alpha, \kappa, k),$$

which estimates the maximal growth of w(k) due to its interaction with the current j(k), see 3.5. In particular, we show that L = 1 if  $\beta \ge \pi$  and if  $\beta < \pi$  we obtain an estimate  $L = L(\alpha, \kappa, k) \le \sqrt{c}$ . We define M and  $M_n$  as

$$M = \sum_{m} 10^{-|m|} (w + \frac{1}{\alpha_{k_m}} j)(k_m, s_0)$$
$$M_n = \sum_{m} 10^{-|m-n|+\chi} (w + \frac{1}{\alpha_{k_m}} j)(k_m, \tilde{s}_0)$$

where  $\chi = -|\operatorname{sgn}(m) - \operatorname{sgn}(n)|$ . We note that

$$\sum_{l\neq 0} \left(\frac{3}{\eta}\right)^{|k_l-k_0|} M_l \le \frac{3}{\eta} M.$$

With these notations the main results of this section are summarized in the following theorem:

**Theorem 3.14.** Let  $c \leq \min(10^{-3}\beta^{\frac{16}{3}}, 10^{-4})$ ,  $\xi \geq 10\kappa^{-1}(1+\beta^{-1})$  and  $\eta = \frac{\xi}{k^2} \geq 10d$  and  $t_k = \frac{\xi}{2}(\frac{1}{k} + \frac{1}{k+1})$ , then it holds that

$$||w, j||_X(t_{k-1}) \le 18\pi L\lambda(c\eta)^{\gamma} ||w, j||_X(t_k).$$

Furthermore, let  $\kappa_k \min(\beta, 1) \geq \frac{1}{c}$  and  $\beta \geq \frac{1}{5}$ 

$$w(k, t_k) \ge \frac{1}{2} \max(w(l, t_k), j(l, t_k)).$$
 (3.21)

Then  $w(k-1, t_{k-1})$  satisfies (3.21) with k replaced by k-1 and

$$|w(k-1, t_{k-1})| \ge \min(\beta, \pi)(c\eta)^{\gamma} w(k, t_k).$$

To prove the estimates of Theorem 3.14 it is convenient to rescale  $\tilde{j}(k) = \alpha_k j(k)$  in (3.16) to obtain

$$\begin{split} \partial_t w(k) &= -\tilde{j}(k) \\ &- c \frac{\xi}{(k+1)^2} \frac{1}{1 + (\frac{\xi}{k+1} - t)^2} w(k+1) \\ &+ c \frac{\xi}{(k-1)^2} \frac{1}{1 + (\frac{\xi}{k-1} - t)^2} w(k-1) \\ \partial_t \tilde{j}(k) &= (2 \frac{t - \frac{\xi}{k}}{1 + (\frac{\xi}{k} - t)^2} - \kappa k^2 (1 + (t - \frac{\xi}{k}^2)) \tilde{j}(k) + \frac{\kappa k^2}{\beta} w(k), \end{split}$$

where we used that  $\kappa = \beta \alpha^2$ . With respect to these unknowns the norm on our space X changes slightly

$$\|w, j\|_X^2 = \sum \lambda_k (w^2(k) + \frac{\beta}{\kappa k^2} \tilde{j}^2(k))$$
  
=:  $\|w, \tilde{j}\|_{\tilde{X}}^2$ .

In the following sections, with slight abuse of notation we omit writing the tilde symbols both for j and X.

Given a choice of time interval  $I_{k_0}$ , considering  $k_0$  as arbitrary but fixed (and unrelated to  $k_0$  of Section 3.3) we further introduce the relative frequencies

$$k_n := k_0 + n,$$

where  $n \in \mathbb{Z}_{>-k_0}$  and also shift our time variable

$$t = \frac{\xi}{k_0} + s.$$

Introducing the coefficient functions

$$a(k) = c\eta \frac{k_0^2}{(k)^2} \frac{1}{1 + (\eta \frac{k_0(k_0 - k)}{k} - s)^2},$$
  

$$b(k) = 2 \frac{(s - \eta \frac{k_0(k_0 - k)}{k})}{1 + (\eta \frac{k_0(k_0 - k)}{k} - s)^2} - \kappa_k (1 + (\eta \frac{(k_0 - k)k_0}{k} - s)^2),$$
(3.22)

the system (3.16) then reads

$$\partial_s w(k) = -j(k)$$

$$-a(k+1)w(k+1)$$

$$+a(k-1)w(k-1),$$

$$\partial_t j(k) = \frac{\kappa_k}{\beta} w(k) + b(k)j(k).$$
(3.23)

For later reference, we note that the coefficient function a satisfies the following estimates:

$$a(k_0) = c\eta \frac{1}{1+s^2},$$
  

$$a(k_{\pm 1}) \le 4\frac{c}{\eta},$$
  

$$a(k_n) \le \frac{c}{\eta},$$
  
(3.24)

for all  $|n| \ge 2$ .

Finally, in view of cancellations of -a(k-1) and +a(k+1) on any given time interval  $I^k$  it is convenient to work with the unknowns

$$u_1 = w(k_0), u_2 = w(k_1) - w(k_{-1}), u_3 = w(k_1) + w(k_{-1}).$$

We then consider (3.23) as a forced system for these three modes (and a separate equation for all other modes):

$$\partial_{s} \begin{pmatrix} u_{1} \\ u_{2} \\ u_{3} \\ j(k_{0}) \end{pmatrix} = \begin{pmatrix} 0 & -a_{1} & a_{2} & -1 \\ 2c\eta \frac{1}{1+s^{2}} & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \\ \frac{\kappa_{k}}{\beta} & 0 & 0 & \frac{2s}{1+s^{2}} - \kappa_{k}(1+s^{2}) \end{pmatrix} \begin{pmatrix} u_{1} \\ u_{2} \\ u_{3} \\ j(k_{0}) \end{pmatrix} + \begin{pmatrix} 0 \\ -a(k_{\pm 2})w(k_{\pm 2}) \mp j(k_{\pm 1}) \\ \mp a(k_{\pm 2})w(k_{\pm 2}) - j(k_{\pm 1}) \\ 0 \end{pmatrix} \tag{3.25}$$

where  $a_1 = \frac{1}{2}(a(k_1) + a(k_{-1}))$  and  $a_2 = \frac{1}{2}(a(k_1) - a(k_{-1}))$ .

The analysis of this system is split into multiple subsections, where we also split the time interval  $I^k$  as

 $I^{k_0} = [s_0, -d] \cup [-d, d] \cup [d, s_1] =: I_1 \cup I_2 \cup I_3,$ 

where  $s_0 = -\frac{\eta}{2} \frac{k_0 - 1}{k_0}$  and  $s_1 = \frac{\eta}{2} \frac{k_0 + 1}{k_0}$ . Similarly to the setting of the Euler equations [DZ21] here the interaction between growth and decay of various modes interacts to determine the over all norm inflation.

#### 3.4.1 Proof of Theorem 3.3

Before proceeding to the proof of Theorem 3.14, in this subsection we discuss how it can be used to establish Theorem 3.3. We split the proof into two auxiliary theorems.

**Theorem 3.15** (technical statement). Let  $c \leq \min(10^{-3}\beta^{\frac{16}{3}}, 10^{-4}), \xi \geq 10\kappa^{-1}(1+\beta^{-1})$  and  $\frac{\xi}{k^2} \geq 10d$ . Then there exists exists a constant  $C = C(\kappa, \alpha, c)$  such that for a fixed  $\xi$  we obtain

$$||w, j||_X(t,\xi) \le \exp(C\sqrt{\xi})||w, j||_X(0,\xi).$$

Furthermore, let  $\xi \geq 10^4 \frac{d^2}{\beta \kappa}$ ,  $\beta \geq \frac{1}{5}$ ,  $k_0 \approx \frac{c}{10}\sqrt{\xi}$  and  $k_1 \approx \frac{4}{\sqrt{\beta \kappa}}$ , then there exists a constant  $C^* = C^*(\kappa, \alpha, c)$  such that for initial data  $w(k, 0) = \delta_{k_0, k}$  and j(k, 0) = 0 we obtain

$$w(k_1, t) \ge \exp(\tilde{C}\sqrt{\xi}).$$

for  $t \in [t_{k_1} - 1, t_{k_1} + 1]$ .

Proof of Theorem 3.15. For fixed  $\xi$ , t and  $k_0 =: 10d\sqrt{\xi}$  we consider  $w(\cdot, \xi, t)$  as an element in X. On X we define the operator  $S_{\tau_1,\tau_2}: X \to X$  as the solution operator of (3.16) on  $[\tau_1, \tau_2]$ , i.e.

$$S_{\tau_1,\tau_2}[w(\cdot,\xi,\tau_1)] = w(\cdot,\xi,\tau_2) S_{\tau_1,\tau_2} \circ S_{\tau_2,\tau_3} = S_{\tau_1,\tau_3}.$$

By Lemma 3.13, Theorem 3.14 and Lemma 3.10 this  ${\cal S}$  then satisfies the following norm estimates:

$$\begin{split} \|S_{0,t_{k_0+1}}\|_{X \to X} &= \exp(C_1 \sqrt{\xi}), \\ \|S_{t_k,t_{k-1}}\|_{X \to X} &= 3\pi c(\frac{\xi}{k^2})^{\gamma}, \\ \|S_{t_1,t}\|_{X \to X} &= 2\frac{1}{1-c\sqrt{\lambda}}. \end{split}$$

Combining these estimates with Stirling's approximation formula we thus obtain the desired upper bound:

$$||S_{0,t}||_{X \to X} \le 2 \exp(C_1 \sqrt{\xi}) \prod_{k=1}^{k_0} 3\pi c(\frac{\xi}{k^2})^{\gamma} \le \exp(C\sqrt{\xi}).$$

Concerning the lower bound, we use first use Lemma 3.13 and then Theorem 3.14 to deduce that

$$w(k_0, t_{k_0}) \ge \frac{1}{2}$$
  
 $w(k-1, t_{k-1}) \ge (c\frac{\xi}{k^2})^{\gamma} \min(\beta, \pi) w(k, t_k)$ 

for  $\sqrt{\frac{c}{10}\xi} \approx k_0 \ge k \ge k_1 \approx \frac{4}{\sqrt{\beta\kappa}}$ . Thus, by again using Stirling's approximation, we conclude that

$$w(k_1, t_{k_1}) \ge \frac{1}{2} \prod_{k=k_1}^{k_2} (c\frac{\xi}{k^2})^{\gamma} \min(\beta, \pi)$$
$$\approx \exp(\tilde{C}\sqrt{\xi}).$$

**Theorem 3.16** (Stability and blow-up). Let  $c \leq \min(10^{-3}\beta^{\frac{16}{3}}, 10^{-4})$  and w, j be a solution to (3.16), then there exists a constant  $C = C(\kappa, \alpha, c)$  such that for all  $C_1 > C$  and initial data which satisfy

$$\int \exp(C_1\sqrt{\xi}) \|w_0, j_0\|_X^2(\xi) \ d\xi < \infty,$$

the solution remains Gevrey 2 regular in the sense that

$$\sup_{t} \int \exp(C_2\sqrt{\xi}) \|w, j\|_X^2(\xi, t) \ d\xi \le \tilde{C} \int \exp(C_1\sqrt{\xi}) \|w_0, j_0\|_X^2(\xi) \ d\xi$$

where  $C_2 = C_1 - C$  and  $\tilde{C} > 0$  is a universal constant.

Furthermore, additionally suppose that  $\beta \geq \frac{1}{5}$ , then there exist a constant  $0 < C^* < C$  and initial data  $w_0, j_0$  which satisfy

$$\int \exp(C^*\sqrt{\xi}) \|w_0, j_0\|_X^2(\xi) \ d\xi < \infty,$$

such that for a subsequence  $k_{n,1}$  the solution diverges in  $L^2$ :

$$||w(\cdot, t_{k_{n,1}})||_{L^2\ell^2} \to \infty$$

Proof of Theorem 3.16. The first part follows directly from Theorem 3.15. For the second part we fix  $\xi_1 = 10^4 \frac{d^2}{\beta\kappa}$  and define the sequence  $\xi_n = n\xi_1$  with the associated  $k_0^{\xi_n} \approx \frac{c}{10}\sqrt{\xi_n}$  and  $k_1 \approx \frac{4}{\sqrt{\beta\kappa}}$ . Note that the starting mode  $k_0^{\xi_n}$  is  $\xi_n$ dependent, but the final mode  $k_1$  is independent of  $\xi_n$ . Furthermore, let  $z_n(\xi)$ be a function in  $C^{\infty} \cap L^2$ , such that

$$\operatorname{supp} z_n(\cdot) \subset [\xi_n - 1, \xi_n + 1]$$
$$\int z_n(\xi)^2 d\xi = 1.$$

We then define the initial data

$$w(k,\xi,0) = \sum_{n=1}^{\infty} \frac{1}{n} z_n(\xi) \exp(-\frac{1}{2}C^*\sqrt{\xi}) \delta_{k_{\xi_n,0},k}$$

We observe that it satisfies the estimates

$$\|w(\cdot,\xi,0)\|_{l^2}^2 = \sum_{n=1}^{\infty} \frac{1}{n^2} z_n(\xi)^2 \exp(-C^*\sqrt{\xi}),$$
$$\int \exp(C^*\sqrt{\xi}) \|w(\cdot,\xi,0)\|_{l^2}^2 d\xi = 2\sum_{n=1}^{\infty} \frac{1}{n^2} = \frac{\pi^2}{3}.$$

Furthermore, by the norm inflation results for each mode at each time  $t_{k_{n,1}}$  we obtain that

$$||w(k_{n,1},\xi,t_{k_{n,1}})||_{l^2} \ge \frac{9}{10} \frac{1}{n^2} z(\xi,n) \exp((\tilde{C}-C^*)\sqrt{\xi}),$$

and integrating in  $\xi$  we conclude that

$$||w(\cdot, t_{k_{n,1}})||_{L^2l^2} \ge \frac{9}{10} \frac{1}{n^2} \exp((\tilde{C} - C^*)\sqrt{\xi_n}) \to \infty.$$

# **3.4.2** Asymptotic Behavior on the Intervals $I_1$ and $I_3$

In this section we consider the equation (3.25) on the outer intervals  $I_1 = [s_0, -d]$ and  $I_3 = [d, s_1]$ . Since a lot of calculations are similar on both intervals we in general write the interval as  $[\tilde{s}_0, \tilde{s}_1]$ , where we only need to distinguish the two cases on a few occasions and in the statement of the conclusions. For the interval  $I_1$  we prove the following proposition: **Proposition 3.17** (Interval  $I_1$ ). Let  $c < \max(10^{-4}, 10^{-1}\beta)$  and  $\xi \ge 10 \max(\kappa^{-1}(1+\beta^{-1}), k_0^2 d)$ . Then for a solution of (3.25) on the interval  $I_1$  the following estimates hold at the time d:

$$\begin{aligned} |u_1(d)| &\leq 2M(c\eta)^{-\gamma_2}, \\ |u_2(d)| &\leq 2M(c\eta)^{\gamma_1}, \\ |u_3|(-d) &\leq 2M_1, \\ |w(k_n, -d)| &\leq 2M_n, \\ |j|(k_0, -d) &\leq \frac{c}{\beta}(c\eta)^{-\gamma_2}M\inf(c, \kappa_{k_0}c^{-2}), \\ |j|(k_{\pm 1}, -d) &\leq \frac{4}{\beta\eta^2}M, \\ |j|(k_n, -d) &\leq \frac{4}{\beta\eta^2}M_n. \end{aligned}$$

If we additionally assume that

$$w(k_0, t_{k_0}) \ge \frac{1}{2} \sup_{l} (w(l, t_l), j(l, t_l)),$$
(3.26)

we obtain that

$$\begin{aligned} |u_{1}(-d) - (c\eta)^{-\gamma_{2}}u_{1}(s_{0})| &= 50cu_{1}(\tilde{s}_{0})(c\eta)^{-\gamma_{2}} \\ |u_{2}(-d)| &\leq 50cu_{1}(\tilde{s}_{0})(c\eta)^{\gamma_{1}}, \\ |u_{3}|(-d), |w|(k_{m}, -d) &\leq 2|u|(s_{0}) \quad for \ |m| \geq 2, \\ |j|(k_{m}, -d) &\leq \frac{4}{\eta}|u|(s_{0}) \quad for \ |m| \geq 1, \\ |j|(k_{0}, -d) &\leq \frac{2c^{2}}{\beta}|u|(s_{0})(c\eta)^{-\gamma_{2}}. \end{aligned}$$
(3.27)

The proof of this proposition is split into several lemmas and concludes at the end of this subsection. For the interval  $I_3$  in a first step we only establish asymptotic estimates. The final conclusion for interval  $I_3$  will be postponed to the proof of Theorem 3.14. On both intervals  $I_1$  and  $I_2$  the interaction of  $u_1$ and  $u_2$  is the main effect to be analyzed. Therefore, we consider the equations for  $u_1$  and  $u_2$  as an inhomogeneous linear system

$$\partial_s \left(\begin{array}{c} u_1\\ u_2 \end{array}\right) = \left(\begin{array}{c} 0 & -\frac{c}{\eta}\\ 2c\eta \frac{1}{s^2} & 0 \end{array}\right) \left(\begin{array}{c} u_1\\ u_2 \end{array}\right) + F, \tag{3.28}$$

where F is a force term. Equation (3.28) with F = 0 has a explicit homogeneous solution and we aim to show that (3.25) can be treated as a perturbation. In the following we denote  $\tilde{u}$  as the homogeneous solution of (3.28). Furthermore, we split the forcing as

$$F =: F_{all} = F_{3mode} + F_j + F_{u_3} + F_{j(k_0 \pm 1)} + F_{\tilde{w}}$$

where we define

$$F_{3mode} = \left(\frac{c}{\eta} - a_1\right)e_1u_2 - 2c\eta \frac{1}{s^2(s^2+1)}e_2u_1,$$

as the 3 mode forcing

$$F_j = -e_1 j(k_0)$$

as the  $k_0$ -th current forcing and

$$F_{u_3} + F_{j(k_0\pm 1)} + F_{\tilde{w}} = e_1 a_2 u_3 \mp e_2 j(k_{\pm 1}) - e_2 a(k_{\pm 2}) w(k_{\pm 2})$$

as the forcings due to  $u_3$ ,  $j(k_0 \pm 1)$  and  $\tilde{w}$ , respectively. The corresponding  $R[F_*]$  are the called r changes. We also define  $\gamma = \sqrt{1 - 8c^2}$  and  $\gamma_1 = \frac{1}{2}(1 + \gamma)$  and  $\gamma_2 = \frac{1}{2}(1 - \gamma)$  and note the following equalities:

$$\begin{split} \gamma_1 \gamma_2 &= 2c^2, \\ \gamma_1 + \gamma_2 &= 1, \\ \gamma &= 1 + \mathcal{O}(c^2), \\ \gamma_1 &= 1 + \mathcal{O}(c^2), \\ \gamma_2 &= \frac{1}{\gamma_1} 2c^2 = 2c^2 + \mathcal{O}(c^4). \end{split}$$

**Lemma 3.18.** Consider (3.28) with F = 0, then the solution is given by

$$\tilde{u}(s) = S(s)r$$

with

$$S(s) = \begin{pmatrix} |\frac{s}{\eta}|^{\gamma_1} & |\frac{s}{\eta}|^{\gamma_2} \\ -\frac{\gamma_1}{c}\frac{s}{\eta}|\frac{s}{\eta}|^{\gamma_1-2} & -\frac{\gamma_2}{c}\frac{s}{\eta}|\frac{s}{\eta}|^{\gamma_2-2} \end{pmatrix},$$

and  $r = S^{-1}(\tilde{s}_0)\tilde{u}(s_0)$ .

Furthermore, we define the operator  $S^*$  as

$$S^*(s) = \begin{pmatrix} |\frac{s}{\eta}|^{\gamma_1} & |\frac{s}{\eta}|^{\gamma_2} \\ \frac{\gamma_1}{c} |\frac{s}{\eta}|^{\gamma_1 - 1} & \frac{\gamma_2}{c} |\frac{s}{\eta}|^{\gamma_2 - 1} \end{pmatrix},$$

which gives the estimate

$$|S(s)r| \le S^*r \qquad \forall r \in (\mathbb{R}_+)^2$$

The inverse of S can be computed as

$$S^{-1}(s) = \operatorname{sgn}(s)c\gamma^{-1} \begin{pmatrix} -\frac{\gamma_2}{c}\frac{s}{\eta}|\frac{s}{\eta}|^{\gamma_2-2} & -|\frac{s}{\eta}|^{\gamma_2}\\ \frac{\gamma_1}{c}\frac{s}{\eta}|\frac{s}{\eta}|^{\gamma_1-2} & |\frac{s}{\eta}|^{\gamma_1} \end{pmatrix}$$
$$= \begin{pmatrix} -\frac{\gamma_2}{\gamma}|\frac{s}{\eta}|^{\gamma_2-1} & -\frac{c}{\gamma}\frac{s}{\eta}|\frac{s}{\eta}|^{\gamma_2-1}\\ \frac{\gamma_1}{\gamma}|\frac{s}{\eta}|^{\gamma_1-1} & \frac{c}{\gamma}\frac{s}{\eta}|\frac{s}{\eta}|^{\gamma_1-1} \end{pmatrix}.$$

**Lemma 3.19.** Let  $u_1, u_2$  be a solution to (3.28) with given  $F = (F_1, F_2)$ , then for

$$R_1[F] = (1+10c^2)2c^2\eta^{1-\gamma_2} \int_{\tilde{s}_0}^s \tau^{\gamma_2-1}F_1(\tau) \ d\tau + c\eta^{-\gamma_2} \int_{\tilde{s}_0}^s \tau^{\gamma_2}F_2(\tau) \ d\tau$$
$$R_2[F] = (1+10c^2)\eta^{1-\gamma_1} \int_{\tilde{s}_0}^s \tau^{\gamma_1-1}F_1(\tau) \ d\tau + c\eta^{-\gamma_1} \int_{\tilde{s}_0}^s \tau^{\gamma_1}F_2(\tau) \ d\tau,$$

 $we \ estimate$ 

$$|u - \tilde{u}| \le S^*(s)R[F].$$

*Proof.* Since S has an inverse, we write

$$u = S(s)r(s)$$

and our aim is to control the evolution of r(s). Therefore, we calculate

$$\begin{aligned} |\partial_s r| &= S^{-1}F \\ |\partial_s r_1| &\le 2c^2 |\frac{s}{\eta}|^{\gamma_2 - 1}F_1 + c |\frac{s}{\eta}|^{\gamma_2}F_2 \\ |\partial_s r_2| &\le |\frac{s}{\eta}|^{\gamma_1 - 1}F_1 + c |\frac{s}{\eta}|^{\gamma_1}F_2 \end{aligned}$$

and so

$$|r_1(s) - r_1(d)| \le 2c^2(1+10c^2)\eta^{1-\gamma_2} \int \tau^{\gamma_2-1}F_1(\tau) + c\eta^{-\gamma_2} \int \tau^{\gamma_2}F_2(\tau)$$
  
$$|r_2(s) - r_2(d)| \le (1+10c^2)\eta^{1-\gamma_1} \int \tau^{\gamma_1-1}F_1(\tau) + c\eta^{-\gamma_1} \int \tau^{\gamma_1}F_2(\tau)$$

In the following we always assume that there exists  $c_1, c_2, \tilde{c}_1, \tilde{c}_2 \ge 0$  such that

$$|u| \leq S^*(s)C(s)$$

$$C_1(s) = c_1 + \tilde{c}_1(\frac{s}{\eta})^{-\gamma}$$

$$C_2(s) = c_2 + \tilde{c}_2(\frac{s}{\eta})^{\gamma}$$
(3.29)

on a maximal interval  $[\tilde{s}_0, s^*]$ . We will establish some estimates on the  $R_i$  depending on  $c_i$  and  $\tilde{c}_i$  and then we will determine specific  $c_i$  and  $\tilde{c}_i$  such that we prove that the maximal  $s^*$  will be greater than  $\tilde{s}_1$ . Later it will be sufficient to choose  $\tilde{c}_1 = 0$  on  $I_1$  and  $\tilde{c}_2 = 0$  on  $I_3$ . We thus deduce

$$\begin{aligned} |u_1(s)| &\leq (c_1 + \tilde{c}_2) |\frac{s}{\eta}|^{\gamma_1} + (\tilde{c}_1 + c_2) |\frac{s}{\eta}|^{\gamma_2} \\ |u_2(s)| &\leq (\frac{\gamma_1}{c} c_1 + \frac{\gamma_2}{c} \tilde{c}_2) |\frac{s}{\eta}|^{\gamma_1 - 1} + (\frac{\gamma_1}{c} \tilde{c}_1 + \frac{\gamma_2}{c} c_2) |\frac{s}{\eta}|^{\gamma_2 - 1} \\ &\leq c_1^* |\frac{s}{\eta}|^{\gamma_1 - 1} + c_2^* |\frac{s}{\eta}|^{\gamma_2 - 1}. \end{aligned}$$

where  $c_1^* = \frac{\gamma_1}{c}c_1 + \frac{\gamma_2}{c}\tilde{c}_2$  and  $c_2^* = \frac{\gamma_1}{c}\tilde{c}_1 + \frac{\gamma_2}{c}c_2$ . For sake of simplicity we will often omit absolute values for the estimates.

**Lemma 3.20** (3 mode forcing estimate). Let u(s) = S(s)r(s) be a solution of (3.25) on  $[\tilde{s}_0, s^*]$ , such that  $|u(s)| \leq S^*(s)C(s)$ , then we estimate

$$R_1[F_{3mode}] \le 20c^2c_1 + (20 + c^4(\frac{s \wedge \tilde{s}_0}{\eta})^{-\gamma})\tilde{c}_1 + (20c^2 + c^4(\frac{s \wedge \tilde{s}_0}{\eta})^{-\gamma})c_2 + 20c^4\tilde{c}_2$$
  

$$R_2[F_{3mode}] \le 20(\frac{s \vee \tilde{s}_0}{\eta})^{\gamma}(c_1 + 2c^2\tilde{c}_2) + 20(\tilde{c}_1 + c^2c_2).$$

*Proof.* The forcing term is given by

$$F_{3mode} = \left(\frac{c}{\eta} - a_1\right)e_1u_2 - 2c\eta \frac{1}{s^2(s^2+1)}e_2u_1$$

Therefore, we estimate

$$R_{1}[e_{2}2c\eta \frac{1}{s^{2}(1+s^{2})}u_{1}] \leq 2c^{2}\eta^{\gamma_{1}} \int_{\tilde{s}_{0}}^{s} \tau^{\gamma_{2}-4}((c_{1}+\tilde{c}_{2})(\frac{\tau}{\eta})^{\gamma_{1}}+(\tilde{c}_{1}+c_{2})(\frac{\tau}{\eta})^{\gamma_{2}})$$

$$\leq c^{4}(c_{1}+\tilde{c}_{2})+c^{4}(\frac{s\wedge\tilde{s}_{0}}{\eta})^{-\gamma}(\tilde{c}_{1}+c_{2})$$

$$R_{2}[e_{2}2c\eta \frac{1}{s^{2}(1+s^{2})}u_{1}] = 2c^{2}\eta^{\gamma_{2}} \int_{\tilde{s}_{0}}^{s} \tau^{\gamma_{1}-4}((c_{1}+\tilde{c}_{2})(\frac{\tau}{\eta})^{\gamma_{1}}+(\tilde{c}_{1}+c_{2})(\frac{\tau}{\eta})^{\gamma_{2}})$$

$$\leq 2c^{3}\eta^{-\gamma}(c_{1}+\tilde{c}_{2})+c^{4}(\tilde{c}_{1}+c_{2}).$$

By Taylor formula we obtain  $|c\frac{1}{\eta}-a_1|\leq 18c\frac{|s|}{\eta^2}$  and so

$$\begin{aligned} R_1[(c\frac{1}{\eta} - a_1)u_2e_1] &\leq (1 + 10c^2)c^2\eta^{1-\gamma_2} \int \tau^{\gamma_2 - 1} 18c\frac{\tau}{\eta^2}((\frac{\tau}{\eta})^{\gamma_1 - 1}c_1^* + (\frac{\tau}{\eta})^{\gamma_2 - 1}c_2^*) \\ &\leq 20c^3c_1^* + 20cc_2^* \\ &\leq 20c^2c_1 + 20\tilde{c}_1 + 20c^2c_2 + 20c^4\tilde{c}_2, \end{aligned}$$

$$R_2[(c\frac{1}{\eta} - a_1)u_2e_1] &= (1 + 10c^2)\eta^{1-\gamma_1} \int \tau^{\gamma_1 - 1} 18c\frac{\tau}{\eta^2}((\frac{\tau}{\eta})^{\gamma_1 - 1}c_1^* + c(\frac{\tau}{\eta})^{\gamma_2 - 1}c_2^*)) \\ &\leq 20c|\frac{s \vee \tilde{s}_0}{\eta}|^{\gamma}c_1^* + 20cc_2^* \\ &\leq 20|\frac{s \vee \tilde{s}_0}{\eta}|^{\gamma}(c_1 + 2c^2\tilde{c}_2) + 20\tilde{c}_1 + 20c^2c_2. \end{aligned}$$

**Lemma 3.21** ( $k_0$ -th current estimate ). Let u(s) = S(s)r(s) be a solution of (3.25) on  $[\tilde{s}_0, s^*]$  such that  $|u(s)| \leq S^*(s)C(s)$ , then we estimate

$$\begin{split} R_1[F_j] &\leq \frac{c^3}{\beta}(c_1 + \tilde{c}_2) + \frac{c^3}{\beta}(\frac{s \wedge s_0}{\eta})^{-\gamma}(\tilde{c}_1 + c_2) + \begin{cases} \frac{4c^{2+\gamma_1}}{\kappa_{k_0}\eta^{1+\gamma_2}}j(k_0, \tilde{s}_0) & \text{on } I_1\\ \eta^{\gamma_1}\frac{c^{4+\gamma_1}}{\kappa_{k_0}}j(k_0, \tilde{s}_0) & \text{on } I_3 \end{cases} \\ R_2[F_j] &\leq \min(\frac{1}{\beta}\frac{1}{2c^{1+\gamma}}\eta^{-\gamma}, \frac{c}{\beta}(\frac{s \vee \tilde{s}_0}{\eta})^{\gamma})(c_1 + \tilde{c}_2) + \frac{1}{\beta}c(c_2 + \tilde{c}_1) \\ &+ \frac{1}{\beta}c(c_2 + \tilde{c}_1) + \begin{cases} 4\frac{c^{\gamma_2}}{\kappa_{k_0}\eta^{1+\gamma_1}}j(k_0, \tilde{s}_0) & \text{on } I_1\\ \eta^{\gamma_2}\frac{c^{2+\gamma_2}}{\kappa_{k_0}}j(k_0, \tilde{s}_0) & \text{on } I_3 \end{cases} . \end{split}$$

Furthermore, on  $I_1$  we estimate

$$|j(k_0, \tilde{s}_1)| \le \frac{2d^2}{\eta^2} \exp(-\frac{\kappa}{2^5} \xi \eta^2) j(k_0, \tilde{s}_0) + \frac{c^2}{\beta} \left( (c_1 + \tilde{c}_2) (\frac{d}{\eta})^{\gamma_1} + (c_2 + \tilde{c}_1) (\frac{d}{\eta})^{\gamma_2} \right)$$

and on  $I_3$ 

$$|j(k_0, \tilde{s}_1)| \le c^2 \eta^2 \exp(-\kappa_{k_0} \eta^3) j(k_0, \tilde{s}_0) + 2 \frac{16^2}{\beta} \frac{1}{\eta^2} (c_1 + c_2 + \tilde{c}_1 + \tilde{c}_2)$$

*Proof.* The equation

$$\partial_s j(k_0) = \left(\frac{2s}{1+s^2} - \kappa_{k_0}(1+s^2)\right)j(k_0) + u_1$$

leads to

$$\begin{aligned} j(k_0) &= \frac{1+s^2}{1+s_0^2} \exp(-\kappa_{k_0}(s-s_0+\frac{1}{3}(s^3-s_0^3)))j(k_0,\tilde{s}_0) \\ &+ \frac{\kappa_k}{\beta} \int_{s_0}^s d\tau_2 \ \frac{1+s^2}{1+\tau_2^2} \exp(-\kappa_{k_0}(s-\tau_2+\frac{1}{3}(s^3-\tau_2^3)))u_1(\tau_2) \\ &= j_1 + j_2. \end{aligned}$$

Therefore, we estimate

$$\begin{split} R_{1}[F_{j_{2}}] &= \frac{\kappa_{k_{0}}}{\beta}c^{2}\eta^{1-\gamma_{2}}\int_{s_{0}}^{s} d\tau_{1}\int_{s_{0}}^{\tau_{1}} d\tau_{2} \ \tau_{1}^{\gamma_{2}-1}\frac{1+\tau_{1}^{2}}{1+\tau_{2}^{2}}\exp(-\kappa_{k_{0}}(\tau_{1}-\tau_{2}+\frac{1}{3}(\tau_{1}^{3}-\tau_{2}^{3})))u_{1}(\tau_{2}) \\ &= \frac{\kappa_{k_{0}}}{\beta}c^{2}\eta^{1-\gamma_{2}}\int_{s_{0}}^{s} d\tau_{1}\int_{s_{0}}^{\tau_{1}} d\tau_{2} \ \tau_{1}^{\gamma_{2}-1}\frac{1+\tau_{1}^{2}}{1+\tau_{2}^{2}} \\ &\quad \exp(-\kappa_{k_{0}}(\tau_{1}-\tau_{2}+\frac{1}{3}(\tau_{1}^{3}-\tau_{2}^{3})))((c_{1}+\tilde{c}_{2})(\frac{\tau_{2}}{\eta})^{\gamma_{1}}+(c_{2}+\tilde{c}_{1})(\frac{\tau_{2}}{\eta})^{\gamma_{2}}) \\ &\leq \frac{1}{\beta}c^{2}\eta^{1-\gamma_{2}}\int_{s_{0}}^{s} d\tau_{2} \ \frac{((c_{1}+\tilde{c}_{2})(\frac{\tau_{2}}{\eta})^{\gamma_{1}}+(c_{2}+\tilde{c}_{1})(\frac{\tau_{2}}{\eta})^{\gamma_{2}})\tau_{2}^{-\gamma_{1}}}{1+\tau_{2}^{2}} \\ &\quad \cdot\int_{\tau_{2}}^{s} d\tau_{1}\kappa_{k_{0}}(1+\tau_{1}^{2})\exp(-\kappa_{k_{0}}(\tau_{1}-\tau_{2}+\frac{1}{3}(\tau_{1}^{3}-\tau_{2}^{3})))) \\ &= \frac{1}{\beta}c^{2}\eta^{1-\gamma_{2}}\int_{s_{0}}^{s} d\tau_{2} \ ((c_{1}+\tilde{c}_{2})(\frac{\tau_{2}}{\eta})^{\gamma_{1}}+(c_{2}+\tilde{c}_{1})(\frac{\tau_{2}}{\eta})^{\gamma_{2}})\frac{\tau_{2}^{-\gamma_{1}}}{1+\tau_{2}^{2}} \\ &\quad \cdot\left[-\exp(-\kappa_{k}(\tau_{1}-\tau_{2}+\frac{1}{3}(\tau_{1}^{3}-\tau_{2}^{3})))\right]_{\tau_{1}=\tau_{2}}^{\tau_{1}=\tau_{2}} \\ &\leq \frac{1}{\beta}c^{2}\eta^{1-\gamma_{2}}\int_{s_{0}}^{s} d\tau_{2} \ (c_{1}+\tilde{c}_{2})\eta^{-\gamma_{1}}\tau_{2}^{-2}+(c_{2}+\tilde{c}_{1})\eta^{-\gamma_{2}}\tau_{2}^{-\gamma-2} \\ &\leq (c_{1}+\tilde{c}_{2})\frac{1}{\beta}c^{2}\eta^{1-\gamma_{2}-\gamma_{1}}[-\tau^{-1}]_{s_{0}}^{s}+(c_{2}+\tilde{c}_{1})\frac{1}{\beta}c^{2}\eta^{\gamma}[-\tau^{-\gamma-1}]_{s_{0}}^{s} \\ &\leq \frac{c_{3}}{\beta}(c_{1}+\tilde{c}_{2})+\frac{c_{3}}{\beta}(\frac{s\Lambda s_{0}}{\eta})^{-\gamma}(\tilde{c}_{1}+c_{2}) \end{split}$$

$$\begin{split} R_{2}[F_{j_{2}}] &= \frac{\kappa_{k_{0}}}{\beta} \eta^{1-\gamma_{1}} \int_{s_{0}}^{s} d\tau_{1} \int_{s_{0}}^{\tau_{1}} d\tau_{2} \ \tau_{1}^{\gamma_{1}-1} \frac{1+\tau_{1}^{2}}{1+\tau_{2}^{2}} \exp(-\kappa_{k_{0}}(\tau_{1}-\tau_{2}+\frac{1}{3}(\tau_{1}^{3}-\tau_{2}^{3})))u_{1}(\tau_{2}) \\ &= \frac{\kappa_{k_{0}}}{\beta} \eta^{1-\gamma_{1}} \int_{s_{0}}^{s} d\tau_{1} \int_{s_{0}}^{\tau_{1}} d\tau_{2} \ \tau_{1}^{-\gamma_{2}} \frac{1+\tau_{1}^{2}}{1+\tau_{2}^{2}} \\ &\quad \cdot \exp(-\kappa_{k_{0}}(\tau_{1}-\tau_{2}+\frac{1}{3}(\tau_{1}^{3}-\tau_{2}^{3})))((c_{1}+\tilde{c}_{2})(\frac{\tau_{2}}{\eta})^{\gamma_{1}} + (c_{2}+\tilde{c}_{1})(\frac{\tau_{2}}{\eta})^{\gamma_{2}}) \\ &\leq \frac{1}{\beta} \eta^{\gamma_{2}} \int_{s_{0}}^{s} d\tau_{2} \ \frac{((c_{1}+\tilde{c}_{2})(\frac{\tau_{2}}{\eta})^{\gamma_{1}+(c_{2}+\tilde{c}_{1})(\frac{\tau_{2}}{\eta})^{\gamma_{2}})\tau_{2}^{-\gamma_{2}}}{1+\tau_{2}^{2}} \\ &\quad \cdot \int_{s_{0}}^{\tau_{2}} d\tau_{1} \ \kappa_{k}(1+\tau_{1}^{2}) \exp(-\kappa_{k}(\tau_{1}-\tau_{2}+\frac{1}{3}(\tau_{1}^{3}-\tau_{2}^{3}))) \\ &= \frac{1}{\beta} \int_{s_{0}}^{s} d\tau_{2} \ ((c_{1}+\tilde{c}_{2})\tau_{2}^{\gamma-2}\eta^{-\gamma} + (c_{2}+\tilde{c}_{1})\tau_{2}^{-2}\eta^{-\gamma_{2}}) \\ &\quad \cdot [\exp(-\kappa_{k_{0}}(\tau_{1}-\tau_{2}+\frac{1}{3}(\tau_{1}^{3}-\tau_{2}^{3})))]_{\tau_{1}=\tau_{2}}^{\tau_{1}=\tau_{2}} \\ &\leq \frac{1}{\beta} \int_{s_{0}}^{s} d\tau_{2} \ ((c_{1}+\tilde{c}_{2})\tau_{2}^{\gamma-2}\eta^{-\gamma} + (c_{2}+\tilde{c}_{1})\tau_{2}^{-2}\eta^{-\gamma_{2}}). \end{split}$$

We note that for the first term we obtain

$$\frac{1}{\beta} \int_{s_0}^s d\tau_2 \ \tau_2^{\gamma-2} \eta^{-\gamma} \le \min(\frac{c}{\beta} (\frac{s \lor \tilde{s}_0}{\eta})^{\gamma}, \frac{1}{\beta c} (c\eta)^{-\gamma}),$$

since we can either integrate it directly or first pull out  $s^\gamma$  and then integrate. Finally, we obtain the following estimate

$$R_1[F_{j_2}] \le \min(\frac{1}{\beta} \frac{1}{2c^{1+\gamma}} \eta^{-\gamma}, \frac{c}{\beta} (\frac{s \lor \tilde{s}_0}{\eta})^{\gamma})(c_1 + \tilde{c}_2) + \frac{1}{\beta} c(c_2 + \tilde{c}_1).$$

On  $I_1$  we estimate the  $j(k_0)$  influence by

$$R_{1}[F_{j_{1}}] = c^{2} \eta^{1-\gamma_{2}} j(s_{0}) \int \tau^{\gamma_{2}-1} \frac{1+\tau^{2}}{1+s_{0}^{2}} \exp(-\kappa_{k_{0}}(\tau-s_{0}+\frac{1}{3}(\tau^{3}-s_{0}^{3})))$$
  
$$\leq 4c^{2+\gamma_{1}} \eta^{-1-\gamma_{2}} j(s_{0}) \int (1+\tau^{2}) \exp(-\kappa_{k_{0}}(\tau-s_{0}+\frac{1}{3}(\tau^{3}-s_{0}^{3})))$$
  
$$\leq \frac{4c^{2+\gamma_{1}}}{\kappa_{k_{0}} \eta^{1+\gamma_{2}}} j(s_{0})$$

and

$$R_{2}[F_{j_{2}}] = \eta^{1-\gamma_{1}} j(s_{0}) \int \tau^{\gamma_{1}-1} \frac{1+\tau^{2}}{1+s_{0}^{2}} \exp(-\kappa_{k_{0}}(\tau-s_{0}+\frac{1}{3}(\tau^{3}-s_{0}^{3})))$$
  
$$= 4\eta^{-1-\gamma_{1}} c^{\gamma_{2}} j(s_{0}) \int (1+\tau^{2}) \exp(-\kappa_{k_{0}}(\tau-s_{0}+\frac{1}{3}(\tau^{3}-s_{0}^{3})))$$
  
$$= 4\frac{c^{\gamma_{2}}}{\kappa_{k_{0}} \eta^{1+\gamma_{1}}} j(s_{0}).$$

and

We estimate  $j(k_0)$  by

$$\begin{aligned} j(k_0,s) &= \frac{1+s^2}{1+\tilde{s}_0^2} \exp(-\kappa_{k_0}(s-\tilde{s}_0+\frac{1}{3}(s^3-\tilde{s}_0^3)))j(k_0.\tilde{s}_0) \\ &+ \frac{\kappa_{k_0}}{\beta} \int_{\tilde{s}_0}^s d\tau \; \frac{1+s^2}{1+\tau^2} \exp(-\kappa_{k_0}(s-\tau+\frac{1}{3}(s^3-\tau^3)))u_1(\tau) \\ &\leq \frac{4d^2}{\eta^2} \exp(-\frac{\kappa}{2^5}\xi\eta^2)j(k_0,\tilde{s}_0) + \frac{c^2}{\beta}((c_1+\tilde{c}_2)(\frac{d}{\eta})^{\gamma_1} + (c_2+\tilde{c}_1)(\frac{d}{\eta})^{\gamma_2}). \end{aligned}$$

On  $I_3$  we estimate the  $j(k_0)$  influence by

$$R_{1}[F_{j_{1}}] = c^{2} \eta^{1-\gamma_{2}} \int \tau^{\gamma_{2}-1} \frac{1+\tau^{2}}{1+s_{0}^{2}} \exp(-\kappa_{k_{0}}(\tau-s_{0}+\frac{1}{3}(\tau^{3}-s_{0}^{3})))j(s_{0})$$
  
$$\leq c^{4+\gamma_{1}} \eta^{\gamma_{1}} \int (1+\tau^{2}) \exp(-\kappa_{k_{0}}(\tau-s_{0}+\frac{1}{3}(\tau^{3}-s_{0}^{3})))j(s_{0})$$
  
$$\leq \eta^{\gamma_{1}} \frac{c^{4+\gamma_{1}}}{\kappa_{k_{0}}} j(s_{0})$$

and

$$\begin{aligned} R_2[F_{j_2}] &= \eta^{1-\gamma_1} \int \tau^{\gamma_1 - 1} \frac{1 + \tau^2}{1 + s_0^2} \exp(-\kappa_{k_0}(\tau - s_0 + \frac{1}{3}(\tau^3 - s_0^3)))j(s_0) \\ &= \eta^{\gamma_2} c^{2+\gamma_2} \int (1 + \tau^2) \exp(-\kappa_{k_0}(\tau - s_0 + \frac{1}{3}(\tau^3 - s_0^3)))j(s_0) \\ &= \eta^{\gamma_2} \frac{c^{2+\gamma_2}}{\kappa_{k_0}} j(s_0). \end{aligned}$$

Next we want to estimate the evolution of  $j(k_0)$ 

$$j(k_0, \tilde{s}_1) = \frac{1+\tilde{s}_1^2}{1+d^2} \exp(-\kappa_{k_0}(\tilde{s}_1 - d + \frac{1}{3}(\tilde{s}_1^3 - d^3)))j(k_0, \tilde{s}_0) \\ + \frac{\kappa_{k_0}}{\beta} \int_d^{\tilde{s}_1} d\tau_2 \ \frac{1+s^2}{1+\tau_2^2} \exp(-\kappa_k(s - \tau_2 + \frac{1}{3}(s^3 - \tau_2^3)))((c_1 + \tilde{c}_2)(\frac{\tau_2}{\eta})^{\gamma_1} + (c_2 + \tilde{c}_1)(\frac{\tau_2}{\eta})^{\gamma_2}).$$

Therefore, we deduce

$$\begin{split} \frac{\kappa_{k_0}}{\beta} \int_{d}^{\tilde{s}_1} d\tau_2 \, \frac{1+s^2}{1+\tau_2^2} \exp(-\kappa_{k_0}(s-\tau_2+\frac{1}{3}(s^3-\tau_2^3)))(\frac{\tau_2}{\eta})^{\gamma_i} \\ &\leq \frac{\kappa_{k_0}}{\beta} \left( \int_{d}^{\frac{1}{2}\tilde{s}_1} + \int_{\frac{1}{2}\tilde{s}_1}^{\tilde{s}_1} \right) \frac{1+s^2}{1+\tau_2^2} \exp(-\kappa_{k_0}(s-\tau_2+\frac{1}{3}(s^3-\tau_2^3)))(\frac{\tau_2}{\eta})^{\gamma_i} \\ &\leq \frac{\kappa_{k_0}}{\beta} \eta^{-\gamma_i} \frac{c^{1-\gamma_i}}{\gamma_i} (1+\eta^2) \exp(-\frac{\kappa_{k_0}}{25}\eta^3) + \frac{\kappa_{k_0}}{\beta} \frac{2^4}{\kappa_{k_0}\eta^2} \\ &\leq \frac{2^5}{\beta} \frac{1}{\eta^2}, \end{split}$$

which leads to

$$|j(k_0, \tilde{s}_1)| \le c^2 \kappa_{k_0} \eta^2 \exp(-\kappa_{k_0} \eta^3) j(k_0, \tilde{s}_0) + \frac{2^5}{\beta} \frac{1}{\eta^2} (c_1 + c_2 + \tilde{c}_1 + \tilde{c}_2).$$

**Lemma 3.22** (Forcing estimate ). Let u(s) = S(s)r(s) be a solution of (3.25) on  $[\tilde{s}_0, s^*]$  such that  $|u(s)| \leq S^*(s)C(s)$ . We define for  $|n| \geq 2$ 

$$\tilde{w}(n) = 2 \sum_{|m| \ge 2} (2c)^{|m-n|+\chi} (w + \frac{4}{\kappa\xi} j)(k_m, \tilde{s}_0) + (2c)^{||m|-2|} c(2c_1^* + \frac{1}{c^2} c_2^*) + (2c)^{|m|-1} (u_3(\tilde{s}_0) + \frac{2}{\kappa\xi\eta} (j(k_{\pm 1}, \tilde{s}_0)))$$

where  $\chi = \chi(m, n) = -|sgn(m) - sgn(n)|$ . Then we estimate

$$R_1[F_{\tilde{w}}] = 2c^2(\tilde{w}(2) + \tilde{w}(-2))$$
  

$$R_2[F_{\tilde{w}}] = c^2(\tilde{w}(2) + \tilde{w}(-2))(\frac{s \vee \tilde{s}_0}{\eta})^{\gamma_1}$$

and

$$R_1[F_{j(k_{\pm 1})}] = \frac{2c}{\kappa\xi\eta} j(k_{\pm 1}, \tilde{s}_0) + \frac{2c}{\beta\kappa\xi} (\tilde{w}(1) + c_1^* + c_2^*)$$
  

$$R_2[F_{j(k_{\pm 1})}] = \frac{2c}{\kappa\xi\eta} j(k_{\pm 1}, \tilde{s}_0) (\frac{s\vee\bar{s}_0}{\eta})^{\gamma_1} + \frac{c}{\beta\kappa\xi} (\tilde{w}(1) + c_1^* + c_2^*) (\frac{s\vee\bar{s}_0}{\eta})^{\gamma_1}$$

and

$$\begin{split} R_1[F_{u_3}] &= 2c\tilde{w}(1) \\ R_2[F_{u_3}] &= 2c\tilde{w}(1)(\frac{s\vee\tilde{s}_0}{\eta})^{\gamma_1}. \end{split}$$

 $Furthermore, \ we \ estimate$ 

$$\begin{aligned} |w(k_n, s)| &\leq \tilde{w}(n) & |n| \geq 2\\ |u_3| &\leq \tilde{w}(1) = \tilde{w}(-1) \\ |j(k_n)| &\leq 2e^{-\frac{1}{2}\kappa\xi\eta(s-\tilde{s}_0)}j(k_n, \tilde{s}_0) + 4\frac{1}{\beta\eta^2}\tilde{w}(n). \end{aligned}$$

*Proof.* To estimate  $w(k_n, s)$  we without loss of generality assume that  $n \ge 2$ . We begin with the case  $n \ge 3$ , where we deduce that

$$\begin{aligned} \partial_s w(k_n) &= a(k_{n+1})w(k_{n+1}) - a(k_{n-1})w(k_{n-1}) - j(k_n) \\ &\leq \frac{c}{\eta}(\tilde{w}(n-1) + \tilde{w}(n+1)) + 2e^{-\kappa\xi\eta(s-\tilde{s}_0)}j(k_0+n,\tilde{s}_0) + 4\frac{1}{\beta\eta^2}\tilde{w}(n). \end{aligned}$$

We estimate

$$2\int e^{-\frac{1}{2}\kappa\xi\eta(\tau-\tilde{s}_0)}j(k_0+n,\tilde{s}_0) \le 4\frac{1}{\kappa\xi\eta}j(k_n,\tilde{s}_0).$$

Thus integrating  $\partial_s w(k_0 + n)$  over time yields

$$w(k_n) \le w(k_n, \tilde{s}_0) + c(\tilde{w}(n-1) + \tilde{w}(n+1)) + 4\frac{1}{\kappa\xi\eta}j(k_0 + n, \tilde{s}_0) + \frac{4}{\beta\kappa\xi}\tilde{w}(n) < \tilde{w}(n).$$

For the case n = 2 we deduce

$$\begin{aligned} \partial_s w(k_2) &= a(k_3)w(k_3) - a(k_1)\frac{1}{2}(u_3 + u_2) - j(k_2) \\ &\leq \frac{c}{\eta}(\tilde{w}(3) + 2\tilde{w}(1) + 2c_1^*(\frac{s}{\eta})^{\gamma_1 - 1} + 2c_2^*(\frac{s}{\eta})^{\gamma_2 - 1}) + 2e^{-\frac{1}{2}\kappa\eta\xi(s - \tilde{s}_0)}j(k_2, \tilde{s}_0) + 4\frac{1}{\beta\kappa\xi\eta}\tilde{w}(2) \\ &w(k_2) &\leq w(k_2, \tilde{s}_0) + \frac{4}{\kappa\xi\eta}j(k_2, \tilde{s}_0) + c(\tilde{w}(3) + 2\tilde{w}(1) + 2c_1^* + \frac{1}{c^2}c_2^*) + 4\frac{1}{\beta\kappa\xi}\tilde{w}(2) \\ &< \tilde{w}(2). \end{aligned}$$

We estimate  $u_3$  by

$$\begin{aligned} \partial_s u_3 &= a(k+2)w(k_2) - a(k-2)w(k_{-2}) - j(k_1) + j(k_{-1}) \\ &\leq \frac{2c}{\eta}(w(2) + w(-2)) + 2e^{-\frac{1}{2}\kappa\xi\eta(s-\tilde{s}_0)}j(k_{\pm 1},\tilde{s}_0) + \frac{4}{\beta\kappa\xi\eta}(\tilde{w}(1) + c_1^* + c_2^*) \\ &|u_3| \leq |u_3(\tilde{s}_0)| + c(\tilde{w}(2) + \tilde{w}(-2)) + \frac{4}{\kappa\xi\eta}(j(k_{\pm 1},\tilde{s}_0) + \frac{1}{\beta}\tilde{w}(1) + \frac{1}{\beta}c_1^* + \frac{1}{\beta}c_2^*) \\ &\leq \tilde{w}(1). \end{aligned}$$

Non-resonant j will often be estimated similarly. Therefore we will use the following notation frequently. We estimate  $j(k_n)$  for  $n \ge 2$  by writing  $\hat{s} = s - \frac{k_0(k_0-k)}{k+1}\eta$  and  $\hat{\tau} = \tau - \frac{k_0(k_0-k)}{k+1}\eta$ 

$$\partial_s j(k_n) = -\kappa_{k_n} (1 + \hat{s}^2) j(k_n) + 2 \frac{\hat{s}}{1 + \hat{s}^2} j(k_n) + \frac{1}{\beta} \kappa_{k_n} w(k_n)$$

which gives

$$j(k_n) \le \frac{1+\hat{s}_0^2}{1+\hat{s}_0^2} e^{-\kappa_{k_n}((\hat{s}-\hat{s}_0+\frac{1}{3}(\hat{s}^3-\hat{s}_0^3))}j(k_n,\tilde{s}_0) + \frac{1}{\beta}\kappa_{k_n} \int d\tau \ \frac{1+\hat{s}^2}{1+\hat{\tau}^2} e^{-\kappa_{k_n}((\hat{s}-\hat{\tau}+\frac{1}{3}(\hat{s}^3-\hat{\tau}^3)))}\tilde{w}(n)$$

For  $\tilde{s}_0 \leq \tau \leq s \leq \tilde{s}_1$  we obtain

$$\kappa_{k_n}(\hat{s} - \hat{\tau} + \frac{1}{3}(\hat{s}^3 - \hat{\tau}^3)) = \kappa_{k_n} \frac{1}{3}(s - \tau)(\hat{s}^2 + \hat{s}\hat{\tau} + \hat{\tau}^2 + 1)$$
  

$$\geq \frac{1}{2}\kappa \max(k_n^2, k_0^2)\eta^2(s - \tau)$$
  

$$\frac{1 + \hat{s}^2}{1 + \hat{\tau}} \leq 2.$$

So we infer

$$\begin{aligned} j(k_n) &\leq 2e^{-\frac{1}{2}\kappa\xi\eta(s-\tilde{s}_0)}j(k_n,\tilde{s}_0) \\ &+ 2\kappa_{k_n}\frac{1}{\beta}\int d\tau \ e^{-\frac{1}{2}\kappa\xi\eta(s-\tau)}\tilde{w}(n) \\ &\leq 2e^{-\frac{1}{2}\kappa\xi\eta(s-\tilde{s}_0)}j(k_n,\tilde{s}_0) + \frac{4}{\beta\eta^2}\tilde{w}(n). \end{aligned}$$

We next turn to the estimate of  $j(k_{\pm 1})$ , where we without loss of generality consider  $j(k_1)$ . With the equation

$$\partial_s j(k_1) = \left(2\frac{\hat{s}}{1+\hat{s}^2} - \kappa_{k_1}(1+\hat{s}^2)\right)j(k_1) + \frac{\kappa_{k_1}}{2\beta}\kappa_{k_1}(u_3+u_2)$$

we estimate

$$\begin{split} j(k_1) &\leq 2e^{-\frac{1}{2}\kappa\xi\eta(s-\tilde{s}_0)}j(k_1,\tilde{s}_0) \\ &+ \frac{\kappa_{k_1}}{2\beta}\int d\tau \ e^{-\frac{1}{2}\kappa\xi\eta(s-\tau)}(\tilde{w}(1) + c_1^*(\frac{\tau}{\eta})^{\gamma_1-1} + c_2^*(\frac{\tau}{\eta})^{\gamma_2-1}) \\ &\leq 2e^{-\frac{1}{2}\kappa\xi\eta(s-\tilde{s}_0)}j(k_1,\tilde{s}_0) + \frac{2}{\beta\eta^2}(\tilde{w}(1) + c_1^* + c_2^*). \end{split}$$

Given these estimates, we next consider the effects on  $R[\cdot]$  by forcing:

$$F_{\tilde{w}} = e_1(a(k_2)w(k_2) + a(k_{-2})w(k_{-2}))$$
  
$$\leq e_1 \frac{c}{\eta} \tilde{w}(2) + e_1 \frac{c}{\eta} \tilde{w}(-2).$$

For constant  $e_2$  functions we estimate

$$R_1[e_2] \le \frac{c}{2}\eta$$
  

$$R_2[e_2] \le \frac{c}{3}\eta(\frac{s\vee\tilde{s}_0}{n})^{\gamma_1}.$$

Therefore, we can control  $F_{\tilde{w}}$  by

$$R_1[F_{\tilde{w}}] = c^2(\tilde{w}(2) + \tilde{w}(-2))$$
  

$$R_2[F_{\tilde{w}}] = c^2(\tilde{w}(2) + \tilde{w}(-2))(\frac{s \vee \tilde{s}_0}{\eta})^{\gamma_1}.$$

For  $F_{j(k_{\pm 1})}$  we use

$$F_{j(k_{\pm 1})} = -e_2 j(k_{\pm 1})$$
  
$$\leq e_2 (2e^{-\frac{1}{2}\kappa\xi\eta(s-\tilde{s}_0)}j(k_{\pm 1},\tilde{s}_0) + \frac{2}{\beta\kappa\xi\eta}(\tilde{w}(1) + c_1^* + c_2^*)),$$

to estimate

$$\begin{aligned} R_1[F_{j(k_{\pm 1})}] &= \frac{2c}{\kappa\xi\eta} j(k_{\pm 1}, \tilde{s}_0) + \frac{2c}{\beta\kappa\xi} (\tilde{w}(1) + c_1^* + c_2^*) \\ R_2[F_{j(k_{\pm 1})}] &= \frac{2c}{\kappa\xi\eta} j(k_{\pm 1}, \tilde{s}_0) (\frac{s\vee\tilde{s}_0}{\eta})^{\gamma_1} + \frac{c}{\beta\kappa\xi} (\tilde{w}(1) + c_1^* + c_2^*) (\frac{s\vee\tilde{s}_0}{\eta})^{\gamma_1}. \end{aligned}$$

Furthermore, for  $F_{u_2}$  we estimate

$$F_{u_3} = e_1 a_2 u_3$$
$$\leq e_1 \frac{c}{\eta} \tilde{w}(1)$$

and

$$R_1[e_1] \le \eta$$
$$R_2[e_1] \le \eta(\frac{s \vee \tilde{s}_0}{\eta})^{\gamma_1}.$$

to deduce

$$R_1[F_{u_3}] \le c\tilde{w}(1)$$
  

$$R_2[F_{u_3}] \le c\tilde{w}(1)(\frac{s \lor \tilde{s}_0}{\eta})^{\gamma_1}.$$

Proof of Proposition 3.17. For the interval  $I_1$  we have  $\tilde{s}_0 = s_0$ ,  $\tilde{s}_1 = -d$ . The initial data of r can be calculated by  $r(\tilde{s}_0) = S^{-1}(s_0)u(s_0)$  and so

$$\begin{aligned} r_1(\tilde{s}_0) &= -\frac{\gamma_2}{\gamma} (\frac{k_0}{2(k_0+1)})^{\gamma_2 - 1} u_1(\tilde{s}_0) + \frac{c}{\gamma} (\frac{k_0}{2(k_0+1)})^{\gamma_2} u_2(\tilde{s}_0) \\ &\approx -4c^2 u_1(\tilde{s}_0) + c u_2(\tilde{s}_0), \\ r_2(\tilde{s}_0) &= \frac{\gamma_1}{\gamma} (\frac{k_0}{2(k_0+1)})^{\gamma_1 - 1} u_1(\tilde{s}_0) - \frac{c}{\gamma} (\frac{k_0}{2(k_0+1)})^{\gamma_1} u_2(\tilde{s}_0) \\ &\approx u_1(\tilde{s}_0) - \frac{c}{2} u_2(\tilde{s}_0). \end{aligned}$$

For other initial data we define

$$N = \sum_{|m| \ge 2} (2c)^{|m|} (w + \frac{8}{\kappa \eta \xi} j)(k_m, \tilde{s}_0) + 2c(u_3(\tilde{s}_0) + \frac{8}{\kappa \eta \xi} j(k_{\pm 1}, \tilde{s}_0)) + \frac{2c}{\xi \kappa} j(k_0, s_0),$$

to bound the impact of the less important terms in the following bootstrap. Let C(s) be defined by the terms

$$c_{1} = 45c^{2}u_{1}(s_{0}) + 2cu_{2}(s_{0}) + 2N,$$
  

$$\tilde{c}_{1} = 2\frac{c^{3}}{\beta \vee 1}c_{2},$$
  

$$c_{2} = 2u_{1}(s_{0}) + 45cu_{2}(s_{0}) + 2N,$$
  

$$\tilde{c}_{2} = 0.$$

AS  $c_1 > r_1(\tilde{s}_0)$  and  $c_2 > r_2(\tilde{s}_0)$  and we have a smooth solution, the estimate  $|u| \leq S^*(s)C(s)$  holds at least for a small time. Let  $s^*$  be the maximal time such that  $|u| \leq S^*(s)C(s)$ . We then aim to show that necessarily  $s^* \geq -d$ , since otherwise the estimate improves, which contradicts the maximality. By Lemma 3.20, Lemma 3.21 and Lemma 3.22 we estimate

$$\begin{split} R_{1}[F_{all}] &= R_{1}[F_{3mode}] + R_{1}[F_{j}] + R_{1}[F_{\tilde{w}}] + R_{1}[F_{j(k_{0}\pm1)}] + R_{1}[F_{u_{3}}] \\ &= 20c^{2}c_{1} + (20 + c^{4}(\frac{s}{\eta})^{-\gamma})\tilde{c}_{1} + (20c^{2} + c^{4}(\frac{s}{\eta})^{-\gamma})c_{2} \\ &+ \frac{c^{3}}{\beta}c_{1} + \frac{c^{3}}{\beta}(\frac{s}{\eta})^{-\gamma}(\tilde{c}_{1} + c_{2}) + \frac{4c^{2+\gamma_{1}}}{\kappa_{k_{0}}\eta^{1+\gamma_{2}}}j(k_{0}, s_{0}) \\ &+ 2c^{2}(\tilde{w}(2) + \tilde{w}(-2)) \\ &+ \frac{2c}{\kappa\xi\eta}j(k_{\pm1}, \tilde{s}_{0}) + \frac{2c}{\beta\kappa\xi}(\tilde{w}(1) + c_{1}^{*} + c_{2}^{*}) \\ &+ 2c\tilde{w}(1) \\ &< 21c^{2}c_{1} + \tilde{c}_{1}(21 + \frac{c^{3}}{\beta}(\frac{s}{\eta})^{-\gamma}) + c_{2}(21c^{2} + \frac{c^{3}}{\beta}(\frac{s}{\eta})^{-\gamma}) + N \end{split}$$

and

$$\begin{split} R_2[F_{all}] &= R_2[F_{3mode}] + R_2[F_j] + R_2[F_{\tilde{w}}] + R_2[F_{j(k_0\pm 1)}] + R_2[F_{u_3}] \\ &= 20c_1 + 20\tilde{c}_1 + 20c^2c_2 \\ &+ \frac{c}{\beta}c_1 + \frac{c}{\beta}(c_2 + \tilde{c}_1) + 4\frac{c^{\gamma_2}}{\kappa_{k_0}\eta^{1+\gamma_1}}j(k_0, s_0) \\ &+ c^2(\tilde{w}(2) + \tilde{w}(-2)) \\ &+ \frac{2c}{\kappa\xi\eta}j(k_{\pm 1}, \tilde{s}_0) + \frac{c}{\beta\kappa\xi}(\tilde{w}(1) + c_1^* + c_2^*) \\ &+ 2c\tilde{w}(1) \\ &< 21c_1 + 21\tilde{c}_1 + \frac{c}{\beta\wedge 1}c_2 + N. \end{split}$$

We split  $R_1$  as

$$R_1[all] = R_1[all][1] + R_1[all][(\frac{s}{n})^{\gamma}],$$

into the part with and without a  $(\frac{s}{\eta})^{\gamma}$  term, respectively. We then estimate

$$\begin{aligned} r_1(\tilde{s}_0) + R_1[all][1] &< r_1(\tilde{s}_0) + 21c^2c_1 + 21\tilde{c}_1 + 21c^2c_2 + N, \\ R_1[all][(\frac{s}{\eta})^{\gamma}] &< \frac{c^3}{1\wedge\beta}\tilde{c}_1 + \frac{c^3}{1\wedge\beta}c_2, \\ r_2(\tilde{s}_0) + R_2[all][1] &< r_2(\tilde{s}_0) + 21c_1 + 21\tilde{c}_1 + 2\frac{c}{1\wedge\beta}c_2 + N, \end{aligned}$$

and thus we conclude the bootstrap that

$$r_1(\tilde{s}_0) + R_1[all][1] < c_1$$
  

$$R_1[all][(\frac{s}{\eta})^{\gamma}] < \tilde{c}_1$$
  

$$r_2(\tilde{s}_0) + R_2[all][1] < c_2.$$

We can therefore extend the estimates past the time  $s^*$ , which contradicts the maximally. Therefore, we obtain that for all times  $s \leq -d$  it holds that

$$|u(s)| \le S^*(s)C(s),$$

which yields the upper bound

$$|u(-d)| \le \begin{pmatrix} (c\eta)^{-\gamma_1}c_1 + (\tilde{c}_1 + c_2)(c\eta)^{-\gamma_2} \\ \frac{1}{c}(c\eta)^{1-\gamma_1}c_1 + (\frac{1}{c}\tilde{c}_1 + cc_2)(c\eta)^{1-\gamma_2} \end{pmatrix} \\ \le 2M \begin{pmatrix} (c\eta)^{-\gamma_2} \\ (c\eta)^{1-\gamma_2} \end{pmatrix}.$$

We next aim to establish an estimate on  $\tilde{w}(n)$ . For this purpose we note that

$$c(2c_1^* + \frac{1}{c^2}c_2^*) \approx 2c_1 + \frac{1}{c^2}\tilde{c}_1 + 2c_2$$
  
$$\approx 2c_1 + 2c_2$$
  
$$\leq 4u_1(s_0) + 100cu_2(s_0) + 3N$$

and

$$\begin{split} \tilde{w}(n) &\leq 2 \sum_{|m| \geq 2} (2c)^{|m-n| + \chi} (w + \frac{4}{\kappa \xi \eta} j) (k_m, \tilde{s}_0) \\ &+ (2c)^{||n| - 2|} c (2c_1^* + \frac{1}{c^2} c_2^*) \\ &+ (2c)^{|n| - 1} (u_3(\tilde{s}_0) + \frac{2}{\kappa \xi \eta} j(k_{\pm 1}, \tilde{s}_0)). \end{split}$$

We hence deduce that

$$\begin{split} \tilde{w}(n) &\leq 2 \sum_{|m| \geq 2} (2c)^{|m-n| + \chi} (w + \frac{4}{\kappa \xi \eta} j) (k_m, \tilde{s}_0) \\ &+ (2c)^{||n| - 2|} (4u_1(s_0) + 100 c u_2(s_0) + 3N) \\ &+ (2c)^{|n| - 1} (u_3(\tilde{s}_0) + \frac{2}{\kappa \xi \eta} j(\tilde{s}_0, k_{\pm 1})) \\ &\leq 2M_n, \end{split}$$

when  $\chi = -|\operatorname{sgn}(m) - \operatorname{sgn}(n)|$ . To prove (3.27) under the condition (3.26) we estimate

$$\begin{aligned} |u(-d) - \tilde{u}(-d)| &\leq S(-d)R[all] \\ &\leq u_1(\tilde{s}_0) \begin{pmatrix} (c\eta)^{-\gamma_2} \\ 5c(c\eta)^{\gamma_1} \end{pmatrix}. \end{aligned}$$

Furthermore, we use

$$\begin{split} \tilde{u}(-d) &= \begin{pmatrix} (c\eta)^{-\gamma_1} & (c\eta)^{-\gamma_2} \\ -\frac{\gamma_1}{2c}(c\eta)^{1-\gamma_1} & -\frac{\gamma_2}{2c}(c\eta)^{1-\gamma_2} \end{pmatrix} \begin{pmatrix} 4c^2u_1(\tilde{s}_0) - 2cu_2(\tilde{s}_0) \\ u_1(\tilde{s}_0) + cu_2(\tilde{s}_0) \end{pmatrix} \\ &\approx u_1(\tilde{s}_0) \begin{pmatrix} (c\eta)^{-\gamma_2} \\ O(c)(c\eta)^{\gamma_1} \end{pmatrix} \end{split}$$

and thus

$$\begin{aligned} |u_1(-d) - (c\eta)^{-\gamma_2} u_1(\tilde{s}_0)| &= 10c u_1(\tilde{s}_0)(c\eta)^{-\gamma_2} \\ |u_2(-d)| &\le 10c u_1(\tilde{s}_0)(c\eta)^{\gamma_1}. \end{aligned}$$

The remaining terms can be estimated by

$$M \le \frac{1}{1 - 10^{-1}} u_1(\tilde{s}_0),$$
  
$$M_n \le \frac{4}{1 - 10^{-1}} u_1(\tilde{s}_0).$$

## **3.4.3** The Resonance and Upper Bounds in $I_2$

The bounds on the evolution of (3.25) on the interval  $I_2 = [-d, d]$  are summarized in the following proposition:

**Proposition 3.23.** Let  $c \leq \min((8\pi)^{-\frac{4}{3}}\beta^{\frac{16}{3}}, 10^{-4})$ . Consider a solution of (3.25) on the interval  $I = [s_0, d]$ , then it holds that

$$\begin{aligned} |u_1(d)| &\leq 3(c\eta)^{-\gamma_2} LM, \\ |u_2(d)| &\leq 7\pi (c\eta)^{\gamma_1} LM, \\ |u_3(d)| &\leq 7\pi (\frac{5}{\eta})^2 (c\eta)^{\gamma_1} LM + 2M_1, \\ |w(k_n, d)| &\leq 7\pi (\frac{5}{\eta})^{|n|-1} (c\eta)^{\gamma_1} LM + 2M_n, \\ |j(k_n, d)| &\leq \frac{4}{\beta \eta^2} (7\pi (\frac{5}{\eta})^{|n|-1} (c\eta)^{\gamma_1} LM + 2M_n), \\ |j(k_0, d)| &\leq \frac{4}{\beta} \min(\kappa_{k_0} \pi d^2, 1) (c\eta)^{-\gamma_2} LM. \end{aligned}$$

For interval  $I_2$  we are mostly concerned with the interaction between  $j(k_0)$ and  $u_1$  and in particular the growth this induces for  $u_2$ . Therefore, consider the ODE system

$$\partial_s u_1 = -j(k_0) + F$$
  
$$\partial_s j(k_0) = \frac{\kappa_{k_0}}{\beta} u_1 + (\frac{2s}{1+s^2} - \kappa_{k_0}(1+s^2))j(k_0),$$
  
(3.30)

our aim is to bound the growth of  $j(k_0)$  and  $u_1$  by a factor. Let  $U(\tau, s)$  be the solution of (3.30) with initial data  $u_1(\tau) = 1$  and  $j(\tau) = 0$  and L as the constant which satisfies

$$|U(\tau, s)| \le L = L(\beta, \kappa, k). \tag{3.31}$$

With the restriction

$$c \le (8\pi)^{-\frac{4}{3}} \beta^{\frac{16}{3}},$$

L is estimated by the following two cases, if  $\beta \ge \pi$  we obtain L = 1 and if  $\beta < \pi$  we obtain a  $L = L(\alpha, \kappa, k) \le \sqrt{c}$ . A proof and more specific bounds can be found in 3.5 and for simplicity of presentation we here only consider two cases.

**Lemma 3.24.** Let  $u_1$  be a solution of (3.30) on [-d, d] such that (3.31) holds, then we estimate

$$\frac{1}{L}|u_1| \le u_1(-d) + \int_{-d}^s |F(\tau)| \ d\tau + |j(-d)| \int \frac{1+\tau^2}{1+d^2} \exp(-\kappa_k(\tau+d+\frac{1}{3}(s^3+d^3))).$$

*Proof.* We may without loss of generality restrict to the case j(-d) = 0, since we can choose  $\tilde{F} = F + \frac{1+s^2}{1+d^2} \exp(-\kappa_k(\tau + d + \frac{1}{3}(s^3 + d^3)))j(-d)$ . By Duhamel' principle the equation (3.30) is solved by

$$u_1(s) = U(-d, s)u_1(-d) + \int U(\tau, s)F(\tau)$$

which yields the desired bound.

Proof of Proposition 3.23. With Proposition 3.17 we estimate until time -d

$$\begin{aligned} &|u_1|(-d) \le 2M(c\eta)^{-\gamma_2} \\ &|u_2|(-d) \le 2M(c\eta)^{\gamma_1}, \\ &|u_3|(-d) \le 2M_1, \\ &|w(k_n, -d)| \le 2M_n, \\ &|j|(k_0, -d) \le \frac{c}{\beta}M(c\eta)^{-\gamma_2}\min(\kappa_{k_0}c^{-2}, 1), \\ &|j|(k_{\pm 1}, -d) \le \frac{4}{\beta\eta^2}M, \\ &|j|(k_n, -d) \le \frac{4}{\beta\eta^2}M_n. \end{aligned}$$

We next aim to prove by a bootstrap that

$$\begin{aligned} |u_1| &\leq 3L(c\eta)^{-\gamma_2} M, \\ |u_2| &\leq 7\pi L(c\eta)^{\gamma_1} M, \\ |u_3| &\leq 15\pi L(\frac{5}{\eta})^2 (c\eta)^{\gamma_1} M + 2M_1, \\ |w(k_n)| &\leq 7\pi L(\frac{5}{\eta})^{|n|-1} (c\eta)^{\gamma_1} M + 2M_n, \\ \int j(k_{\pm 1}) &\leq 7\pi L \frac{3d}{\beta\eta^2} (c\eta)^{\gamma_1} M, \\ \int j(k_n) &< \frac{2}{\beta} \frac{d}{\eta^2} (7\pi L \frac{5}{\eta} (c\eta)^{\gamma_1} M + 2M_n). \end{aligned}$$

To estimate  $u_1$  we use Lemma 3.24 to deduce

$$\begin{aligned} |u_1| &\leq L \left( u_1(-d) + \int a_1 u_2 + a_2 u_3 + \frac{1+s^2}{1+d^2} \exp(-\kappa_{k_0}(s-\tau + \frac{1}{3}(s^3-\tau^3)))j(-d) \right) \\ &\leq 2L(c\eta)^{-\gamma_2} M + L\frac{1}{\eta}(1+\eta^{-2})(7\pi L(c\eta)^{\gamma_1} M + 2M_1) \\ &+ \frac{L}{1+d^2} \min(\frac{1}{\kappa_{k_0}}, d^3)j(k_0, -d) \\ &\leq 2L(c\eta)^{-\gamma_2} M + L\frac{1}{\eta}(1+\eta^{-2})(7\pi L(c\eta)^{\gamma_1} M + 2M_1) + L\frac{c}{\beta}(c\eta)^{-\gamma_2} M \\ &< 3L(c\eta)^{-\gamma_2} M, \end{aligned}$$

where we used that  $\frac{40}{\eta}M_1 \leq \frac{1}{10}M(c\eta)^{-\gamma_2}$  since  $\eta \geq \frac{1}{10c}$  and that  $7\pi Lc < \frac{1}{2}$ . We estimate  $u_2$  by

$$\begin{aligned} |u_2| &\leq 2(c\eta)^{\gamma_1} M + \int 2c\eta \frac{1}{1+s^2} u_1 + a(k_{\pm 2})w(k_{\pm 2}) + j(k_{\pm 1}) \\ &\leq 2(c\eta)^{\gamma_1} M + 2\pi c\eta |u_1|_{L^{\infty}_s} + \frac{4}{\eta} |w(k_{\pm 2})|_{L^{\infty}_s} + \int j(k_{\pm 2}) \\ &< 7\pi L(c\eta)^{\gamma_1} M. \end{aligned}$$
In order to control  $u_3$ , we integrate  $\partial_s u_3$  in time, which yields

$$\begin{aligned} |u_3| &\leq |u_3|(-d) + \int a(k_{\pm 2})w(k_{\pm 2}) + j(k_{\pm 1}) \\ &\leq 2M_1 + \frac{5}{\eta}(7\pi L\frac{4}{\eta}(c\eta)^{\gamma_1}M + 2M_2) + 7\pi L\frac{3d}{\beta\eta^2}(c\eta)^{\gamma_1}M \\ &< 8\pi L(\frac{5}{\eta})^2(c\eta)^{\gamma_1}M + 2M_1. \end{aligned}$$

For  $w(k_n)$  we first consider  $|n| \ge 3$ . By integrating  $\partial_s w(k_n)$  we deduce

$$|w(k_n)| \le |w(k_n, -d)| + \frac{2}{\eta} (|w(k_{n+1})|_{L_s^{\infty}} + |w(k_{n-1})|_{L_s^{\infty}}) + \int j(k_n) < 7\pi L(\frac{5}{\eta})^{|n|-1} (c\eta)^{\gamma_1} M + 2M_n.$$

For the cases  $n = \pm 2$  we similarly conclude that

$$\begin{aligned} |w(k_{\pm 2})| &\leq |w(k_{\pm 2}, -d)| + \frac{2}{\eta} (|u_2|_{L_s^{\infty}} + |u_3|_{L_s^{\infty}} + |w(k_{\pm 3})|_{L_s^{\infty}}) \\ &< 7\pi L \frac{5}{\eta} (c\eta)^{\gamma_1} M + 2M_{\pm 2}. \end{aligned}$$

For the estimates on the current j we argue similarly as on the interval  $I_1$  and introduce  $\hat{s}$  as the shifted time coordinates. To estimate  $j(k_{\pm 1})$  we integrate  $\partial_s j(k_{\pm 1})$  in time:

$$j(k_{\pm 1}) = 2 \exp(-\frac{1}{2}\kappa\xi\eta(s+d))j(k_{\pm 1}, -d) + \frac{\kappa_{k_{\pm 1}}}{\beta} \int \frac{1+\hat{s}^2}{1+\hat{\tau}^2} \exp(-\kappa_{k_{\pm 1}}(\hat{s}-\hat{\tau}+\frac{1}{3}(\hat{s}^3-\hat{\tau}^3)))(u_2\pm u_3).$$

The impact of  $j(k_{\pm 1})$  is bounded by

$$\int j(k_{\pm 1}) \leq \frac{4}{\kappa \xi \eta} j(k_{\pm 1}, -d) + \frac{2d}{\beta \eta^2} (|u_2|_{L_s^{\infty}} + |u_3|_{L_s^{\infty}}) \\ \leq 7\pi L \frac{3d}{\beta \eta^2} (c\eta)^{\gamma_1} M$$

and hence yields the estimate

$$\begin{aligned} j(k_{\pm 1}) &\leq 2 \exp(-\frac{1}{2} d\kappa \xi \eta) j(k_{\pm 1}, -d) + \frac{4}{\beta \eta^2} (|u_3|_{L^{\infty}_s} + |u_2|_{L^{\infty}_s}) \\ &< \frac{4}{\beta \eta^2} (7\pi L(c\eta)^{\gamma_1} M + 2M_1). \end{aligned}$$

By integrating we thus obtain the following estimate for  $j(k_n)$ :

$$j(k_n) = 2 \exp(-\kappa \xi \eta(s+d)) j(k_n, -d) + \frac{\kappa_{k_n}}{\beta} \int \frac{1+\hat{s}^2}{1+\hat{\tau}^2} \exp(-\kappa_{k_n}(\hat{s}-\hat{\tau}+\frac{1}{3}(\hat{s}^3-\hat{\tau}^3))) w(k_n),$$

which leads to

$$\int j(k_n) = \frac{1}{\kappa \xi \eta} j(k_n, -d) + \frac{1}{\beta} \frac{d}{\eta^2} |w(k_n)|_{L^{\infty}_s}$$
$$\leq \frac{2}{\beta} \frac{d}{\eta^2} (7\pi L \frac{4}{\eta} (c\eta)^{\gamma_1} M + 2M_n)$$

and

$$j(k_n) \le 2 \exp(-\kappa \xi d\eta) j(k_n, -d) + \frac{4}{\beta \eta^2} |w(k_n)|_{L_s^\infty} < \frac{4}{\beta \eta^2} (7\pi L(\frac{5}{\eta})^{|n|-1} (c\eta)^{\gamma_1} M + 2M_n).$$

We estimate  $j(k_0)$  by integrating

$$j(k_0) = \frac{1+s^2}{1+d^2} \exp(-\kappa_{k_0}(s+d+\frac{1}{3}(s^3+d^3))j(k_0,-d) + \frac{\kappa_{k_0}}{\beta} \int \frac{1+s^2}{1+\tau^2} \exp(-\kappa_{k_0}(s-\tau+\frac{1}{3}(s^3-\tau^3)))u_1(\tau).$$

The second term can be estimated by

$$\frac{\kappa_{k_0}}{\beta} \int \frac{1+s^2}{1+\tau^2} \exp(-\kappa_{k_0}(s-\tau+\frac{1}{3}(s^3-\tau^3))) u_1(\tau) \\ \leq \frac{1}{\beta} \min(\kappa_{k_0} \pi d^2, 1) |u_1|_{L_s^{\infty}}$$

and thus

$$j(k_{0},d) = \exp(-\kappa_{k_{0}}(2d + \frac{2}{3}d^{3})j(k_{0},-d) + \frac{1}{\beta}\min(\kappa_{k_{0}}\pi d^{2},1)|u_{1}|_{L_{s}^{\infty}} \leq \exp(-\kappa_{k_{0}}(2d + \frac{2}{3}d^{3})\frac{c^{2}}{\beta}M(c\eta)^{-\gamma_{2}}\min(\kappa_{k_{0}},1) + \frac{3LM}{\beta}\min(\kappa_{k_{0}}\pi d^{2},1)(c\eta)^{-\gamma_{2}} < \frac{4LM}{\beta}\min(\kappa_{k_{0}}\pi d^{2},1)(c\eta)^{-\gamma_{2}}.$$

### 3.4.4 The Echo and Lower Bounds in the Interval $I_2$

In this section we establish the echo mechanism on the interval  $I_2$ , i.e. our aim is to show that the mode  $u_1$  induces growth of the  $u_2$  mode. For this echo mechanism we need the additional assumption

$$\kappa k_0^2 \min(\beta, 1) > \frac{1}{c}.\tag{3.32}$$

As shown in Subsection 3.2.3, this is not only a technical assumption. When  $k_0$  is too small, the  $u_1$  term can become negative due to the action of j and hence negate the growth of  $u_2$  and we could even obtain  $u_2(d) \approx 0$ . We will use initial data of the form

$$u_{1}(-d) = 1,$$
  

$$u_{2}(-d) \leq 50c^{2}\eta,$$
  

$$|j|(k_{0}, -d) \leq \frac{2c^{2}}{\beta},$$
  

$$|w|(k, -d), |u_{3}|(-d) \leq 5(c\eta)^{\gamma_{2}},$$
  

$$|j|(k, -d) \leq \frac{20}{\eta}(c\eta)^{\gamma_{2}}.$$
  
(3.33)

Which corresponds to the echoes on  $I_1$  normalized in terms of u(-d). We will prove that u closely matches the following asymptotics:

$$\tilde{u}_1 = \exp(-\frac{1}{\beta}(\tan^{-1}(s) + \tan^{-1}(d))),$$
  
$$\tilde{u}_2 = u_2(-d) + 2c\eta\beta(1 - \exp(-\frac{1}{\beta}(\tan^{-1}(s) + \tan^{-1}(d)))).$$

**Proposition 3.25.** Consider a solution of (3.25) with initial data (3.33), then the following estimates hold:

$$|u_{1}(d) - \tilde{u}_{1}(d)| = 12\pi c,$$
  

$$|u_{2}(d) - \tilde{u}_{2}(d)| \leq 24\pi c^{2}\eta,$$
  

$$w(k_{n}, d), u_{3}(d) \leq 6(c\eta)^{\gamma_{2}},$$
  

$$j(k_{n}, d) \leq \frac{25}{\beta\eta^{2}}(c\eta)^{\gamma_{2}},$$
  

$$j(k_{0}, d) \leq \frac{2}{\beta}.$$
  
(3.34)

In the following it is convenient to introduce the good unknown:

$$g(s) = (1+s^2)j - \frac{u_1}{\beta},$$

In terms of g our equations then read

$$\partial_s u_1 = -\frac{1}{\beta} \frac{1}{1+s^2} u_1 - a_1 u_2 + a_2 u_3 - \frac{1}{1+s^2} g$$

 $\quad \text{and} \quad$ 

$$\begin{split} \partial_s g &= 2sj(k_0) + (1+s^2)\partial_s j(k_0) - \frac{1}{\beta}\partial_s u_1 \\ &= \frac{2s}{1+s^2}(g + \frac{1}{\beta}u_1) \\ &- \kappa_{k_0}(1+s^2)g + \frac{2s}{1+s^2}(g + \frac{1}{\beta}u_1) \\ &+ \frac{1}{\beta^2}\frac{1}{1+s^2}u_1 + \frac{1}{\beta}a_1u_2 - \frac{1}{\beta}a_2u_3 + \frac{1}{\beta}\frac{1}{1+s^2}g \\ &= (\frac{4s+\frac{1}{\beta}}{1+s^2} - \kappa_{k_0}(1+s^2))g \\ &+ \frac{1}{\beta}\frac{4s+\frac{1}{\beta}}{1+s^2}u_1 + \frac{1}{\beta}a_1u_2 - \frac{1}{\beta}a_2u_3. \end{split}$$

Therefore, (3.25) can be equivalently expressed as

$$\partial_{s} \begin{pmatrix} u_{1} \\ u_{2} \\ u_{3} \\ g \end{pmatrix} = \begin{pmatrix} -\frac{1}{\beta} \frac{1}{1+s^{2}} & -a_{1} & a_{2} & -\frac{1}{1+s^{2}} \\ 2c\eta \frac{1}{1+s^{2}} & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \\ \frac{1}{\beta} \frac{4s+\frac{1}{\beta}}{1+s^{2}} & \frac{1}{\beta}a_{1} & \frac{1}{\beta}a_{2} & \frac{4s+\frac{1}{\beta}}{1+s^{2}} - \kappa_{k}(1+s^{2}) \end{pmatrix} \begin{pmatrix} u_{1} \\ u_{2} \\ u_{3} \\ g \end{pmatrix} \\ + \begin{pmatrix} 0 \\ a(k\pm 2)w(k\pm 2) - j(k\pm 1) \\ \pm a(k\pm 2)w(k\pm 2) \mp j(k\pm 1) \\ 0 \end{pmatrix}.$$
(3.35)

The homogeneous system with respect to (3.35) is given by

$$\partial_s \left(\begin{array}{c} \tilde{u}_1\\ \tilde{u}_2 \end{array}\right) = \left(\begin{array}{c} -\frac{1}{\beta}\frac{1}{1+s^2} & 0\\ 2c\eta\frac{1}{1+s^2} & 0 \end{array}\right) \left(\begin{array}{c} \tilde{u}_1\\ \tilde{u}_2 \end{array}\right)$$
(3.36)

with the explicit solution

$$\tilde{u}_1 = \exp(-\frac{1}{\beta}(\tan^{-1}(s) + \tan^{-1}(d)))u_1(-d)$$
  

$$\tilde{u}_2 = u_2(-d) + 2c\eta\beta(1 - \exp(-\frac{1}{\beta}(\tan^{-1}(s) + \tan^{-1}(d))))u_1(-d).$$
(3.37)

In the following, we prove that the solution of (3.35) can be treated as a perturbation of (3.37). Note that we can approximate

$$\beta(1 - \exp(-\frac{1}{\beta}(\tan^{-1}(s) + \tan^{-1}(d)))) \approx \min(\beta, (\tan^{-1}(s) + \tan^{-1}(d))),$$

where " $\approx$ " in this case corresponds to the explicit bounds

$$\frac{1}{2}\min(\beta,\cdot) \le \beta(1 - \exp(-\frac{1}{\beta}\cdot)) \le \min(\beta,\cdot)$$

Proof of Proposition 3.25. We want to show by a bootstrap that

$$|u_{1} - \tilde{u}_{1}| \leq c_{1} = 12\pi c$$

$$|u_{2} - \tilde{u}_{2}| \leq c_{2} = (2\pi + 1)c\eta c_{1}$$

$$|u_{3}|, |w(k_{n})| \leq 6(c\eta)^{\gamma_{2}}$$

$$\int j(k_{n}) \leq \frac{13d}{\beta\eta^{2}}(c\eta)^{\gamma_{2}}$$

$$\int j(s, k_{\pm 1}) \leq \frac{10\pi}{\beta\eta}.$$
(3.38)

Let  $s^*$  be the maximal time such that (3.38) holds. We assume that  $s^* \leq d$  and show that this leads to a contradiction by improving (3.38). The estimates of  $j(k_n)$  for  $n \neq 0$  are done similarly as in Proposition 3.23 and we hence omit them here. First, we estimate g:

$$g_{0}(s) = \frac{(1+s^{2})^{2}}{(1+d^{2})^{2}} \exp\left(\frac{1}{\beta}(\tan^{-1}(s) + \tan^{-1}(d)) - \kappa_{k_{0}}(s + d + \frac{1}{3}(s^{3} + d^{3})))g(-d),\right.$$
  
$$g(s) - g_{0}(s) = \frac{1}{\beta} \int \frac{(1+s^{2})^{2}}{(1+\tau^{2})^{2}} \exp\left(\frac{1}{\beta}(\tan^{-1}(s) - \tan^{-1}(\tau)) - \kappa_{k_{0}}(s - \tau + \frac{1}{3}(s^{3} - \tau^{3}))\right) \left. \left(\frac{4\tau + \frac{1}{\beta}}{1+\tau^{2}}u_{1}(\tau) + \frac{1}{\beta}a_{2}u_{2}(\tau) - \frac{1}{\beta}a_{3}u_{3}(\tau)\right) d\tau.$$

Next we estimate the size of the perturbations by

$$\int d\tau_2 \; \frac{\exp(-\frac{1}{\beta}(\tan^{-1}(s)-\tan^{-1}(\tau_2)))}{1+\tau_2^2} (g-g_0)(\tau_2)$$

$$= \frac{1}{\beta} \int d\tau_2 \int d\tau_1 \; \exp\left(-\frac{1}{\beta}(\tan^{-1}(s)-\tan^{-1}(\tau_1)) - \kappa_{k_0}(\tau_2-\tau_1+\frac{1}{3}(\tau_2^3-\tau_1^3))\right)$$

$$\cdot \frac{1+\tau_2^2}{(1+\tau_1^2)^2} \left(\frac{4\tau_1+\frac{1}{\beta}}{1+\tau_1^2}u_1(\tau_1) + \frac{1}{\beta}a_2u_2(\tau_1) - \frac{1}{\beta}a_3u_3(\tau_1)\right)$$

$$\leq \frac{2}{\beta\kappa_{k_0}} \int d\tau_1 \; \exp(-\frac{1}{\beta}(\tan^{-1}(s)-\tan^{-1}(\tau_1)))$$

$$\cdot \frac{1}{(1+\tau_1^2)^2} \left(\frac{4\tau_1+\frac{1}{\beta}}{1+\tau_1^2}u_1(\tau_1) + \frac{1}{\beta}a_2u_2(\tau_1) - \frac{1}{\beta}a_3u_3(\tau_1)\right)$$

$$\leq \frac{2}{\kappa_{k_0\beta}}(2|u_1|_{L_s^\infty} + \frac{4c}{\eta}(|u_2|_{L_s^\infty} + |u_3|_{L_s^\infty}))$$

and

$$\int \frac{\exp(-\frac{1}{\beta}(\tan^{-1}(s)-\tan^{-1}(\tau_{2})))}{1+\tau_{2}^{2}}g_{0}(\tau_{2}) d\tau_{2}$$

$$= c^{4}\exp(-\frac{1}{\beta}(\tan^{-1}(s)+\tan^{-1}(d)))g(-d)$$

$$\cdot \int (1+\tau_{2}^{2})\exp(-\kappa_{k_{0}}(\tau_{2}+d+\frac{1}{3}(\tau_{2}^{3}+d^{3})))g_{0}(-d)$$

$$= \frac{c^{4}}{\kappa_{k_{0}}}\exp(-\frac{1}{\beta}(\tan^{-1}(s)+\tan^{-1}(d)))\frac{3}{\beta}g(-d) \leq \frac{3}{\beta}\frac{c^{4}}{\kappa_{k_{0}}},$$

where we used (3.33) to obtain  $g(-d) \leq \frac{3}{\beta}$ . To estimate  $u_1$  we look at the difference to the homogeneous system,

$$\partial_s(u_1 - \tilde{u}_1) = -\frac{1}{\beta} \frac{1}{1+s^2} (u_1 - \tilde{u}_1) - a_1 u_2 + a_3 u_3 - \frac{1}{1+s^2} g$$

which leads after integrating to

$$\begin{aligned} |u_1 - \tilde{u}_1| &\leq \frac{4}{\eta} (|u_2|_{L_s^{\infty}} + |u_3|_{L_s^{\infty}}) + \frac{1}{\kappa_{k_0}} (\frac{2}{\beta} |u_1|_{L_s^{\infty}} + \frac{4c}{\eta\beta} (|u_2|_{L_s^{\infty}} + |u_3|_{L_s^{\infty}})) + \frac{3}{\beta} \frac{c^4}{\kappa_{k_0}} \\ &\leq 8c \min(\beta, \pi) + 4(2\pi + 1)cc_1 + \frac{4}{\kappa_{k_0}\beta} (1 + c_1) + \frac{5}{\eta} 6(c\eta)^{\gamma_2} < c_1 \end{aligned}$$

since  $\kappa k_0^2 \ge \frac{1}{\beta c}$ . We estimate  $u_2 - \tilde{u}_2$  by

$$\partial_s(u_2 - \tilde{u}_2) = 2c\eta \frac{1}{1+s^2}(u_1 - \tilde{u}_1) + a(k_{\pm 2})w(k_{\pm 2}) - j(k_{\pm 1})$$

which implies by integrating in s, that

$$\begin{aligned} |u_2 - \tilde{u}_2| &\leq 2\pi c\eta c_1 + \frac{2}{\eta} |w(k_{\pm 2})|_{L^{\infty}_s} + \int j(k_{\pm 1}) \\ &\leq 2\pi c\eta c_1 + (12 + 10\pi \frac{1}{\beta}) c^{\gamma_2} \eta^{-\gamma_1} \\ &< (2\pi + 1) c\eta c_1. \end{aligned}$$

Next we estimate  $w(k_n)$  for  $|n| \ge 3$ . We remark that the estimates for  $u_3$  and  $w_{k_{\pm 2}}$  are similar and hence we omit them. By integrating over the derivative we deduce

$$w(k_n, -d) \le w(k_n, d) + \frac{2}{\eta} (|w(k_{n+1})|_{L_s^{\infty}} + |w(k_{n-1})|_{L_s^{\infty}}) + \int j(k_n) < 6(c\eta)^{\gamma_2}.$$

So the bootstrap is concluded. It is left to estimate  $j(k_0)$ . We write

$$\partial_s j(k_0) = \frac{\kappa_{k_0}}{\beta} u_1 + (\frac{2s}{1+s^2} - \kappa_{k_0}(1+s^2))j(k_0) \\ \leq \frac{\kappa_{k_0}}{\beta} u_1 - \frac{8}{9}\kappa_{k_0}j(k_0)$$

where in the second line we used (3.32). By integrating, we obtain

$$j(k_0, s) \le \exp(-\frac{8}{9}\kappa_{k_0}(s+d))j(k_0, -d) + \frac{\kappa_{k_0}}{\beta} \int_{-d}^{s} d\tau \exp(-\frac{8}{9}\kappa_{k_0}(s-\tau))u_1(\tau)$$

which leads to

$$j(k_0, d) \leq \exp(-2d\frac{8}{9}\kappa_{k_0})j(k_0, -d)$$
  
+  $\frac{9}{8}\frac{1}{\beta}|u_1|_{L^{\infty}_s}$   
 $\leq \frac{2}{\beta}.$ 

#### 3.4.5 Proof of Theorem 3.14

In Subsections 3.4.2, 3.4.3 and 3.4.4 we proved lower and upper bounds until the time s = d. Furthermore, in Subsection 3.4.2 we already showed the asymptotic behavior on the interval  $I_3$ . In this subsection we need to combine the results of these subsections to obtain the final lower and upper bounds for the complete interval  $I^k$ . This will be achieved in two steps: first we conclude the bootstrap on  $I_3$ , afterwards we show that all terms result in the desired estimates.

Proof of Theorem 3.14. Following we proceed similarly as in the proof of Proposition 3.17, just for  $I_3$ . In particular we use the tools from Subsection 3.4.2. We thus need to prove the missing estimate on  $[d, s_1]$ . Let  $r_i(d)$  be the initial data of r(s). We define the  $c_i$  terms by

$$c_{1} = 2(r_{1}(d) + (21c^{2} + 2\frac{c^{3}}{\beta}(c\eta)^{\gamma})r_{2}(s_{0}) + N + N_{j})$$

$$\tilde{c}_{1} = 0$$

$$c_{2} = \frac{1}{1-2\frac{c}{\beta}}(r_{2}(d) + N + N_{j})$$

$$\tilde{c}_{2} = 22c_{1} + \frac{c}{\beta}\tilde{c}_{2}.$$
(3.39)

and

$$N = 2c \frac{1}{\kappa \xi \eta^{\gamma_2}} j(k_0 \pm 1, d) + 2c u_3(d)$$
  
+  $2 \sum_{|m| \ge 2} (2c)^{|m|} (w + \frac{4}{\kappa \eta \xi} j)(k_m, d)$   
 $N_j = 4(c\eta)^{\gamma_2} \frac{c^2}{\kappa_{k_0}} j(k_0, d).$ 

We prove by bootstrap that

$$|u|(s) \le S^*(s)C(s). \tag{3.40}$$

Since  $c_i \ge r_i(d)$  this estimate holds locally, and we again let  $s^*$  be the maximal time such that (3.40) holds. We assume that  $s^* \le s_1$  and improve the estimate, which gives a contradiction and thus proves that (3.40) holds on  $[d, s_1]$ . For the  $R_i$  we obtain with the Lemmas 3.20, 3.21, 3.22 that

$$\begin{split} R_1[F_{all}] &= R_1[F_{3mode}] + R_1[F_j] + R_1[F_{\tilde{w}}] + R_1[F_{j(k_0\pm 1)}] + R_1[F_{u_3}] \\ &\leq 20c^2c_1 + 20c^4\tilde{c}_2 + (20c^2 + c^4(c\eta)^{\gamma})c_2 \\ &+ \frac{c^3}{\beta}(c_1 + \tilde{c}_2) + \frac{c^3}{\beta}(c\eta)^{\gamma}c_2 + (c\eta)^{\gamma_1}\frac{4c^4}{\kappa_{k_0}}j(k_0,d) \\ &+ 2c^2(\tilde{w}(2) + \tilde{w}(-2)) \\ &+ \frac{2c}{\kappa\xi\eta}j(k_{\pm 1},d) + \frac{2c}{\beta\kappa\xi}(\tilde{w}(1) + c_1^* + c_2^*) \\ &+ 2c\tilde{w}(1) \\ &\leq 21c^2c_1 + 2\frac{c^3}{\beta}\tilde{c}_2 + (21c^2 + \frac{c^3}{\beta}(c\eta)^{\gamma})c_2 + N + c^2(c\eta)^{\gamma}N_j, \end{split}$$

and

$$\begin{split} R_{2}[F_{all}] &= R_{2}[F_{3mode}] + R_{2}[F_{j}] + R_{2}[F_{\tilde{w}}] + R_{2}[F_{j(k_{0}\pm1)}] + R_{2}[F_{u_{3}}] \\ &\leq 20(\frac{s}{\eta})^{\gamma}(c_{1}+2c^{2}\tilde{c}_{2}) + 20c^{2}c_{2} \\ &+ \frac{c}{\beta}(\frac{s}{\eta})^{\gamma}(c_{1}+\tilde{c}_{2}) + \frac{c}{\beta}c_{2} + (c\eta)^{\gamma_{2}}\frac{c^{2}}{\kappa_{k_{0}}}j(k_{0},d)) \\ &+ c^{2}(\tilde{w}(2) + \tilde{w}(-2))(\frac{s}{\eta})^{\gamma_{1}} \\ &+ \frac{2c}{\kappa\xi\eta}j(k_{\pm1},d)(\frac{s}{\eta})^{\gamma_{1}} + \frac{c}{\beta\kappa\xi}(\tilde{w}(1) + c_{1}^{*} + c_{2}^{*})(\frac{s}{\eta})^{\gamma_{1}} \\ &+ 2c\tilde{w}(1)(\frac{s}{\eta})^{\gamma_{1}} \\ &\leq 21(\frac{s}{\eta})^{\gamma}c_{1} + 2\frac{c}{\beta}c_{2} + \frac{c}{\beta}(\frac{s}{\eta})^{\gamma}\tilde{c}_{2} + N(\frac{s}{\eta})^{\gamma} + N_{j}. \end{split}$$

Therefore, we deduce that

$$\begin{aligned} r_1(s_0) + R_1[all][1] &\leq r_1(d) + 21c^2(c_1 + \tilde{c}_2) + (21c^2 + \frac{c^3}{\beta}(c\eta)^{\gamma})c_2 \\ &+ N + c^2(c\eta)^{\gamma}N_j < c_1, \\ r_2(s_0) + R_2[all][1] &\leq r_2(d) + 2\frac{c}{\beta}c_2 + N + N_j < c_2, \\ R_2[all][(\frac{s}{\eta})^{\gamma}] &< 20c_1 + 2c^2\tilde{c}_2 < \tilde{c}_2. \end{aligned}$$

This concludes the bootstrap and we estimated  $|u|(s) \leq S^*(s)C(s)$  for  $s \leq s_1$ . To finish the proof of the theorem we need to establish the norm estimate at the final time. With Proposition 3.23 we obtain the following bounds:

$$\begin{aligned} &|u_1|(d) \le 3(c\eta)^{-\gamma_2} LM, \\ &|u_2|(d) \le 7\pi (c\eta)^{\gamma_1} LM, \\ &|u_3|(d) \le 7\pi (\frac{5}{\eta})^2 (c\eta)^{\gamma_1} LM + 2M_1, \\ &|w(k_n, d)| \le 7\pi (\frac{5}{\eta})^{|n|-1} (c\eta)^{\gamma_1} LM + 2M_n, \\ &j(k_n, d) \le \frac{4}{\eta^2 \beta} (7\pi (\frac{4}{\eta})^{|n|-1} (c\eta)^{\gamma_1} LM + 2M_n), \\ &j(k_0, d) \le \frac{4LM}{\beta} \min(\kappa_{k_0} \pi d^2, 1) (c\eta)^{-\gamma_2}. \end{aligned}$$

This in turn yields

$$N = 2c \frac{1}{\kappa \xi \eta^{\gamma_2}} j(k_0 \pm 1, d) + 2cu_3(d) + 2 \sum_{|m| \ge 2} (2c)^{|m-k_0|} (w + \frac{8}{\kappa \eta \xi n^2} j)(k_m, d) \le c(c\eta)^{\gamma} LM, N_j \le 4(c\eta)^{\gamma_2} \frac{c^2}{\kappa_{k_0}} \frac{4LM}{\beta} \min(\kappa_{k_0} \pi d^2, 1)(c\eta)^{-\gamma_2} \le \frac{16}{\beta} \min(\pi, \frac{c^2}{\kappa_{k_0}}) LM.$$

Using these bounds, we consider Afterwards, we estimate

$$r(d) = S^{-1}(d)u(d)$$

$$= -2c\gamma^{-1} \begin{pmatrix} -\frac{\gamma_2}{2c}|c\eta|^{-\gamma_2+1} & -|c\eta|^{-\gamma_2} \\ \frac{\gamma_1}{2c}|c\eta|^{-\gamma_1+1} & |c\eta|^{-\gamma_1} \end{pmatrix} u(d)$$

$$\begin{vmatrix} r_1 \\ r_2 \end{vmatrix} \end{pmatrix} (d) \le LM \begin{pmatrix} 15\pi c(c\eta)^{\gamma} \\ 4 \end{pmatrix}$$

and hence deduce that

$$c_1 = (c\eta)^{\gamma} (30\pi c + 30\frac{c^2}{\beta} + c)LM$$
  
$$\leq 31\pi c (c\eta)^{\gamma}LM$$
  
$$c_2 = (5 + \frac{16\pi}{\beta})LM.$$

This implies the estimate

$$u(s_1) \leq S^*(s_1)C(s_1)$$
  

$$\leq LM \left(\begin{array}{cc} \frac{1}{2} & 1\\ \frac{1}{2c} & 2c \end{array}\right) \left(\begin{array}{c} 31\pi c(c\eta)^{\gamma}\\ (5+\frac{16\pi}{\beta}) \end{array}\right)$$
  

$$\leq LM(c\eta)^{\gamma} \left(\begin{array}{c} 16c + (5+\frac{16\pi}{\beta})(c\eta)^{-\gamma}\\ 16\pi \end{array}\right),$$

where we used that  $(c\eta)^{-\gamma} \frac{1}{\kappa_{k_0}} = (\frac{k_0^2}{c\xi})^{\gamma} \frac{1}{\kappa k_0^2} = \frac{1}{(c\xi)^{\gamma} \kappa k_0^{2\gamma_2}} \ll \beta c$ . For  $\tilde{w}(n)$  we obtain

$$\begin{split} \tilde{w}(n) &= 2 \sum_{|m| \ge 2} (2c)^{|m-n| + \chi} (w + \frac{4}{\kappa \eta \xi} j)(k_m, d) \\ &+ (2c)^{||n| - 2|} c(c_1^* + \frac{1}{c^2} c_2^*) \\ &+ (2c)^{|c| - 1} u_3(d) \\ &\le L(2c)^{|n|} M + M_n \\ u_3(n) &\le L(2c)^{|n| + 2} M + M_1. \end{split}$$

Furthermore, by integrating over  $\partial_s j(k_n)$  we obtain

$$j(k_n, s_1) \le L \frac{5}{\kappa \xi \eta} ((2c)^{|n|} M + M_n),$$
  
$$j(k_{\pm 1}, s_1) \le L \frac{5}{\kappa \xi \eta} ((2c)^{|n|+2} M + M_1).$$

In order to estimate  $j(k_0)$  we use Lemma 3.21:

$$\begin{aligned} |j(k_0, s_1)| &\leq Lc^2 \eta^2 \exp(-\kappa_{k_0} \eta^3) j(k_0, d) \\ &+ 2 \frac{16^2}{\beta} \frac{1}{\eta^2} (c_1 + c_2 + \tilde{c}_1 + \tilde{c}_2) \\ &\leq 3\pi \frac{\kappa_{k_0}}{\beta} \eta^2 \exp(-\kappa_{k_0} \eta^3) (\frac{d}{\eta})^{\gamma_2} M \\ &+ L4 \frac{16^2}{\beta} \frac{1}{\eta^2} 2M (c\eta)^{-\gamma_2} \\ &\leq \frac{2^{11}}{\beta} \frac{1}{\eta^2} M (c\eta)^{-\gamma_2}. \end{aligned}$$

We further estimate

$$\begin{split} M^{2} &= \sum_{m,n \geq 1} 10^{-m-n} (w + \frac{1}{\alpha_{k_{n}}} j)(k_{n})(w + \frac{1}{\alpha_{k_{m}}} j)(k_{m}) \\ &\leq \frac{2}{1-10^{-1}} \sum_{n \geq 1} 10^{-n} (w^{2} + \frac{1}{\alpha_{k_{n}}^{2}} j^{2})(k_{n}) \\ &\leq \frac{2}{1-10^{-1}} \frac{1}{\lambda_{k_{0}}} \sum_{n \geq 1} (10^{-n} \frac{\lambda_{k_{0}}}{\lambda_{k_{n}}}) \lambda_{k_{n}} (w^{2} + \frac{1}{\alpha_{k_{n}}^{2}} j^{2})(k_{n}) \\ &\leq \frac{2}{1-10^{-1}} \frac{1}{\lambda_{k_{0}}} \|w, j\|_{X} (s_{0})^{2} \\ \sum_{|n|\geq 1} \lambda_{k_{n}} M_{n}^{2} &= \sum_{n} \lambda_{k_{n}} \sum_{m,l} 10^{-|m-l|-|l-n|-\chi_{l}-\chi_{m}} (w + \frac{1}{\alpha_{k_{m}}} j)^{2}(k_{m}) \\ &\leq \frac{2}{1-10^{-1}} \sum_{n} \lambda_{k_{n}} \sum_{m} 10^{-|m-n|-\chi_{m}} (w^{2} + \frac{1}{\alpha_{k_{m}}^{2}} j^{2})(k_{m}) \\ &\leq \frac{2}{1-10^{-1}} \sum_{n} \lambda_{k_{m}} (w + \frac{1}{\alpha_{k_{m}}} j)^{2}(k_{m}) \sum_{n} 10^{-|m-n|-\chi_{m}} \frac{\lambda_{k_{n}}}{\lambda_{k_{m}}} \\ &\leq \frac{2\hat{\lambda}^{2}}{1-10^{-1}} \|w, j\|_{X}^{2}. \end{split}$$

Combining these bounds we infer the norm estimate

$$\begin{split} \|w, j\|_X^2(s_1) &\leq 16\pi L^2 M^2(c\eta)^{2\gamma} (\lambda_{k\pm 1} + \lambda_{k_0} (16\pi + 5(c\eta)^{-2\gamma})^2) \\ &+ \sum_{|n|\geq 1} L^2 \lambda_{k_n} (10^{-|n|} M + M_n)^2 \\ &= M^2(c\eta)^{-2\gamma} (\lambda_{k_{\pm 1}^2} (2c)^2 + \lambda_{k_0}^2 + 2\sum_{|n|\geq 1} \lambda_{k_n}^2 10^{-2|n|} + L^2 \sum_{|n|\geq 1} \lambda_{k_n} M_n^2 \\ &\leq L^2 (\hat{\lambda} (16\pi)^2 + 3\hat{\lambda}^2) (c\eta)^{2\gamma} \|w, j\|_X^2(s_0). \end{split}$$

This finally allows us to complete the proof of the upper bound and obtain that

$$||w, j||_X(s_1) \le 18\pi L\hat{\lambda}(c\eta)^{\gamma} ||w, j||_X(s_0).$$

To prove the lower bound we use Proposition 3.17 and Proposition 3.25 and obtain that at time s = d it holds that

$$|u_1(d) - \exp(-\frac{\pi}{\beta})(c\eta)^{\gamma_2}| = O(c)$$
  

$$|u_2(d) - 2\beta(1 - \exp(-\frac{\pi}{\beta}))(c\eta)^{\gamma_1}| \le O(c)$$
  

$$w(k_n, d), u_3(d) \le 6$$
  

$$j(k_n, d) \le \frac{10\pi}{\beta\eta} \frac{1}{\eta}$$
  

$$j(k_0, d) \le \frac{2}{\beta}.$$

We calculate  $\tilde{u}_2$  by

$$\begin{split} \tilde{u}_{2}(s_{1}) &= (0 \ 1)S(s_{1})S^{-1}(d)u(d) \\ &\approx (\frac{1}{2c} \ 2c)2c \left( \begin{array}{cc} -c(c\eta)^{\gamma_{1}} & -(c\eta)^{-\gamma_{2}} \\ \frac{1}{2c}(c\eta)^{\gamma_{2}} & (c\eta)^{-\gamma_{1}} \end{array} \right)u(d) \\ &\approx (-c(c\eta)^{\gamma_{1}} + 2c(c\eta)^{\gamma_{2}})u_{1}(d) + (-(c\eta)^{-\gamma_{2}} + 4c^{2}(c\eta)^{-\gamma_{1}})u_{2}(d) \\ &\approx c(c\eta)^{\gamma_{1}}u_{1}(d) + (c\eta)^{\gamma_{2}}u_{2}(d) \\ &\approx 2(c\eta)^{\gamma_{1}}\beta(1 - \exp(-\frac{\pi}{\beta}))u_{1}(-d) \\ &\approx 2(c\eta)^{\gamma}\beta(1 - \exp(-\frac{\pi}{\beta}))u_{1}(s_{0}). \end{split}$$

The difference  $u_2 - \tilde{u}_2$  is estimated by

$$\begin{aligned} |u_2 - \tilde{u}_2| &\leq (0 \ 1) S^*(s_1) R[F] \\ &\leq \frac{1}{2c} R_1[F] + 2c R_2[F] \\ &\leq (c\eta)^{\gamma_2} u_2(d) + O(c) \\ &= 2(c\eta)^{\gamma} \beta (1 - \exp(-\frac{\pi}{\beta})) u_1(s_0) + O(c). \end{aligned}$$

Furthermore, we obtain

$$M \le \frac{1}{1 - 10^{-1}} u_1(\tilde{s}_0)$$
$$M_n \le \frac{4}{1 - 10^{-1}} u_1(\tilde{s}_0).$$

So we finally obtain since  $\beta \geq \frac{1}{5}$ 

$$w(k_{-1}, t_{k_{-1}}) \approx 2(c\eta)^{\gamma} \beta (1 - \exp(-\frac{\pi}{\beta})) u_1(s_0)$$
  
 
$$\geq \frac{1}{2} \max_l(w(k_l, t_{k_l}),$$

which gives

$$w(k_{-1}, t_{k_{-1}}) \ge (c\eta)^{\gamma} \min(\beta, \pi) w(k_{-1}, t_{k_{-1}}).$$

In this article we have studied the asymptotic (in)stability of the magnetohydrodynamic equations with a shear, a constant magnetic field and magnetic dissipation. Here multiple effects compete to determine the long time behavior of solutions:

- Echoes in the inviscid fluid equations may lead to large norm inflation.
- The underlying magnetic field leads to an exchange between kinetic and magnetic energy. In particular, for large magnetic fields oscillation my diminish norm inflation.
- Magnetic dissipation may stabilize the flow. Hence, a priori, it is not clear whether stability requires Gevrey regularity (as for the Euler equations) or Sobolev regularity (as for the fully dissipative problem) and how the evolution depends on the size of the magnetic field  $\alpha$  and on the resistivity  $\kappa$ .

As the main result of this article we show that the balance between these effects is parametrized by the parameter  $\beta = \frac{\kappa}{\alpha^2} > 0$  and that the behavior for finite, positive  $\beta$  strongly differs from both the fully non-dissipative case and the large dissipation limit (which reduces to the Euler equations). In particular, we show that in this regime the magnetic dissipation is not strong enough to stabilize the evolution in Sobolev regularity and establish Gevrey regularity as optimal both in terms of upper and lower bounds. It remains an interesting problem for future research to determine the optimal stability classes for other partial dissipation regimes and to study the inviscid limit  $\kappa \downarrow 0$ .

# 3.5 Estimating the Growth Factor

In Section 3.4.3 we observe the evolution of (3.25) on the interval  $I_2 = [-d, d]$ . Here we observe the interaction between j and  $u_1$ 

$$\partial_s u_1 = -j \partial_s j = \frac{K}{\beta} u_1 + (\frac{2s}{1+s^2} - K(1+s^2))j,$$
(3.41)

with  $\kappa_k$  replaced by K for simplicity In particular we bound the growth of  $u_1$  by a factor. Let  $U(\tau, s)$  be the solution of (3.41) with initial data  $u_1(\tau) = 1$  and  $j(\tau) = 0$ . We show that

- $|U(\tau,s)| \le 1$  for  $\beta \ge \frac{\pi}{2}$
- $|U(\tau,s)| \le L = L(\beta,K)$  for  $\beta < \frac{\pi}{2}$ .

With the restriction

$$c \le (8\pi)^{\frac{4}{3}} \beta^{\frac{16}{3}}.$$
(3.42)

we obtain

$$L(\beta, K) = \begin{cases} 1 & 1 \le K \\ \sqrt{d} & \frac{1}{2}c^{\frac{3}{4}} \le K \le 1 \\ 2(1 + \frac{\pi}{\beta}) & \frac{2\pi}{\beta}c^3 \le K \le \frac{1}{2}c^{\frac{3}{4}} \\ 1 & K \le \frac{2\pi}{\beta}c^3 \end{cases}$$
(3.43)

We note that (3.42) is not optimal, in the sense that Section 3.4.3 we need  $Lc \ll 1$  and we could optimize the  $\frac{1}{2}c^{\frac{3}{4}}$  term to obtain a larger L but better (3.42). However, this would yield a lot dependencies which would make the final theorem more technical to state. The most important part of this estimates is to verify that  $\beta$  can be very small if c is chosen small enough. First we do an energy estimate, let

$$E = u_1^2 + \frac{\beta}{K}j$$

which leads to

$$\frac{1}{2}\partial_s E \le (\frac{2s}{1+s^2} - K(1+s^2))_+ E.$$

Therefore, we obtain for  $K \ge 1$  that  $\partial_s E \le 0$ , which proves our first estimate. Furthermore, we infer for  $K \le 1$ 

$$E(s) \le E(\tau) \begin{cases} 1 & s \le 0\\ (1+s^2)^2 & 0 \le s \le (\frac{K}{2})^{\frac{-}{1}3}\\ 4(K)^{-\frac{4}{3}} & (\frac{K}{2})^{-\frac{1}{3}} \le s, \end{cases}$$

We conclude

$$u_1(s) \le \begin{cases} 1 & s \le 0\\ 1+s^2 & 0 \le s \le (\frac{K}{2})^{\frac{1}{3}}\\ 2(K)^{-\frac{2}{3}} & (\frac{K}{2})^{\frac{1}{3}} \le s \end{cases}$$

which proves (3.43) for  $\frac{1}{2}c^{\frac{3}{4}} \leq K \leq 1$ . For small K we need to make a different ansatz. We write j as,

$$j(s) = \frac{K}{\beta} \int_{-d}^{s} \frac{1+s^2}{1+\tau^2} \exp(-K(s-\tau + \frac{1}{3}(s^3 - \tau^3)))u(\tau) d\tau$$

and so

$$\begin{split} u(s) - 1 &= -\frac{K}{\beta} \iint_{-d \le \tau_1 \le \tau_2 \le s} d(\tau_1, \tau_2) \frac{1 + \tau_2^2}{1 + \tau_1^2} \exp(-K(\tau_2 - \tau_1 + \frac{1}{3}(\tau_2^3 - \tau_1^3))) u(\tau_1) \\ &= -\frac{1}{\beta} \int_{-d \le \tau_1 \le s} d\tau_1 \ u(\tau_1) \frac{1}{1 + \tau_1^2} [\exp(-K(\tau_2 - \tau_1 + \frac{1}{3}(\tau_2^3 - \tau_1^3)))]_{\tau_2 = \tau_1}^{\tau_2 = s} \\ &= -\frac{1}{\beta} \int_{-d \le \tau_1 \le s} d\tau_1 \ u(\tau_1) \frac{1}{1 + \tau_1^2} (1 - \exp(-K(s - \tau_1 + \frac{1}{3}(s^3 - \tau_1^3)))). \end{split}$$

Now we exploit that u is decreasing till the smallest time such that u(s) = 0. This holds, since if u is positive, then j is positive and so  $\partial_s u = -j \leq 0$ . Therefore, we bound

$$\frac{1}{\beta} \int_{-d \le \tau_1 \le s} d\tau_1 \ u(\tau_1) \frac{1}{1 + \tau_1^2} (1 - \exp(-K(s - \tau_1 + \frac{1}{3}(s^3 - \tau_1^3)))))$$

by 1 to deduce  $0 \le u(s) \le 1$ . Let s be positive, then we estimate

$$\begin{split} \frac{1}{\beta} \int_{-d \le \tau_1 \le s} d\tau_1 \frac{1}{1 + \tau_1^2} (1 - \exp(-K(s - \tau_1 + \frac{1}{3}(s^3 - \tau_1^3)))) \\ &= \frac{1}{\beta} \int_{-d \le \tau_1 \le -s} d\tau_1 \frac{1}{1 + \tau_1^2} (1 - \exp(-K(s - \tau_1 + \frac{1}{3}(s^3 - \tau_1^3)))) \\ &+ \frac{1}{\beta} \int_{-s \le \tau_1 \le s} d\tau_1 \frac{1}{1 + \tau_1^2} (1 - \exp(-K(s - \tau_1 + \frac{1}{3}(s^3 - \tau_1^3)))) \\ &\leq \frac{1}{\beta s} + \frac{\pi}{\beta} (1 - \exp(-K(2s + \frac{2}{3}s^3))) \\ &\leq \frac{1}{\beta s} + \frac{\pi}{\beta} K(2s + \frac{2}{3}s^3) \\ &\leq \frac{1}{\beta s} + \frac{\pi}{\beta} Ks^3. \end{split}$$

This term is less than zero if  $\frac{2}{\beta} \leq s \leq (\frac{\beta}{2\pi K})^{\frac{1}{3}}$ . We choose  $s = \min((\frac{\beta}{2\pi K})^{\frac{1}{3}}, d)$  maximal. When s = d, then  $(\frac{\beta}{2\pi K})^{\frac{1}{3}} \geq d$  which is satisfied if  $K \leq \frac{2\pi}{\beta}c^3$  and so we obtain the last estimate of (3.43). Now we need to prove the case if  $\frac{2\pi}{\beta}c^3 \leq K \leq \frac{1}{2}c^{\frac{3}{4}}$ , with the previous calculation we obtain for  $s_1 = (\frac{\beta}{2\pi K})^{\frac{1}{3}}$ , that  $0 \leq u(s_1) \leq 1$ . Then for  $s \geq s_1$  we have

$$\begin{aligned} u(s) - 1 &= \frac{1}{\beta} \int_{-d \le \tau_1 \le s} d\tau \ u(\tau) \frac{1}{1 + \tau^2} (1 - \exp(-K(s - \tau + \frac{1}{3}(s^3 - \tau^3)))) \\ |u(s) - 1| &\le \frac{1}{\beta} \int_{-d \le \tau \le s_1} d\tau \ \frac{1}{1 + \tau^2} + \frac{1}{\beta} \int_{t_1 \le \tau_1 \le s} d\tau \ u(\tau) \frac{1}{1 + \tau^2} \\ &\le \frac{\pi}{\beta} + \frac{1}{\beta s_1} |u|_{L_s^{\infty}}. \end{aligned}$$

Due to  $K \leq \frac{1}{2}c^{\frac{4}{3}}$  and (3.42) we obtain  $s_1\beta = (\frac{\beta^4}{2\pi K})^{\frac{1}{3}} \geq 2$  and so

$$|u(s)| \le \frac{1}{1 - \frac{1}{\beta s_1}} (1 + \frac{\pi}{\beta})$$
  
 $\le 2(1 + \frac{\pi}{\beta}).$ 

# 3.6 Nonlinear Instability of Waves

In this appendix we consider the nonlinear instability of the traveling waves.

$$\partial_t w + (v\nabla w)_{\neq} = \alpha \partial_x j + (b\nabla j)_{\neq} - (2c\sin(x)\partial_y \Delta_t^{-1}w)_{\neq}$$
  
$$\partial_t j + (v\nabla j)_{\neq} = \kappa \Delta_t j + \alpha \partial_x w + (b\nabla w)_{\neq} - 2\partial_x \partial_y^t \Delta_t^{-1} j - (2(\partial_i v\nabla)\partial_i \Delta^{-1} j)_{\neq},$$
  
(3.44)

For brevity of notation let us denote the Gevrey 2 norm with constant C by

$$\|(w,j)\|_{\mathcal{G}_C}^2 = \int \sum_k \exp(C\sqrt{|\xi|}) |\mathcal{F}(w,j)|^2 d\xi.$$

Then the norm inflation result of Theorem 3.3 further implies the nonlinear instability of any non-trivial traveling wave for C sufficiently small.

**Corollary 3.26.** Let  $0 < c < \min(10^{-4}, 10^{-3} \frac{\kappa}{\alpha^2})$  be given and consider a traveling wave as in Lemma 3.2 and let  $0 < C_2 < C_*$  where  $C_* = C_*(c)$  is as in Theorem 3.3. Then the nonlinear evolution equations around the traveling wave are unstable for small initial data in  $\mathcal{G}_{C_2}$  in the sense that for any  $0 < C_1 < C_2$ ,  $\epsilon > 0$  and N > 1 there exists initial data with

$$\|(w_0, j_0)\|_{\mathcal{G}_{C_2}} < \epsilon$$

but such that for some time T > 0 it holds that

$$||(w,j)|_{t=T}||_{\mathcal{G}_{C_2}} \ge N ||(w_0,j_0)||_{\mathcal{G}_{C_1}}$$

We stress that this results considers the instability of the traveling waves and that the space with respect to which instability is established depends on the size c of the wave. A nonlinear instability result for the underlying stationary state (3.2) in the spirit of [DM23, Bed20, DZ21] further requires that the size c of the traveling is comparable to  $\epsilon$ .

Proof of Corollary 3.26. We argue by contradiction. Thus suppose that the nonlinear solution is uniformly controlled in  $\mathcal{G}_{C_1}$  for all times:

$$\sup_{t>0} \|(w,j)\|_{\mathcal{G}_{C_1}} \le D\epsilon.$$

for some constant D > 0. Given this a priori control of regularity we may consider the nonlinear equations as a forced linear problem

$$\partial_t(w,j) + L(w,j) = F$$

where L is the linear operator considered throughout this article and F is the quadratic nonlinearity. If we denote by  $S(t, \tau)$  the solution operator associated to L it then follows that for any T > 0

$$(w,j)_{t=T} = S(T,0)(w_0,j_0) + \int_0^T S(T,\tau)F(\tau)d\tau.$$

By the norm inflation results of Theorem 3.3 for any  $C_2 < C_\ast$  there exists initial data and a time T>0 such that

$$\|S(T,0)(w_0,j_0)\|_{L^2} \ge N \|(w_0,j_0)\|_{\mathcal{G}_{C_2}}.$$
(3.45)

Since this estimate is linear after multiplication with a factor we may assume that this initial data also has size smaller than  $\epsilon$ . On the other hand, by the results of Section 3.3 and of Theorem 3.3 for any fixed time T,  $S(T, \tau)$  is uniformly bounded as a map from  $L^2$  to  $L^2$ . More precisely, we recall that  $S(T, \tau)$  decouples with respect to the frequency  $\xi$  in y.

- For  $\xi$  with  $|\xi| \gg T^2$  by the results of Section 3.3 the time interval (0, T) is considered "small time" and hence  $S(T, \tau)$  is bounded uniformly.
- If instead  $|\xi| \leq T^2$  then Theorem 3.3 provides an upper bound of the operator norm by  $\exp(C\sqrt{\xi}) \leq \exp(CT)$ .

Thus there exists an extremely large constant E (depending on T) such that

$$\|\int_0^T S(T,\tau)F(\tau)d\tau\|_{L^2} \le E\int_0^T \|F(\tau)\|_{L^2}d\tau.$$

Finally, we note that by assumption

$$\|F(\tau)\|_{L^2} \le D^2 \epsilon^2.$$

Hence, choosing  $\epsilon \ll \frac{1}{ED^2NT}$  the Duhamel integral can be treated as a perturbation of (3.45), which concludes the proof.

# Chapter 4

# On the Sobolev Stability Threshold for the 2D MHD Equations with Horizontal Magnetic Dissipation

This chapter is the preprint [KZ2] and is a joint work with Christian Zillinger.

#### NIKLAS KNOBEL AND CHRISTIAN ZILLINGER

Abstract. In this chapter we consider the stability threshold of the 2D magnetohydrodynamics (MHD) equations near a combination of Couette flow and large constant magnetic field. We study the partial dissipation regime with full viscous and only horizontal magnetic dissipation. In particular, we show that this regime behaves qualitatively differently than both the fully dissipative and the non-resistive setting.

# 4.1 Introduction

The equations of magnetohydrodynamics (MHD)

$$\partial_t V + V \cdot \nabla V + \nabla \Pi = (\nu_x \partial_x^2 + \nu_y \partial_y^2) V + B \cdot \nabla B,$$
  

$$\partial_t B + V \cdot \nabla B = (\kappa_x \partial_x^2 + \kappa_y \partial_y^2) B + B \cdot \nabla V,$$
  

$$\nabla \cdot v = \nabla \cdot b = 0,$$
  

$$(t, x, y) \in \mathbb{R}^+ \times \mathbb{T} \times \mathbb{R},$$
(4.1)

model the evolution of the velocity V of conducting, non-magnetic fluids interacting with a magnetic field B. The MHD equations are commonly used in

applications ranging from astrophysics and the description of plasmas to control problems for liquid metals in industrial applications [Dav16]. Similarly to the Navier-Stokes and Euler equations, questions of hydrodynamic stability and the behavior for high Reynolds numbers (that is, for  $\nu$ ,  $\kappa$  tending to zero) are a very active area of research both inner-mathematically and in view of applications.

Motivated by stability results for the isotropic full-dissipation case ( $\nu_x = \nu_y = \kappa_x = \kappa_y > 0$ ) and instability results for the non-resistive case ( $\kappa_x = \kappa_y = 0$ ), we are interested in the behavior of the two-dimensional magnetohydrodynamic (MHD) equations with partial dissipation, where some of the dissipation coefficients

$$\kappa_y, \kappa_x, \nu_x, \nu_y \ge 0,$$

are allowed to vanish. More specifically, we study the behavior near the stationary solution given by the combination of Couette flow and a (large) constant magnetic field

$$V_s = ye_1, \quad B_s = \alpha e_1, \tag{4.2}$$

for the case of vanishing vertical resistivity,  $\kappa_y = 0$ . By the symmetry  $B \mapsto -B$ , we consider the case  $\alpha \ge 0$ . For the related case of the Navier-Stokes equations (that is, without any magnetic field) the (in)stability of Couette flow at high Reynolds number is known as the Sommerfeld paradox [MB01] and is related to nonlinear instability of the Euler equations [BM15a, DM23, DZ21].

However, for the case of sufficiently small data it was proven in [BVW18] that (mixing enhanced) dissipation can counteract this instability in the Navier-Stokes equations and that (long time asymptotic) stability holds in Sobolev spaces for initial data with

$$\|\omega\|_{H^N} \le \epsilon \ll \nu^{\gamma}$$

with  $\gamma \geq \frac{1}{2}$ . Later in [MZ22] this has been improved to  $\gamma = \frac{1}{3}$ . This is an example of a stability threshold result, which establishes stability for small data and determines suitable (optimal) exponents  $\gamma$  for given norms.

Since the addition of the magnetic field is known to possibly destabilize the dynamics (see the following discussion), our main questions concern the MHD equations (4.1) in terms of perturbations moving with the underlying shear flow:

$$v(x, y, t) = V(x - yt, y, t) - V_s,$$
  
$$b(x, y, t) = B(x - yt, y, t) - B_s.$$

The corresponding perturbed equations in these new variables read

$$\partial_t v + v_2 e_1 - 2\partial_x \Delta_t^{-1} \nabla_t v_2 = \nu \cdot \Delta_t v + \alpha \partial_x b + b \nabla_t b - v \nabla_t v - \nabla_t \pi,$$
  

$$\partial_t b - b_2 e_1 = \kappa \cdot \Delta_t b + \alpha \partial_x v + b \nabla_t v - v \nabla_t b,$$
  

$$\nabla_t \cdot v = \nabla_t \cdot b = 0.$$
(4.3)

Here, we introduce the time-dependent derivatives  $\partial_y^t = \partial_y - t\partial_x$ ,  $\nabla_t = (\partial_x, \partial_y^t)$ and  $\Delta_t = \partial_x^2 + (\partial_y^t)^2$ . Furthermore, we use the following short notation for the dissipation operator:

$$\nu \cdot \Delta_t = \nu_x \partial_x^2 + \nu_y (\partial_y^t)^2,$$
  
$$\kappa \cdot \Delta_t = \kappa_x \partial_x^2 + \kappa_y (\partial_y^t)^2.$$

In this article we aim to establish a Sobolev stability threshold for (4.3) for the specific anisotropic, partial dissipation case

$$\kappa_y = 0, \ \kappa_x = \nu_x = \nu_y > 0.$$

In particular, we show that this setting exhibits qualitatively different behavior than the fully dissipative and the non-resistive case.

Following a similar notation as [Lis20] we make the following definition.

**Definition 4.1** (Stability threshold). Consider the MHD equations (4.1) with anisotropic dissipation  $0 < \nu_x = \nu_y = \kappa_x =: \mu \ll 1$  and  $\kappa_y = 0$  and let X be a Banach space with norm  $||(v, b)||_X$ . We then say that the exponent  $\gamma = \gamma(X)$  is a stability threshold for the space X if for initial data with

$$\|(v_{in}, b_{in})\|_X \le \epsilon \ll \mu^{\gamma},$$

the corresponding solution of (4.3) remains uniformly bounded for all future times with a quantitative control

$$\sup_{t>0} \|(v,b)\|_X \lesssim \epsilon.$$

We remark that this definition does not require optimality (that is, instability for smaller choices of  $\gamma$ ). Optimal stability thresholds quantify the appearance of instability in the large Reynolds number limit and are an active area of research for many fluid systems. In view of the large literature, the interested reader is referred to the following articles for the Navier-Stokes equations [BVW18, BGM17] and the Boussinesq equations [ZZ23, LWX<sup>+</sup>21, TWZZ20] for a discussion and further references.

For the (isotropic) MHD equations ( $\nu := \nu_x = \nu_y$  and  $\kappa := \kappa_x = \kappa_y$ ), there exists several results for non-vanishing magnetic dissipation.

- When considering full isotropic dissipation  $\nu = \kappa > 0$ , Liss [Lis20] established a Sobolev threshold in the 3D case. Under a Diophantine condition on the magnetic field, he establishes stability for  $||(v, b)||_{H^N}$  with  $\gamma = 1$ . For the 2D case an improvement to  $\gamma = \frac{2}{3}$  is expected due to the lack of lift-up instability. Indeed, in a very recent paper, [Dol24], Dolce establishes such a threshold for the regime  $0 < C\kappa^3 \le \nu \le \kappa$ .
- In the 2D inviscid case with isotropic magnetic dissipation,  $\nu = 0$  and  $\kappa > 0$ , in [KZ1] the authors established linear instability of nearby (in analytic regularity) so-called traveling wave type solutions in Gevrey 2 regularity. As an (almost) matching nonlinear result, [ZZ24] established a stability threshold  $\gamma \geq 1$  for Gevrey  $2 \delta$  regularity for any  $0 < \delta < 1$ .

• The setting with only an underlying magnetic field but without shear flow exhibits qualitatively different behavior and was studied for the case of the whole space in [BSS88, RWXZ14] in the full dissipation case and in [CRW13, JLWY19] for the partially dissipative case.

To the authors' knowledge there are no such results in the literature for the nonresistive case  $\kappa = 0$  with Couette flow, both for the viscous or inviscid regime  $\nu = 0$  or  $\nu > 0$ , and neither for partial dissipation regimes. In view of linear instability results [HHKL18] (see also Proposition 4.2), for these equations any stability threshold results would need to consider unknowns different from (v, b).

As a step towards understanding this non-resistive regime, in this article we consider the 2D MHD equations with isotropic viscosity but only horizontal resistivity (while [Lis20, Dol24] consider full dissipation). In particular, we ask to which extent, as quantified by Sobolev stability thresholds, this partial dissipation regime behaves or does not behave like these extremal cases.

In the (ideal) MHD equations ( $\nu = \kappa = 0$ ) the interaction of shear flows and the magnetic field has been shown to possibly cause instabilities, with arguments both on physical [CM91, HTY05] and mathematical grounds [HT01, ZZ221].

As our first result, we show that this instability also persists in the viscous but non-resistive MHD. These equations exhibit norm inflation in  $H^N$  for all choices of  $\nu > 0$ .

**Proposition 4.2** (Instability for the non-resistive MHD equations). Consider the isotropic equation with  $1 \ge \nu > 0$ ,  $\kappa = 0$ ,  $\alpha > \frac{1}{2}$  and  $N \ge 3$ , then the stationary solution (4.2) is linearly unstable in  $H^N$ . More precisely, there exists initial data  $(v, b)_{in} \in H^N$  such that the solution to the linearized problem satisfies

$$\|(v,b)\|_{H^N} \approx \langle \nu t \rangle \|(v,b)_{in}\|_{H^N}$$

as  $t \to \infty$ . This is optimal in the sense that for all initial data  $(v, b)_{in} \in H^N$ the solution to the linearized problem satisfies

$$\|(v,b)\|_{H^N} \lesssim \langle \nu t \rangle \|(v,b)_{in}\|_{H^N}.$$

The implicit constants here may depend on  $\alpha$ . As a consequence, the nonlinear equations also exhibit arbitrarily large norm inflation in  $H^N$ . That is, for any  $C = C(\nu) > 0$  there exists an  $\varepsilon_0 > 0$  such that for all  $\varepsilon < \varepsilon_0$  there exists initial data  $(v, b)_{in}$  and a time T such that

$$\|(v,b)_{in}\|_{H^N} = \varepsilon, \|(v,b)|_{t=T}\|_{H^N} \ge C \|(v,b)_{in}\|_{H^N}.$$

In particular, there cannot exist a Sobolev threshold for  $||(v, b)||_{H^N}$ .

We remark that following the same argument also instability in suitable Gevrey spaces can be established.

As mentioned above, the isotropic fully dissipative case is known to be stable in Sobolev regularity [Lis20, Dol24]. For the associated partial dissipation regimes, in view of the underlying shear dynamics the associated vertical dissipation case is expected to behave similarly as the full dissipation case. The effects of partial dissipation are a very actively studied field of research in other fluid systems, such as the Boussinesq equations [DWZ21, CW13, ABSPW22]), but, to the authors' knowledge, is largely open in the MHD equations near Couette flow.

In the present case of horizontal resistivity,  $\kappa_y = 0$  and  $\nu_x = \nu_y = \kappa_x$ , the lack of vertical dissipation leads to stronger instabilities, requiring finer control and use of the coupling by a strong magnetic field. Our main results are summarized in the following theorem.

**Theorem 4.3.** Consider the MHD equations with horizontal resistivity,  $\mu := \nu_x = \nu_y = \kappa_x > 0$  and  $\kappa_y = 0$ , near the stationary solution (4.2) with  $\alpha > \frac{1}{2}$  and let  $N \ge 6$  be given.

Then there exist constants  $c_0 = c(\alpha) > 0$ , such that for all initial data  $(v, b)_{in}$ which satisfy

$$||(v,b)_{in}||_{H^N} = \varepsilon \le c_0 \mu^{\frac{3}{2}}$$

the corresponding solution (v, b) of (4.4) satisfies the estimates

$$\begin{aligned} \|v\|_{L^{\infty}H^{N}} + \mu^{\frac{1}{2}} \|\nabla_{t}v\|_{L^{2}H^{N}} &\lesssim \varepsilon, \\ \|b\|_{L^{\infty}H^{N}} + \mu^{\frac{1}{2}} \|\partial_{x}b\|_{L^{2}H^{N}} &\lesssim \varepsilon. \end{aligned}$$

Let us comment on these results:

• Proposition 4.2 shows instability in terms of (v, b) for the non-resistive case. Hence, the (horizontal) magnetic dissipation is shown to be necessary for long-time stability results for (v, b).

However, similarly as in the Boussinesq equations [BBZD23, Zil23], in principle stability results in terms of other unknowns such as the magnetic potential  $\phi = (-\Delta_t)^{-1} \nabla_t^{\perp} b$  could hold for longer or even infinite times, which remains an exciting question for future research.

• Theorem 4.3 establishes a stability threshold  $\gamma = \frac{3}{2}$ . In particular, we stress that the lack of vertical magnetic dissipation not only poses a key challenge of our analysis but results in a different threshold value than the fully dissipative setting [Lis20, Dol24].

Indeed, the main constraint on our stability threshold is given by the control of the nonlinearity  $v \cdot \nabla_t b$  and the reduced decay rates already at the linearized level (see Section 4.2). As we show in Section 4.3.3, our estimates of the so-called reaction terms (4.23) and (4.27) require a lower bound on the threshold by  $\frac{3}{2}$  and are expected to be optimal for this partial dissipation case.

- Theorem 4.3 considers the case  $\mu := \nu_x = \nu_y = \kappa_x$ . As we discuss in Sections 4.2 and 4.3, we expect that instead of equality it suffices to require that  $\frac{1}{2\alpha}\nu_y \leq \kappa_x \leq C\nu_y^{\frac{1}{3}}$ , similarly as in the full dissipation case studied in [Dol24]. These constraints naturally arise in the linearized problem studied in Proposition 4.4. Furthermore, we expect that results can be extended to the case of purely vertical viscous dissipation with additional technical effort.
- Due to missing vertical dissipation, we obtain no decay of the x-averaged magnetic field b<sub>=</sub> which is forced by the nonlinearity.

To prove our results, it is convenient to work with the unknowns

$$p_1 = \Lambda_t^{-1} \nabla_t^{\perp} \cdot v, \ p_2 = \Lambda_t^{-1} \nabla_t^{\perp} \cdot b; \quad \Lambda_t := \sqrt{-\Delta_t}.$$

Similarly to the vorticity and current, the curl operator  $\nabla_t^{\perp}$  eliminates the pressure and yields a scalar quantity, while the operator  $\Lambda_t^{-1} \nabla_t^{\perp} \cdot$  is of order 0. Moreover, since v and b are divergence-free, similarly to viscosity formulations of the 2D Navier-Stokes equations, it can be shown by integration by parts that

$$||Av||_{L^2} = ||Ap_1||_{L^2},$$
  
$$||Ab||_{L^2} = ||Ap_2||_{L^2},$$

for all Fourier multiplier A which commute with  $\nabla_t$  and  $\Lambda_t$ . This, in particular, includes  $\langle \nabla \rangle^N$  which corresponds to the Sobolev norm  $\| \cdot \|_{H^N}$ .

In terms of these unknowns our equations read

$$\partial_t p_1 - \partial_x \partial_x^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \mu \Delta_t p_1 + \Lambda_t^{-1} \nabla_t^{\perp} (b \nabla_t b - v \nabla_t v), \partial_t p_2 + \partial_x \partial_x^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = \mu \partial_x^2 p_2 + \Lambda_t^{-1} \nabla_t^{\perp} (b \nabla_t v - v \nabla_t b),$$

$$(4.4)$$

The remainder of the article is structured as follows:

- In Section 4.2, as a first step we establish linear stability of the equations (4.4). In view of the lack of vertical resistivity we here crucially rely on the interaction of  $p_1$  and  $p_2$  due the the underlying constant magnetic field. Moreover, we discuss the effects of partial dissipation and the resulting limited (optimal) decay rates in time.
- In Section 4.3, we introduce a bootstrap method for the proof of Theorem 4.3. Decomposing into low and high frequency contributions here yields several error terms, which are handled in different subsections. In particular, we need to distinguish between the evolution of the x-average (which does not experience enhanced dissipation due to the shear) and its  $L^2$ -orthogonal complement, as well as different frequency decompositions of the nonlinear terms (called reaction and transport terms in the literature).

- More precisely, in Subsection 4.3.2 we collect all nonlinear terms which can be estimated in a straightforward way. In view of partial magnetic dissipation a main challenge is given by the effect of  $v\nabla_t b$  on  $p_2$  at high frequencies. Here, we distinguish between terms without *x*-average in Subsection 4.3.3 and with average in Subsection 4.3.4 and perform a decomposition into a transport and a reaction term. The low frequency regime is discussed in Subsection 4.3.5 and does not require a very precise analysis.
- As a complementary result, in Section 4.4 we establish instability of the non-resistive, viscous MHD equations and prove Proposition 4.2. Here we first prove linear algebraic instability and then deduce a nonlinear norm inflation result as a corollary.

#### 4.1.1 Notations and conventions

For two real numbers  $a, b \in \mathbb{R}$  we denote the minimum and maximum as

$$\min(a, b) = a \wedge b,$$
$$\max(a, b) = a \lor b.$$

We use the notation  $f \leq g$  if there exists a constant C independent of all relevant parameters such that  $|f| \leq C|g|$ . Furthermore, we write  $f \approx g$  if  $f \leq g$  and  $g \leq f$ .

Moreover, for any vector or scalar v we define

$$\langle v \rangle = (1 + |v|^2)^{\frac{1}{2}}.$$

For a function  $f(x, y) \in L^2(\mathbb{T} \times \mathbb{R})$  we denote the x-average and its  $L^2$ -orthogonal complement as

$$f_{\pm}(y) = \int_{\mathbb{T}} f(x, y) dx,$$
  
$$f_{\neq} = f - f_0.$$

Throughout this text, unless noted otherwise, the spatial variables  $(x, y) \in \mathbb{T} \times \mathbb{R}$  are periodic in the horizontal direction and the respective Fourier variables are denoted as

$$(k,\xi) \in (\mathbb{Z},\mathbb{R})$$

or  $(l,\eta)$ . The norms  $\|\cdot\|_{L^p}$  and  $\|\cdot\|_{H^N}$  refer to the standard Lebesgue and Sobolev norms for functions on  $\mathbb{T} \times \mathbb{R}$ . For time-dependent functions we denote  $L^p H^s = L^p_t H^s$  as the space with the norm

$$||f||_{L^p H^s} = |||f||_{H^s(\mathbb{T} \times \mathbb{R})}||_{L^p(0,T)}.$$

We define the weight  $A^N$  and  $A^{N'}_{\mu}$  by the Fourier multipliers

$$A^{N} = M \langle \nabla \rangle^{N},$$
  
$$A^{N'}_{\mu} = M \langle \nabla \rangle^{N'} e^{c\mu t \mathbf{1}_{k \neq 0}},$$

for  $3 < N' \le N - 2$  and  $0 < c < \frac{1}{2}(1 - \sqrt{\frac{2}{3}})$ . With slight abuse of notation we identify the multiplier operators with their Fourier symbols. The operator M is a time dependent Fourier multiplier, introduced in [BVW18], and is defined to satisfy the following equation:

$$\begin{split} -\frac{\dot{M}}{M} &= \frac{|k|}{k^2+|\xi-kt|^2},\\ M(0,k,\xi) &= 1. \end{split}$$

That is, M is given as

$$M(t,k,\xi) = \exp\left(-\int_0^t d\tau \frac{|k|}{k^2 + (\xi - k\tau)^2}\right).$$

In particular, the operator M is comparable to the identity in the sense that

 $1 \geq M(t,k,\xi) \geq c$ 

for some constant c and all  $k \neq 0$  (and  $M(t, 0, \xi) := 1$  for k = 0).

The operators A thus define energies comparable to Sobolev (semi)norms:

$$\|A^{N} \cdot \|_{L^{2}} \approx \| \cdot \|_{H^{N}},$$
$$\|A^{N'}_{\mu} \cdot \|_{L^{2}} \approx \|e^{c\mu t \mathbf{1}_{k\neq 0}} \cdot \|_{H^{N}}.$$

In particular, since N is sufficiently large, the norm defined by  $A^N$  satisfies an algebra property.

# 4.2 Linear Stability

In this section we study the stability of the linearized version of the equations (4.4):

$$\partial_t p_1 - \partial_x \partial_x^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \nu \cdot \Delta_t p_1,$$
  

$$\partial_t p_2 + \partial_x \partial_x^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = \kappa \cdot \Delta_t p_2,$$
  

$$\nu = (\mu, \mu), \ \kappa = (\mu, 0).$$
(4.5)

The ode tools to establish stability of such systems are well-known in related systems such as the Boussinesq equations [LWX<sup>+</sup>21, BBZD23, BZD20, MZZ23, Zil21b].

Our main results are summarized in the following proposition.

**Proposition 4.4** (Linear stability). Let  $\mu > 0$ ,  $\alpha > \frac{1}{2}$  and  $N \ge 6$  be as in Theorem 4.3. Then the equations (4.5) are stable in  $H^N$  in the sense that for any choice of initial data  $p_{in} \in H^N$  the corresponding solution satisfies

$$\|p\|_{L^{\infty}H^{N}} + \mu^{1/2} \|\nabla_{t} p_{1}\|_{L^{2}H^{N}} + \mu^{1/2} \|\partial_{x} p_{2}\|_{L^{2}H^{N}} \lesssim e^{-C\mu t} \|p_{in}\|_{H^{N}}$$

As we discuss after the proof, in the case  $\frac{1}{2\alpha}\nu \leq \kappa \leq \nu^{1/3}$  the optimal decay rate for large times is given by  $\mu = \min(\nu^{1/3}, \kappa)$ . In particular, the coupling induced by the underlying magnetic field cannot yield enhanced dissipation rates for both components once the viscous dissipation becomes too large.

Proof of Proposition 4.4. We note that in this linear evolution equation (4.5) all coefficient functions are independent of both x and y. Therefore the equations decouple after a Fourier transform and we may equivalently consider the ode system

$$\partial_t \hat{p}_1 - \frac{k(\xi - kt)}{k^2 + (\xi - kt)^2} \hat{p}_1 - \alpha i k \hat{p}_2 = -\nu (k^2 + (\xi - kt)^2) \hat{p}_1,$$
  

$$\partial_t \hat{p}_2 + \frac{k(\xi - kt)}{k^2 + (\xi - kt)^2} \hat{p}_2 - \alpha i k \hat{p}_1 = -\kappa k^2 \hat{p}_2,$$
(4.6)

for an arbitrary but fixed frequency  $(k, \eta) \in \mathbb{Z} \times \mathbb{R}$ . Since the equations are trivial for k = 0, in the following we further without loss of generality may assume that  $k \neq 0$ . Furthermore, after shifting t by  $\frac{\xi}{k}$ , we may assume that  $\xi = 0$  and thus obtain a system of the form

$$\partial_t \begin{pmatrix} p_1 \\ p_2 \end{pmatrix} = \begin{pmatrix} -\frac{t}{1+t^2} - \nu k^2 (1+t^2) & i\alpha k \\ i\alpha k & \frac{t}{1+t^2} - \kappa k^2 \end{pmatrix} \begin{pmatrix} p_1 \\ p_2 \end{pmatrix}, \quad (4.7)$$

where we dropped the hats for simplicity of notation.

In a first naive estimate, we can test this equations with  $(p_1, p_2)$  and obtain that

$$\partial_t(|p_1|^2 + |p_2|^2) \le (\frac{|t|}{1+t^2} - \mu k^2)(|p_1|^2 + |p_2|^2),$$

which already yields the desired decay for times  $|t| \gg (\mu k^2)^{-1}$ . However, a Gronwall-type estimate on the remaining interval would only yield a very rough upper bound on the possible growth by

$$\exp\left(\int_{|t| \lesssim (\mu k^2)^{-1}} \frac{|t|}{1+t^2} dt\right) \lesssim (1+(\mu k^2)^{-1})^2.$$

In order to improve this estimate, a common trick is to make use of the fact that  $|\alpha|$  is relatively large and to consider

$$E = |p_1|^2 + |p_2|^2 - \frac{t}{1+t^2} \Re\left(\frac{1}{i\alpha k} p_1 \overline{p_2}\right).$$

Since  $|\alpha| > \frac{1}{2}$  this energy is positive definite and comparable to  $|p_1|^2 + |p_2|^2$ , with a constant which degenerates as  $|\alpha| \downarrow \frac{1}{2}$ .

Computing the time derivative of the last term, we note that

$$\begin{aligned} &\frac{t}{1+t^2} \partial_t \Re\left(\frac{1}{i\alpha k} p_1 \overline{p_2}\right) \\ &\leq \frac{t}{1+t^2} (|p_1|^2 - |p_2|^2) \\ &+ \frac{|t|}{1+t^2} \frac{1}{|\alpha|} \nu k (1+t^2) |p_1| |p_2| \\ &+ \frac{|t|}{1+t^2} \frac{1}{|\alpha|} \kappa k |p_1| |p_2| \\ &+ \mathcal{O}(t^{-2}) |p_1| |p_2|. \end{aligned}$$

The first term exactly cancels out the possibly large contribution in  $\partial_t (|p_1|^2 + |p_2|^2)$ . For the second and third term we use the fact that  $\frac{1}{\alpha} < 2$  and that these terms can hence be absorbed into the dissipation terms at the cost of a slight loss of constants, provided that

$$\frac{1}{2\alpha}\nu \le \kappa$$

We estimate

$$\begin{aligned} |t| \frac{1}{|\alpha|} \nu k |p_1| |p_2| &\leq \frac{2}{3} \nu k^2 (1+t^2) |p_2|^2 + \frac{3}{8} \frac{1}{|\alpha|k} \nu k^2 |p_2| \\ &\leq \frac{2}{3} \nu k^2 (1+t^2) |p_2|^2 + \frac{3}{4} \kappa k^2 |p_2| \end{aligned}$$

and

$$\frac{|t|}{1+t^2}\frac{1}{|\alpha|}\kappa k|p_1||p_2| \le \frac{8}{1+t^2}\frac{1}{|\alpha|^2}|p_1|^2 + \frac{1}{8}\kappa k^2|p_2|^2$$

Noting that  $\partial_t \frac{|t|}{1+t^2} = \mathcal{O}(t^{-2})$  is integrable in time, we thus arrive at

$$\partial_t E \lesssim \mathcal{O}(t^{-2})E - \nu k^2 (1+t^2) |p_1|^2 - \kappa k^2 |p_2|^2.$$

Further defining

$$\tilde{E} = E \exp(\int^t \mathcal{O}(\tau^{-2}) d\tau),$$

it follows that  $\tilde{E} \approx E$  decays exponentially in time and that the damping terms are integrable in time, which yields the desired result.

We further remark that for t (corresponding to times  $t + \frac{\xi}{k}$ ) such that  $|t| \leq |\alpha|(\mu k^2)^{-1/2}$  the system (4.7) exhibits fast damping in both components (due to the coupling by  $\alpha$ ). However, for times much larger than this (that is, far away

from  $\frac{\xi}{k}$ ) the decay is limited. Indeed, after relabeling  $p_1 \mapsto ip_1$  and introducing the energy E to control contributions by  $\frac{t}{1+t^2}$ , this follows from the fact that the eigenvalues of the matrix

$$\begin{pmatrix} -\mu k^2(1+t^2) & -\alpha \\ \alpha & -\mu k^2 \end{pmatrix}$$

are given by

$$\lambda_{1,2} = -\frac{\mu k^2 (2+t^2)}{2} \pm \sqrt{\frac{1}{4} (\mu k^2 t^2)^2 - \alpha^2}.$$

In the first case, the square root is purely imaginary and hence  $\Re(\lambda_1) = \Re(\lambda_2)$  is comparable to the stronger dissipation term

$$-\mu k^2(1+t^2).$$

For large times, instead the same eigenvalue computation yields

$$\lambda_1 \approx -\mu k^2 \langle t \rangle^2, \ \lambda_2 \approx -\mu k^2$$

and hence the decay estimates of Proposition 4.4 are not improved to  $\mu^{1/3}$ . This linear result thus highlights the effects of the coupling induced by the underlying constant magnetic field and shows which optimal decay estimates can be expected. In particular, it clearly illustrates that the loss of vertical magnetic dissipation incurs a change of decay rate compared to the fully dissipative case.

# 4.3 Bootstrap Hypotheses and Outline of Proof

We next turn to the full nonlinear problem (4.4), where we intend to treat the nonlinear contributions as errors and make use of the smallness of our initial data.

Our approach here follows a bootstrap argument, which is by now standard in the field (see, for instance, [BVW18]). In the notation of Section 4.1.1 we assume that at the initial time

$$\|A^{N}p\|_{L^{2}}^{2} + \|A_{\mu}^{N'}p\|_{L^{2}}^{2} \le c_{0}\epsilon^{2}$$

$$(4.8)$$

for  $3 < N' \leq N - 2$ . The constant  $c_0 = c_0(\alpha) > 0$  will later be chosen small enough and tends to 0 as  $\alpha \to \frac{1}{2}$ . Given this estimate at the initial time, our aim in the remainder of this section is to establish the following estimates for the corresponding solution:

#### • High frequency estimates

$$\|A^{N}p_{1}\|_{L^{\infty}L^{2}}^{2} + \mu \|A^{N}\nabla_{t}p_{1}\|_{L^{2}L^{2}}^{2} + \|\sqrt{-\frac{\dot{M}}{M}}A^{N}p_{1}\|_{L^{2}L^{2}}^{2} < \varepsilon^{2},$$

$$\|A^{N}p_{2}\|_{L^{\infty}L^{2}}^{2} + \mu \|A^{N}\partial_{x}p_{2}\|_{L^{2}L^{2}}^{2} + \|\sqrt{-\frac{\dot{M}}{M}}A^{N}p_{2}\|_{L^{2}L^{2}}^{2} < \varepsilon^{2}.$$

$$(4.9)$$

• Low frequency estimates

$$\begin{aligned} \|A_{\mu}^{N'}p_{1}\|_{L^{\infty}L^{2}}^{2} + \mu\|A_{\mu}^{N'}\nabla_{t}p_{1}\|_{L^{2}L^{2}}^{2} + \|\sqrt{-\frac{\dot{M}}{M}}A_{\mu}^{N'}p_{1}\|_{L^{2}L^{2}}^{2} < \varepsilon^{2}, \\ \|A_{\mu}^{N'}p_{2}\|_{L^{\infty}L^{2}}^{2} + \mu\|A_{\mu}^{N'}\partial_{x}p_{2}\|_{L^{2}L^{2}}^{2} + \|\sqrt{-\frac{\dot{M}}{M}}A_{\mu}^{N'}p_{2}\|_{L^{2}L^{2}}^{2} < \varepsilon^{2}. \end{aligned}$$
(4.10)

By local well-posedness and our assumptions on the initial data, these estimates are satisfied at least on some (small) time interval (0, T). In our bootstrap approach we assume for the sake of contradiction that the maximal time Twith this property is finite. We then show that on that same time interval all estimates hold with improved bounds instead, which however would imply that the estimates could be extended for a small additional time, contradicting the maximality of T.

With this understanding, we suppress T in our notation (see Section 4.1.1) and all  $L^p$  norms in time are understood to be norms on  $L^p(0,T)$ .

The splitting into high and low frequencies is essential to close the estimates in Subsection 4.3.3 and Subsection 4.3.4. In particular, we need the  $e^{-c\mu t}$  decay to bound the so-called reaction error. Moreover, we require strong control of commutators involving A in order to control the so-called transport error. Both error terms impose strong restrictions on the energies and do not allow to close estimates in an easy way. We overcome this difficulty by linking separate energy estimates in the high frequency part and the low frequency part. On the one hand, we can use the additional  $e^{-c\mu t}$  in the low frequency part to our benefit in the analysis of the high frequency part. On the other hand, the difference in regularity allows us to control derivatives in the low frequency estimate by the using high frequency estimate.

Given a solution  $(p_1, p_2)$  of (4.4) and letting  $A = A^N, A^{N'}_{\mu}$ , computing time derivatives we need to control

$$\begin{split} \partial_t \|Ap_1\|_{L^2}^2 &+ 2(1-c)\mu \|A\nabla_t p_1\|_{L^2}^2 + 2\|\sqrt{-\frac{\dot{M}}{M}}Ap_1\|_{L^2}^2 \\ &\leq 2\langle A^2p_1, \partial_x \partial_x^t \Delta_t^{-1}p_1 + \Lambda_t^{-1} \nabla_t^{\perp} (b\nabla_t b - v\nabla_t v)\rangle =: L[p_1] + NL[p_1], \\ \partial_t \|Ap_2\|_{L^2}^2 &+ 2(1-c)\mu \|A\partial_x p_2\|_{L^2}^2 + 2\|\sqrt{-\frac{\dot{M}}{M}}Ap_2\|_{L^2}^2 \\ &\leq 2\langle A^2p_2, -\partial_x \partial_x^t \Delta_t^{-1}p_2 + \Lambda_t^{-1} \nabla_t^{\perp} (b\nabla_t v - v\nabla_t b)\rangle =: L[p_2] + NL[p_2]. \end{split}$$

Here we have split contributions into linear (that is, quadratic integrals) and nonlinear terms (that is, trilinear integrals). Note that the choice of  $0 < c < \frac{1}{2}(1-\sqrt{\frac{2}{3}})$  is made such that 1-c is not too small to absorb linear effects for  $\alpha$  close to  $\frac{1}{2}$ . For later reference, we note that the bootstrap assumptions (4.9) and (4.10) yield the following estimates:

$$\|\partial_x^2 \Lambda_t^{-1} \Lambda^{-1} p\|_{H^N} \lesssim \frac{1}{t} \|p_{\neq}\|_{H^N}$$
(4.11)

and for  $A = A^N, A^{N'}_{\mu}$ 

$$\|Ap_{1,\neq}\|_{L^{2}L^{2}} \lesssim \mu^{-\frac{1}{2}}\varepsilon, \|Ap_{2,\neq}\|_{L^{2}L^{2}} \lesssim \mu^{-\frac{1}{2}}\varepsilon.$$
(4.12)

Furthermore, for the nonlinear terms we will often use the equality

$$||Av||_{L^2} = ||Ap_1||_{L^2}, ||Ab||_{L^2} = ||Ap_2||_{L^2}.$$

Throughout the following sections, we aim to establish smallness of the contributions by the linear terms  $L[\cdot]$  and nonlinear terms  $NL[\cdot]$ . More precisely, we establish the following proposition.

**Proposition 4.5** (Control of errors). Under the assumptions of Theorem 4.3 suppose that the initial data satisfies the smallness condition (4.8) and let T > 0 be such the high and low frequency estimates (4.9), (4.10) are satisfied. Then on the same time interval it holds that

$$\int_{0}^{T} L[p_{1}] + L[p_{2}]dt \leq \frac{1}{2\alpha}(c_{0}+1)\varepsilon^{2} + O(\mu^{-1}\varepsilon^{3}),$$
$$\int_{0}^{T} NL[p_{1}] + NL[p_{2}]dt \leq \mu^{-\frac{3}{2}}\varepsilon^{3}.$$

As a consequence, supposing that  $\alpha > \frac{1}{2}$  and  $\epsilon \ll \mu^{3/2}$ , this implies that both the high frequency and low frequency estimates (4.9), (4.10) improve and thus T can only have been maximal if  $T = \infty$ , which proves Theorem 4.3. Thus proving Proposition 4.5 is our main concern in this section and our proof is split over the following subsections. The most important estimates, highlighting the effects of partial dissipation, are established in Subsections 4.3.1, 4.3.3 and 4.3.4.

We note that the nonlinear terms

$$\langle Ap_1, \Lambda_t^{-1} \nabla_t^{\perp} A(b \nabla_t b - v \nabla_t v) \rangle = - \langle Av, A(b \nabla_t b - v \nabla_t v) \rangle, \\ \langle Ap_2, \Lambda_t^{-1} \nabla_t^{\perp} A(b \nabla_t v - v \nabla_t b) \rangle = - \langle Ab, A(b \nabla_t v - v \nabla_t b) \rangle,$$

for  $A = A^N, A^{N'}_{\mu}$  are all trilinear products involving

$$a^1a^2a^3 \in \{vvv, vbb, bbv, bvb\}$$

and we will use this notation to refer to the specific terms. Since the x-averages do not experience fast (mixing enhanced) decay under the dissipation, we split these products as

$$\begin{split} \langle Aa^1, A(a^2 \nabla_t a^3) \rangle &= \langle Aa^1_{\neq}, A(a^2_{\neq} \nabla_t a^3_{\neq})_{\neq} \rangle \\ &+ \langle Aa^1_{\neq}, A(a^2_{=} \nabla_t a^3_{\neq}) \rangle \\ &+ \langle Aa^1_{\neq}, A(a^2_{\neq} \nabla_t a^3_{=}) \rangle \\ &+ \langle Aa^1_{=}, A(a^2_{\neq} \nabla_t a^3_{=}) \rangle \end{split}$$

where the full splitting is only used for the *bvb* term.

#### 4.3.1 Estimate of the linear error

In this subsection we establish the estimate of the linear terms in Proposition 4.5. Here, we use some of the same techniques as in the proof of linear stability in Section 4.2, but instead focus on establishing quantitative bounds on the time integral.

Taking a Fourier transform of (4.4) yields

$$\partial_t \hat{p}_1 - \frac{k(\xi - kt)}{k^2 + (\xi - kt)^2} \hat{p}_1 - \alpha i k \hat{p}_2 = -\mu (k^2 + (\xi - kt)^2) \hat{p}_1 + \mathcal{F}[NL[p_1]],$$
  

$$\partial_t \hat{p}_2 + \frac{k(\xi - kt)}{k^2 + (\xi - kt)^2} \hat{p}_2 - \alpha i k \hat{p}_1 = -\mu k^2 \hat{p}_2 + \mathcal{F}[NL[p_2]].$$
(4.13)

Recalling the various contributions, we aim to estimate

$$\begin{split} \langle A^2 p_2, -\partial_x \partial_y^t \Delta_t^{-1} p_2 \rangle + \langle A^2 p_1, \partial_x \partial_y^t \Delta_t^{-1} p_1 \rangle \\ = \sum_k \int d\xi A^2 \frac{k(\xi - kt)}{k^2 + (\xi - kt)^2} (|\hat{p}_1|^2 - |\hat{p}_2|^2). \end{split}$$

In the following, with slight abuse of notation, we omit the hat denoting the Fourier transform and only consider  $k \neq 0$ , since for k = 0 this term vanishes.

Similarly as in the linear stability results of Section 4.2, we note that the Fourier multiplier a priori is not integrable in time and cannot easily be estimated by the partial dissipation. Hence, we rely on the coupling induced by the underlying magnetic field to eliminate some of this contribution and to provide better decay. More precisely, multiplying the equations (4.13) with  $\hat{p}_2, \hat{p}_1$  and omitting the hats for simplicity of notation, we obtain the following identity:

$$\begin{split} &|p_1(k)|^2 - |p_2(k)|^2 \\ &= -\frac{1}{i\alpha k} (p_1 \overline{i\alpha k p_1} + i\alpha k p_2 \overline{p_2}) \\ &= -\frac{1}{i\alpha k} p_1 (\partial_t \overline{p}_2 + \frac{k(\xi - kt)}{k^2 + (\xi - kt)^2} \ \overline{p}_2 + \mu k^2 \overline{p}_2 - \overline{\mathcal{F}}[NL[p_2]])) \\ &- \frac{1}{i\alpha k} \overline{p}_2 (\partial_t p_1 - \frac{k(\xi - kt)}{k^2 + (\xi - kt)^2} p_1 + (\mu k^2 + \mu(\xi - kt)^2) p_1 - \mathcal{F}[NL[p_1]] \\ &= \frac{-1}{i\alpha k} (\partial_t (p_1 \overline{p}_2) + \mu (k^2 + (\xi - kt)^2) p_1 \overline{p}_2 + \mu k^2 p_1 \overline{p}_2) \\ &- \frac{1}{\alpha i k} (p_1, p_2) \cdot \mathcal{F}[\Lambda_t^{-1} \nabla_t^{\perp} (b \nabla_t b - v \nabla_t v, b \nabla_t v - v \nabla_t b)]. \end{split}$$

Thus we split L into two linear terms and one nonlinear term:

$$L = 2\sum_{k} \int d\xi A^{2} \frac{k(\xi - kt)}{k^{2} + (\xi - kt)^{2}} \frac{-1}{i\alpha k} \partial_{t}(p_{1}\overline{p}_{2}) + 2\sum_{k} \int d\xi A^{2} \frac{k(\xi - kt)}{k^{2} + (\xi - kt)^{2}} \frac{-1}{i\alpha k} (2\mu k^{2} + \mu(\xi - kt)^{2}) p_{1}\overline{p}_{2}$$
(4.14)  
$$- \frac{2}{\alpha} \langle A \partial_{y}^{t} \Delta_{t}^{-1}(p_{1}, p_{2})_{\neq}, A \Lambda_{t}^{-1} \nabla_{t}^{\perp} (b \nabla_{t} b - v \nabla_{t} v, b \nabla_{t} v - v \nabla_{t} b)_{\neq} \rangle$$
$$= L_{1} + L_{\mu} + ONL.$$

We estimate  $L_{\mu}$  by

$$\begin{split} L_{\mu} &= \frac{2}{\alpha} \mu \sum_{k \neq 0} \int d\xi A^2 \frac{(2k^2 + (\xi - kt)^2)(\xi - kt)}{k^2 + (\xi - kt)^2} p_1 \overline{p}_2 \\ &= \frac{2}{\alpha} \mu \sum_{k \neq 0} \int d\xi A^2 \frac{(2k^2 + (\xi - kt)^2)(\xi - kt)}{(k^2 + (\xi - kt)^2)^{\frac{3}{2}}} p_1 (k^2 + (\xi - kt)^2)^{\frac{1}{2}} \overline{p}_2 \\ &\leq \frac{2}{\alpha} \mu \sup_s \left( \frac{(2 + s^2)s}{(1 + s^2)^{\frac{3}{2}}} \right) \|A \partial_x p_2\|_{L^2} \|A \nabla_t p_1\|_{L^2} \\ &\leq \sqrt{\frac{2}{3}} \frac{1}{\alpha} \mu (\|A \partial_x p_2\|_{L^2}^2 + \|A \nabla_t p_1\|_{L^2}^2), \end{split}$$

where we used that

$$\begin{aligned} \left| \frac{(2k^2 + (\xi - kt)^2)(\xi - kt)}{(k^2 + (\xi - kt)^2)^{\frac{3}{2}}} \right| &= \left| \frac{(2 + (\frac{\xi}{k} - t)^2)(\frac{\xi}{k} - t)}{(1 + (\frac{\xi}{k} - t)^2)^{\frac{3}{2}}} \right| \\ &\leq \sup_s \left( \frac{(2 + s^2)s}{(1 + s^2)^{\frac{3}{2}}} \right) \\ &\leq \sqrt{\frac{2}{3}}. \end{aligned}$$

To estimate  $L_1$ , we integrate by parts in time to deduce that

$$\begin{split} &\int d\tau \sum_{k} \int d\xi A^{2} \frac{k(\xi-kt)}{k^{2}+(\xi-kt)^{2}} \frac{-1}{i\alpha k} \partial_{t}(p_{1}\overline{p}_{2}) \\ &= \left[ \frac{-1}{i\alpha} \sum_{k} \int d\xi A^{2} \frac{(\xi-kt)}{k^{2}+(\xi-kt)^{2}} p_{1}\overline{p}_{2} \right]_{0}^{t} \\ &+ \int d\tau \frac{1}{i\alpha} \sum_{k} \int d\xi p_{1}\overline{p}_{2} \partial_{t} \left( A^{2} \frac{(\xi-kt)}{k^{2}+(\xi-kt)^{2}} \right) \\ &= \left[ \frac{-1}{i\alpha} \sum_{k} \int d\xi A^{2} \frac{(\xi-kt)}{k^{2}+(\xi-kt)^{2}} p_{1}\overline{p}_{2} \right]_{0}^{t} \\ &+ \int d\tau \frac{2}{i\alpha} \sum_{k} \int d\xi p_{1}\overline{p}_{2} \frac{\dot{M}}{M} A^{2} \frac{(\xi-kt)}{k^{2}+(\xi-kt)^{2}} \\ &+ c\mu \mathbf{1}_{A=A_{\mu}^{N'}} \int d\tau \frac{2}{i\alpha} \sum_{k} \int d\xi p_{1}\overline{p}_{2} A^{2} \frac{(\xi-kt)}{k^{2}+(\xi-kt)^{2}} \\ &+ \int d\tau \frac{1}{i\alpha} \sum_{k} \int d\xi p_{1}\overline{p}_{2} A^{2} \frac{k(k^{2}-(kt-\xi)^{2})}{(k^{2}+(\xi-kt)^{2})^{2}}. \end{split}$$

So we infer by Hölder's inequality that

$$\begin{split} &\int d\tau \sum_{k} \int d\xi A^{2} \frac{k(\xi-kt)}{k^{2}+(\xi-kt)^{2}} \frac{-1}{i\alpha k} \partial_{t}(p_{1}\overline{p}_{2}) \\ &\leq \frac{1}{\alpha} (\|Ap_{1}(0)\|_{L^{2}} \|Ap_{2}(0)\|_{L^{2}} + \|Ap_{1}(t)\|_{L^{2}} \|Ap_{2}(t)\|_{L^{2}}) \\ &+ \mu \|A\partial_{x}p_{1}\|_{L^{2}L^{2}} \|A\sqrt{-\frac{\dot{M}}{M}}p_{2}\|_{L^{2}L^{2}} \\ &+ \frac{1}{\alpha} \|A\sqrt{-\frac{\dot{M}}{M}}p_{1}\|_{L^{2}L^{2}} \|A\sqrt{-\frac{\dot{M}}{M}}p_{2}\|_{L^{2}L^{2}} \end{split}$$

and thus

$$\begin{split} &\int Ld\tau - \int ONLd\tau \\ &\leq \frac{1}{2\alpha} (\|Ap_1(0)\|_{L^2}^2 + \|Ap_2(0)\|_{L^2}^2) \\ &+ \frac{1}{2\alpha} (\|Ap_1\|_{L^{\infty}L^2}^2 + \|Ap_2(t)\|_{L^{\infty}L^2}^2) \\ &+ \frac{1}{\alpha} (\mu(1-c)\|\partial_x Ap\|_{L^2}^2 + \mu(1-c)\|\partial_y^t Ap\|_{L^2}^2 + \|\sqrt{-\frac{\dot{M}}{M}}Ap\|_{L^2}^2). \end{split}$$

Using the dissipation estimates (4.12), we therefore obtain

$$\int Ld\tau \le \frac{1}{2\alpha}(c_0+1)\varepsilon^2 + \int ONLd\tau, \qquad (4.15)$$

where the ONL part will be estimated at the beginning of the next subsection.

# **4.3.2** Immediate nonlinear estimates for $A^N$

In this subsection, we collect some estimates which can be obtained in a straight forward approach using standard techniques (e.g. see [BVW18]). In particular, for these terms we are not constrained by the lack of vertical resistivity. For most estimates we do not aim to establish optimal (mixing enhanced) bounds, since these bounds are in any case better than the ones involving horizontal resistivity and hence do not affect the over all stability threshold. In the following we write  $A = A^N$ .

**ONL estimate:** Using integration by parts in space and Hölder's inequality, the nonlinear contribution in (4.14) can be estimated by

$$ONL = \frac{2}{\alpha} \langle A \partial_y^t \Delta_t^{-1} v_{\neq}, A(b \nabla_t b - v \nabla_t v)_{\neq} \rangle \\ + \frac{2}{\alpha} \langle A \partial_y^t \Delta_t^{-1} b_{\neq}, A(b \nabla_t v - v \nabla_t b)_{\neq} \rangle \\ = \frac{2}{\alpha} \langle A \partial_y^t \Delta_t^{-1} (\nabla_t^{\perp} \otimes v_{\neq}), A(b \otimes b - v \otimes v)_{\neq} \rangle \\ + \frac{2}{\alpha} \langle A \partial_y^t \Delta_t^{-1} (\nabla_t^{\perp} \otimes b_{\neq}), A(b \otimes v - v \otimes b)_{\neq} \rangle \\ \lesssim \frac{2}{\alpha} \|A(v, b)_{\neq}\|_{L^2}^2 \|A(v, b)\|_{L^2}.$$

Recalling the bounds (4.12) and integrating in time we thus obtain that

$$\int ONL d\tau \lesssim \mu^{-1} \varepsilon^3. \tag{4.16}$$

Estimates with an *x*-average in the second component: Let  $a^1a^2a^3 \in$  $\{vvv, vbb, bbv, bvb\}$ , then we need to estimate the trilinear products

$$\begin{aligned} \langle Aa_{\neq}^{1}, A(a_{=}^{2}\nabla_{t}a_{\neq}^{3})\rangle &= \langle Aa_{\neq}^{1}, A(a_{=,1}^{2}\partial_{x}a_{\neq}^{3})\rangle \\ &\lesssim \|Aa_{\neq}^{1}\|_{L^{2}}\|Aa_{=,1}^{2}\|_{L^{2}}\|A\partial_{x}a_{\neq}^{3}\|_{L^{2}}. \end{aligned}$$

Integrating in time and again using the bound (4.12) yields a control by

$$\int d\tau \langle Aa^1, A(a_{\pm}^2 \nabla_t a^3) \rangle \lesssim \mu^{-1} \varepsilon^3.$$
(4.17)

The influence of the underlying x-averaged velocity and magnetic field on the average-less parts can thus be easily controlled by the dissipation, provided  $\epsilon \ll \mu$ . In the following we focus on terms involving  $a_{\neq}^2$ . **vvv estimate:** We first discuss the velocity non-linearity and use the alge-

bra property of  $H^N$  and the bounds on A to estimate

$$\langle Av, Av_{\neq} \nabla_t v \rangle \le \|Av\|_{L^2} \|Av_{\neq}\|_{L^2} \|A\nabla_t v\|_{L^2}.$$

Here, the contribution by  $||A\nabla_t v||_{L^2}$  is square integrable in time due to the dissipation (4.12), while  $||Av_{\neq}||_{L^2}$  is square integrable in time by the inviscid damping estimates (4.11). Integrating in time thus yields a bound by

$$\int d\tau \langle Av, A(v_{\neq} \nabla_t v) \rangle \lesssim \mu^{-1} \varepsilon^3.$$
(4.18)

**vbb estimate:** For the contributions by the *vbb* nonlinearity we intend to argue similarly, but have to account for the lack of vertical magnetic dissipation (which we compensate for by using the full fluid dissipation). We may split the integral as

$$\begin{aligned} \langle Av, A(b_{\neq} \nabla_t b) \rangle &= \int Av_1 A(b_{1,\neq} \partial_x + b_{2,\neq} \partial_y^t) b_1 \\ &+ \int Av_2 A(b_{1,\neq} \partial_x + b_{2,\neq} \partial_y^t) b_2. \end{aligned}$$

For the second term we integrate by parts to obtain

$$\int Av_1 A(b_{2,\neq} \partial_y^t b_1) = -\int A \partial_y^t v_1 A(b_{2,\neq} b_1) - \int Av_1 A(\partial_y^t b_{2,\neq} b_1).$$

Furthermore, since b is divergence-free, it holds that  $\partial_y^t b_2 = -\partial_x b_1$  and hence

$$\langle Av, A(b_{\neq}\nabla_{t}b)\rangle \leq \|Av\|_{L^{2}}\|Ab_{\neq}\|_{L^{2}}\|A\partial_{x}b\|_{L^{2}} + \|\partial_{y}^{t}v\|_{L^{2}}\|Ab_{\neq}\|_{L^{2}}\|Ab_{2}\|_{L^{2}}.$$

We may therefore estimate this term using the full fluid and horizontal magnetic dissipation (4.12) and integrating in time yields a bound by

$$\int d\tau \langle Av, A(b_{\neq} \nabla_t b) \rangle \lesssim \mu^{-1} \varepsilon^3.$$
(4.19)

**bbv estimate:** Finally, for the *bbv* contribution, we may again use the full fluid dissipation and the algebra property of A (and  $H^N$ ) to obtain a bound

$$\langle Ab, A(b_{\neq} \nabla_t v) \rangle \lesssim \|Ab\|_{L^2} \|Ab_{\neq}\|_{L^2} \|A \nabla_t v\|_{L^2}.$$

Integrating in time and using (4.12) we thus obtain a bound by

$$\int d\tau \langle Ab, A(b_{\neq} \nabla_t v) \rangle \lesssim \mu^{-1} \varepsilon^3.$$
(4.20)

#### 4.3.3 High frequency *bvb* term without *x*-average

Having established several straightforward estimates using the full fluid dissipation, in this and the following subsections we establish bounds for the high frequency (that is,  $A^N$  terms as in (4.9)) terms involving *bvb*. For simplicity, we write  $A = A^N$  and aim to establish the estimate

$$\langle Ab, A(v_{\neq} \nabla_t b) \rangle \lesssim \mu^{-\frac{3}{2}} \varepsilon^3.$$

We split the bvb term according to (non)vanishing x-averages:

$$\begin{split} \langle Ab, A(v_{\neq} \nabla_t b) \rangle &= \langle Ab_{\neq}, A(v_{\neq} \nabla_t b_{\neq})_{\neq} \rangle \\ &+ \langle Ab_{\neq}, A(v_{\neq} \nabla_t b_{=})_{\neq} \rangle \\ &+ \langle Ab_{=}, A(v_{\neq} \nabla_t b_{\neq})_{=} \rangle. \end{split}$$

Let us first consider the term without any x-averages, which can be written as

$$\begin{split} \langle Ab_{\neq}, A(v_{\neq}\nabla_t b_{\neq}) \rangle &= \int Ab_{1,\neq} A((v_{1,\neq}\partial_x + v_{2,\neq}\partial_y^t)b_{1,\neq}) \\ &+ \int Ab_{2,\neq} A((v_{1,\neq}\partial_x + v_{2,\neq}\partial_y^t)b_{2,\neq}) \end{split}$$

We estimate the second contribution using the algebra property of  $H^N$  and that  $\partial_y^t b_2 = -\partial_x b_1$ , since b is divergence-free:

$$\int d\tau \int Ab_{2,\neq} A(v_{1,\neq}\partial_x + v_{2,\neq}\partial_y^t) b_{2,\neq}$$

$$\leq \int d\tau \|Ab_{2,\neq}\|_{L^2} (\|Av_{1,\neq}\|_{L^2} \|A\partial_x b_{2,\neq}\|_{L^2} + \|Av_{2,\neq}\|_{L^2} \|A\partial_y^t b_{2,\neq}\|_{L^2})$$

$$\leq \int d\tau \|Ab_{2,\neq}\|_{L^2} (\|Av_{1,\neq}\|_{L^2} \|\partial_x b_{2,\neq}\|_{L^2} + \|Av_{2,\neq}\|_{L^2} \|A\partial_x b_{1,\neq}\|_{L^2}).$$

Employing Hölder's inequality this contribution can thus be estimated as

$$\int d\tau \int Ab_{2,\neq} A((v_{1,\neq}\partial_x + v_{2,\neq}\partial_y^t)b_{2,\neq})$$

$$\leq \int d\tau \|Ab_{2,\neq}\|_{L^2} \|Av_{\neq}\|_{L^2} \|A\partial_x b_{\neq}\|_{L^2}$$

$$\leq \|Ab_{2,\neq}\|_{L^2L^2} \|Av_{\neq}\|_{L^{\infty}L^2} \|A\partial_x b_{\neq}\|_{L^2L^2}$$

$$\leq \mu^{-1} \varepsilon^3.$$
(4.21)

It remains to control the contribution by  $b_{1,\neq}$ , which in view to the lack of vertical resistivity is the hardest term to control. Since the velocity field v is divergence-free, we observe that

$$\int Ab_{1,\neq}(v_{\neq}\nabla_t Ab_{1,\neq}) = 0.$$

Therefore, we obtain the following cancellations and introduce a splitting in Fourier space:

$$\begin{split} \int Ab_{1,\neq} A(v_{\neq} \nabla_t b_{1,\neq}) &= \int Ab_{1,\neq} (A(v \nabla_t b_{1,\neq}) - (v_{\neq} \nabla_t Ab_{1,\neq})) \\ &= \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) A(k,\xi) b_1(k,\xi) \frac{(A(k,\xi) - A(l,\eta))(\xi l - \eta k)}{\sqrt{(k-l)^2 + (\xi - \eta - (k-l)t)^2}} \\ &\quad p_1(k-l,\xi-\eta) b_1(l,\eta) \\ &= T + R + \mathcal{R}. \end{split}$$

Here, the Fourier regions

$$\Omega_T = \{ |k - l, \xi - \eta| \le \frac{1}{8} |l, \eta| \}, 
\Omega_R = \{ |l, \eta| \le \frac{1}{8} |k - l, \xi - \eta| \}, 
\Omega_R = \{ \frac{1}{8} |l, \eta| \le |k - l, \xi - \eta| \le 8 |l, \eta| \},$$

correspond to the the transport (T) or low-high term, reaction (R) or high-low term and the remainder  $(\mathcal{R})$  or high-high term. In the following we omit the  $\neq$  subscripts.

**Transport term:** Since  $|k - l, \xi - \eta| \leq \frac{1}{8}|l, \eta|$  we obtain that  $|l, \eta| \approx |k, \xi|$ . Without loss of generality we assume that  $\xi \leq \eta$ , since we can use either of the following splittings

$$\xi l - k\eta = (\xi - \eta)l - (k - l)\eta$$
$$= (\xi - \eta)k - \xi(k - l).$$

Thus using the second equality we estimate

$$T \leq \|\partial_{y}\Lambda_{t}^{-1}p_{1}\|_{L^{\infty}}\|Ab_{1}\|_{L^{2}}\|\partial_{x}Ab_{1}\|_{L^{2}} + \sum_{k,l\neq 0} \iint d(\xi,\eta)\mathbf{1}_{\Omega_{T}}(\mathbf{1}_{2\langle t\rangle(k\vee l)\geq\xi} + \mathbf{1}_{2\langle t\rangle(k\vee l)\leq\xi}) \cdot A(k,\xi)b_{1}(k,\xi)\frac{(A(k,\xi)-A(l,\eta))\xi(l-k)}{\sqrt{(k-l)^{2}+(\xi-\eta-(k-l)t)^{2}}}p_{1}(k-l,\xi-\eta)b_{1}(l,\eta),$$

where we distinguished between  $2\langle t \rangle (k \vee l) \ge \xi$  and  $2\langle t \rangle (k \vee l) \le \xi$ .

The first case is estimated by using the dissipation and (4.11):

$$\sum_{k,l\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \mathbf{1}_{\xi \leq 2(k \vee l)\langle t \rangle} A(k,\xi) b_{1}(k,\xi) \\ \cdot \frac{(A(k,\xi) - A(l,\eta))\xi(l-k)}{\sqrt{(k-l)^{2} + (\xi-\eta - (k-l)t)^{2}}} p_{1}(k-l,\xi-\eta) b_{1}(l,\eta) \\ \lesssim \langle t \rangle \|Ab_{1}\|_{L^{2}} \|\Lambda_{t}^{-1}\partial_{x}p_{1}\|_{L^{\infty}} \|\partial_{x}Ab_{1}\|_{L^{2}} \\ \lesssim \|Ab_{1}\|_{L^{2}} \|\Lambda\partial_{x}p_{1}\|_{L^{\infty}} \|\partial_{x}Ab_{1}\|_{L^{2}} \\ \lesssim \|Ab_{1}\|_{L^{2}} \|Ap_{1}\|_{L^{2}} \|\partial_{x}Ab_{1}\|_{L^{2}}.$$

For the second case,  $2\langle t \rangle (k \vee l) \leq \xi$ , we need to estimate

$$\begin{split} (A^N(k,\xi) - A^N(l,\eta)) &= (M(k,\xi)|k,\xi|^N - M(l,\eta)|l,\eta|^N) \\ &= M(k,\xi)(|k,\xi|^N - |l,\eta|^N) \\ &+ M(l,\eta)(\frac{M(k,\xi)}{M(l,\eta)} - 1)|l,\eta|^N. \end{split}$$

By the mean value theorem, we obtain

$$\begin{split} |k,\xi|^{N} - |l,\eta|^{N} &\leq N|k - \theta l, \xi - \theta \eta|^{N-1} |k - l, \xi - \eta| \\ &\lesssim |k - l, \xi - \eta| (|l,\eta|^{N-1} + |k - l, \xi - \eta|^{N-1}) \\ &\lesssim |k - l, \xi - \eta| |l,\eta|^{N-1}. \end{split}$$

For the differences in M we use that for a,b>0 it holds that  $|e^{a-b}-1|\leq e^{a+b}-1$  and hence

$$\begin{aligned} |\frac{M_1(k,\xi)}{M_1(l,\eta)} - 1| &= |\exp\left(\int_0^t \frac{|l|}{l^2 + (\eta - ls)^2} - \frac{|k|}{k^2 + (\xi - ks)^2} ds\right) - 1| \\ &\leq |\exp\left(\int_0^t \frac{|l|}{l^2 + (\eta - ls)^2} + \frac{|k|}{k^2 + (\xi - ks)^2} ds\right) - 1|.\end{aligned}$$

Thus for  $\eta \ge \xi \ge 2t(k \lor l)$  by integrating we obtain that

$$\begin{aligned} |\frac{M_1(k,\xi)}{M_1(l,\eta)} - 1| &\leq \exp\left(\frac{1}{|l|} \int_0^t \frac{1}{1 + (\frac{\eta}{l} - s)^2} ds + \frac{1}{|k|} \int_0^t \frac{1}{1 + (\frac{\xi}{k} - s)^2} ds\right) - 1 \\ &\leq \exp(\frac{1}{\eta} + \frac{1}{\xi}) - 1 \\ &\lesssim \frac{1}{\eta} + \frac{1}{\xi}. \end{aligned}$$

Therefore, we deduce that

$$\sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_T} \mathbf{1}_{\xi \ge 2(k\vee l)t} A(k,\xi) b_1(k,\xi) \frac{(A(k,\xi) - A(l,\eta))\xi(l-k)}{\sqrt{(k-l)^2 + (\xi-\eta - (k-l)t)^2}}$$

$$p_1(k-l,\xi-\eta) b_1(l,\eta)$$

$$\lesssim \|Ab_1\|_{L^2} \|\Lambda_t^{-1}\partial_x p_1\|_{L^\infty} \|Ab_1\|_{L^2}$$

$$\lesssim \langle t \rangle^{-1} \|Ab_1\|_{L^2} \|\Lambda\partial_x p_1\|_{L^\infty} \|Ab_1\|_{L^2}$$

$$\lesssim \langle t \rangle^{-1} \|Ab_1\|_{L^2} \|Ap_1\|_{L^2} \|Ab_1\|_{L^2},$$

where we used the estimate (4.11). Combining all estimates, we have derived the following estimate of the transport term:

$$\int T d\tau \lesssim \|Ab_1\|_{L^{\infty}L^2} \|Ap_1\|_{L^{\infty}L^2} \|Ab_1\|_{L^2L^2}$$

$$\lesssim \mu^{-\frac{1}{2}} \varepsilon^3.$$
(4.22)

**Reaction term:** Since  $|l, \eta| \leq \frac{1}{8}|k-l, \xi - \eta|$  we obtain  $|k-l, \xi - \eta| \approx |k, \xi|$ . With the identity

$$\xi l - k\eta = l(\xi - \eta - (k - l)t) - (k - l)(\eta - lt)$$

and  $A(k,\xi) - A(l,\eta) \lesssim A(k-l,\xi-\eta)$  we infer

~

$$\begin{split} R &= \sum_{k,l,k-l \neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_R} A(k,\xi) b_1(k,\xi) \frac{(A(k,\xi) - A(l,\eta))(l(\xi - \eta - (k-l)t) - (k-l)(\eta - lt)))}{\sqrt{(k-l)^2 + (\xi - \eta - (k-l)t)^2}} \\ & \cdot p_1(k-l,\xi - \eta) b_1(l,\eta) \\ &\leq \|Ab_1\|_{L^2} \|A\partial_y^t \Lambda_t^{-1} p_1\|_{L^2} \|\partial_x b_1\|_{L^\infty} \\ & + \|Ab_1\|_{L^2} \|A\Lambda_t^{-1} p_1\|_{L^2} \|\partial_y^t \partial_x^2 b_1\|_{L^\infty} \\ & + \|\partial_x Ab_1\|_{L^2} \|A\Lambda_t^{-1} p_1\|_{L^2} \|\partial_y^t \partial_x b_1\|_{L^\infty}. \end{split}$$

We split  $\partial_y^t = \partial_y - t\partial_x$  and use the definition of the low-frequency multiplier  $A_\mu^{N'}$  to estimate

$$\begin{split} \|\langle \partial_x \rangle^2 \partial_y^t b_1 \|_{L^{\infty}} &\leq \|\langle \partial_x \rangle^2 \partial_y b_1 \|_{L^{\infty}} + \|\langle \partial_x \rangle^2 t \partial_x b_1 \|_{L^{\infty}} \\ &\leq t \|\Lambda^{N'} b_1 \|_{L^2} \\ &\lesssim t e^{-c\mu t} \|A_\mu^{N'} b_1 \|_{L^2} \\ &\lesssim \mu^{-1} \|A_\mu^{N'} b_1 \|_{L^2}. \end{split}$$

Therefore, integrating in time yields the estimate

$$\int Rd\tau \lesssim \|Ab_1\|_{L^2L^2} \left( \|A\partial_y^t \Lambda_t^{-1} p_1\|_{L^2L^2} \|Ab_1\|_{L^{\infty}L^2} \right) + \mu^{-1} \|A\partial_x b_1\|_{L^2L^2} \|A\Lambda_t^{-1} p_1\|_{L^2L^2} \|A_{\mu}^{N'} b_1\|_{L^{\infty}L^2} \lesssim \varepsilon^3 \mu^{-\frac{3}{2}}.$$

$$(4.23)$$

 $\mathcal{R}$  term: We consider the Fourier region where  $\frac{1}{8}|l,\eta| \leq |k-l,\xi-\eta| \leq 8|l,\eta|$ . Thus, we have the bounds  $|k,\xi| \leq |l,\eta|$  and  $A(k,\xi) \leq A(l,\eta) \approx A(k-l,\xi-\eta)$ . Furthermore, we note that

$$\xi l - \eta k \le |l, \eta|^2,$$
and thus estimate the remainder terms as

$$\mathcal{R} = \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_{\mathcal{R}}} A(k,\xi) b_1(k,\xi)$$
$$\frac{(A(k,\xi) - A(l,\eta))(\xi l - \eta k)}{\sqrt{(k-l)^2 + (\xi - \eta - (k-l)t)^2}} p_1(k-l,\xi-\eta) b_1(l,\eta)$$
$$\lesssim \|Ab_1\|_{L^2} \|A\Lambda_t^{-1}p_1\|_{L^2} \|\Lambda^2 b_1\|_{L^{\infty}}$$
$$\lesssim \|Ab_1\|_{L^2} \|A\Lambda_t^{-1}p_1\|_{L^2} \|Ab_1\|_{L^2}.$$

Hence after integrating in time, we deduce that

$$\int \mathcal{R} \lesssim \|Ab_1\|_{L^2 L^2} \|\sqrt{-\frac{\dot{M}}{M}} Ap_1\|_{L^2 L^2} \|Ab_1\|_{L^\infty L^2} \lesssim \mu^{-\frac{1}{2}} \varepsilon^3.$$
(4.24)

Combining the estimates (4.21), (4.22), (4.23) and (4.24), we finally conclude that

$$\langle Ab_{\neq}, A(v_{\neq}\nabla_t b_{\neq})_{\neq} \rangle \lesssim \mu^{-\frac{3}{2}} \varepsilon^3.$$
 (4.25)

#### 4.3.4 High frequency estimates for *bvb* terms with *x*-averages

In this subsection we aim to estimate the remaining terms in the bvb integrals, which involve x-averages. We consider the two terms

$$\begin{aligned} \langle Ab_{\neq}, A(v_{\neq}\nabla_t b_{=})_{\neq} \rangle + \langle Ab_{=}, A(v_{\neq}\nabla_t b_{\neq})_{=} \rangle \\ &= \langle Ab_{1,\neq}, A(v_{\neq}\nabla_t b_{1,=})_{\neq} \rangle + \langle Ab_{1,=}, A(v_{\neq}\nabla_t b_{1,\neq})_{=} \rangle, \end{aligned}$$

where we used that  $b_{2,=} = 0$ , since b is divergence-free. Using integration by parts and the fact that v is divergence-free, we obtain that

$$\begin{split} \langle Ab_{1,\neq}, v_{\neq} \nabla_t Ab_{1,=} \rangle + \langle Ab_{1,=}, v_{\neq} \nabla_t Ab_{1,\neq} \rangle \\ &= \langle v_{\neq}, \nabla_t (Ab_{1,=}Ab_{1,\neq}) \rangle = 0, \end{split}$$

and thus

$$\begin{split} \langle Ab_{1,\neq}, A(v_{\neq}\nabla_{t}b_{1,=})\rangle + \langle Ab_{1,=}, A(v_{\neq}\nabla_{t}b_{1,\neq})\rangle \\ &= \langle Ab_{1,\neq}, A(v_{\neq}\nabla_{t}b_{1,=}) - v_{\neq}\nabla_{t}Ab_{1,=}\rangle + \langle Ab_{1,=}, A(v_{\neq}\nabla_{t}b_{1,\neq}) - v_{\neq}\nabla_{t}Ab_{1,\neq}\rangle \\ &= \sum_{k\neq 0} \iint d(\xi,\eta)A(k,\xi)b_{1}(k,\xi)\frac{(A(k,\xi)-A(0,\eta))(-k\eta)}{\sqrt{k^{2}+(\xi-\eta-kt)^{2}}}p_{1}(k,\xi-\eta)b_{1}(0,\eta) \\ &+ \sum_{k\neq 0} \iint d(\xi,\eta)A(0,\xi)b_{1}(0,\xi)\frac{(A(0,\xi)-A(k,\eta))(-k\xi)}{\sqrt{k^{2}+(\xi-\eta-kt)^{2}}}p_{1}(k,\xi-\eta)b_{1}(-k,\eta). \end{split}$$

Again we split this integrals into the transport T, reaction R and remainder terms  $\mathcal{R}$  with the same definition of sets in Fourier space:

$$\Omega_T = \{ |\xi - \eta| \le \frac{1}{8} |\eta| \},$$
  

$$\Omega_R = \{ |\eta| \le \frac{1}{8} |\xi - \eta| \},$$
  

$$\Omega_R = \{ \frac{1}{8} |\eta| \le |\xi - \eta| \le 8 |\eta| \}.$$

**Transport term:** Since  $|\xi - \eta| \leq \frac{1}{8}|\eta|$  we obtain that  $|\eta| \approx |\xi|$ . In our estimates, we distinguish the cases  $\xi \lor \eta \leq 2k\langle t \rangle$  and  $\xi \lor \eta \geq 2k\langle t \rangle$ . In the first case,  $\xi \lor \eta \leq 2k\langle t \rangle$  we obtain a bound by

$$\sum_{k\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_T} \mathbf{1}_{\xi \vee \eta \leq k \langle t \rangle} A(k,\xi) b_1(k,\xi) \frac{(A(k,\xi) - A(0,\eta))k\eta}{\sqrt{k^2 + (\xi - \eta - kt)^2}} p_1(k,\xi - \eta) b_1(0,\eta) + \sum_{k\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_T} \mathbf{1}_{\xi \vee \eta \leq k \langle t \rangle} A(0,\xi) b_1(0,\xi) \frac{(A(0,\xi) - A(k,\eta))k\xi}{\sqrt{k^2 + (\xi - \eta - kt)^2}} p_1(k,\xi - \eta) b_1(-k,\eta) \leq t \|Ab_{\pm}\|_{L^2} \|\partial_x^2 \Lambda_t^{-1} p_{1,\neq}\|_{L^\infty} \|Ab_{1,\neq}\|_{L^2} \lesssim \|Ab_{\pm}\|_{L^2} \|Ap_{1,\neq}\|_{L^2} \|Ab_{1,\neq}\|_{L^2}.$$

In the case  $\xi \lor \eta \ge 2k\langle t \rangle$ , we instead estimate

$$A(k,\xi) - A(0,\eta) \le M(k,\xi)(\xi^2 + k^2)^{\frac{N}{2}} - \eta^N$$
  
=  $(M(k,\xi) - 1)(\xi^2 + k^2)^{\frac{N}{2}} + ((\xi^2 + k^2)^{\frac{N}{2}} - \eta^N).$ 

Since  $\xi \geq 2k\langle t \rangle$ , in the first summand we may bound

$$M(k,\xi) - 1 = \exp\left(-\int_0^t \frac{|k|}{k^2 + (\xi - ks)^2} ds\right) - 1$$
$$\lesssim \frac{1}{\xi} \lesssim \frac{1}{\eta}.$$

By the mean value theorem we further infer

$$(\xi^2 + k^2)^{\frac{N}{2}} - \eta^N \le ((\xi - \theta\eta)^2 + k^2)^{\frac{N-1}{2}} |k, \xi - \eta| \lesssim |k, \xi - \eta| (\xi^2 + k^2)^{\frac{N-1}{2}}.$$

Thus, using that  $k \leq \xi \lesssim \eta$ , we deduce that

$$A(k,\xi) - A(0,\eta) \lesssim |k,\xi - \eta|\eta^{N-1},$$
  
$$A(k,\eta) - A(0,\xi) \lesssim |k,\xi - \eta|\eta^{N-1},$$

where the proof for  $A(k,\eta)-A(0,\xi)$  is analogous. Finally, we obtain

$$\begin{split} \langle Ab_{\neq}, \mathbf{1}_{\Omega_{T}} \mathbf{1}_{\eta \geq kt} A(v_{\neq} \nabla_{t} b_{=}) \rangle + \langle Ab_{=}, \mathbf{1}_{\Omega_{T}} \mathbf{1}_{\eta \geq kt} A(v_{\neq} \nabla_{t} b_{\neq}) \rangle \\ \lesssim \sum_{k \neq 0} \iint d(\xi, \eta) A(k, \xi) b_{1}(k, \xi) \frac{|k, \xi - \eta| \eta^{N-1}}{\sqrt{k^{2} + (\xi - \eta - kt)^{2}}} p_{1}(k, \xi - \eta) b_{1}(0, \eta) \\ + \sum_{k \neq 0} \iint d(\xi, \eta) A(0, \xi) b_{1}(0, \xi) \frac{|k, \xi - \eta| \eta^{N-1}}{\sqrt{k^{2} + (\xi - \eta - kt)^{2}}} p_{1}(k, \xi - \eta) b_{1}(k, \eta) \\ \lesssim \|Ab_{=}\|_{L^{\infty}} \|A\Lambda_{t}^{-1} p_{1, \neq}\|_{L^{2}} \|Ab_{1, \neq}\|_{L^{2}} \\ \lesssim \|Ab_{=}\|_{L^{\infty}} \|A\Lambda_{t}^{-1} p_{1, \neq}\|_{L^{2}} \|Ab_{1, \neq}\|_{L^{2}}, \end{split}$$

and integrating in time yields the desired bound:

$$\int \langle Ab_{\neq}, \mathbf{1}_{\Omega_T} A(v_{\neq} \nabla_t b_{=}) \rangle + \langle Ab_{=}, \mathbf{1}_{\Omega_T} A(v_{\neq} \nabla_t b_{\neq}) \rangle d\tau$$

$$\lesssim \mu^{-1} \varepsilon^3.$$
(4.26)

**Reaction term:** Since  $|\eta| \leq \frac{1}{8} |\xi - \eta|$  we obtain  $|\xi - \eta| \approx |\xi|$  and thus

$$\begin{split} R &= \langle Ab_{\neq}, \mathbf{1}_{\Omega_{R}} A((v_{\neq} \nabla_{t} b_{=}) - v_{\neq} \nabla_{t} Ab_{=}) \rangle + \langle Ab_{=}, \mathbf{1}_{\Omega_{R}} (A(v_{\neq} \nabla_{t} b_{\neq}) - v_{\neq} \nabla_{t} Ab_{\neq})_{=} \rangle \\ &\leq \sum_{k \neq 0} \iint d(\xi, \eta) \mathbf{1}_{\Omega_{R}} A(k, \xi) b_{1}(k, \xi) \frac{(A(k, \xi) - A(0, \eta))k\eta}{\sqrt{k^{2} + (\xi - \eta - kt)^{2}}} p_{1}(k, \xi - \eta) b_{1}(0, \eta) \\ &+ \sum_{k \neq 0} \iint d(\xi, \eta) \mathbf{1}_{\Omega_{R}} A(0, \xi) b_{1}(0, \xi) \frac{(A(0, \xi) - A(-k, \eta))k\xi}{\sqrt{k^{2} + (\xi - \eta - kt)^{2}}} p_{1}(k, \xi - \eta) b_{1}(-k, \eta) \\ &\lesssim \|Ab_{1, \neq}\|_{L^{2}} \|A\partial_{x} \Lambda_{t}^{-1} p_{1, \neq}\|_{L^{2}} \|\partial_{y} b_{1, =}\|_{L^{\infty}} \\ &+ \|Ab_{1, =}\|_{L^{2}} \|A\partial_{y} \partial_{x}^{-1} \Lambda_{t}^{-1} p_{1, \neq}\|_{L^{2}} \|\partial_{x}^{2} b_{1, \neq}\|_{L^{\infty}}. \end{split}$$

Expressing  $\partial_y = \partial_y^t + t \partial_x$  in terms of the time-dependent derivatives, at this point we require the splitting into high and low frequency estimates. More precisely, using the additional time decay of the low-frequency part, we estimate

$$\begin{aligned} \|A\partial_y\partial_x^{-1}\Lambda_t^{-1}p_{1,\neq}\|_{L^2} &\leq \|A\partial_y^t\partial_x^{-1}\Lambda_t^{-1}p_{1,\neq}\|_{L^2} + t\|A\Lambda_t^{-1}p_{1,\neq}\|_{L^2} \\ &\lesssim \|Ap_{1,\neq}\|_{L^2} + t\|A\Lambda_t^{-1}p_{1,\neq}\|_{L^2} \end{aligned}$$

and using the definition of  $A^{N'}_{\mu}$  we can absorb the growth of the factor t at the cost of a power of  $\mu$ :

$$\begin{aligned} \|\partial_x^2 b_{1,\neq}\|_{L^{\infty}} &\leq \|\Lambda^{N'} b_{1,\neq}\|_{L^2} \\ &\lesssim e^{-c\mu t} \|A_{\mu}^{N'} b_{1,\neq}\|_{L^2} \\ &\lesssim \mu^{-1} \langle t \rangle^{-1} \|A_{\mu}^{N'} b_{1,\neq}\|_{L^2} \end{aligned}$$

Thus we obtain

$$\begin{split} R \lesssim \|A^{N}p_{1,\neq}\|_{L^{2}}\|A^{N}b_{1,=}\|_{L^{2}}\|A^{N}b_{1,\neq}\|_{L^{2}} \\ + \mu^{-1}\|A^{N}b_{1,=}\|_{L^{2}}\|A\Lambda_{t}^{-1}p_{1,\neq}\|_{L^{2}}\|A\mu^{N'}b_{1,\neq}\|_{L^{2}} \end{split}$$

Integrating in time then yields the estimate

$$\int R d\tau \lesssim \mu^{-\frac{3}{2}} \varepsilon^3. \tag{4.27}$$

 $\mathcal{R}$  term: The remainder term  $\mathcal{R}$  can be estimated by the same argument as in the case without x-averages in Subsection 4.3.3.

Combining the estimates (4.26), (4.27) and (4.25), we conclude that the *bvb* term can be controlled as

$$\langle Ab, A(v_{\neq}\nabla_t b) \rangle \lesssim \mu^{-\frac{3}{2}} \varepsilon^3.$$
 (4.28)

#### 4.3.5 Low frequency estimates

In this subsection we establish the estimates on the low frequency errors. For simplicity of presentation we present the proof of these estimates for the bvb

nonlinearity. The estimates with an x-average in the second component are analogous to the ones in Subsection 4.3.2. The arguments for the vvv, vbb, bbv or ONL trilinear terms are also analogous.

We aim to establish the bound

$$\langle A^{N'}_{\mu}b, A^{N'}_{\mu}(v_{\neq}\nabla_t b)\rangle \lesssim \mu^{-\frac{1}{2}}\varepsilon^3,$$

and, as in the previous section, separately discuss the transport, reaction and remainder term.

For the transport term, we note that

$$v_{\neq} \nabla_t = \nabla_t^{\perp} \Lambda_t^{-1} p_1 \nabla_t$$
$$= \nabla^{\perp} \Lambda_t^{-1} p_1 \nabla.$$

Hence, we may rewrite

$$\langle A^{N'}_{\mu}b, A^{N'}_{\mu}(v_{\neq}\nabla_{t}b)\rangle = \langle A^{N'}_{\mu}b, A^{N'}_{\mu}(\nabla^{\perp}\Lambda^{-1}_{t}p_{1,\neq}\nabla b)\rangle.$$

In a first step, we estimate the  $b_{\neq}$  term by using the algebra property of  $A^{N'}$ :

$$\begin{split} \langle A_{\mu}^{N'}b, A_{\mu}^{N'}(\nabla^{\perp}\Lambda_{t}^{-1}p_{1,\neq}\nabla b_{\neq})\rangle \\ &\leq \|A_{\mu}^{N'}b\|_{L^{2}}e^{c\mu_{x}t} \big(\|A^{N'}\nabla^{\perp}\Lambda_{t}^{-1}p_{1,\neq}\|_{L^{2}}\|\nabla b_{\neq}\|_{L^{\infty}} + \\ \|\nabla^{\perp}\Lambda_{t}^{-1}p_{1,\neq}\|_{L^{\infty}}\|A^{N'}\nabla b_{\neq}\|_{L^{2}}\big) \\ &\leq \|A_{\mu}^{N'}b\|_{L^{2}} \big(\|A^{N}\Lambda_{t}^{-1}p_{1,\neq}\|_{L^{2}}\|A_{\mu}^{N'}b_{\neq}\|_{L^{2}} + \|A_{\mu}^{N'}\Lambda_{t}^{-1}p_{1,\neq}\|_{L^{2}}\|A^{N}b_{\neq}\|_{L^{2}}\big). \end{split}$$

Integrating in time then yields the estimate

$$\int d\tau \langle A^{N'}_{\mu} b, A^{N'}_{\mu} (v_{\neq} \nabla_t b_{\neq}) \rangle \lesssim \mu^{-\frac{1}{2}} \varepsilon^3.$$
(4.29)

Furthermore, we estimate the  $b_{=}$  term by partial integration and the algebra property of  $A^{N^{\prime}}$ 

$$\begin{split} \langle A_{\mu}^{N'} b, A_{\mu}^{N'} (\nabla^{\perp} \Lambda_{t}^{-1} p_{1, \neq} \nabla b_{=}) \rangle \\ &= - \langle A_{\mu}^{N'} b_{1, \neq}, A_{\mu}^{N'} (\partial_{x} \Lambda_{t}^{-1} p_{1, \neq} \partial_{y} b_{1,=}) \rangle \\ &= \langle \partial_{x} A_{\mu}^{N'} b_{1, \neq}, A_{\mu}^{N'} (\Lambda_{t}^{-1} p_{1, \neq} \partial_{y} b_{1,=}) \rangle \\ &\leq \| \partial_{x} A_{\mu}^{N'} b_{1, \neq} \|_{L^{2}} e^{c \mu t} \left( \| A^{N'} \Lambda_{t}^{-1} p_{1, \neq} \|_{L^{2}} \| \partial_{y} b_{1,=} \|_{L^{\infty}} \right. \\ &\qquad + \| \Lambda_{t}^{-1} p_{1, \neq} \|_{L^{\infty}} \| \partial_{y}^{N'+1} b_{1,=} \|_{L^{2}} \right) \\ &\lesssim \| \partial_{x} A_{\mu}^{N'} b_{1, \neq} \|_{L^{2}} \left( \| A_{\mu}^{N'} \Lambda_{t}^{-1} p_{1, \neq} \|_{L^{2}} \| A^{N'} b_{1,=} \|_{L^{2}} \right. \\ &\qquad + \| A_{\mu}^{N'} \Lambda_{t}^{-1} p_{1, \neq} \|_{L^{2}} \| A^{N} b_{1,=} \|_{L^{2}} \right). \end{split}$$

Integrating in time then yields that

$$\int d\tau \langle A^{N'}_{\mu} b_{\neq}, A^{N'}_{\mu} (v_{\neq} \nabla_t b_{=}) \rangle \lesssim \mu^{-\frac{1}{2}} \varepsilon^3.$$
(4.30)

This concludes our proof of Proposition 4.5 and hence of Theorem 4.3. More precisely, the claimed estimates for both  $A^N$  and  $A^{N'}_{\mu}$  are obtained by combining the respective linear estimate (4.15), the high frequency nonlinear estimates (4.16), (4.17), (4.18), (4.19), (4.20), (4.28), and the low frequency estimates given in (4.29) and (4.30).

We emphasize that the stability threshold of  $\frac{3}{2}$  is determined by the estimates for the action of the  $v \cdot \nabla_t b$  nonlinearity in the estimate (4.28) and, in particular, by the estimates of the reaction terms (4.23) and (4.27). These estimates are expected to be optimal and together with the linear estimates of Section 4.2 highlight the effects of the lack of vertical resistivity.

The partial dissipation case considered in this article

$$\kappa_y = 0, \ \nu_x = \nu_y = \kappa_x > 0,$$

shows the large impact of (partial) magnetic resistivity on the behavior of the MHD equations and the (de)stabilizing role of the magnetic field. As mentioned following Theorem 4.3, more generally our methods of proof extend to the case where  $\kappa_x$  is bounded below in terms of  $\nu$ :

$$\nu_y^{1/3} \ge \kappa_x \ge \frac{1}{2\alpha} \nu_y.$$

The complementary regime, where  $\kappa_x$  tends to zero quicker than  $\nu_y$  remains an interesting topic for future work. The limiting case,  $\kappa_x = 0$ , and the associated instability is discussed in the following section.

## 4.4 Instability of the Non-Resistive MHD System

As a complementary result, in this section we consider the non-resistive MHD equations and establish the instability estimates of Proposition 4.2.

#### 4.4.1 Linear instability

We begin by studying the linearized MHD equations with isotropic viscosity and vanishing resistivity:

$$\partial_t p_1 - \partial_x \partial_x^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \nu \Delta_t p_1, 
\partial_t p_2 + \partial_x \partial_x^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = 0.$$
(4.31)

**Lemma 4.6** (Quantitative linear instability of the non-resistive MHD equations). Under the assumptions of Proposition 4.2, for the linearized equations (4.31) there exists initial data  $p_{in}$  such that

$$\|p(t)\|_{H^{N}} \ge t \frac{\nu}{8\alpha^{2}} \|p_{in}\|_{H^{N}},$$

$$\|p(t)\|_{H^{N-1}} \ge t \frac{\nu^{2}}{32\alpha^{4}} \|p_{in}\|_{H^{N}}.$$

$$(4.32)$$

Furthermore, for all solutions such that at time  $\tau$  it holds  $p(\tau) \in H^N$ , then we obtain

$$\|p(t)\|_{H^N} \lesssim \langle \nu(t-\tau) \rangle \|p(\tau)\|_{H^N}$$
(4.33)

for all  $t > \tau$ .

Proof of Lemma 4.6. After a Fourier transform (4.31) yields

$$\partial_t p_1(k) = -\frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} p_1(k) + \alpha k p_2(k) - \nu (k^2 + (\xi - kt)^2) p_1(k),$$
  

$$\partial_t p_2(k) = \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} p_2(k) - \alpha k p_1(k).$$
(4.34)

Here, in order to simplify notation we have relabeled  $p_2 \mapsto ip_2$  so that we obtain only real-valued coefficient functions.

For the lower bound we fix k = -1 and  $\xi \ge 3\frac{\alpha^2}{\nu}$  and choose  $p_1(0, k, \xi) = 0$ ,  $p_2(0, k, \xi) = 1$ . In this case, the Duhamel integral formula yields that

$$p_1 = -\alpha \int_0^t d\tau_1 \sqrt{\frac{1 + (\tau_1 + \xi)^2}{1 + (t + \xi)^2}} \exp(-\nu(t - \tau + \frac{1}{3}((t + \xi)^3 - (\tau_1 + \xi)^3)))p_2(\tau_1),$$

and that

$$p_{2} - \sqrt{\frac{1 + (t+\xi)^{2}}{1+\xi^{2}}}$$

$$= -\alpha k \int_{0}^{t} d\tau_{2} \sqrt{\frac{1 + (t+\xi)^{2}}{1+(\tau_{2}+\xi)^{2}}} p_{1}(\tau_{2})$$

$$= -\alpha^{2} \int_{0}^{t} d\tau_{2} \int_{0}^{\tau_{1}} d\tau_{1} \frac{\sqrt{1 + (t+\xi)^{2}} \sqrt{1 + (\tau_{1}+\xi)^{2}}}{1+(\tau_{2}+\xi)^{2}} p_{2}(\tau_{1})$$

$$\cdot \exp(-\nu(\tau_{2} - \tau_{1} + \frac{1}{3}((\tau_{2} + \xi)^{3} - (\tau_{1} + \xi)^{3})))$$

Denoting  $|p_2|_{\infty}(t) = \sup_{0 \le \tau \le t} |p_2(\tau)|$ , the double integral term can be bounded by

$$\alpha^{2}|p_{2}|_{\infty} \int_{0}^{t} d\tau_{1} \int_{\tau_{2}}^{t} d\tau_{2} \exp(-\nu(\tau_{2}-\tau_{1}+\frac{1}{3}((\tau_{2}+\xi)^{3}-(\tau_{1}+\xi)^{3}))).$$

Furthermore, we may estimate

$$\begin{split} \int_0^t d\tau_1 \int_{\tau_2}^t d\tau_2 \exp(-\nu(\tau_2 - \tau_1 + \frac{1}{3}((\tau_2 + \xi)^3 - (\tau_1 + \xi)^3))) \\ &= \int_0^t d\tau_1 \int_{\tau_2}^t d\tau_2 \frac{1 + (\tau_2 + \xi)^2}{1 + (\tau_2 + \xi)^2} \exp(-\nu(\tau_2 - \tau_1 + \frac{1}{3}((\tau_2 + \xi)^3 - (\tau_1 + \xi)^3))) \\ &\leq \int_0^t d\tau_1 \int_{\tau_2}^t d\tau_2 \frac{1 + (\tau_2 + \xi)^2}{1 + (\tau_1 + \xi)^2} \exp(-\nu(\tau_2 - \tau_1 + \frac{1}{3}((\tau_2 + \xi)^3 - (\tau_1 + \xi)^3))) \\ &\leq \frac{1}{\nu} \int_0^t d\tau_1 \frac{1}{1 + (\tau_1 + \xi)^2} [\exp(-\nu(\tau_2 - \tau_1 + \frac{1}{3}((\tau_2 + \xi)^3 - (\tau_1 + \xi)^3)))]_{\tau_2 = \tau_1}^{\tau_2 = t} \\ &\leq \frac{1}{\nu\xi}. \end{split}$$

Hence, we obtain that

$$p_2 - \sqrt{\frac{1 + (t + \xi)^2}{1 + \xi^2}} p_{2,in} | \le \frac{\alpha^2}{\nu \xi} |p_2|_{\infty} =: c |p_2|_{\infty},$$

For later reference we note that  $c=\frac{\alpha^2}{\nu\xi}$  satisfies  $0< c\leq \frac{1}{3}$  and hence  $0<\frac{c}{1-c}\leq$  $\frac{1}{2}$ .

Then it follows that

$$|p_2| \le c|p_2|_{\infty} + \sqrt{\frac{1+(t+\xi)^2}{1+\xi^2}}|p_{2,in}|,$$

and, since  $\xi \ge 0$ , the function  $\sqrt{\frac{1+(t+\xi)^2}{1+\xi^2}}$  is monotonly increasing in time. This implies that

$$|p_2|_{\infty} \le \frac{1}{1-c} \sqrt{\frac{1+(t+\xi)^2}{1+\xi^2}} |p_{2,in}|$$

Hence we infer

$$\begin{aligned} |p_2 - \sqrt{\frac{1 + (t + \xi)^2}{1 + \xi^2}} p_{2,in}| &\leq \frac{c}{1 - c} \sqrt{\frac{1 + (t + \xi)^2}{1 + \xi^2}} |p_{2,in}| \\ &\leq \frac{1}{2} \sqrt{\frac{1 + (t + \xi)^2}{1 + \xi^2}} |p_{2,in}|. \end{aligned}$$

Since  $0 < \frac{c}{1-c} \leq \frac{1}{2}$ ,  $p_2$  is comparable to  $\sqrt{\frac{1+(t+\xi)^2}{1+\xi^2}}p_{2,in}$ . We next keep k = -1 fixed but combine this construction for different  $\xi$ . More precisely, let  $a(\xi)$  be such that  $\operatorname{supp}_{\xi}(a(\xi)) \subset [3\frac{\alpha^2}{\nu}, 4\frac{\alpha^2}{\nu}]$  and  $\int (2+\xi^2)^{\frac{N}{2}}a^2(\xi) = 0$ . 1. Then for the initial data

$$p_{in}(k,\xi) = \mathbf{1}_{k=-1}a(\xi)$$

it holds that

$$\begin{aligned} \|p_{in}\|_{H^N} &= 1, \\ \|p(t)\|_{H^N} \geq t \frac{\nu}{8\alpha^2}, \\ |p(t)\|_{H^{N-1}} \geq t \frac{\nu^2}{32\alpha^4}, \end{aligned}$$

which proves (4.32).

We prove the upper bound in three steps

- 1. Let  $t \ge \tau \ge \nu^{-1} + \frac{\xi}{k}$ , then we estimate  $|p|(t) \lesssim \langle \nu(t-\tau) \rangle |p|(\tau)$ .
- 2. Let  $\nu^{-1} + \frac{\xi}{k} \ge t \ge \tau \ge -\nu^{-1} + \frac{\xi}{k}$ , then we estimate  $|p|(t) \lesssim |p|(\tau)$ .
- 3. Let  $-\nu^{-1} + \frac{\xi}{k} \ge t \ge \tau$ , then we estimate  $|p|(t) \lesssim |p|(\tau)$ .

From (1-3) estimate (4.33) follows directly. In the following, we prove (1-3).

1. Let  $t \ge \tau \ge \nu^{-1} + \frac{\xi}{k}$ . Then we obtain

$$\begin{aligned} \partial_t |p|^2 &\leq 2 \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} |p_2|^2 \\ &+ \left(-2 \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} - 2\nu k^2 (1 + (t - \frac{\xi}{k})^2)) |p_1|^2 \\ &\leq 2 \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} |p_2|^2. \end{aligned}$$

Thus we obtain by Gronwall's Lemma

$$\begin{split} |p|^{2}(t) &\leq \frac{1 + (t - \frac{\xi}{k})^{2}}{1 + (\tau - \frac{\xi}{k})^{2}} |p|^{2}(\tau) \\ &\leq \frac{1 + (\tau - \frac{\xi}{k})^{2} + (t - \frac{\xi}{k})^{2} - (\tau - \frac{\xi}{k})^{2}}{1 + (\tau - \frac{\xi}{k})^{2}} |p|^{2}(\tau) \\ &\leq (1 + \frac{(t - \frac{\xi}{k})^{2} - (\tau - \frac{\xi}{k})^{2}}{1 + (\tau - \frac{\xi}{k})^{2}}) |p|^{2}(\tau) \\ &\leq 2 \langle \nu(t - \tau) \rangle^{2} |p|^{2}(\tau). \end{split}$$

2. Let  $\nu^{-1} + \frac{\xi}{k} \ge t \ge \tau \ge -\nu^{-1} + \frac{\xi}{k}$ . We define the energy

$$E = |p|^2 + \frac{2}{\alpha k} \frac{s}{1+s^2} p_1 p_2$$

As  $\alpha > \frac{1}{2}, E$  is positive definite with

$$(1 - \frac{1}{2\alpha})|p|^2 \le E \le (1 + \frac{1}{2\alpha})|p|^2.$$

Then we derive in time and infer

$$\begin{split} \partial_t E + \nu k^2 (1+s^2)) p_1^2 &= \frac{2}{\alpha k} \frac{1-s^2}{(1+s^2)^2} p_1 p_2 \\ &+ \frac{2}{\alpha} \nu k s p_1 p_2 \\ &\leq \frac{2}{\alpha k} \frac{1}{1+s^2} p_1 p_2 \\ &+ \frac{2}{\alpha} \nu p_2^2 + \nu k (1+s^2) p_1^2. \end{split}$$

This further implies that

$$\partial_t E \lesssim \left(\frac{1}{1+s^2} + \nu\right) E.$$

By Gronwall's lemma we infer

$$E(t) \le \exp\left(C \int_{\tau}^{t} \frac{1}{1+\tau_{1}^{2}} + \nu d\tau_{1}\right) E(\tau)$$
  
$$\le \exp\left(C(\pi + 2\nu\nu^{-1})E(\tau)\right)$$
  
$$\lesssim E(\tau).$$

Therefore, we obtain that

 $|p|(t) \lesssim |p|(\tau).$ 

3. Let  $-\nu^{-1} + \frac{\xi}{k} \ge t \ge \tau$ . Then we obtain that

$$\begin{split} \partial_t |p|^2 &\leq 2 \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} |p_2|^2 \\ &+ (-2 \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} - 2\nu k^2 (1 + (t - \frac{\xi}{k})^2)) |p_1|^2 \\ &\leq 0. \end{split}$$

Thus we arrive at the desired estimate

$$|p|(t) \lesssim |p|(\tau).$$

#### 4.4.2 Nonlinear norm inflation

We next consider the nonlinear non-resistive MHD equations in their perturbative form around the stationary solution (4.2):

$$\partial_t p_1 - \partial_x \partial_x^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \nu \Delta_t p_1 + \nabla_t^{\perp} \Lambda_t^{-1} (b \nabla_t b - v \nabla_t v), 
\partial_t p_2 + \partial_x \partial_x^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = \nabla_t^{\perp} \Lambda_t^{-1} (b \nabla_t v - v \nabla_t b).$$
(4.35)

The following lemma establishes the norm inflation result of Proposition 4.2.

**Lemma 4.7** (Nonlinear norm inflation for the non-resistive MHD equations). Under the same assumptions of Proposition 4.2, we consider the non-resistive nonlinear MHD equations (4.35). Then for all  $C = C(\nu) > 1$  there exists  $\varepsilon_0 > 0$ such that for all  $0 < \varepsilon < \varepsilon_0$  there exists initial data  $p_{in}$  such that

$$\|p_{in}\|_{H^N} = \varepsilon$$

and

$$\|p\|_{L^{\infty}H^N} \ge \varepsilon C.$$

*Proof.* For the sake of contradiction we assume that there exists  $\varepsilon_0 > 0$  such that for all  $0 < \varepsilon \leq \varepsilon_0$  and for any choice of initial data with  $\|p_{in}\|_{H^N} = \epsilon$  it holds that

$$\|p\|_{L^{\infty}H^N} \le \varepsilon C.$$

Our plan is to choose initial data such that for a choice of  $\varepsilon$  and t we obtain a contradiction to this bound. In particular, we choose  $p_{in}$  as the data of the linear instability result, Lemma 4.6, such that the associated linear solution  $p_{lin}$ satisfies

$$\|p_{in}\|_{H^N} = \varepsilon,$$
  
$$\|p_{lin}(t)\|_{H^{N-1}} \ge t \frac{\nu^2}{32\alpha^4}.$$

Let  $S(\tau, t)$  be the solution operator for the linearized system. Then in view of (4.33) we have the estimate

$$\|S(\tau,t)\|_{H^N \to H^N} \lesssim \langle \nu(t-\tau) \rangle \le \langle \nu t \rangle.$$
(4.36)

Thus, since

$$\partial_t (p - p_{lin}) \le L(p - p_{lin}) + NL[p],$$

we deduce that

$$\begin{split} \|p - p_{lin}\|_{H^{N-1}}^2 &\leq \int_0^t \|S(\tau, t)\|_{H^N \to H^N} \|p - p_{lin}\|_{H^{N-1}} \|p\|_{H^{N-1}} \|\nabla_t p\|_{H^{N-1}} \\ &\lesssim \|p - p_{lin}\|_{L^{\infty} H^{N-1}} \|p\|_{L^{\infty} H^{N-1}} \|p\|_{L^{\infty} H^N} 2 \int_0^t \tau \langle \nu \tau \rangle \\ &\lesssim t^2 \langle \nu t \rangle \varepsilon^2 C^2 \|p - p_{lin}\|_{L^{\infty} H^{N-1}}. \end{split}$$

This yields the estimate

$$\|p - p_{lin}\|_{L^{\infty}H^{N-1}} \le \tilde{C}t^2 \langle \nu t \rangle \varepsilon^2 C^2,$$

for some  $\tilde{C}$ . Finally, we obtain

$$\begin{aligned} \|p\|_{H^{N-1}} &\geq \|p_{lin}\|_{H^{N-1}} - \|p - p_{lin}\|_{L^{\infty}H^{N-1}} \\ &\geq \|p_{lin}\|_{H^{N-1}} - t^2 \langle \nu t \rangle \varepsilon^2 \tilde{C} C^2 \\ &\geq t \varepsilon (\frac{\nu^2}{32\alpha^4} - t^2 \langle \nu t \rangle \varepsilon C^2). \end{aligned}$$

This completes our proof by contradiction provided this term is large enough for a given small  $\varepsilon$  and suitable time. Indeed for the choice  $\varepsilon \leq \frac{1}{8} \frac{\nu^6}{32^3 C^4 \tilde{C} \alpha^{10}}$  at the time  $t = 2C \frac{32\alpha^4}{\nu^2}$  it holds that

$$\|p\|_{H^{N-1}} \ge \frac{1}{2}t \frac{\nu^2}{32\alpha^4} \varepsilon \ge C\varepsilon.$$

This concludes our proof of the nonlinear norm inflation and hence completes our proof of Proposition 4.2.  $\hfill \Box$ 

The behavior of the MHD equations and, in particular, the interaction of shear flows, the magnetic field and dissipation are an area of current active research [Lis20, Dol24, ZZ24, KZ1]. However, prior works have focused on cases where the resistivity is at least as strong as the fluid viscosity and where thus the behavior is closely related to that of the Navier-Stokes equations. In contrast, the non-resistive MHD equations exhibit additional instability, as for instance shown in Proposition 4.2.

Motivated by this dichotomy, in this article we have studied the anisotropic, partial dissipation regime

$$\kappa_y = 0, \ \kappa_x = \nu_x = \nu_y$$

and the associated stability threshold in the inviscid limit. As shown in Theorem 4.3 and highlighted in the estimates of Sections 4.2, 4.3.4 and 4.3.3, this partial dissipation regime behaves qualitatively differently than both the fully dissipative case and the non-resistive case. Moreover, our analysis crucially used the coupling of the velocity field and magnetic field induced by the underlying magnetic field, which allowed us to obtain improved estimates for the magnetic field despite the lack of the symmetry of the dissipation.

Partial, anisotropic dissipation in the MHD equations is thus shown to give rise to distinct regimes with different (in)stability properties and demonstrates an intricate interplay of shear dynamics, magnetic interaction and anisotropic dissipation. A more complete understanding of all these regimes, the case of resistivity vanishing faster than viscosity and a characterization of the (in)stability properties of the ideal MHD equations remain exciting questions for future research.

# Chapter 5

# Sobolev Stability for the 2D MHD Equations in the Non-Resistive Limit

This chapter is the preprint [K].

#### NIKLAS KNOBEL

Abstract. This chapter considers the stability of the 2D magnetohydrodynamics (MHD) equations close to a combination of Couette flow and a constant magnetic field. We consider the ideal conductor limit for the case when viscosity  $\nu$  is larger than resistivity  $\kappa$ ,  $\nu \geq \kappa > 0$ . For this regime, we establish a bound on the Sobolev stability threshold. Furthermore, for  $\kappa \leq \nu^3$  this system exhibits instability, which leads to norm inflation of size  $\nu \kappa^{-\frac{1}{3}}$ .

### 5.1 Introduction

The equations of magnetohydrodynamics (MHD)

$$\partial_t V + V \cdot \nabla V + \nabla \Pi = \nu \Delta V + B \cdot \nabla B,$$
  

$$\partial_t B + V \cdot \nabla B = \kappa \Delta B + B \cdot \nabla V,$$
  

$$\nabla \cdot V = \nabla \cdot B = 0,$$
  

$$(t, x, y) \in \mathbb{R}^+ \times \mathbb{T} \times \mathbb{R} =: \Omega,$$
  
(5.1)

model the evolution of a magnetic field  $B: \Omega \to \mathbb{R}^2$  interacting with the velocity  $V: \Omega \to \mathbb{R}^2$  of a conducting fluid. The MHD equations are a common model used in astrophysics, planetary magnetism and controlled nuclear fusion [Dav16]. The quantities  $\nu, \kappa \geq 0$  correspond to fluid viscosity and magnetic resistivity.

The pressure  $\Pi : \Omega \to \mathbb{R}$  ensures that the velocity remains divergence-free. A fundamental problem of fluid dynamics and plasma physics is the stability and long-time behavior of solutions to equation (5.1) and in particular stability of specific solutions. We consider the combination of an affine shear flow, called Couette flow, and a constant magnetic field:

$$V_s = ye_1,$$
$$B_s = \alpha e_1.$$

In particular, the solution combines the effects of mixing due to shear and coupling by the magnetic field. The Couette flow mixes any perturbation, which leads to increased dissipation rates, called enhanced dissipation, and stabilizes the equation. The coupling with a constant magnetic field propagates this mixing to magnetic perturbations. However, the magnetic field weakens the mixing, especially if viscosity is larger than resistivity, inviscid damping gets counteracted by algebraic growth for specific time regimes.

In the related case of the Navier-Stokes equation, that is when no magnetic field is present, one observes turbulent solutions as viscosity reaches small values. In contrast, the linearized problem around Couette flow is stable for all values of the viscosity. These phenomena are known as the Sommerfeld paradox [LL11] and highlight instability due to nonlinear effects. In [BM15a, DM23, DZ21, BM14, IJ13] various authors show sharp stability in Gevrey 2 spaces (spaces between  $C^{\infty}$  and analytic) for the inviscid case  $\nu = 0$ . The nonlinear instability can be suppressed by the viscosity for initial data sufficiently small in Sobolev spaces, ensuring stability [BVW18, MZ22, BGM17].

When considering the MHD equations without Couette flow, the constant magnetic field stabilizes the equation. The dynamics of small initial perturbations of the ideal MHD equation around a strong enough magnetic field is close to the linearized system [BSS88]. For stability in several dissipation regimes we refer to [WZ17, RWXZ14, HXY18, RWXZ14, Sch88, CF23, Koz89] and references therein. However, global in time wellposedness for the non-resistive case is still open (see the discussion in [CF23]). Furthermore, a shear flow leads to qualitatively different behavior and instabilities [HT01, HHKL18].

Recently, the MHD equation around Couette flow has gathered significant interest [Lis20, KZ1, ZZ24, Dol24, KZ2]. Already on a linear level, the behavior of the MHD changes for different values of  $\nu$  and  $\kappa$ . In [Lis20] Liss proved the first stability threshold for the MHD equations. He considered the full dissipative regime of  $\kappa = \nu > 0$  and proved the stability of the three-dimensional MHD equation for initial data which is sufficiently small in Sobolev spaces. For the analogous two-dimensional problem, Dolce [Dol24] proved stability in the more general setting of  $0 < \kappa^3 \leq \nu \leq \kappa$ . In [KZ2] Zillinger and the author considered the case of only horizontal resistivity and full viscosity and established stability for small data in Sobolev spaces. For the regime of vanishing viscosity  $\nu = 0$ and non-vanishing resistivity  $\kappa > 0$ , in [KZ1] we constructed a linear stability and instability mechanism around nearby traveling waves in Gevrey 2 spaces. In a corresponding nonlinear stability result, Zhao and Zi [ZZ24] proved the almost matching nonlinear result of Gevrey  $\sigma$  stability for  $1 \leq \sigma < 2$  and for sufficiently small perturbations.

The results mentioned above on stability around Couette flow focus on the setting when resistivity is larger than viscosity  $\nu \leq \kappa$ . Indeed in the setting  $\nu > 0$  and  $\kappa = 0$ , the magnetic effects dominate leading to a linear instability mechanism and thus a growth of the magnetic field by  $\nu t$  for specific initial data [KZ2].

In this paper, we consider the setting  $0 < \kappa \leq \nu$ . In particular, this also includes the non-resistive limit  $\kappa \downarrow 0$  independent of  $\nu$ . To the author's knowledge the stability of the regime  $\kappa < \nu$  has not previously been studied for the MHD equation around Couette flow. To state the main result, we define the perturbative unknowns

$$\begin{split} v(x,y,t) &= V(x+yt,y,t) - V_s,\\ b(x,y,t) &= B(x+yt,y,t) - B_s, \end{split}$$

where the change of variables  $x \mapsto x + yt$  follows the characteristics of the Couette flow. For these unknowns, equation (5.1) becomes

$$\partial_t v + v_2 e_1 - 2 \partial_x \Delta_t^{-1} \nabla_t v_2 = \nu \Delta_t v + \alpha \partial_x b + b \nabla_t b - v \nabla_t v - \nabla_t \pi,$$
  

$$\partial_t b - b_2 e_1 = \kappa \Delta_t b + \alpha \partial_x v + b \nabla_t v - v \nabla_t b,$$
  

$$\nabla_t \cdot v = \nabla_t \cdot b = 0.$$
(5.2)

Due to the change of variables the spatial derivatives become time-dependent, i.e.  $\partial_y^t = \partial_y - t\partial_x$ ,  $\nabla_t = (\partial_x, \partial_y^t)^T$  and  $\Delta_t = \partial_x^2 + (\partial_y^t)^2$ .

For equation (5.2) we establish Lipschitz stability for initial data which is sufficiently small in Sobolev spaces, in the sense that there exists a bound on the initial data  $\varepsilon_0 = \varepsilon_0(\nu, \kappa)$  and a Lipschitz constant  $L = L(\nu, \kappa)$  such that for initial data which satisfies

$$||(v,b)_{in}||_{H^N} = \varepsilon \le \varepsilon_0,$$

the corresponding solution is globally bounded in time by

$$\|(v,b)(t)\|_{H^N} \le L\varepsilon.$$

For the non-resistive case,  $\kappa = 0$ , global wellposedness is an open problem and so Lipschitz stability in Sobolev spaces is unclear. Thus, naturally the question arises, which  $\varepsilon_0$  and L are optimal and how they behave in the limit  $\nu, \kappa \downarrow 0$ . We denote a Sobolev stability threshold as  $\gamma_1, \gamma_2 \in \mathbb{R}$ , such that for  $\varepsilon_0 = c_0 \nu^{\gamma_1} \kappa^{\gamma_2}$  with small  $c_0 > 0$  we obtain

$$\begin{aligned} \|(v,b)_{in}\|_{H^N} &\leq c_0 \nu^{\gamma_1} \kappa^{\gamma_2} \to \text{ stability,} \\ \|(v,b)_{in}\|_{H^N} &\gg c_0 \nu^{\gamma_1} \kappa^{\gamma_2} \to \text{possible instability.} \end{aligned}$$

This extends the common convention in the field (eg. see [BVW18]) to allow for two independent parameters  $\nu$  and  $\kappa$ . In particular, it agrees with the common convention when restricting to the case  $\nu \approx \kappa$ . It allows us to discuss cases where  $\kappa$  tends to zero much quicker than  $\nu$ . Establishing a possible instability is highly nontrivial since for the nonlinear setting it is difficult to construct solutions that exhibit norm inflation. To the author's knowledge, there does not exist any nonlinear instability result for the MHD equation around Couette flow in Sobolev spaces.

For accessibility and simplicity of notation, we state our main result as the following theorem (see Theorem 1 for a detailed description).

**Theorem 5.1.** Consider  $\alpha > \frac{1}{2}$ ,  $N \ge 5$  and a small enough constant  $c_0 = c_0(\alpha) > 0$ . Let  $0 < \kappa \le \nu \le \frac{1}{40}(1 - \frac{1}{2\alpha})^{\frac{6}{5}}$ , then we obtain Sobolev stability for initial data which is sufficiently small in Sobolev spaces, where the estimates qualitatively differ for the regimes  $\kappa \ge \nu^3$  and  $\kappa \le \nu^3$ . More precisely:

• In the regime of  $\nu^3 \lesssim \kappa$ , for all initial data which satisfy

$$||(v,b)_{in}||_{H^N} = \varepsilon \le c_0 \nu^{\frac{1}{12}} \kappa^{\frac{1}{2}}$$

the global in time solution (v, b) of (5.2) satisfies the Lipschitz bound

$$\sup_{t>0} \|(v,b)(t)\|_{H^N} \lesssim \varepsilon.$$

• In the regime of  $\nu^3 \gtrsim \kappa$ , for all initial data which satisfy

$$||(v,b)_{in}||_{H^N} = \varepsilon \le c_0 \nu^{-\frac{11}{12}} \kappa^{\frac{5}{6}}$$

the global in time solution (v, b) of (5.2) satisfies the Lipschitz bound

$$\sup_{t>0} \|(v,b)(t)\|_{H^N} \lesssim \nu \kappa^{-\frac{1}{3}} \varepsilon.$$

In particular, we obtain Lipschitz stability for the Lipschitz constant  $L \approx \max(1, \nu \kappa^{-\frac{1}{3}})$ for the smallness parameter  $\varepsilon_0 \approx \min(\nu^{\frac{1}{12}}\kappa^{\frac{1}{2}}, \nu^{-\frac{11}{12}}\kappa^{\frac{5}{6}}).$ 

In the proof, we employ an energy method similar to [BBZD23, MZZ23, Zil21b, Dol24, KZ2]. In the following, we outline the main challenges and novelties of the proof:

- The imbalance of resistivity  $\kappa$  and viscosity  $\nu$  yields two cases  $\nu^3 \lesssim \kappa$  and  $\nu^3 \gtrsim \kappa$  (or equivalently  $1 \lesssim \nu \kappa^{-\frac{1}{3}}$  or  $1 \gtrsim \nu \kappa^{-\frac{1}{3}}$ ). These cases give different values for L, namely 1 and  $\nu \kappa^{-\frac{1}{3}}$ . By Proposition 5.3, the norm inflation of  $\nu \kappa^{-\frac{1}{3}}$  appears in the linear dynamics and thus is sharp.
- We consider the case  $\nu^3 \gtrsim \kappa$ . On certain time scales the viscosity is so strong that fluid effects get suppressed while the effects of the magnetic field dominate. Thus, the term  $\partial_t b = e_1 b_2$  in (5.1) generates algebraic growth in specific regimes (see Subsection 5.2). Estimating this linear effect yields the norm inflation by  $L = \nu \kappa^{-\frac{1}{3}}$ . The algebraic growth appears on different time scales depending on the frequency, a precise estimate of the nonlinear terms is necessary.

- For the case  $\nu^3 \lesssim \kappa$  the algebraic growth is bounded by a finite constant. In the subcase  $\nu = \kappa$  the sum of the threshold parameters is  $\gamma_1 + \gamma_2 = \frac{7}{12}$  which is a slight improvement over  $\frac{2}{3}$  in [Dol24]. This improvement is attained by the choice of our adapted unknowns which changes the structure of the nonlinearity.
- In the proof of Theorem 5.1 we perform a low and high frequency decomposition  $a = a_{hi} + a_{low}$ . For high frequencies, the nonlinear term consist of  $a_{low} \nabla_t a_{hi}$ , called transport term and  $a_{hi} \nabla_t a$ , called reaction term (including hi - hi interactions). Compared to the Navier-Stokes equation, in the case of the MHD equation, it is vital to bound the transport term precisely. In particular, for  $\kappa \leq \nu^3$  the previously mentioned algebraic growth affects the estimate of the transport term strongly.
- The threshold is determined by the nonlinear term  $v\nabla_t b = \Lambda_t^{-1}\nabla^{\perp}p_1\nabla b$ acting on b in (5.2), for the natural unknown  $p_1 = \Lambda_t^{-1}\nabla^{\perp}v$  (which we discuss later in more detail). In our estimates we rely on two stabilizing effects, the strong viscosity of v and the  $\Lambda_t^{-1}$  in front of  $p_1$ . For the nonlinear term  $v\nabla_t b$  both effects fall onto v. Due to the weaker integrability of the b this term determines the threshold after integrating in time.

With the main challenges in mind, let us comment on the results:

- The size of the constant magnetic field  $\alpha > \frac{1}{2}$  results in a strong interaction between v and b. Due to this interaction, the decay in v and growth in b are in balance (see Lemma 5.2). Constants may depend on  $\alpha$  and degenerate as  $\alpha \downarrow \frac{1}{2}$ . For example we obtain  $\lim_{\alpha \downarrow \frac{1}{2}} c_0(\alpha) = 0$ .
- Figure 5.1 shows which areas stability has been proven. The graphic shows only qualitative behavior and after rescaling we obtain the same graphic. The resistivity  $\kappa$  is on the vertical axis and the viscosity  $\nu$  is on the horizontal axis. We prove stability for the regime  $0 < \kappa \leq \nu$ , which we divide into two segments:  $\nu^3 \lesssim \kappa$  in orange and  $\nu^3 \gtrsim \kappa$  in red. In [Dol24] Dolce considered the regime of  $0 < (\frac{16}{\alpha}\kappa)^3 \leq \nu \leq \kappa$ , which is in blue. The authors of [ZZ24] considered the line  $\nu = 0$  which is in purple. The black line corresponds to  $\nu = \kappa > 0$  of [Lis20].

Stability for the regimes  $0 < \nu \leq (\frac{16}{\alpha}\kappa)^3$ ,  $\kappa = 0 < \nu$  and  $\kappa = \nu = 0$  remain open. For the set  $0 < \nu \ll \kappa^3$ , we expect that an adjusted application of the methods used in this article yield stability. We expect stability for the case  $\kappa = 0$  and  $0 < \nu$  to be very difficult since we obtain linear growth for the  $p_2$  variable. For  $\Lambda_t^{-1}p_2$  we obtain linear stability but then there is no time decay in the magnetic field and so we lack an important stabilizing effect. In the inviscid case,  $\kappa = \nu = 0$  the linearized system is stable in the p variables. However, due to the lack of dissipation, it is very challenging to bound the nonlinear terms.

• Our threshold consists of parameters  $\gamma_1$  and  $\gamma_2$ . An alternative notation is to impose the relation  $\nu \approx \kappa^{\delta}$  for some  $0 \leq \delta \leq 1$ . With that convention



Figure 5.1: Sketch of areas with results for stability.

we obtain stability if  $\varepsilon \leq c_0 \kappa^{\gamma(\delta)}$  for

$$\gamma = \begin{cases} \frac{1}{2} + \frac{\delta}{12} & \delta \ge \frac{1}{3} \\ \frac{5}{6} - \frac{11}{12}\delta & \text{otherwise.} \end{cases}$$

The remainder of this article is structured as follows:

- In Section 5.2 we discuss the linearized system. We identify two different time regions where "circular movement" or "strong viscosity" determine the linearized behavior. We estimate both effects separately and then establish the estimates for the linearized system.
- In Section 5.3 we prove the main theorem. We employ a bootstrap approach, where we control errors in Proposition 5.4. The main difficulty is to bound the linear growth and the nonlinear effect of  $v\nabla_t b$  acting on b.

#### **Notations and Conventions**

For  $a,b\in\mathbb{R}$  we denote their minimum and maximum as

$$\min(a, b) = a \land b,$$
$$\max(a, b) = a \lor b.$$

We write  $f \lesssim g$  if  $f \leq Cg$  for a constant C independent of  $\nu$  and  $\kappa$ . Furthermore, we write  $f \approx g$  if  $f \lesssim g$  and  $g \lesssim f$ . We denote the Lebesgue spaces  $L^p =$ 

 $L^p(\mathbb{T} \times \mathbb{R})$  and the Sobolev spaces  $H^N = H^N(\mathbb{T} \times \mathbb{R})$  for some  $N \in \mathbb{N}$ . For time-dependent functions, we denote  $L^p H^s = L^p_t H^s$  as the space with the norm

$$\|f\|_{L^p H^s} = \left\|\|f\|_{H^s(\mathbb{T} \times \mathbb{R})}\right\|_{L^p(0,T)},\tag{5.3}$$

where omit writing the T. We write the time-dependent spatial derivatives

$$\begin{aligned} \partial_y^t &= \partial_y - t \partial_x, \\ \nabla_t &= (\partial_x, \partial_y^t)^T, \\ \Delta_t &= \partial_x^2 + (\partial_y^t)^2, \end{aligned}$$

and the half Laplacians as

$$\begin{split} \Lambda &= (-\Delta)^{\frac{1}{2}}, \\ \Lambda_t &= (-\Delta_t)^{\frac{1}{2}}. \end{split}$$

The function  $f \in H^N$  is decomposed into its x average and the orthogonal complement

$$f_{=}(y) = \int f(x, y) dx,$$
  
$$f_{\neq} = f - f_{=}.$$

#### The adapted unknowns

For the following, it is useful to change to the unknowns  $p_{1,\neq} = \Lambda_t^{-1} \nabla_t^{\perp} v_{\neq}$  and  $p_{2,\neq} = \Lambda_t^{-1} \nabla_t^{\perp} b_{\neq}$ . However, since  $\Lambda_t^{-1} \nabla_t^{\perp}$  is not a bounded operator on the x average, we define

$$\begin{split} p_{1,\neq} &= \Lambda_t^{-1} \nabla_t^{\perp} v_{\neq}, \\ p_{1,=} &= v_{1,=}, \\ p_{2,\neq} &= \Lambda_t^{-1} \nabla_t^{\perp} b_{\neq}, \\ p_{2,=} &= b_{1,=}. \end{split}$$

Thus (5.2) can be equivalently expressed as

$$\partial_t p_1 - \partial_x \partial_y^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \nu \Delta_t p_1 + \Lambda_t^{-1} \nabla_t^{\perp} (b \nabla_t b - v \nabla_t v),$$
  

$$\partial_t p_2 + \partial_x \partial_y^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = \kappa \Delta_t p_2 + \Lambda_t^{-1} \nabla_t^{\perp} (b \nabla_t v - v \nabla_t b), \qquad (5.4)$$
  

$$p|_{t=0} = p_{in}.$$

These unknowns are particularly useful since

$$||Ap_1||_{L^2} = ||Av||_{L^2}, ||Ap_2||_{L^2} = ||Ab||_{L^2},$$

for all Fourier multipliers A such that one side is finite. We note that we can recover  $v_{\neq}$  and  $b_{\neq}$  from  $p_{\neq}$  by

$$\begin{aligned} v_{\neq} &= -\Lambda_t^{-1} \nabla_t^{\perp} p_{1,\neq} \\ b_{\neq} &= -\Lambda_t^{-1} \nabla_t^{\perp} p_{2,\neq}. \end{aligned}$$

The operator  $\Lambda_t^{-1} \nabla_t^{\perp}$  can be seen as the perpendicular Riesz transform shifted in time on frequency space applied to either a vector or a scalar. It satisfies  $(\Lambda_t^{-1} \nabla_t^{\perp}) \circ (\Lambda_t^{-1} \nabla_t^{\perp}) = - \operatorname{Id} \operatorname{in} H^N$ .

#### 5.2 Linear Stability and Norm Inflation

In this section, we consider the behavior of the linearized version of (5.4):

$$\partial_t p_1 - \partial_x \partial_y^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \nu \Delta_t p_1, 
\partial_t p_2 + \partial_x \partial_y^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = \kappa \Delta_t p_2.$$
(5.5)

For this equation, we establish the following proposition:

**Proposition 5.2** (Linear energy estimate). Consider  $\alpha > \frac{1}{2}$ ,  $N \ge 0$  and  $0 < \kappa \le \nu$ . Let  $p_{in} \in H^N$  with  $p_{in,=} = \int p_{in} dx = 0$ , then the solution p of (5.5) satisfies the bound

$$\|p(t)\|_{H^N} \lesssim e^{-c\kappa^{\frac{1}{3}}t} (1 + \nu\kappa^{-\frac{1}{3}}) \|p_{in}\|_{H^N}.$$
(5.6)

Furthermore, we obtain for specific initial data norm inflation of  $\nu \kappa^{-\frac{1}{3}}$ . The proof uses Lemma 3 from [KZ2].

**Proposition 5.3** (Linear norm inflation). Consider  $\alpha > \frac{1}{2}$ ,  $N \ge 1$  and  $\nu \ge \max(2\kappa, \kappa^{\frac{1}{3}})$ , then there exist initial data  $p_{in}$  and such that at the time  $T = \kappa^{-\frac{1}{3}}$  the solution p of (5.5) satisfies

$$||p(T)||_{H^N} \gtrsim \nu \kappa^{-\frac{1}{3}} ||p_{in}||_{H^N}.$$
 (5.7)

For the proof of the propositions, we perform a Fourier transform  $(x, y) \mapsto (k, \xi)$  and replace  $p_1$  by  $ip_1$ . Hence the system (5.5) can be equivalently (for  $k \neq 0$ ) written as

$$\partial_t p_1 = -\frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} p_1 - \alpha k p_2 - \nu k^2 (1 + (t - \frac{\xi}{k})^2) p_1,$$
  

$$\partial_t p_2 = \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} p_2 + \alpha k p_1 - \kappa k^2 (1 + (t - \frac{\xi}{k})^2) p_2.$$
(5.8)

Here, with slight abuse of notation, we omit writing the Fourier transformation. This equation has several effects that appear in different regimes of  $t - \frac{\xi}{k}$ , which we discuss in the following. The effect of circular movement appears for  $|t - \frac{\xi}{k}| \lesssim \nu^{-1}$  and the effect of strong viscosity for  $\nu^{-1} \lesssim |t - \frac{\xi}{k}| \lesssim \kappa^{-\frac{1}{3}}$ . Then before proceeding to the proof of the propositions, we briefly sketch these effects.

#### Circular movement

To highlight the effect of the constant magnetic field  $\alpha$  in (5.5) we consider the toy model

$$\partial_t p_1 = -\alpha k p_2,$$
  

$$\partial_t p_2 = \alpha k p_1.$$
(5.9)

This is solved by

$$p(t) = \begin{pmatrix} \cos(\alpha kt) & -\sin(\alpha kt) \\ \sin(\alpha kt) & \cos(\alpha kt) \end{pmatrix} p_{in}.$$

We call this effect of the constant magnetic field (5.9) circular movement, which leads to a transfer between  $p_1$  and  $p_2$ . This circular movement is counteracted by viscosity for times away from  $\frac{\xi}{k}$ .

#### Effect of strong viscosity

Let us consider the case when  $0 < \kappa \ll \nu$  and for simplicity of notation let k = 1and  $\xi = 0$ . Due to the viscosity, we obtain  $p_1 \approx 0$  for large times  $t \ge t_0 \gg 1$ . Then from (5.8) we deduce the toy model

$$\partial_t p_2 = \left(\frac{t}{1+t^2} - \kappa (1+t^2)\right) p_2. \tag{5.10}$$

The first term in (5.10) leads to linear growth until the resistivity is strong enough for the second term to take over. This is seen in the explicit solution of (5.10)

$$p_2(t) = \frac{\langle t \rangle}{\langle t_0 \rangle} \exp\left(-\kappa \int_{t_0}^t 1 + \tau^2 \ d\tau\right) p_2(t_0).$$

This is estimated by

$$p_2(t) \lesssim t_0^{-1} \kappa^{-\frac{1}{3}} e^{-c\kappa^{\frac{1}{3}}(t-t_0)} p_2(t_0),$$

which corresponds to the maximal growth which we obtain. In the following, we will see that  $t_0 \approx \nu^{-1}$  is the time after which viscosity dominates. The reader may expect that the enhanced dissipation timescale  $\nu^{-\frac{1}{3}}$  would be the relevant timescale, but the combination of circular movement and the viscosity gives enough decay for  $p_2$  such that the linear growth gets suppressed until the time  $\nu^{-1}$ .

#### Proof of Proposition 5.2

*Proof.* For simplicity of notation, we introduce the new variable  $s = t - \frac{\xi}{k}$  and initial time  $s_{in} = -\frac{\xi}{k}$ . Then equation (5.8) reads

$$\partial_s p_1 = -\frac{s}{1+s^2} p_1 - \alpha k p_2 - \nu k^2 (1+s^2) p_1,$$
  
$$\partial_s p_2 = \frac{s}{1+s^2} p_2 + \alpha k p_1 - \kappa k^2 (1+s^2) p_2.$$

Further we change the unknown to  $\tilde{p} = \exp(\frac{\kappa}{2}k^2(s - s_{in} + \frac{1}{3}(s^3 - s_{in}^3)))p$ . For  $\tilde{\kappa} = \frac{\kappa}{2}$  and  $\tilde{\nu} = \nu - \frac{\kappa}{2}$ , this yield the equation

$$\partial_s \tilde{p}_1 = -\frac{s}{1+s^2} \tilde{p}_1 - \alpha k \tilde{p}_2 - \tilde{\nu} k^2 (1+s^2) \tilde{p}_1, \partial_s \tilde{p}_2 = \frac{s}{1+s^2} \tilde{p}_2 + \alpha k \tilde{p}_1 - \tilde{\kappa} k^2 (1+s^2) \tilde{p}_1.$$

Let us denote  $s_0 := \nu^{-1}$  and in the following, we distinguish between times  $|s| \leq s_0$  and  $|s| \geq s_0$ . We first consider the case  $s_{in} \leq -s_0$ . For  $|s| \leq s_0$ , the circular movement is not suppressed by the viscosity.

circular movement is not suppressed by the viscosity. We define the energy  $E = |\tilde{p}|^2 + \frac{1}{\alpha k} \frac{2s}{1+s^2} \tilde{p}_1 \tilde{p}_2$ , then *E* is a positive quadratic form due to our assumption  $\alpha > \frac{1}{2}$  and satisfies

$$(1 - \frac{1}{2\alpha k})|\tilde{p}|^2 \le E \le (1 + \frac{1}{2\alpha k})|\tilde{p}|^2.$$

We calculate the time derivative

$$\begin{aligned} \partial_s E + \tilde{\nu} k^2 (1+s^2) \tilde{p}_1^2 + \tilde{\kappa} k^2 (1+s^2) \tilde{p}_2^2 \\ &= \frac{1}{\alpha k} \partial_s (\frac{2s}{1+s^2}) \tilde{p}_1 \tilde{p}_2 - 2s \frac{(\tilde{\nu} - \tilde{\kappa})k}{\alpha} \tilde{p}_1 \tilde{p}_2 \\ &\leq \frac{1}{\alpha k} \partial_s (\frac{2s}{1+s^2}) \tilde{p}_1 \tilde{p}_2 + \frac{1}{2} \tilde{\nu} k^2 (1+s^2) \tilde{p}_1^2 + \frac{2\tilde{\nu}}{\alpha^2} \tilde{p}_2^2 \end{aligned}$$

and so with  $|\tilde{p}|^2 \leq \frac{2\alpha}{2\alpha - 1}E$  we infer

$$|\partial_s E| \le \frac{\alpha}{\alpha - \frac{1}{2}} \left(\frac{1}{1 + s^2} + 2\frac{\tilde{\nu}}{\alpha^2}\right) E.$$

Gronwall's lemma implies

$$E(s_0) \le \exp\left(\frac{\alpha}{\alpha - \frac{1}{2}} (\pi + 2\frac{\tilde{\nu}}{\alpha^2} |s_0|) r\right) E(-s_0).$$

Since  $\nu s_0 = 1$ , we deduce

$$E(s_0) \lesssim E(-s_0)$$

and thus

$$|\tilde{p}(s_0)| \lesssim |\tilde{p}(-s_0)|. \tag{5.11}$$

Consider the case  $|s| \ge s_0$ , we calculate

$$\begin{split} &\frac{1}{2}\partial_s |\tilde{p}|^2 \leq (-\frac{s}{1+s^2} - \tilde{\nu}k^2(1+s^2))\tilde{p}_1^2 \\ &+ (\frac{s}{1+s^2} - \tilde{\kappa}k^2(1+s^2))\tilde{p}_2^2, \end{split}$$

and since  $\left(-\frac{s}{1+s^2} - \tilde{\nu}k^2(1+s^2)\right) \leq 0$  for all  $|s| \geq s_0$  we conclude

$$\partial_s |\tilde{p}|^2 \le (\frac{2s}{1+s^2} - \tilde{\kappa}k^2(1+s^2))_+ \tilde{p}_2^2$$

Thus we obtain the estimate

$$|\tilde{p}(s)|^{2} \leq \begin{cases} |\tilde{p}(s_{in})|^{2} & s \leq -s_{0}, \\ \frac{1+s^{2}}{1+s_{0}^{2}}|\tilde{p}(s_{0})|^{2} & s_{0} \leq s \leq 2(\kappa k^{2})^{-\frac{1}{3}}, \\ (1+4\nu^{2}\kappa^{-\frac{2}{3}}k^{-\frac{4}{3}})|\tilde{p}(s_{0})|^{2} & s_{0} \vee 2(\kappa k^{2})^{-\frac{1}{3}} \leq s. \end{cases}$$

Combining this with (5.11) we infer

$$|\tilde{p}(s)| \lesssim (1 + \nu \kappa^{-\frac{1}{3}} k^{-\frac{2}{3}}) |p(s_{in})|.$$

The case  $s_{in} \geq -s_0$  is established similarly since we only bound the growth. With

$$\exp(-\frac{\kappa}{2}k^2(s-s_{in}+\frac{1}{3}(s^3-s_{in}^3)))) \lesssim e^{-c\kappa^{\frac{1}{3}t}}$$

we deduce

$$\begin{aligned} |p(s)| &\lesssim e^{-c\kappa^{\frac{1}{3}t}} |\tilde{p}|(s) \\ &\lesssim (1 + \nu \kappa^{-\frac{1}{3}} k^{-\frac{2}{3}}) e^{-c\kappa^{\frac{1}{3}t}} |p|(s_{in}). \end{aligned}$$

Equation (5.8) decouples in  $\xi$  and k, so we infer the proposition with this estimate.

#### **Proof of Proposition 5.3**

*Proof.* We introduce the notations  $\tilde{p}(t,k,\xi) = \exp(\kappa k^2 \int_0^t (1+(\tau-\frac{\xi}{k})^2)d\tau)p(t,k,\xi)$ and  $\tilde{\nu} := \nu - \kappa$ . Then the equations (5.8) read

$$\partial_t \tilde{p}_1 = -\frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} \tilde{p}_1 - \alpha k \tilde{p}_2 - \tilde{\nu} k^2 (1 + (t - \frac{\xi}{k})^2) \tilde{p}_1,$$
  

$$\partial_t \tilde{p}_2 = \frac{t - \frac{\xi}{k}}{1 + (t - \frac{\xi}{k})^2} \tilde{p}_2 + \alpha k \tilde{p}_1,$$
  

$$\tilde{p}_{in} = p_{in}.$$
(5.12)

We point out that this exactly agrees with the linearized equation with the non-resistive case. Specifically in Lemma 3 of [KZ2] it was shown that there exists frequency localized initial data such that for all t > 0 and frequencies  $\xi \in [2\frac{\alpha^2}{\tilde{\nu}}, 4\frac{\alpha^2}{\tilde{\nu}}]$  it holds that

$$\tilde{p}_2(t, -1, \xi) \ge \frac{t}{2\xi} p_{2,in}(-1, \xi) \gtrsim \tilde{\nu} t p_{2,in}(-1, \xi).$$
(5.13)

From this lower bound, we deduce the norm inflation for the non-resistive limit. For times  $\tau \in [0, \kappa^{-\frac{1}{3}}]$  and frequencies  $\xi \in [2\frac{\alpha^2}{\tilde{\nu}}, 4\frac{\alpha^2}{\tilde{\nu}}]$  we obtain

$$|\tau + \xi| \lesssim \nu^{-1} + \kappa^{-\frac{1}{3}}.$$

Then for  $T = \kappa^{-\frac{1}{3}}$  there exist a  $C = C(\alpha)$ , such that

$$1 \ge \exp(-\kappa \int_0^T (1 + (t+\xi)^2) dt) \gtrsim \exp(-C\kappa (\nu^{-1} + \kappa^{-\frac{1}{3}})^2 \kappa^{-\frac{1}{3}}) \ge \exp(-2C) \gtrsim 1$$

Where we used in the last estimate, that  $\kappa \leq \nu^3$  yields  $\kappa(\nu^{-1} + \kappa^{-\frac{1}{3}})^2 \kappa^{-\frac{1}{3}} \leq 2$ . From (5.13) and  $\tilde{\nu} = \nu - \kappa \geq \frac{1}{2}\nu$ , since  $\nu \geq 2\kappa$ , we deduce, that

$$p_2(T, -1, \xi) = \exp(-\kappa \int_0^T (1 + (t + \xi)^2) dt) \tilde{p}_2(T, -1, \xi) \gtrsim \nu \kappa^{-\frac{1}{3}} p_{2,in}(-1, \xi),$$

From this, we infer the norm inflation

$$||p(T)||_{H^N} \gtrsim \nu \kappa^{-\frac{1}{3}} ||p_{in}||_{H^N}.$$

### 5.3 Sobolev Stability for the Nonlinear System

The following theorem is a more general statement of Theorem 5.1. We dedicate the remainder of the section to the proof.

**Theorem 1.** Let  $\alpha > \frac{1}{2}$  and  $N \ge 5$ , then there exist  $c_0, c > 0$ , such that for all  $0 < \kappa \le \nu \le \frac{1}{40} (1 - \frac{1}{2\alpha})^{\frac{6}{5}}$  there exist  $L = \max(1, \nu \kappa^{-\frac{1}{3}})$ , such that for all initial data, which satisfy

$$\begin{aligned} \|(v,b)_{in,\neq}\|_{H^N} &= \varepsilon \le c_0 L^{-1} \nu^{\frac{1}{12}} \kappa^{\frac{1}{2}}, \\ \|(v,b)_{in,=}\|_{H^N} \le \tilde{\varepsilon}, \qquad \text{with } \varepsilon \le \tilde{\varepsilon} \le \nu^{-\frac{1}{12}} \varepsilon \end{aligned}$$
(5.14)

the corresponding solution of (5.2) satisfies the bound

$$\begin{aligned} \|(v,b)_{\neq}(t)\|_{L^{\infty}H^{N}} + \|\nabla_{t}(\nu v,\kappa b)_{\neq}\|_{L^{2}H^{N}} &\lesssim Le^{-c\kappa^{\frac{3}{5}}t}\varepsilon, \\ \|(v,b)_{=}(t)\|_{L^{\infty}H^{N}} + \|\partial_{y}(\nu v,\kappa b)_{=}\|_{L^{2}H^{N}} &\lesssim \tilde{\varepsilon}. \end{aligned}$$

$$(5.15)$$

Furthermore, we obtain the following enhanced dissipation estimates

$$\begin{split} \|v_{\neq}\|_{L^{2}H^{N}} &\lesssim L\nu^{-\frac{1}{6}}e^{-c\kappa^{\frac{1}{3}}t}\varepsilon, \\ \|b_{\neq}\|_{L^{2}H^{N}} &\lesssim L\kappa^{-\frac{1}{6}}e^{-c\kappa^{\frac{1}{3}}t}\varepsilon. \end{split}$$

This theorem implies Theorem 5.1. With slight abuse of notation, we write L as the  $\nu$  and  $\kappa$  dependent part of the Lipschitz constant. We prove this theorem by using a bootstrap method. Let A be the Fourier weight

$$A:=M|\nabla|^N e^{c\kappa^{\frac{1}{3}}t\mathbf{1}_{\neq}},$$

where  $M = M_L M_1 M_\kappa M_\nu M_{\nu^3}$  are defined as

$$\frac{-\dot{M}_L}{M_L} = \frac{t - \frac{\xi}{k}}{1 + (\frac{\xi}{k} - t)^2} \mathbf{1}_{\{\nu^{-1} \le t - \frac{\xi}{k} \le (c_1 \kappa k^2)^{-\frac{1}{3}}\}} \qquad k \neq 0$$

$$\frac{-\dot{M}_1}{M_1} = C_\alpha \frac{|k| + \nu^{\frac{1}{12}} |k|^2}{k^2 + (\xi - kt)^2} \qquad \qquad k \neq 0,$$

$$\frac{-\dot{M}_{\nu}}{M_{\nu}} = \frac{\nu^{\frac{1}{3}}}{1 + \nu^{\frac{2}{3}}(t - \frac{\xi}{k})^2} \qquad \qquad k \neq 0,$$

$$\frac{-\dot{M}_{\kappa}}{M_{\kappa}} = \frac{\kappa^{\frac{1}{3}}}{1+\kappa^{\frac{2}{3}}(t-\frac{\xi}{k})^2} \qquad \qquad k \neq 0,$$

$$\begin{split} & \frac{-\dot{M}_{\nu^3}}{M_{\nu^3}} = \frac{C_{\alpha}\nu}{1+\nu^2(t-\frac{\xi}{k})^2} & k \neq 0, \\ & M_{\cdot}(t=0) = M_{\cdot}(k=0) = 1. \end{split}$$

The weight  $M_L$  is an adaption of the weight  $m^{\frac{1}{2}}$  in [Lis20] to our setting. The method of using time-dependent Fourier weights is common when working with solutions around Couette flow and the other weights are modifications of previously used weights (cf. [BVW18, MZ22, Lis20, ZZ24]). For simplicity, here we only state their main properties and refer to Appendix 5.4 for a detailed description. The constants  $C_{\alpha} = \frac{2}{\min(1,\alpha-\frac{1}{2})}$ ,  $c = \frac{1}{200}(1-\frac{1}{2\alpha})^2$  and  $c_1 = \frac{1}{20}(1-\frac{1}{2\alpha})$  are determined through the linear estimates. For the weights we obtain

$$L^{-1} \leq \min(1, \nu^{-1} \kappa^{\frac{1}{3}} k^{\frac{2}{3}}) \lesssim M_L \leq 1,$$
  

$$M_1 \approx M_\kappa \approx M_\nu \approx M_{\nu^3} \approx 1.$$
(5.16)

We note that the weight  $M_L$  is distinct from the others due to its lower bound  $L^{-1}$ , which depends on  $\nu$  and  $\kappa$ . The weight  $M_L$  is necessary to bound the linear growth in the region  $\nu^{-1} \leq t - \frac{\xi}{k} \leq (\kappa k^2)^{-\frac{1}{3}}$ . Controlling the effects of  $M_L$  is one of the main challenges in the proof of Theorem 1. We recall the unknowns p and equation (5.4)

$$\partial_t p_1 - \partial_x \partial_y^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \nu \Delta_t p_1 + \Lambda_t^{-1} \nabla_t^{\perp} (b \nabla_t b - v \nabla_t v),$$
  

$$\partial_t p_2 + \partial_x \partial_y^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = \kappa \Delta_t p_2 + \Lambda_t^{-1} \nabla_t^{\perp} (b \nabla_t v - v \nabla_t b),$$
  

$$p|_{t=0} = p_{in}.$$
(5.17)

Let  $\chi \in C^{\infty}(\mathbb{R}_+ \times \mathbb{Z} \times \mathbb{R})$  be a Fourier multiplier defined by

$$\chi = \chi(k,\xi) = \begin{cases} 1 & |t - \frac{\xi}{k}| \le \nu^{-1} \\ 0 & |t - \frac{\xi}{k}| \ge 2\nu^{-1} \end{cases}$$
(5.18)

$$\partial_t \chi \le 2\nu.$$
 (5.19)

We define the main energy

$$E := \|Ap_{\neq}\|_{L^2}^2 + \frac{2}{\alpha} \Re \langle \partial_y^t \Delta_t^{-1} \chi Ap_{1,\neq}, Ap_{2,\neq} \rangle.$$

Here  $\Re$  denotes the real part, in the following, we omit writing the symbol  $\Re$  since we derive an upper bound. As  $\alpha > \frac{1}{2}$ , this energy is positive definite and

satisfies

$$(1 - \frac{1}{2\alpha}) \|Ap\|_{L^2} \le E \le (1 + \frac{1}{2\alpha}) \|Ap\|_{L^2}.$$
(5.20)

In the following, we assume initial data as in Theorem 1, i.e. (5.14). We use a bootstrap approach to prove the following two estimates globally in time: The **energy estimate without** *x***-average** 

$$|E||_{L^{\infty}} + ||A\nabla_{t} \otimes (\nu p_{1,\neq}, \kappa p_{2,\neq})||_{L^{2}L^{2}}^{2} + \sum_{j=1,\nu,\kappa,\nu^{3}} ||\sqrt{\frac{-\dot{M}_{j}}{M_{j}}} Ap_{\neq}||_{L^{2}L^{2}}^{2} \leq (C\varepsilon)^{2}$$
(5.21)

The energy estimate with *x*-average

$$\|p_{\pm}\|_{L^{\infty}H^{N}}^{2} + \|\partial_{y}(\nu p_{1,\pm}, \kappa p_{2,\pm})\|_{L^{2}H^{N}}^{2} \le (C\tilde{\varepsilon})^{2}.$$
(5.22)

We then prove that the equality in the estimates is not attained at time T. By local wellposedness, the estimates thus remain valid at least for a short additional time. This contradicts the maximality and thus T has to be infinite. We note that we suppress in our notation the T in the estimates (see (5.3)). With  $1 \le \kappa^{-\frac{1}{3}} \left( \frac{-\dot{M}_{\kappa}}{M_{\kappa}} + \kappa k^2 (1 + (t - \frac{\xi}{k})^2) \right)$  and  $1 \le \nu^{-\frac{1}{3}} \left( \frac{-\dot{M}_{\nu}}{M_{\nu}} + \nu k^2 (1 + (t - \frac{\xi}{k})^2) \right)$  we infer from (5.20) and (5.21) the enhanced dissipation estimates

$$\|Ap_{1,\neq}\|_{L^2L^2} \le 2(1-\frac{1}{2\alpha})^{-1}\nu^{-\frac{1}{6}}C\varepsilon,$$
(5.23)

$$\|Ap_{2,\neq}\|_{L^{2}L^{2}} \le 2(1-\frac{1}{2\alpha})^{-1}\kappa^{-\frac{1}{6}}C\varepsilon.$$
(5.24)

By the construction of  $M_1$  we obtain

$$\|\partial_x \Lambda_t^{-1} A p\|_{L^2} \lesssim \nu^{-\frac{1}{12}} \varepsilon$$

We obtain the energy estimate by deriving the energy E

$$\begin{aligned} \partial_{t}E + 2 \|A\nabla_{t} \otimes (\nu p_{1,\neq}, \kappa p_{2,\neq})\|_{L^{2}}^{2} + 2 \|\sqrt{\frac{-\dot{M}}{M}}Ap_{\neq}\|_{L^{2}}^{2} \\ = & 2c\kappa^{\frac{1}{3}} \|Ap_{\neq}\|_{L^{2}}^{2} \\ & - 2\langle A(1-\chi)p_{1,\neq}, \partial_{x}\partial_{x}^{t}\Delta_{t}^{-1}Ap_{1,\neq}\rangle \\ & + 2\langle A(1-\chi)p_{2,\neq}, \partial_{x}\partial_{x}^{t}\Delta_{t}^{-1}Ap_{2,\neq}\rangle \\ & + \frac{4}{\alpha}\langle\chi\partial_{y}^{t}\Delta_{t}^{-1}Ap_{1,\neq}, \dot{A}p_{2,\neq}\rangle \\ & + \frac{2}{\alpha}\langle\chi\partial_{t}(\partial_{y}^{t}\Delta_{t}^{-1})Ap_{1,\neq}, Ap_{2,\neq}\rangle \\ & + \frac{2}{\alpha}\langle\partial_{t}(\chi)\partial_{y}^{t}\Delta_{t}^{-1}Ap_{1,\neq}, Ap_{2,\neq}\rangle \\ & + \frac{2}{\alpha}\langle\lambda\partial_{t}(\chi\partial_{y}^{t}Ap_{1,\neq}, Ap_{2,\neq}\rangle \\ & + 2\langle Av_{\neq}, A(b\nabla_{t}b - v\nabla_{t}v)\rangle \\ & + 2\langle Ab_{\neq}, A(b\nabla_{t}v - v\nabla_{t}b)\rangle \\ & + \frac{2}{\alpha}\langle\chiA\partial_{y}^{t}\Delta_{t}^{-1}b_{\neq}, A(b\nabla_{t}v - v\nabla_{t}b)\rangle \\ & = L_{1} + L_{NR} + L_{R} + NL_{\neq} + ONL. \end{aligned}$$

Where we denoted by ONL all the terms which include the operator  $\partial_y^t \Delta_t^{-1}$ and NL the one which does not. Furthermore, for the energy of x-averages, we obtain

$$\begin{aligned} \partial_t \|p_{\pm}\|_{H^N} + \|\partial_y(\nu p_{1,\pm}, \kappa p_{2,\pm})\|_{H^N} \\ &\leq \langle \langle \partial_y \rangle^N v_{1,\pm}, \langle \partial_y \rangle^N (b\nabla_t b - v\nabla_t v)_{\pm} \rangle \\ &+ \langle \langle \partial_y \rangle^N b_{1,\pm}, \langle \partial_y \rangle^N (b\nabla_t v - b\nabla_t v)_{\pm} \rangle \\ &= NL_{\pm}. \end{aligned}$$
(5.26)

In the following subsections, we establish the following proposition:

**Proposition 5.4** (Control of errors). Under the assumptions of Theorem 1, there exists a constant  $C = C(\alpha) > 0$  such that if (5.21) and (5.22) are satisfied for T > 0, then the following estimate holds

$$\int_{0}^{T} L_{1} + L_{R} + L_{NR} dt \leq \frac{17 + \frac{3}{2}\alpha}{10} (C\varepsilon)^{2} + 2 \|\sqrt{\frac{-\dot{M}_{L}}{M_{L}}} Ap_{2}\|_{L^{2}L^{2}}^{2}, 
\int_{0}^{T} NL_{\neq} + ONL dt \leq L\nu^{-\frac{1}{12}} \kappa^{-\frac{1}{2}} \varepsilon^{3} + (L\kappa^{-\frac{1}{3}} + \kappa^{-\frac{1}{2}}) \tilde{\varepsilon} \varepsilon^{2}, \qquad (5.27) 
\int_{0}^{T} NL_{=} dt \leq L\nu^{-\frac{1}{12}} \kappa^{-\frac{1}{2}} \tilde{\varepsilon} \varepsilon^{2}.$$

With this proposition we deduce Theorem 1:

Proof of Theorem 1. By a standard application of the Banach fixed-point theorem we obtain local well-posedness, see Appendix 5.5. Thus for all initial data, there exists a time interval [0, T] such that (5.21) and (5.22) hold. Let  $T^*$  be the maximal time such that (5.21) and (5.22) hold. Let  $c_0$  be a given, small constant and suppose for the sake of contradiction that  $T^* < \infty$ . With the estimates (5.25), (5.26) and (5.27) and since  $c_0$  is small we obtain that the estimates (5.21) and (5.22) do not attain equality. Thus by local existence,  $T^*$  is not the maximal time and thus we obtain a contradiction. Therefore, for small enough  $c_0$ , (5.21) and (5.22) hold global in time and so we infer Theorem 1.  $\Box$ 

The remainder of the section is dedicated to the proof of Proposition 5.4. We rearrange and use partial integration to infer that

$$\langle Av_{\neq}, b\nabla_t Ab_{\neq} - v\nabla_t Av_{\neq} \rangle + \langle Ab_{\neq}, b\nabla_t Av_{\neq} - v\nabla_t Ab_{\neq} \rangle$$
  
=  $\langle b, \nabla_t (Av_{\neq} Ab_{\neq}) \rangle - \frac{1}{2} \langle v, \nabla_t (Av_{\neq} Av_{\neq}) + \nabla_t (Ab_{\neq} Ab_{\neq}) \rangle$  (5.28)  
= 0.

The NL term consists of trilinear products with the unknowns

$$a^1 a^2 a^3 \in \{vvv, vbb, bbv, bvb\}.$$
(5.29)

Thus, we denote the nonlinear terms

$$\begin{split} NL_{\neq}[a^{1}a^{2}a^{3}] = & \langle Aa_{\neq}^{1}, A(a_{\neq}^{2}\nabla_{t}a_{\neq}^{3})_{\neq} - a_{\neq}^{2}\nabla_{t}Aa_{\neq}^{3} \rangle, \\ & + \langle Aa_{\neq}^{1}, A(a_{=}^{2}\nabla_{t}a_{\neq}^{3}) - a_{=}^{2}\nabla_{t}a_{\neq}^{3} \rangle, \\ & + \langle Aa_{\neq}^{1}, A(a_{\neq}^{2}\nabla_{t}a_{=}^{3}) \rangle, \\ NL_{=}[a^{1}a^{2}a^{3}] = & \langle \langle \partial_{y} \rangle^{N}a_{=}^{1}, \langle \partial_{y} \rangle^{N}(a_{\neq}^{2}\nabla_{t}a_{\neq}^{3})_{=} \rangle. \end{split}$$

If we do not use specific choices for  $a^1 a^2 a^3$  we write just NL. Similarly, we use  $a^1 a^2 a^3 \in \{bvv, bbb, vbv, vvb\}$  for ONL. Furthermore, we always use i such that  $p_i = \Lambda_t^{-1} \nabla_t^{\perp} a^2$  in the sense that i = 1 if  $a^2 = v$  and i = 2 if  $a^2 = b$ . We perform the energy estimates in the next subsections:

- In Subsection 5.3.1 we estimate the linear error terms. In this subsection, the split with  $\chi$  into resonant and non-resonant regions depending on  $\nu$  is vital.
- In Subsection 5.3.2 we conclude the energy estimate for the nonlinear term without x average. Here it is necessary to perform a low and high frequency decomposition. This gives us a reaction and a transport term. In particular, for  $\kappa \downarrow 0$  bounding the transport term is very challenging due to the linear growth.
- In Subsections 5.3.3, 5.3.4 and 5.3.5 we estimate nonlinear terms with an *x*-average component.
- In Subsection 5.3.6 we estimate nonlinear term which arise due  $\chi$  in the resonant regions. For these terms, we obtain an additional  $\Lambda_t^{-1}$ , which has a stabilizing effect. This stabilizing effect is necessary due to a nonlinear term consisting of only magnetic components.

#### 5.3.1 Linear estimates

In this section, we establish estimates of the linear errors  $L_1$ ,  $L_R$  and  $L_{NR}$  of (5.25). In order to estimate  $L_1$ , we use (5.23) and (5.24) to deduce

$$\int L_1 d\tau = 2c\kappa^{\frac{1}{3}} \|Ap_{\neq}\|_{L^2 L^2}^2 \le 8(1 - \frac{1}{2\alpha})^{-1} c(C\varepsilon)^2.$$

For the  $L_{NR}$  terms in (5.25), we infer

$$\begin{aligned} \langle A(1-\chi)p_{1,\neq}, A\partial_x \partial_x^t \Delta_t^{-1} p_{1,\neq} \rangle &= \sum_{k\neq 0} \int d\xi (1-\chi) \frac{t-\frac{\xi}{k}}{1+(t-\frac{\xi}{k})^2} A^2 p_1^2 \\ &\leq \nu^3 \|A\nabla_t p_{1,\neq}\|_{L^2}^2, \end{aligned}$$

since  $(1-\chi)\frac{t-\frac{\xi}{k}}{1+(t-\frac{\xi}{k})^2} \leq (1-\chi)(\nu)^3(1+(t-\frac{\xi}{k})^2)$  due to  $\chi = 1$  for  $|t-\frac{\xi}{k}| \leq \nu^{-1}$ . Furthermore, using (5.34) we estimate

$$\begin{split} \langle A(1-\chi)p_{2,\neq}, A\partial_x \partial_x^t \Delta_t^{-1} p_2 \rangle &= \sum_{k\neq 0} \int d\xi (1-\chi) \frac{t-\frac{\xi}{k}}{1+(t-\frac{\xi}{k})^2} A^2 p_2^2 \\ &\leq \sum_{k\neq 0} \int d\xi (1-\chi) (\frac{-\dot{M}_L}{M_L} + \kappa c_1 (1+(t-\frac{\xi}{k})^2)) A^2 p_2^2 \\ &\leq \|\sqrt{\frac{-\dot{M}_L}{M_L}} A p_{2,\neq}\|_{L^2}^2 + \kappa c_1 \|\nabla_t A p_{2,\neq}\|_{L^2}^2. \end{split}$$

Thus with (5.21) we deduce

$$\int L_{NR} d\tau \le (2\nu^2 + 2c_1)(C\varepsilon)^2 + 2\|\sqrt{\frac{-\dot{M}_L}{M_L}}Ap_{2,\neq}\|_{L^2L^2}^2.$$

For  $L_R$ , we estimate in frequency space

$$\begin{split} |(\mathbf{1}_{\neq}\partial_{y}^{t}\Delta_{t}^{-1})^{\wedge}| &= |(\frac{\xi-kt}{k^{2}+(\xi-kt)^{2}})_{k\neq0}| \leq \frac{1}{2}, \\ \frac{-\dot{M}_{L}}{M_{L}}|(\partial_{y}^{t}\Delta_{t})^{\wedge}| \leq \left| \left(\frac{(t-\frac{\xi}{k})^{2}}{k(1+(t-\frac{\xi}{k})^{2})^{2}}\right)_{k\neq0} \right| \leq C_{\alpha}^{-1}\frac{-\dot{M}_{1}}{M_{1}}. \end{split}$$

So it follows that

$$\begin{split} &\frac{4}{\alpha} \langle \chi \partial_y^t \Delta_t^{-1} A p_{1,\neq}, \dot{A} p_{2,\neq} \rangle \\ &= \frac{4}{\alpha} \langle \chi A p_{1,\neq}, (c\kappa^{\frac{1}{3}} + \frac{\dot{M}_1}{M_1} + \frac{\dot{M}_L}{M_L} + \frac{\dot{M}_\kappa}{M_\kappa} + \frac{\dot{M}_\nu}{M_\nu} + \frac{\dot{M}_{\nu^3}}{M_{\nu^3}}) \partial_y^t \Delta_t^{-1} A p_{2,\neq} \rangle \\ &\leq \frac{2c}{\alpha} \kappa^{\frac{1}{3}} \|Ap_{\neq}\|_{L^2}^2 + (1 + C_{\alpha}^{-1}) \frac{1}{\alpha} \|\sqrt{\frac{-\dot{M}_1}{M_1}} A p_{\neq}\|_{L^2}^2 \\ &\quad + \frac{1}{\alpha} \|\sqrt{\frac{-\dot{M}_\kappa}{M_\kappa}} A p_{\neq}\|_{L^2}^2 + \frac{1}{\alpha} \|\sqrt{\frac{-\dot{M}_\nu}{M_\nu}} A p_{\neq}\|_{L^2}^2 + \frac{1}{\alpha} \|\sqrt{\frac{-\dot{M}_\nu}{M_\nu^3}} A p_{\neq}\|_{L^2}^2. \end{split}$$

We use the estimate in frequency space

$$\left| \left( \left( (\partial_x^2 - (\partial_y^t)^2) \Delta_t^{-2} \right)_{\neq} \right)^{\wedge} \right| = \left| \left( \frac{1 - (t - \frac{\xi}{k})^2}{k^2 (1 + (t - \frac{\xi}{k})^2)^2} \right)_{k \neq 0} \right| \le C_{\alpha}^{-1} \frac{-\dot{M}_1}{M_1},$$

to infer that

$$\frac{1}{\alpha} \langle \chi A p_{1,\neq}, A \partial_x^{-1} (\partial_x^2 - (\partial_y^t)^2) \Delta_t^{-2} p_{2,\neq} \rangle \leq C_{\alpha}^{-1} \frac{1}{\alpha} \| \sqrt{\frac{-\dot{M}_1}{M_1}} A p_{1,\neq} \|_{L^2} \| \sqrt{\frac{-\dot{M}_1}{M_1}} A p_{2,\neq} \|_{L^2} \\ \leq C_{\alpha}^{-1} \frac{1}{2\alpha} \| \sqrt{\frac{-\dot{M}_1}{M_1}} A p_{\neq} \|_{L^2}^2.$$

With (5.19) we deduce

$$\langle \partial_y^t \Delta_t^{-1} \partial_t(\chi) A p_{1,\neq}, A p_{2,\neq} \rangle \le \nu \|A p_{1,\neq}\|_{L^2} \|A \Lambda_t^{-1} p_{2,\neq}\|_{L^2}.$$

By the Fourier support of  $\chi$  (see (5.18)) and the definition of  $M_{\nu^3}$  we obtain  $\chi \leq 2C_{\alpha}^{-1}\nu^{-1}\frac{-\dot{M}_{\nu^3}}{M_{\nu^3}}\chi$ , which yields

$$\frac{|\nu+\kappa|}{\alpha} \langle \chi A \partial_y^t p_{1,\neq}, A p_{2,\neq} \rangle \le 2C_{\alpha}^{-1} \frac{\nu^{\frac{1}{2}}}{\alpha} \|A \partial_y^t p_{1,\neq}\|_{L^2} \|\sqrt{\frac{-M_{\nu^3}}{M_{\nu^3}}} A p_{2,\neq}\|_{L^2}.$$

Thus for the linear error  $L_R$  we infer

$$\int L_R d\tau \leq \left(\frac{2c}{\alpha} + \nu^{\frac{5}{6}}\right) (C\varepsilon)^2 + \frac{1 + C_{\alpha}^{-1}}{\alpha} \|\sqrt{\frac{-\dot{M_1}}{M_1}} p_{\neq}\|_{L^2 L^2}^2 + \frac{1}{\alpha} \|\sqrt{\frac{-\dot{M_{\kappa}}}{M_{\kappa}}} p_{\neq}\|_{L^2 L^2}^2 + \frac{1}{\alpha} \|\sqrt{\frac{-\dot{M_{\nu}}}{M_{\nu}}} p_{\neq}\|_{L^2 L^2}^2 + \frac{1 + C_{\alpha}^{-1}}{2\alpha} \|\sqrt{\frac{-\dot{M_{\nu}3}}{M_{\nu^3}}} A p_{2,\neq}\|_{L^2 L^2} + C_{\alpha}^{-1} \frac{1}{2\alpha} \nu \|A\partial_y^t p_{1,\neq}\|_{L^2 L^2}^2.$$

Combining the estimates for all linear terms, we obtain

$$\begin{split} \int L + L_R + L_{NR} \, d\tau \\ &\leq \left( (8 + \frac{2}{\alpha})(1 - \frac{1}{2\alpha})^{-1}c + 2\nu^2 + 2c_1 + \nu^{\frac{5}{6}} \right) (C\varepsilon)^2 \\ &+ 2 \| \sqrt{\frac{-\dot{M}_L}{M_L}} Ap_2 \|_{L^2 L^2}^2 \\ &+ (1 + \frac{3}{2}C_\alpha^{-1}) \frac{1}{\alpha} \| \chi \sqrt{\frac{-\dot{M}_1}{M_1}} p_{\neq} \|_{L^2 L^2}^2 + \frac{1}{\alpha} \| \sqrt{\frac{-\dot{M}_\kappa}{M_\kappa}} p_{\neq} \|_{L^2 L^2}^2 + \frac{1}{\alpha} \| \sqrt{\frac{-\dot{M}_\nu}{M_\nu}} p_{\neq} \|_{L^2 L^2}^2 \\ &+ \frac{1 + C_\alpha^{-1}}{2\alpha} \| \sqrt{\frac{-M_{\nu^3}}{M_{\nu^3}}} Ap_{2,\neq} \|_{L^2 L^2} + C_\alpha^{-1} \frac{1}{2\alpha} \nu \| A \partial_y^t p_{1,\neq} \|_{L^2 L^2}^2 \\ &\leq (12(1 - \frac{1}{2\alpha})^{-1}c + 2\nu^2 + 2c_1 + \nu^{\frac{5}{6}} + \frac{1 + 2C_\alpha^{-1}}{2\alpha}) (C\varepsilon)^2 \\ &+ 2 \| \sqrt{\frac{-\dot{M}_L}{M_L}} Ap_{2,\neq} \|_{L^2 L^2}^2. \end{split}$$

Since  $\alpha > \frac{1}{2}$  we deduce  $\frac{1+2C_{\alpha}^{-1}}{\alpha} < 1 + \frac{1}{2\alpha}$ . Choosing the constants such that

$$c = \frac{1}{200} (1 - \frac{1}{2\alpha})^2,$$
  
$$c_1 = \frac{1}{20} (1 - \frac{1}{2\alpha}),$$

and recalling that

$$\nu \le \frac{1}{40} (1 - \frac{1}{2\alpha})^{\frac{6}{5}},$$

we conclude that  $(12(1-\frac{1}{2\alpha})^{-1}c+2\nu^2+2c_1+\nu^{\frac{5}{6}}+\frac{1+2C_{\alpha}^{-1}}{2\alpha}<\frac{17+\frac{3}{2\alpha}}{20}$ . Thus we obtain the estimate

$$\int L_1 + L_R + L_{NR} d\tau \le \frac{17 + \frac{3}{2\alpha}}{20} (C\varepsilon)^2 + 2 \|\sqrt{\frac{-\dot{M}_L}{M_L}} A p_{2,\neq}\|_{L^2 L^2}^2$$

This yields the first estimate of Proposition 5.4.

#### 5.3.2 Nonlinear terms without an *x*-average

We apply the notation of (5.29) and aim to estimate terms of the form

$$\langle Aa_{\neq}^{1}, A(a_{\neq}^{2}\nabla_{t}a_{\neq}^{3}) - a_{\neq}^{2}\nabla_{t}Aa_{\neq}^{3} \rangle$$

$$= \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \frac{A(k,\xi) - A(l,\eta)}{A(k-l,\xi-\eta)A(l,\eta)} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi-\eta - (k-l)t)^{2})^{\frac{1}{2}}}$$

$$(Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta)$$

$$= T + R.$$

Here, we split the integral into the reaction  ${\cal R}$  and the transport T terms which correspond to the sets

$$\Omega_R = \{ |k - l, \xi - \eta| \ge \frac{1}{8} |l, \eta| \},\$$
  
$$\Omega_T = \{ |k - l, \xi - \eta| < \frac{1}{8} |l, \eta| \},\$$

in Fourier space. We split the weights

1

$$\begin{split} A(k,\xi) - A(l,\eta) &= e^{ct\kappa^{\frac{1}{3}}} (M_L(k,\xi) - M_L(l,\eta)) M_1(k,\xi) M_\kappa(k,\xi) M_\nu(k,\xi) M_{\nu^3}(k,\xi) |k,\xi|^N \\ &+ e^{ct\kappa^{\frac{1}{3}}} (|k,\xi|^N - |l,\eta|^N) M_L(l,\eta) M_1(k,\xi) M_\kappa(k,\xi) M_\nu(k,\xi) M_{\nu^3}(k,\xi) |l,\eta|^N \\ &+ e^{ct\kappa^{\frac{1}{3}}} (M_1(k,\xi) - M_1(l,\eta)) M_L(l,\eta) M_\kappa(k,\xi) M_\nu(k,\xi) M_{\nu^3}(k,\xi) |l,\eta|^N \\ &+ e^{ct\kappa^{\frac{1}{3}}} (M_\kappa(k,\xi) - M_\kappa(l,\eta)) M_1(l,\eta) M_L(l,\eta) M_\nu(k,\xi) M_{\nu^3}(k,\xi) |l,\eta|^N \\ &+ e^{ct\kappa^{\frac{1}{3}}} (M_\nu(k,\xi) - M_\nu(l,\eta)) M_1(l,\eta) M_L(l,\eta) M_\kappa(l,\eta) M_{\nu^3}(k,\xi) |l,\eta|^N \\ &+ e^{ct\kappa^{\frac{1}{3}}} (M_{\nu^3}(k,\xi) - M_{\nu^3}(l,\eta)) M_1(l,\eta) M_L(l,\eta) M_\kappa(l,\eta) M_\nu(l,\eta) |l,\eta|^N \end{split}$$

and thus by (5.16) we estimate

$$\frac{A(k,\xi) - A(l,\eta)}{A(k-l,\xi-\eta)A(l,\eta)} \lesssim e^{-ct\kappa^{\frac{1}{3}}} \frac{|M_L(k,\xi) - M_L(l,\eta)|}{M_L(k-l,\xi-\eta)M_L(l,\eta)} \frac{|\xi,\eta|^N}{|l,\eta|^N|k-l,\xi-\eta|^N} \\
+ e^{-ct\kappa^{\frac{1}{3}}} \frac{||k,\xi|^N - |l,\eta|^N|}{|l,\eta|^N|k-l,\xi-\eta|^N} \frac{1}{M_L(k-l,\xi-\eta)} \\
+ e^{-ct\kappa^{\frac{1}{3}}} \sum_{j=1,\kappa,\nu,\nu^3} |M_j(k,\xi) - M_j(l,\eta)| \frac{1}{|k-l,\xi-\eta|^N} \frac{1}{M_L(k-l,\xi-\eta)}.$$
(5.30)

**Reaction term:** On the set  $\Omega_R$  it holds that  $|k - l, \xi - \eta| \geq \frac{1}{8}|l, \eta|$ , thus  $|k, \xi|, |l, \eta| \lesssim |k-l, \xi-\eta|$ . From (5.16), (5.30) and  $\frac{|k, \xi|^N}{|l, \eta|^N |k-l, \xi-\eta|^N}, \frac{||k, \xi|^N - |l, \eta|^N|}{|l, \eta|^N |k-l, \xi-\eta|^N} \lesssim \frac{1}{|l, \eta|^N}$  we infer

$$\frac{A(k,\xi) - A(l,\eta)}{A(k-l,\xi-\eta)A(l,\eta)} \lesssim \frac{1}{M_L(k-l,\xi-\eta)M_L(l,\eta)} \frac{1}{|l,\eta|^N}.$$

With  $\xi l-k\eta=(\xi-\eta-(k-l)t)l-(k-l)(\eta-lt)$  and Hölder's inequality we deduce

$$\begin{split} R &= e^{-ct\kappa^{\frac{1}{3}}} \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_R} \frac{A(k,\xi) - A(l,\eta)}{A(k-l,\xi-\eta)A(l,\eta)} \frac{\xi l - k\eta}{((k-l)^2 + (\xi-\eta - (k-l)\tau)^2)^{\frac{1}{2}}} \\ &\quad (Aa^1)(k,\xi)(Ap_i)(k-l,\xi-\eta)(Aa^3)(l,\eta) \\ &\lesssim e^{-ct\kappa^{\frac{1}{3}}} \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_R} \frac{1}{|l,\eta|^N} \frac{1}{M_L(k-l,\xi-\eta)M_L(l,\eta)} \frac{|(\xi-\eta - (k-l)t)l - (k-l)(\eta-lt)|}{((k-l)^2 + (\xi-\eta - (k-l)\tau)^2)^{\frac{1}{2}}} \\ &\quad (Aa^1)(k,\xi)(Ap_i)(k-l,\xi-\eta)(Aa^3)(l,\eta) \\ &\lesssim \|Aa_l^{\frac{1}{2}}\|_{L^2} \|\frac{1}{M_L}Ap_{i,\neq}\|_{L^2} \|\frac{1}{M_L}Aa_\ell^{\frac{3}{2}}\|_{L^2} \\ &\quad + \|Aa_\ell^{\frac{1}{2}}\|_{L^2} \|\partial_x \frac{1}{M_L}\Lambda_t^{-1}Ap_{i,\neq}\|_{L^2} \|\frac{1}{M_L}A\partial_y^t a_\ell^{\frac{3}{2}}\|_{L^2}. \end{split}$$

We use (5.36) and (5.37) to infer

$$R \lesssim \|Aa_{\neq}^{1}\|_{L^{2}}(\|Ap_{i,\neq}\|_{L^{2}} + \nu\|(\Lambda_{t} \wedge \kappa^{-\frac{1}{3}})Ap_{i,\neq}\|_{L^{2}})(\|Aa_{\neq}^{3}\|_{L^{2}} + \nu\|A(\Lambda_{t} \wedge \kappa^{-\frac{1}{3}})a_{\neq}^{3}\|_{L^{2}}) + L\|Aa_{\neq}^{1}\|_{L^{2}}(\|A\partial_{x}\Lambda_{t}^{-1}p_{i,\neq}\|_{L^{2}} + \nu\|Ap_{i,\neq}\|_{L^{2}})\|\partial_{y}^{t}Aa_{\neq}^{3}\|_{L^{2}}.$$

Integrating in time yields

$$\int R d\tau \lesssim L \nu^{-\frac{1}{12}} \kappa^{-\frac{1}{2}} \varepsilon^3.$$

**Transport term:** On the set  $\Omega_T$  it holds that  $|k - l, \xi - \eta| < \frac{1}{8}|l, \eta|$  and thus it follows that  $|k, \xi| \approx |l, \eta|$ . By the mean value theorem, there exists  $\theta \in [0, 1]$  such that

$$\begin{aligned} \left| |k,\xi|^{N} - |l,\eta|^{N} \right| &\leq N|k-l,\xi-\eta||k-\theta l,\xi-\theta \eta|^{N-1} \\ &\lesssim |k-l,\xi-\eta||l,\eta|^{N-1}. \end{aligned}$$

Thus with (5.30) and Lemma 5.7 we conclude, that

$$\frac{A(k,\xi) - A(l,\eta)}{A(k-l,\xi-\eta)A(l,\eta)} \lesssim \frac{1}{M_L(l,\eta)} \left(\frac{1}{|l|} + \nu^{\frac{1}{12}}\right) \frac{1}{|k-l,\xi-\eta|^{N-1}}$$
(5.31)

+ 
$$\sum_{j=\kappa,\nu,\nu^3} |M_j(k,\xi) - M_j(l,\eta)| \frac{1}{M_L(l,\eta)} \frac{1}{|k-l,\xi-\eta|^N}$$
 (5.32)

$$+ \frac{M_L(k,\xi) - M_L(l,\eta)}{M_L(k-l,\xi-\eta)M_L(l,\eta)} \frac{1}{|k-l,\xi-\eta|^N}.$$
(5.33)

Based on this estimate, in the following we distinguish between different regimes in frequency,

$$\begin{split} T &= \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \frac{A(k,\xi) - A(l,\eta)}{A(k-l,\xi-\eta)A(l,\eta)} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi - \eta - (k-l)\tau)^{2})^{\frac{1}{2}}} \\ &\quad (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ &\lesssim \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \frac{1}{M_{L}(l,\eta)} (\frac{1}{|l|} + \nu^{\frac{1}{12}}) \frac{1}{|k-l,\xi-\eta|^{N-1}} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi - \eta - (k-l)\tau)^{2})^{\frac{1}{2}}} \\ &\quad (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ &+ \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \mathbf{1}_{|\frac{\eta}{l} - l| \geq |\frac{\xi - \eta}{k-l} - l|} \frac{\sum_{j=\kappa,\nu,\nu^{3}} |M_{j}(k,\xi) - M_{j}(l,\eta)|}{M_{L}(l,\eta)} \frac{1}{|k-l,\xi-\eta|^{N}} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi - \eta - (k-l)\tau)^{2})^{\frac{1}{2}}} \\ &\quad (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ &+ \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \mathbf{1}_{|\frac{\eta}{l} - l| \leq |\frac{\xi - \eta}{k-l} - l|} \frac{\sum_{j=\kappa,\nu,\nu^{3}} |M_{j}(k,\xi) - M_{j}(l,\eta)|}{M_{L}(l,\eta)} \frac{1}{|k-l,\xi-\eta|^{N}} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi - \eta - (k-l)\tau)^{2})^{\frac{1}{2}}} \\ &\quad (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ &+ \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \frac{M_{L}(k,\xi) - M_{L}(l,\eta)}{M_{L}(k-l,\xi-\eta)M_{L}(l,\eta)} \frac{1}{|k-l,\xi-\eta|^{N}} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi - \eta - (k-l)\tau)^{2})^{\frac{1}{2}}} \\ &\quad (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ &+ \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \frac{M_{L}(k,\xi) - M_{L}(l,\eta)}{M_{L}(k-l,\xi-\eta)M_{L}(l,\eta)} \frac{1}{|k-l,\xi-\eta|^{N}} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi - \eta - (k-l)\tau)^{2})^{\frac{1}{2}}} \\ &\quad (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ &= T_{1,1} + T_{1,2} + T_{1,3} + T_{2}. \end{split}$$

Here, the  $T_{1,1}$  term is due to estimate (5.31). For (5.32) we distinguish between the frequencies  $|\frac{\eta}{l} - t| \ge |\frac{\xi - \eta}{k - l} - t|$  in  $T_{1,2}$  and  $|\frac{\eta}{l} - t| \le |\frac{\xi - \eta}{k - l} - t|$  in  $T_{1,3}$ . The  $M_L$  commutator (5.33) is  $T_2$ , which requires further splitting. For  $T_{1,1}$  we use  $\xi l - k\eta = (\xi - \eta - (k - l)t)l - (k - l)(\eta - lt)$ , (5.16) and (5.37) to estimate

$$\begin{split} T_{1,1} &= \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \frac{1}{M_{L}(l,\eta)} (\frac{1}{|l|} + \nu^{\frac{1}{12}}) \frac{1}{|k-l,\xi-\eta|^{N-1}} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi-\eta-(k-l)\tau)^{2})^{\frac{1}{2}}} \\ &\quad (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ &\lesssim \|Aa_{\neq}^{1}\|_{L^{2}} \|Ap_{i,\neq}\|_{L^{2}} \|\frac{1}{M_{L}}Aa_{\neq}^{3}\|_{L^{2}} + \|Aa_{\neq}^{1}\|_{L^{2}} \|A\Lambda_{t}^{-1}p_{i,\neq}\|_{L^{2}} \|\frac{1}{M_{L}}A\partial_{y}^{t}a_{\neq}^{3}\|_{L^{2}} \\ &\quad + \nu^{\frac{1}{12}} \|Aa_{\neq}^{1}\|_{L^{2}} \|Ap_{i,\neq}\|_{L^{2}} \|\frac{1}{M_{L}}A\partial_{x}a_{\neq}^{3}\|_{L^{2}} \\ &\leq L \|Aa_{\neq}^{1}\|_{L^{2}} \|Ap_{i,\neq}\|_{L^{2}} \|Aa_{\neq}^{3}\|_{L^{2}} + L \|Aa_{\neq}^{1}\|_{L^{2}} \|A\Lambda_{t}^{-1}p_{i,\neq}\|_{L^{2}} \|A\partial_{y}^{t}a_{\neq}^{3}\|_{L^{2}} \\ &\quad + \nu^{\frac{1}{12}} \|Aa_{\neq}^{1}\|_{L^{2}} \|Ap_{i,\neq}\|_{L^{2}} \|\Lambda_{t}a_{\neq}^{3}\|_{L^{2}}. \end{split}$$

After integrating in time we deduce that

$$\int T_{1,1}d\tau \lesssim (L+\nu^{-\frac{1}{12}})\kappa^{-\frac{1}{2}}\varepsilon^3.$$

For  $T_{1,2}$  we use  $|\frac{\eta}{l} - t| \ge |\frac{\xi - \eta}{k - l} - t|$  to infer that  $|\xi l - k\eta| = |(\xi - \eta - (k - l)t)l - (k - l)(\eta - lt)| \le 2|(k - l)(\eta - lt)|$ . Furthermore, with  $\sum_{i=\nu,\kappa,\nu^3} |M_i(k,\xi) - M_i(l,\eta)| \approx 1$ 

1 and (5.35) we conclude that

$$\begin{split} T_{1,2} &\lesssim \sum_{k,l,k-l \neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \mathbf{1}_{|\frac{\eta}{l} - t| \geq |\frac{\xi - \eta}{k - l} - t|} \frac{1}{M_{L}(l,\eta)} \frac{1}{|k - l,\xi - \eta|^{N}} \frac{|(k - l)(\eta - lt)|}{((k - l)^{2} + (\xi - \eta - (k - l)\tau)^{2})^{\frac{1}{2}}} \\ & (Aa^{1})(k,\xi)(Ap_{i})(k - l,\xi - \eta)(Aa^{3})(l,\eta) \\ &\lesssim \|Aa_{\neq}^{1}\|_{L^{2}} \|A\Lambda_{t}^{-1}p_{i,\neq}\|_{L^{2}} \|\frac{1}{M_{L}}A\partial_{y}^{t}a_{\neq}^{3}\|_{L^{2}} \\ &\lesssim L\|Aa_{\neq}^{1}\|_{L^{2}} \|A\Lambda_{t}^{-1}p_{i,\neq}\|_{L^{2}} \|A\partial_{y}^{t}a_{\neq}^{3}\|_{L^{2}}. \end{split}$$

So after integrating in time, we obtain

$$\int T_{1,2} d\tau \lesssim L \kappa^{-\frac{1}{2}} \varepsilon^3.$$

For  $T_{1,3}$ , we use  $|\frac{\eta}{l} - t| \leq |\frac{\xi - \eta}{k - l} - t|$  to infer  $\xi l - k\eta \leq 2(\xi - \eta - (k - l)t)l$ . Furthermore, with (5.43) we deduce

$$\sum_{j=\kappa,\nu,\nu^3} |M_j(k,\xi) - M_j(l,\eta)| \lesssim \nu^{\frac{1}{3}} \frac{|\xi l - k\eta|}{|kl|}.$$

Combining these two estimates by Hölder's inequality and (5.35) it follows, that

$$\begin{split} T_{1,3} \lesssim \sum_{k,l,k-l \neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \mathbf{1}_{|\frac{\eta}{l}-t| \leq |\frac{\xi-\eta}{k-l}-t|} \frac{1}{M_{L}(l,\eta)} \frac{1}{|k-l,\xi-\eta|^{N}} \frac{\nu^{\frac{1}{3}}(\xi l-k\eta)^{2}}{kl((k-l)^{2}+(\xi-\eta-(k-l)\tau)^{2})^{\frac{1}{2}}} \\ & (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ \lesssim \sum_{k,l,k-l \neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \mathbf{1}_{|\frac{\eta}{l}-t| \leq |\frac{\xi-\eta}{k-l}-t|} \frac{1}{M_{L}(l,\eta)} \frac{1}{|k-l,\xi-\eta|^{N}} \frac{\nu^{\frac{1}{3}}(\xi-\eta-(k-l)t)^{2}l^{2}}{kl((k-l)^{2}+(\xi-\eta-(k-l)\tau)^{2})^{\frac{1}{2}}} \\ & (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ \lesssim \nu^{\frac{1}{3}} \sum_{k,l,k-l \neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \mathbf{1}_{|\frac{\eta}{l}-t| \leq |\frac{\xi-\eta}{k-l}-t|} \frac{1}{M_{L}(l,\eta)} \frac{|\xi-\eta-(k-l)t|}{|k-l,\xi-\eta|^{N-1}} \\ & (Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta) \\ \lesssim \nu^{\frac{1}{3}} \|Aa^{1}_{\#}\|_{L^{2}} \|A\Lambda_{t}p_{i,\#}\|_{L^{2}} \|\frac{1}{M_{L}}Aa^{3}_{\#}\|_{L^{2}} \\ \lesssim L\nu^{\frac{1}{3}} \|Aa^{1}_{\#}\|_{L^{2}} \|A\Lambda_{t}p_{i,\#}\|_{L^{2}} \|Aa^{3}_{\#}\|_{L^{2}}. \end{split}$$

Thus integrating in time yields

$$\int T_{1,3}d\tau \le L\nu^{\frac{1}{6}}\kappa^{-\frac{1}{2}}\varepsilon^3.$$

To estimate the  $T_2$  term, we split the integral into the sets

$$\Omega_{1} = \{\min(t - \frac{\eta}{l}, t - \frac{\xi - \eta}{k - l}) \ge \nu^{-1}\}, 
\Omega_{2} = \{t - \frac{\eta}{l} \ge \nu^{-1} \ge t - \frac{\xi - \eta}{k - l}\}, 
\Omega_{3} = \{t - \frac{\xi - \eta}{k - l} \ge \nu^{-1} \ge t - \frac{\eta}{l}\}, 
\Omega_{4} = \{t - \frac{\xi}{k} \ge \nu^{-1} \ge \max(t - \frac{\eta}{l}, t - \frac{\xi - \eta}{k - l})\}.$$

For frequencies such that  $\nu^{-1} \ge \max(t - \frac{\eta}{l}, t - \frac{\xi}{k})$ , then  $M_L(k, \xi) - M_L(l, \eta) = 0$ and hence the commutator vanishes. Thus the sets  $\Omega_j$  covers all regions of the support. The sets  $\Omega_1, \Omega_2$  and  $\Omega_3$  are chosen to distinguish between  $\frac{1}{M_L} = 1$  and  $\frac{1}{M_L} > 1$  for different frequencies and on set  $\Omega_4$  we use strong dissipation in the first component. We split the set  $T_2$  into

$$T_{2} = \sum_{k,l,k-l\neq 0} \int d(\xi,\eta) \mathbf{1}_{\Omega_{T}} \frac{M_{L}(k,\xi) - M_{L}(l,\eta)}{M_{L}(k-l,\xi-\eta)M_{L}(l,\eta)} \frac{\xi l - k\eta}{((k-l)^{2} + (\xi-\eta - (k-l)\tau)^{2})^{\frac{1}{2}}} \frac{1}{|k-l,\xi-\eta|^{N}}$$

$$(Aa^{1})(k,\xi)(Ap_{i})(k-l,\xi-\eta)(Aa^{3})(l,\eta)(\mathbf{1}_{\Omega_{1}} + \mathbf{1}_{\Omega_{2}} + \mathbf{1}_{\Omega_{3}} + \mathbf{1}_{\Omega_{4}})$$

$$= T_{2,1} + T_{2,2} + T_{2,3} + T_{2,4}.$$

For  $T_{2,1}$  we use (5.35) to deduce

$$T_{2,1} \leq \nu^2 \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_1} \langle t - \frac{\eta}{l} \wedge \kappa^{-\frac{1}{3}} \rangle \langle t - \frac{\xi-\eta}{k-l} \wedge \kappa^{-\frac{1}{3}} \rangle \frac{1}{|k-l,\xi-\eta|^{N-1}} \\ \frac{(\xi-\eta-(k-l)t)l-(k-l)(\eta-lt)}{((k-l)^2+(\xi-\eta-(k-l)\tau)^2)^{\frac{1}{2}}} (Aa^1)(k,\xi)(Ap_i)(k-l,\xi-\eta)(Aa^3)(l,\eta) \\ \lesssim \nu^2 \|Aa_{\neq}^1\|_{L^2} \|(\Lambda_t \wedge \kappa^{-\frac{1}{3}})Ap_{i,\neq}\|_{L^2} \|A\Lambda_t a_{\neq}^3\|_{L^2} \\ + L\nu \|Aa_{\neq}^1\|_{L^2} \|(\Lambda_t \wedge \kappa^{-\frac{1}{3}})\Lambda_t^{-1}\Lambda^{-1}Ap_{i,\neq}\|_{L^2} \|A\partial_y^t a_{\neq}^3\|_{L^2}$$

and so

$$\int T_{2,1} d\tau \lesssim L\nu^{\frac{5}{6}} \kappa^{-\frac{1}{2}} \varepsilon^3.$$

Now we consider  $T_{2,2}$ . By (5.35) we infer

$$T_{2,2} \leq \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_2} \nu \langle t - \frac{\eta}{l} \wedge \kappa^{-\frac{1}{3}} \rangle \frac{1}{|k-l,\xi-\eta|^{N-1}} \frac{(\xi-\eta-(k-l)t)l-(k-l)(\eta-lt)}{((k-l)^2+(\xi-\eta-(k-l)\tau)^2)^{\frac{1}{2}}} (Aa^1)(k,\xi)(Ap_i)(k-l,\xi-\eta)(Aa^3)(l,\eta) \leq \nu \|Aa_{\neq}^1\|_{L^2} \|Ap_{i,\neq}\|_{L^2} \|A\Lambda_t a_{\neq}^3\|_{L^2} + L \|Aa_{\neq}^1\|_{L^2} \|A\Lambda_t^{-1}p_{i,\neq}\|_{L^2} \|A\partial_y^t a_{\neq}^3\|_{L^2}.$$

Integrating in time yields

$$\int T_{2,2} d\tau \lesssim L \kappa^{-\frac{1}{2}} \varepsilon^3.$$

To estimate  $T_{2,3}$ , we need to distinguish between different choices of a. Using (5.35) we estimate

$$\begin{split} T_{2,3} &= \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \mathbf{1}_{\Omega_3} \nu \langle t - \frac{\xi - \eta}{k - l} \wedge \kappa^{-\frac{1}{3}} \rangle \frac{1}{|k - l, \xi - \eta|^{N-1}} \frac{(\xi - \eta - (k - l)t)l - (k - l)(\eta - lt)}{((k - l)^2 + (\xi - \eta - (k - l)\tau)^2)^{\frac{1}{2}}} \\ &\quad (Aa^1)(k,\xi)(Ap_i)(k - l, \xi - \eta)(Aa^3)(l,\eta) \\ &\lesssim \nu \|Aa_{\neq}^1\|_{L^2} \|(\Lambda_t \wedge \kappa^{-\frac{1}{3}})Ap_{i,\neq}\|_{L^2} \|A\partial_x a_{\neq}^3\|_{L^2} \\ &\quad + \nu \|Aa_{\neq}^1\|_{L^2} \|A(\Lambda_t \wedge \kappa^{-\frac{1}{3}})\Lambda^{-1}\Lambda_t^{-1}p_{i,\neq}\|_{L^2} \|A\partial_y^t a_{\neq}^3\|_{L^2} \end{split}$$

and thus after integrating in time

$$\int T_{2,3}[vvv]d\tau \lesssim \nu^{-\frac{1}{2}}\varepsilon^3,$$
$$\int T_{2,3}[bvb]d\tau \lesssim \kappa^{-\frac{1}{2}}\varepsilon^3,$$
$$\int T_{2,3}[bbv]d\tau \lesssim \kappa^{-\frac{1}{2}}\varepsilon^3.$$

In the case of vbb, we use (5.35) to estimate

$$\begin{split} T_{2,3}[vbb] &= \sum_{k,l,k-l \neq 0} \iint d(\xi,\eta) \nu \langle t - \frac{\xi - \eta}{k - l} \rangle \frac{1}{|k - l, \xi - \eta|^{N-1}} \frac{(\xi - \eta - (k - l)t)k - (k - l)(\xi - kt)}{((k - l)^2 + (\xi - \eta - (k - l)\tau)^2)^{\frac{1}{2}}} \\ & (Av)(k,\xi)(Ap_2)(k - l, \xi - \eta)(Ab)(l,\eta) \\ &\lesssim \nu \|\partial_x Av_{\neq}\|_{L^2} \|(\Lambda_t \wedge \kappa^{-\frac{1}{3}})Ap_{2,\neq}\|_{L^2} \|Ab_{\neq}\|_{L^2} \\ &+ \nu \|\partial_y^t Av_{\neq}\|_{L^2} \|(\Lambda_t \wedge \kappa^{-\frac{1}{3}})\Lambda_t^{-1}Ap_{2,\neq}\|_{L^2} \|Ab_{\neq}\|_{L^2} \\ &\leq \nu \|Ab_{\neq}\|_{L^2} \|A\nabla_t v_{\neq}\|_{L^2} \|A\Lambda_t p_{2,\neq}\|_{L^2}. \end{split}$$

Thus after integrating in time, we obtain

$$\int T_{2,3}[vbb]d\tau \lesssim \nu^{\frac{1}{2}} \kappa^{-\frac{1}{2}} \varepsilon^3.$$

For  $T_{2,4}$  we obtain that  $M(l,\eta) = M(k-l,\xi-\eta) = 1$ . We use  $t - \frac{\xi}{k} \ge \nu \ge \max(t - \frac{\eta}{l}, t - \frac{\xi - \eta}{k-l})$  to deduce that

$$1 = \tfrac{kt-\xi}{kt-\xi} = \tfrac{k}{k} \tfrac{t-\frac{\xi}{k}}{t-\frac{\xi}{k}} \leq \nu \tfrac{|\xi-kt|}{|k|}.$$

With  $\xi l - k\eta = (\xi - \eta - (k - l)t)l - (k - l)(\eta - lt)$  we infer that

$$\begin{split} T_{2,4} &= \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \frac{1}{|k-l,\xi-\eta|^N} \frac{\xi l - k\eta}{((k-l)^2 + (\xi-\eta-(k-l)\tau)^2)^{\frac{1}{2}}} \\ &\quad (Aa^1)(k,\xi)(Ap_i)(k-l,\xi-\eta)(Aa^3)(l,\eta) \\ &= \nu \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) |\xi - kt| \frac{1}{|k-l,\xi-\eta|^{N-1}} \frac{(\xi-\eta-(k-l)t)l}{|l|((k-l)^2 + (\xi-\eta-(k-l)\tau)^2)^{\frac{1}{2}}} \\ &\quad (Aa^1)(k,\xi)(Ap_i)(k-l,\xi-\eta)(Aa^3)(l,\eta) \\ &\quad + \sum_{k,l,k-l\neq 0} \iint d(\xi,\eta) \frac{1}{|k-l,\xi-\eta|^{N-1}} \frac{(k-l)(\eta-lt)}{((k-l)^2 + (\xi-\eta-(k-l)\tau)^2)^{\frac{1}{2}}} \\ &\quad (Aa^1)(k,\xi)(Ap_i)(k-l,\xi-\eta)(Aa^3)(l,\eta) \\ &\leq \nu \|A\partial_y^t a_{\neq}^1\|_{L^2} \|Ap_{i,\neq}\|_{L^2} \|Aa_{\neq}^3\|_{L^2} \\ &\quad + \|Aa_{\neq}^{\perp}\|_{L^2} \|\Lambda_t^{-1}Ap_{i,\neq}\|_{L^2} \|A\partial_y^t a_{\neq}^3\|_{L^2}. \end{split}$$

Thus integrating in time yields

$$\int T_{2,4} d\tau \lesssim \kappa^{-\frac{1}{2}} \varepsilon^3.$$

# 5.3.3 Nonlinear terms with an *x*-average in the second component

We apply the notation of (5.29)

$$\langle Aa_{\neq}^{1}, A(a_{1,=}^{2}\partial_{x}a_{\neq}^{3}) - a_{1,=}^{2}\partial_{x}Aa_{\neq}^{3} \rangle$$

$$= \sum_{k\neq 0} \iint d(\xi,\eta)(Aa^{1})(k,\xi)(A(k,\xi) - A(k,\eta))ka_{1}^{2}(0,\xi-\eta)a^{3}(k,\eta)$$

$$= R + T.$$

Here we split into reaction and transport terms according to the sets

$$\Omega_R = \{ |\xi - \eta| \ge \frac{1}{8} |k, \eta| \}, \Omega_T = \{ |\xi - \eta| < \frac{1}{8} |k, \eta| \}.$$

**Reaction term** On the set  $\Omega_R$  it holds that  $|\xi - \eta| \ge \frac{1}{8}|k,\eta|$ , then we obtain  $|A(k,\xi) - A(k,\eta)| \le |\xi - \eta|^N$  and thus with (5.35), it follows that

$$\sum_{k \neq 0} \iint d(\xi, \eta) (Aa^{1})(k, \xi) (A(k, \xi) - A(k, \eta)) ka_{1}^{2}(0, \xi - \eta) a^{3}(k, \eta)$$

$$\lesssim \|Aa_{\neq}^{1}\|_{L^{2}} \|a_{=}^{2}\|_{H^{N}} \|\partial_{x}a_{\neq}^{3}\|_{L^{\infty}}$$

$$\lesssim \|Aa_{\neq}^{1}\|_{L^{2}} \|a_{=}^{2}\|_{H^{N}} \|\frac{1}{M_{L}} Aa_{\neq}^{3}\|_{L^{2}}$$

$$\lesssim L \|Aa_{\neq}^{1}\|_{L^{2}} \|a_{=}^{2}\|_{H^{N}} \|Aa_{\neq}^{3}\|_{L^{2}}.$$

Integrating in time yields a bound

$$\int R \ d\tau \lesssim L \kappa^{-\frac{1}{3}} \varepsilon^2 \tilde{\varepsilon}.$$

**Transport term** On the set  $\Omega_L$  it holds that  $|k,\eta| \ge \frac{1}{8}|\xi - \eta|$ . By the mean value theorem there exists a  $\theta \in [0, 1]$ 

$$||k,\eta|^N - |k,\xi|^N| \lesssim |\xi - \eta||k,\eta - \theta\xi|^{N-1} \lesssim |\xi - \eta||k,\eta|^{N-1}.$$

Thus, we can estimate the difference in A by

$$\begin{aligned} |A(k,\xi) - A(k,\eta)| &\lesssim (M_L(k,\xi) - M_L(k,\eta))|k,\xi|^N \\ &+ M_L(k,\eta)|k,\xi|^N \sum_{j=1,\kappa,\nu,\nu^3} |M_j(k,\xi) - M_j(k,\eta)| \\ &+ M_L(k,\eta)(|k,\eta|^N - |k,\xi|^N) \\ &\lesssim \frac{1}{k} |\xi - \eta| |k,\xi|^N \end{aligned}$$
where we used (5.40), (5.42) and (5.43) to estimate the differences in  $M_j$ . So we infer, that

$$T \le \sum_{k \ne 0} \iint d(\xi, \eta) \mathbf{1}_{\Omega_T} |Aa^1|(k, \xi)| a_1^2 |(0, \xi - \eta) \frac{1}{M_L(k, \eta)} |Aa^3|(k, \eta).$$

and thus integrating in time yields

$$\int T d\tau \lesssim L \|Aa_{\neq}^{1}\|_{L^{2}L^{2}} \|a_{=}^{2}\|_{L^{\infty}H^{N}} \|Aa_{\neq}^{3}\|_{L^{2}L^{2}} \lesssim L \kappa^{-\frac{1}{3}} \varepsilon^{2} \tilde{\varepsilon}.$$

## 5.3.4 Nonlinear terms with an *x*-average in the third component

We aim to estimate

$$\begin{split} \langle Aa_{1,\neq}^{1}, A(a_{2,\neq}^{2}\partial_{y}a_{1,=}^{3}) \rangle \\ &= \sum_{k\neq 0} \iint d(\xi,\eta) (Aa_{1}^{1})(k,\xi) A(k,\xi) \frac{k\eta}{\sqrt{k^{2} + (\xi-\eta-kt)^{2}}} p_{i}(k,\xi-\eta) a_{1}^{3}(0,\eta) \\ &= R+T \end{split}$$

where we split into the reaction and transport terms according to the sets

$$\Omega_R = \{ |k, \xi - \eta| \ge \frac{1}{8} |\eta| \}, \Omega_T = \{ |k, \xi - \eta| < \frac{1}{8} |\eta| \}.$$

**Reaction term** On the set  $\Omega_R$  it holds that  $|k, \xi - \eta| \ge \frac{1}{8} |\eta|$ . With (5.35) we infer

$$\begin{split} R &= \sum_{k \neq 0} \iint d(\xi, \eta) \mathbf{1}_{\Omega_R} (Aa_1^1)(k, \xi) A(k, \xi) \frac{k\eta}{\sqrt{k^2 + (\xi - \eta - kt)^2}} p_i(k, \xi - \eta) a_1^3(0, \eta) \\ &\lesssim \|Aa_{1, \neq}^1\|_{L^2} \|A \frac{1}{M_L} \partial_x \Lambda_t^{-1} p_{i, \neq}\|_{L^2} \|\partial_y a_{1, =}^3\|_{L^{\infty}} \\ &\lesssim \|Aa_{1, \neq}^1\|_{L^2} \|Ap_{i, \neq}\|_{L^2} \|a_{1, =}^3\|_{H^N}. \end{split}$$

Integrating in time then yields

$$\int R d\tau \lesssim \kappa^{-\frac{1}{3}} \varepsilon^2 \tilde{\varepsilon}.$$

**Transport term** On the set  $\Omega_T$  it holds that  $|k, \xi - \eta| \leq \frac{1}{8} |\eta|$ , then with (5.35) we estimate

$$T = \sum_{k \neq 0} \iint d(\xi, \eta) \mathbf{1}_{\Omega_{T}} (Aa_{1}^{1})(k, \xi) A(k, \xi) \frac{k\eta}{\sqrt{k^{2} + (\xi - \eta - kt)^{2}}} p_{i}(k, \xi - \eta) a_{1}^{3}(0, \eta)$$
  
$$\lesssim \|Aa_{\neq}^{1}\|_{L^{2}} \|\partial_{x} \Lambda_{t}^{-1} p_{i, \neq}\|_{L^{\infty}} \|\partial_{y} a_{=}^{3}\|_{H^{N}}$$
  
$$\lesssim \|Aa_{\neq}^{1}\|_{L^{2}} \|\frac{1}{M_{L}} A\Lambda_{t}^{-1} p_{i, \neq}\|_{L^{2}} \|\partial_{y} a_{=}^{3}\|_{H^{N}}$$
  
$$\lesssim \|Aa_{\neq}^{1}\|_{L^{2}} (\|\Lambda_{t}^{-1} Ap_{i}\|_{L^{2}} + \nu \|Ap_{i, \neq}\|_{L^{2}}) \|\partial_{y} a_{=}^{3}\|_{H^{N}}.$$

Integrating in time yields

$$\int T d\tau \lesssim \kappa^{-\frac{1}{2}} \varepsilon^2 \tilde{\varepsilon}.$$

## 5.3.5 Nonlinear terms with an *x*-average in first component

Now we turn to

$$\langle \langle \partial_y \rangle^N a_{\pm,1}^1, \langle \partial_y \rangle^N (a_{\neq}^2 \nabla_t a_{\neq,1}^3)_{\pm} \rangle$$
  
=  $-\sum_{k \neq 0} \iint d(\xi, \eta) \langle \xi \rangle^{2N} a_1^1(0, \xi) \frac{k\xi}{\sqrt{k^2 + (\xi - \eta + kt)^2}} p_i(-k, \xi - \eta) a_1^3(k, \eta).$ 

Applying Hölder's inequality, the Sobolev embedding and the definition of  ${\cal A}$  yields

$$\begin{aligned} &\langle \langle \partial_{y} \rangle^{N} a_{1,=}^{1}, \langle \partial_{y} \rangle^{N} (a_{\neq}^{2} \nabla_{t} a_{\neq,1}^{3}) = \rangle \\ &\leq \|\partial_{y} \langle \partial_{y} \rangle^{N} a_{=}^{1}\|_{L^{2}} (\|\partial_{x} \Lambda_{t}^{-1} p_{i,\neq}\|_{L^{\infty}} \|\langle \partial_{y} \rangle^{N} a_{\neq}^{3}\|_{L^{2}} + \|\langle \partial_{y} \rangle^{N} \Lambda_{t}^{-1} p_{i,\neq}\|_{L^{2}} \|\partial_{x} a_{\neq}^{3}\|_{L^{\infty}} \\ &\leq \|\partial_{y} a_{=}^{1}\|_{H^{N}} \|A \frac{1}{M_{L}} \Lambda_{t}^{-1} p_{i,\neq}\|_{L^{2}} \|A \frac{1}{M_{L}} a_{\neq}^{3}\|_{L^{2}} \end{aligned}$$

With (5.35) we infer

$$\begin{split} &\langle \langle \partial_y \rangle^N a_{1,=}^1, \langle \partial_y \rangle^N (a_{\neq}^2 \nabla_t a_{\neq,1}^3) = \rangle \\ &\lesssim \|\partial_y a_{=}^1\|_{H^N} (\|A\Lambda_t^{-1} p_{i,\neq}\|_{L^2} + \nu \|Ap_{i,\neq}\|_{L^2}) (\|Aa_{\neq}^3\|_{L^2} + \nu \|A(\Lambda_t \wedge \kappa^{-\frac{1}{3}})a_{\neq}^3\|_{L^2}) \\ &\lesssim \|\partial_y a_{=}^1\|_{H^N} \|A\Lambda_t^{-1} p_{i,\neq}\|_{L^2} (\|Aa_{\neq}^3\|_{L^2} + \nu \|(\Lambda_t \wedge \kappa^{-\frac{1}{3}})a_{\neq}^3\|_{L^2}) \\ &+ \nu \|\partial_y a_{=}^1\|_{H^N} \|Ap_{i,\neq}\|_{L^2} (\|Aa_{\neq}^3\|_{L^2} + \nu \|(\Lambda_t \wedge \kappa^{-\frac{1}{3}})a_{\neq}^3\|_{L^2}). \end{split}$$

Integrating in time yields

$$\int \langle \langle \partial_y \rangle^N a_{\pm}^1, \langle \partial_y \rangle^N (a_{\neq}^2 \nabla_t a_{\neq,1}^3)_{\pm} \rangle d\tau \lesssim L \kappa^{-\frac{1}{2}} \varepsilon^2 \tilde{\varepsilon}.$$

#### 5.3.6 Other nonlinear terms

In this subsection, we aim to estimate

$$\begin{split} \langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a^2 \nabla_t a^3) \rangle &= \langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{\neq}^2 \nabla_t a_{\neq}^3) \rangle \\ &+ \langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{\neq}^2 \nabla_t a_{=}^3) \rangle \\ &+ \langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{=}^2 \nabla_t a_{\neq}^3) \rangle. \end{split}$$

with the choices  $a^1a^2a^3\in\{bvv,bbb,vbv,vvb\}.$  We start with the case of no x-averages and use

$$\xi l - k\eta = (\xi - kt)(l - k) + k(\xi - \eta - (k - l)t)$$

and (5.35) to infer

$$\begin{split} &\langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{\neq}^2 \nabla_t a_{\neq}^3) \rangle \\ &= \sum_{k,l,k-l \neq 0} \iint d(\xi,\eta) \frac{\xi - kt}{k^2 + (\xi - kt)^2} \frac{\xi l - k\eta}{\sqrt{(k-l)^2 + (\xi - \eta - (k-l)t)^2}} A^2(k,\xi) a^1(k,\xi) p_i(k-l,\xi-\eta) a^3(l,\eta) \\ &\lesssim \|Aa_{\neq}^1\|_{L^2} \|\frac{1}{M_L} \partial_x \Lambda_t^{-1} A p_{i,\neq}\|_{L^2} \|\frac{1}{M_L} A a_{\neq}^3\|_{L^2} \\ &+ \|\partial_x \Lambda_t^{-1} A a_{\neq}^1\|_{L^2} \|\frac{1}{M_L} A p_{i,\neq}\|_{L^2} \|\frac{1}{M_L} A a_{\neq}^3\|_{L^2} \\ &\lesssim L \|Aa_{\neq}^1\|_{L^2} (\|\partial_x \Lambda_t^{-1} A p_{i,\neq}\|_{L^2} + \nu \|A p_{i,\neq}\|_{L^2}) \|Aa_{\neq}^3\|_{L^2} \\ &+ L(1 + \nu \kappa^{-\frac{1}{3}}) \|\partial_x \Lambda_t^{-1} A a_{\neq}^1\|_{L^2} \|A p_{i,\neq}\|_{L^2} \|A a_{\neq}^3\|_{L^2}. \end{split}$$

Thus integrating in time yields

$$\int \langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{\neq}^2 \nabla_t a_{\neq}^3) \rangle d\tau \lesssim L \kappa^{-\frac{1}{2}} \varepsilon^3.$$

For the case, when the average is in the second component, we use partial integration to estimate

$$\begin{split} &\langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{1,=}^2 \partial_x a_{\neq}^3) \rangle \\ &= -\langle \chi A \partial_x \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{1,=}^2 a_{\neq}^3) \rangle \\ &\lesssim \|\partial_x \Lambda_t^{-1} a_{\neq}^1\|_{L^2} \|a_{=}^2\|_{H^N} \|\frac{1}{M_L} A a_{\neq}^3\|_{L^2} \\ &\lesssim L \|\partial_x \Lambda_t^{-1} a_{\neq}^1\|_{L^2} \|a_{=}^2\|_{H^N} \|A a_{\neq}^2\|_{L^2} \end{split}$$

and thus integrating in time and using  $L = \max(1, \nu \kappa^{-\frac{1}{3}})$  yields

$$\int \langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{1,=}^2 \partial_x a_{\neq}^3) \rangle d\tau \lesssim \kappa^{-\frac{1}{2}} \varepsilon^2 \tilde{\varepsilon}.$$

For the case when the average is in the third component, we obtain

$$\langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{2,\neq}^2 \partial_y a_{=}^3) \rangle$$

$$= \sum_{k \neq 0} \iint d(\xi, \eta) \chi \frac{\xi - kt}{k^2 + (\xi - kt)^2} \frac{k\eta}{\sqrt{k^2 + (\xi - \eta - t)^2}} \frac{A(\xi, k)}{A(\xi - \eta, k)} (Aa^1)(k, \xi) (Ap_i)(k, \xi - \eta) a^3(0, \eta).$$

Thus by  $\eta = \xi - kt - (\xi - \eta - kt)$  we estimate

$$\langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{2,\neq}^2 \partial_y a_{=}^3) \rangle$$
  

$$\leq \| (\partial_y^t)^2 \Delta_t^{-1} A a_{\neq}^1 \|_{L^2} \| \partial_x \Lambda_t^{-1} \frac{1}{M_L} A p_{i,\neq} \|_{L^2} \| a_{=}^3 \|_{H^N}$$
  

$$+ \| \partial_x \partial_y^t \Delta_t^{-1} A a_{\neq}^1 \|_{L^2} \| \frac{1}{M_L} A p_{i,\neq} \|_{L^2} \| a_{=}^3 \|_{H^N}$$
  

$$\leq L \| A a_{\neq}^1 \|_{L^2} \| \partial_x \Lambda_t^{-1} A p_{i,\neq} \|_{L^2} \| a_{=}^3 \|_{H^N}$$
  

$$+ L \| \partial_x \Lambda_t^{-1} A a_{\neq}^1 \|_{L^2} \| A p_{i,\neq} \|_{L^2} \| a_{=}^3 \|_{H^N}$$

Integrating in time and using  $L = \max(1, \nu \kappa^{-\frac{1}{3}})$  yields

$$\int \langle \chi A \partial_y^t \Delta_t^{-1} a_{\neq}^1, A(a_{2,\neq}^2 \partial_y a_{=}^3) \rangle d\tau \lesssim \kappa^{-\frac{1}{2}} \varepsilon^2 \tilde{\varepsilon}.$$

Which concludes the estimate

$$\int ONL d\tau \lesssim L \kappa^{-\frac{1}{2}} \tilde{\varepsilon} \varepsilon^2.$$

Combining the estimates of Subsection 5.3.2 to 5.3.6 completes the proof of Proposition 5.4 and thus Theorem 1.  $\hfill \Box$ 

In this article, we have shown that the MHD equations around Couette flow with magnetic resistivity smaller than fluid viscosity  $\nu \geq \kappa > 0$  are stable for initial data which is small enough in Sobolev spaces. If the resistivity is much smaller than the viscosity,  $\nu \kappa^{-\frac{1}{3}} > 0$ , large viscosity destabilizes the equation, leading to norm inflation of size  $\nu \kappa^{-\frac{1}{3}}$ . Controlling this norm inflation is a major new challenge compared to other dissipation regimes.

#### 5.4 Construction of the Weights

Let A be the Fourier weight

$$A := M \langle \nabla \rangle e^{c \kappa^{\frac{1}{3}} t \mathbf{1}_{\neq}},$$

with  $M = M_1 M_L M_\kappa M_\nu M_{\nu^3}$  defined as

$$\frac{-\dot{M}_L}{M_L} = \frac{t - \frac{\xi}{k}}{1 + (\frac{\xi}{k} - t)^2} \mathbf{1}_{\{\nu^{-1} \le t - \frac{\xi}{k} \le (c_1 \kappa k^2)^{-\frac{1}{3}}\}} \qquad k \ne 0,$$

$$\frac{-\dot{M}_1}{M_1} = C_\alpha \frac{|k| + \nu^{\frac{1}{12}} |k|^2}{k^2 + (\xi - kt)^2} \qquad \qquad k \neq 0,$$

$$\frac{-\dot{M}_{\nu}}{M_{\nu}} = \frac{\nu^{\frac{1}{3}}}{1 + \nu^{\frac{2}{3}}(t - \frac{\xi}{k})^2} \qquad \qquad k \neq 0,$$

$$\frac{-\dot{M}_{\kappa}}{M_{\kappa}} = \frac{\kappa^{\frac{1}{3}}}{1+\kappa^{\frac{2}{3}}(t-\frac{\xi}{k})^2} \qquad \qquad k \neq 0$$

$$\begin{split} & \frac{-\dot{M}_{\nu^3}}{M_{\nu^3}} = \frac{C_{\alpha}\nu}{1+\nu^2(t-\frac{\xi}{k})^2} & k \neq 0, \\ & M_{\cdot}(t=0) = M_{\cdot}(k=0) = 1. \end{split}$$

The weight  $M_L$  is an adaption of the weight  $m^{\frac{1}{2}}$  in [Lis20] to the present setting and  $M_{\nu^3}$  we use to differentiate between resonant and non-resonant regions. The method of using time-dependent Fourier weights is common when working at solutions around Couette flow and the other weights are modifications of previously used weights (cf. [BVW18, MZ22, Lis20, ZZ24] for shear related systems such as Navier-Stokes). The constants  $C_{\alpha} = \frac{2}{\min(1,\alpha-\frac{1}{2})}, c = \frac{1}{20}(1-\frac{1}{2\alpha})^2$  and  $c_1 = \frac{1}{20}(1-\frac{1}{2\alpha})$  are determined through the linear estimates. For the weights we obtain that for all times t > 0, it holds that

$$M_1 \approx M_\kappa \approx M_\nu \approx M_{\nu^3} \approx 1,$$
$$L^{-1} \le \min(1, \nu^{-1} \kappa^{\frac{1}{3}} k^{\frac{2}{3}}) \lesssim M_L \le 1.$$

**Lemma 5.5** ( $M_L$  properties). The weight  $M_L$  satisfies the following bounds

$$\mathbf{1}_{|t-\frac{\xi}{k}| \ge \nu^{-1}} \frac{t-\frac{\xi}{k}}{1+(t-\frac{\xi}{k})^2} \le \frac{-\dot{M}_L}{M_L} + \kappa k^2 c_1 (1+(t-\frac{\xi}{k})^2),$$
(5.34)

$$\frac{1}{M_L(k,\xi)} \le 1 + \nu^{\frac{1}{2}} \langle t - \frac{\xi}{k} \rangle \wedge \kappa^{-\frac{1}{3}}.$$
(5.35)

Furthermore, it follows for  $a \in H^1$ , that

$$\|\frac{1}{M_L}a_{\neq}\|_{L^2} \le \|a_{\neq}\|_{L^2} + \|(\Lambda_t^{-1} \wedge \kappa^{-\frac{1}{3}})a_{\neq}\|_{L^2}, \tag{5.36}$$

$$\|\frac{1}{M_L}\partial_x a_{\neq}\|_{L^2} \le \|\Lambda_t a_{\neq}\|_{L^2}.$$
(5.37)

*Proof.* This follows immediately, from the definition of  $M_L$ .

**Lemma 5.6** (Enhanced dissipation estimates ). The weights  $M_{\nu}$  and  $M_{\kappa}$  satisfy the following bounds

$$\frac{1}{2}\nu^{\frac{1}{3}} \le \frac{-\dot{M}_{\kappa}}{M_{\kappa}} + \nu(k^2 + (\xi - kt)^2), \qquad (5.38)$$

$$\frac{1}{2}\kappa^{\frac{1}{3}} \le \frac{-\dot{M}_{\nu}}{M_{\nu}} + \kappa (k^2 + (\xi - kt)^2).$$
(5.39)

*Proof.* This follows immediately, from the definition of  $M_{\nu}$  and  $M_{\kappa}$ .

**Lemma 5.7** (Difference estimates). Let  $k, l \in \mathbb{Z} \setminus \{0\}$  and  $\xi, \eta \in \mathbb{R}$ , then there hold the following bounds on differences

$$1 - \frac{M_1(k,\xi)}{M_1(k,\eta)} \lesssim \frac{|\xi - \eta|}{|k|}, \tag{5.40}$$

$$1 - \frac{M_1(k,\xi)}{M_1(l,\eta)} \lesssim \frac{|k-l|}{|l|} + \nu^{\frac{1}{12}},\tag{5.41}$$

$$M_L(k,\eta) - M_L(k,\xi) \le 2\frac{|\xi-\eta|}{k},$$
 (5.42)

$$1 - \frac{M_j(k,\xi)}{M_j(l,\eta)} \le 2j^{\frac{1}{3}} \frac{|\xi l - k\eta|}{|kl|}, \qquad j \in \{\kappa, \nu, \nu^3\}$$
(5.43)

*Proof.* We start with the  $M_1$  estimate (5.40) and consider  $M_1(k,\xi) \leq M_1(k,\eta)$ 

$$\begin{split} 1 - \frac{M_1(k,\xi)}{M_1(k,\eta)} &= 1 - \exp\left(-\left|\int_0^t \frac{|k| + |k|^2 \nu^{\frac{1}{12}}}{k^2 + (\xi - k\tau)^2} - \frac{|k| + |k|^2 \nu^{\frac{1}{12}}}{k^2 + (\eta - kt)^2} d\tau\right|\right),\\ &\leq 1 - \exp\left(-\int_{[\frac{\xi}{k}, \frac{\eta}{k}] \cup t - [\frac{\xi}{k}, \frac{\eta}{k}]} 1 \ d\tau\right) \lesssim \frac{|\xi - \eta|}{|k|}. \end{split}$$

The case  $M_1(k,\xi) \ge M_1(k,\eta)$  follows by the same argument and  $M_1(k,\xi) \approx M_1(k,\eta) \approx 1$ . For (5.41) we consider the case  $M_1(k,\xi) \le M_1(l,\eta)$  and infer that

$$1 - \frac{M_1(k,\xi)}{M_1(l,\eta)} = 1 - \exp\left(-\left|\int_0^t \frac{|k| + |k|^2 \nu^{\frac{1}{12}}}{k^2 + (\xi - kt)^2} - \frac{|l| + |l|^2 \nu^{\frac{1}{12}}}{l^2 + (\eta - lt)^2}\right|\right),$$
  
=  $1 - \exp(-2\pi(\frac{1}{l\wedge k} + \nu^{\frac{1}{12}})) \lesssim \frac{1}{l\wedge k} + \nu^{\frac{1}{12}} \lesssim \frac{|k-l|}{|l|} + \nu^{\frac{1}{12}}$ 

The case  $M_1(k,\xi) \ge M_1(l,\eta)$  follows by the same argument and  $M_1(k,\xi) \approx M_1(l,\eta) \approx 1$ . For (5.42) we consider the case  $M_L(k,\xi) \le M_L(k,\eta)$  and thus

$$M_L(k,\eta) - M_L(k,\xi) = M_L(k,\eta)(1 - \frac{M_L(k,\xi)}{M_L(k,\eta)}) \lesssim 1 - \frac{M_L(k,\xi)}{M_L(k,\eta)}.$$

We infer

$$1 - \frac{M_L(k,\xi)}{M_L(k,\eta)} = 1 - \exp\left(-\left|\int_0^t \frac{\tau - \frac{\xi}{k}}{1 + (\tau - \frac{\xi}{k})^2} d\tau - \int_0^t \frac{\tau - \frac{\eta}{k}}{1 + (\tau - \frac{\eta}{k})^2} d\tau\right|\right),\$$
  
=  $1 - \exp\left(-\int_{-\left[\frac{\xi}{k}, \frac{\eta}{k}\right] \cup t - \left[\frac{\xi}{k}, \frac{\eta}{k}\right]} 1 d\tau\right),\$   
 $\leq 2\frac{|\xi - \eta|}{k}.$ 

The case  $M_L(k,\xi) \ge M_L(k,\eta)$  follows by the same argument.

For (5.43) we estimate the  $M_{\kappa}$  difference, since the  $M_{\nu}$  and  $M_{\nu^3}$  differences are done similar. Let  $M_{\kappa}(k,\xi) \geq M_{\kappa}(l,\eta)$ , then it follows

$$\begin{split} 1 - \frac{M_{\kappa}(k,\xi)}{M_{\kappa}(l,\eta)} &= 1 - \exp\left(-\kappa^{\frac{1}{3}} |\int_{0}^{t} \frac{1}{1+\kappa^{\frac{2}{3}}(t-\frac{\xi}{k})^{2}} - \frac{1}{1+\kappa^{\frac{2}{3}}(t-\frac{\eta}{l})^{2}}|\right),\\ &\leq 1 - \exp\left(-\kappa^{\frac{1}{3}} |\int_{0}^{t} \mathbf{1}_{-[\frac{\xi}{k},\frac{\eta}{l}]\cup t-[\frac{\xi}{k},\frac{\eta}{l}]}(\tau)d\tau|\right),\\ &\leq 1 - \exp\left(-2\kappa^{\frac{1}{3}} |\frac{\xi}{k} - \frac{\eta}{l}|\right),\\ &\leq \kappa^{\frac{1}{3}} \frac{|\xi l - k\eta|}{|kl|}. \end{split}$$

The case  $M_{\kappa}(k,\xi) \leq M_{\kappa}(l,\eta)$  follows from the same steps and  $M_{\kappa}(k,\xi) \approx M_{\kappa}(l,\eta)$ .

#### 5.5 Local Wellposedness

We expect the local wellposedness result to be well-known, but were not able to find it stated in the literature. In the following, we prove the local wellposedness by a standard application of the Banach fixed-point theorem.

**Proposition 5.8.** Consider equation (5.17) with initial data  $p_{in} \in H^N$  for  $N \geq 5$ . Then there exists a time T such that there exists a unique solution  $p(t) \in H^N$  to (5.17) for all  $t \in [0, T]$ .

*Proof.* We prove existence with the Banach fixed-point theorem. Let  $T = 1 + 2\|p_{in}\|_{H^N}(1+\frac{8}{\kappa})$  and let X be the space

 $X = \{ p \in L^{\infty} H^{N} \cap CH^{N-2} : \ p(t=0) = p_{in}, \ \|p\|_{L^{\infty} H^{N}}^{2} + \frac{\kappa}{2} \|\nabla_{t}p\|_{L^{\infty} H^{N}}^{2} \le 2\|p_{in}\|_{H^{N}}^{2} \}$ 

with the norm

$$\|p\|_X^2 := \|p\|_{L^{\infty}H^N}^2 + \frac{\kappa}{2} \|\nabla_t p\|_{L^{\infty}H^N}^2$$

We define  $F: X \mapsto X$  as a mapping  $q \mapsto p = F(q)$  such that p solves

$$\begin{split} \partial_t p_1 &- \partial_x \partial_y^t \Delta_t^{-1} p_1 - \alpha \partial_x p_2 = \nu \Delta_t p_1 + \Lambda_t^{-1} \nabla_t^{\perp} (\nabla_t^{\perp} \Lambda_t^{-1} q_2 \nabla_t b - \nabla_t^{\perp} \Lambda_t^{-1} q_1 \nabla_t v), \\ \partial_t p_2 &+ \partial_x \partial_y^t \Delta_t^{-1} p_2 - \alpha \partial_x p_1 = \kappa \Delta_t p_2 + \Lambda_t^{-1} \nabla_t^{\perp} (\nabla_t^{\perp} \Lambda_t^{-1} q_2 \nabla_t v - \nabla_t^{\perp} \Lambda_t^{-1} q_1 \nabla_t b), \\ p|_{t=0} &= p_{in}. \end{split}$$

Then the mapping F satisfies:

- 1. The mapping  $F: X \to X$  is well defined on X.
- 2. The mapping F is a contraction, i.e.  $||F(p) F(\tilde{p})||_{L^{\infty}H^N} \leq \frac{1}{2} ||p \tilde{p}||_{L^{\infty}H^N}$ .

Since X is a complete metric space, if we prove (1) and (2), then it follows that F has a unique fixpoint by the Banach fixed-point theorem.

1. Let  $q \in X$ , then we obtain for p = F(q)

$$\begin{aligned} \partial_t \|p\|_{H^N}^2 + \kappa \|\nabla_t p\|_{H^N}^2 &\leq \|p\|_{H^N}^2 + \langle \Lambda^N v, \Lambda^N (\nabla_t^{\perp} \Lambda_t^{-1} q_2 \nabla_t b - \nabla_t^{\perp} \Lambda_t^{-1} q_1 \nabla_t v) \rangle \\ &+ \langle \Lambda^N b, \Lambda^N (\nabla_t^{\perp} \Lambda_t^{-1} q_2 \nabla_t v - \nabla_t^{\perp} \Lambda_t^{-1} q_1 \nabla_t b) \rangle \\ &\leq \|p\|_{H^N}^2 + \|p\|_{H^N} \|q\|_{H^N} \|\nabla_t p^{n+1}\|_{H^N} \\ &\leq \|p\|_{H^N}^2 + \frac{2}{\kappa} \|p\|_{H^N}^2 \|q\|_{H^N}^2 + \frac{\kappa}{2} \|\nabla_t p\|_{H^N}^2. \end{aligned}$$

Thus we obtain

$$\begin{aligned} \|p\|_X^2 &\leq \|p_{in}\|_{H^N}^2 + T(1 + \frac{1}{\kappa} \|q\|_{L^{\infty}H^N}^2) \|p\|_{L^{\infty}H^N}^2 \\ &\leq \|p_{in}\|_{H^N}^2 + T(1 + \frac{2}{\kappa} \|p_{in}\|_{H^N}^2) \|p\|_{L^{\infty}H^N}^2, \end{aligned}$$

Since

$$T(1+\frac{4}{\kappa}||p_{in}||^2_{H^N}) < \frac{1}{2}$$

we infer the bound

$$\|p\|_X^2 \le 2\|p_{in}\|_{H^N}^2.$$

As  $\partial_t p \in H^{N-2}$ , it follows that  $p \in CH^{N-2}$  and thus  $p \in X$ .

2. We show that F is a contraction. Let  $q,\tilde{q}\in X$  we denote p=F(q) and  $\tilde{p}=F(\tilde{q}).$  We need to show that

$$||p - \tilde{p}||_X < \frac{1}{2} ||q - \tilde{q}||_X,$$

by time estimate we obtain

$$\begin{split} \partial_t \| p - \tilde{p} \|_{H^N}^2 + \kappa \| \nabla_t (p - \tilde{p}) \|_{H^N}^2 &\leq \| p - \tilde{p} \|_{H^N}^2 \\ &+ \langle \Lambda^N (v - \tilde{v}), \Lambda^N (\nabla_t^{\perp} \Lambda_t^{-1} q_2 \nabla_t b - \nabla_t^{\perp} \Lambda_t^{-1} q_1 \nabla_t v) \rangle \\ &+ \langle \Lambda^N (b - \tilde{b}), \Lambda^N (\nabla_t^{\perp} \Lambda_t^{-1} q_2 \nabla_t v - \nabla_t^{\perp} \Lambda_t^{-1} q_1 \nabla_t b) \rangle \\ &- \langle \Lambda^N (v - \tilde{v}), \Lambda^N (\nabla_t^{\perp} \Lambda_t^{-1} \tilde{q}_2 \nabla_t \tilde{b} - \nabla_t^{\perp} \Lambda_t^{-1} \tilde{q}_1 \nabla_t v) \rangle \\ &- \langle \Lambda^N (b - \tilde{b}), \Lambda^N (\nabla_t^{\perp} \Lambda_t^{-1} \tilde{q}_2 \nabla_t \tilde{v} - \nabla_t^{\perp} \Lambda_t^{-1} \tilde{q}_1 \nabla_t \tilde{b}) \rangle \\ &\leq \| p - \tilde{p} \|_{H^N}^2 + \| p - \tilde{p} \|_{H^N} (\| q - \tilde{q} \|_{H^N} \| \nabla_t \tilde{p} \|_{H^N} + \| q \|_{H^N} \| \nabla_t (p - \tilde{p}) \|_{H^N}) \\ &\leq \| p - \tilde{p} \|_{H^N}^2 (1 + \frac{2}{\kappa} \| q \|_{H^N}^2) \\ &+ \| p - \tilde{p} \|_{H^N} \| q - \tilde{q} \|_{H^N} \| \nabla_t \tilde{p} \|_{H^N} + \frac{\kappa}{2} \| \nabla_t (p - \tilde{p}) \|_{H^N}^2. \end{split}$$

Integrating in time yields

$$\begin{split} \|p - \tilde{p}\|_{L^{\infty}H^{N}}^{2} + \frac{\kappa}{2} \|\nabla_{t}(p - \tilde{p})\|_{L^{2}H^{N}}^{2} \\ &\leq \|p - \tilde{p}\|_{L^{\infty}H^{N}}^{2} T(1 + \frac{2}{\kappa} \|q\|_{L^{\infty}H^{N}}^{2}) \\ &+ \sqrt{T} \|p - \tilde{p}\|_{L^{\infty}H^{N}} \|q - \tilde{q}\|_{L^{\infty}H^{N}} \|\nabla_{t}\tilde{p}\|_{L^{2}H^{N}}^{2} \\ &\leq \|p - \tilde{p}\|_{L^{\infty}H^{N}}^{2} T(1 + \frac{2}{\kappa} \|q\|_{L^{\infty}H^{N}}^{2} + 4 \|\nabla_{t}\tilde{p}\|_{L^{2}H^{N}}^{2}) \\ &+ \frac{T}{4} \|q - \tilde{q}\|_{L^{\infty}H^{N}}^{2}. \end{split}$$

Choosing T such that

$$T(1 + \frac{2}{\kappa} \|q\|_{H^N}^2 + \|\nabla_t \tilde{p}\|_{L^2 H^N}^2) \le T + 2T \|p_{in}\|_{H^N} (1 + \frac{8}{\kappa}) < \frac{1}{2},$$

it follows, that

$$\|p - \tilde{p}\|_X^2 \le \frac{1}{2} \|p - \tilde{p}\|_{H^N} + \frac{T}{4} \|q - \tilde{q}\|_{L^{\infty} H^N}^2.$$

We hence conclude, that

$$||p - \tilde{p}||_X^2 \le \frac{1}{2} ||q - \tilde{q}||_X^2$$

### Appendix A

## Mathematical Background and Notation

#### Fourier transformation

We consider the Fourier transformation on the space  $\mathbb{T} \times \mathbb{R}$ . Due to linearity, all the properties can be deduced by the Fourier transformation on  $\mathbb{R}$  and the Fourier series on  $\mathbb{T}$ . In the following, we list the main properties (see [Gra14]). Let  $f \in L^2(\mathbb{T} \times \mathbb{R})$ , we define for  $(h, \xi) \in \mathbb{T} \times \mathbb{R}$  it's Fourier transform as

Let  $f \in L^2(\mathbb{T} \times \mathbb{R})$ , we define for  $(k, \xi) \in \mathbb{Z} \times \mathbb{R}$  it's Fourier transform as

$$\mathcal{F}f(k,\xi) := \hat{f}(k,\xi) := \frac{1}{2\pi} \int_0^{2\pi} dx \int dy \ e^{-i(kx+\xi y)} f(x,y)$$

For a function  $g \in L^2(\mathbb{Z} \times \mathbb{R})$  we define the inverse Fourier transform

$$\mathcal{F}^{-1}g(x,y) := \check{g}(x,y) := \frac{1}{2\pi} \sum_{k} \int d\xi \ e^{i(kx+\xi y)}g(k,\xi).$$

Then, it holds that

$$\mathcal{F} \circ \mathcal{F}^{-1} = \mathrm{Id}_{L^2} \qquad \mathcal{F}^{-1} \circ \mathcal{F} = \mathrm{Id}_{L^2}.$$

The Fourier transformation acts as an isometry on  $L^2$ , i.e. for all  $f \in L^2$ 

$$\|f\|_{L^2} = \|\hat{f}\|_{L^2}.$$

For functions f such that  $\partial_x f \in L^2$  and  $\partial_y f \in L^2$  it holds that

$$(\partial_x f)^{\wedge} = ik\hat{f},$$
$$(\partial_y f)^{\wedge} = i\xi\hat{f}.$$

#### Sobolev and Gevrey spaces

For  $1 \leq p \leq \infty$  and  $s \geq 0$  the spaces  $L^p$  and  $H^s$  corresponds to the classical Lebesgue and Sobolev spaces for functions on the set  $\mathbb{T} \times \mathbb{R}$ . Sobolev spaces satisfy the Sobolev embedding

$$\|f\|_{L^{\infty}} \lesssim \|f\|_{H^s}$$

for s > 1.

For time dependent functions we define  $L^p H^s = L^p(0,T;H^s)$  as the space with the norm

$$\|\cdot\|_{L^pH^s} = \|\|\cdot\|_{H^s}\|_{L^p(0,T)}.$$

There are different definitions of Gevrey classes and spaces. The main concept is that functions in Gevrey spaces decay exponentially in the Fourier variables on the  $L^2$  norm. In the following, we will provide two definitions, the one we use in Chapter 3 and a more general version.

Let  $1 \leq \sigma$ , then a function  $f \in L^2(\mathbb{T} \times \mathbb{R})$  belongs to the Gevrey- $\sigma$  spaces (in sense of Chapter 3) if

$$\sum_{k} \int \exp(\lambda |\xi|^{\frac{1}{\sigma}}) |\hat{f}(k,\xi)|^2 d\xi$$

for some constant  $\lambda > 0$ .

For the more general version, let  $1 \leq \sigma$ , we define the Gevrey- $\sigma$  class as the Gevrey spaces  $G_{\sigma}^{\lambda,N}$  for  $\lambda > 0$  and  $N \geq 0$  with

$$G^{\lambda,N}_{\sigma} := \{ f \in L^2 : \|f\|_{G^{\lambda,N}_{\sigma}} < \infty \}$$
(A.1)

with

$$\|f\|_{G^{\lambda,N}_{\sigma}}^{2} = \sum_{k} \int d\xi \ \langle k,\xi \rangle^{N} \exp(\lambda|k,\xi|^{\frac{1}{\sigma}}) |\hat{f}|^{2}(k,\xi).$$
(A.2)

The  $\lambda$  is called the radius of convergence.

#### **Notations**

For two real numbers  $a, b \in \mathbb{R}$ , we denote the minimum and maximum as

$$\min(a, b) = a \wedge b,$$
$$\max(a, b) = a \lor b.$$

We write  $f \leq g$  if there exists a constant C independent of all relevant parameters such that  $|f| \leq C|g|$ . Furthermore, we write  $f \approx g$  if  $f \leq g$  and  $g \leq f$ .

Moreover, for any vector or scalar v we define

$$\langle v \rangle = (1 + |v|^2)^{\frac{1}{2}}$$

 $\langle v \rangle = (1+|v|^2)^{\frac{1}{2}}.$   $\in L^2(\mathbb{T}^+,\mathbb{T}^+)$ For a function  $f \in L^2(\mathbb{T} \times \mathbb{R})$  we denote the x-average and its  $L^2$ -orthogonal complement as

$$f_{=}(y) = \int_{\mathbb{T}} f(x, y) dx,$$
  
$$f_{\neq} = f - f_{=}.$$

### Appendix B

# Derivation of the MHD Equations from the Navier-Stokes and Maxwell Equations

In this appendix, we derive the MHD equations from the Navier-Stokes and Maxwell Equations. We follow the book of Davidson [Dav16]. The MHD equations model an electrical conduction and non-magnetic fluid and are derived from the Navier-Stokes and Maxwell equations under the vanishing charge density assumption. The MHD equations are derived in three steps. First, we write down the governing equations, then we explain the simplifications and finally we derive the MHD equations.

#### **Governing Equations**

A (conduction) fluid satisfies the Navier-Stokes equation

$$\partial_t V + V \cdot \nabla V + \nabla \Pi_v = \nu \Delta V + F,$$
  
div(V) = 0, (B.1)

Here V is the fluid velocity,  $\Pi_v$  is the fluid pressure,  $\nu$  is the fluid viscosity and F is the applied forces.

The conduction fluid generates an electric and magnetic field which satisfies

Maxwell's equations

$$\nabla \cdot E = \frac{\rho}{\varepsilon},$$
  

$$\nabla \cdot B = 0,$$
  

$$\nabla \times E = -\partial_t B,$$
  

$$\nabla \times B = \mu J + \mu \varepsilon \partial_t E.$$
  
(B.2)

Here E is the electric field, B is the magnetic field, J is the current density,  $\rho$  is the charge density,  $\varepsilon$  is the permittivity of free space and  $\mu$  is the magnetic constant. In a conducting fluid, the current density is proportional to the Lorentz force

$$f = q(E + V \times B) \tag{B.3}$$

on free charges, with q as the charge.

In a stationary conductor, the current density J is proportional to the force applied to free charges, and thus, the Ohmic law  $J = \sigma \tilde{E}$  applies there. For a conducting fluid, we consider the electric field measured in a moving frame with the velocity of the conducting fluid

$$J = \sigma(E + V \times B). \tag{B.4}$$

With the electrical conductivity  $\sigma$ .

The Lorentz force (B.3) affects the moving particles, but we are more interested in the volumetric version. Thus we sum over the charge q of a unit volume of the conductor. Then the sum over charges is the density  $\sum q = \rho$  and the sum over moving charges is the current  $\sum qV = J$  and so we obtain the force

$$F = \rho E + J \times B. \tag{B.5}$$

#### Simplification for the MHD Equations

The charge density  $\rho$  in a conducting fluid is small and negligible compared to other effects. Thus for the MHD equations, we assume that

$$\rho = 0. \tag{B.6}$$

Due to the charge conservation, we obtain that the current is divergent free

$$\nabla \cdot J = -\partial_t \rho = 0.$$

The second simplification is, for the last term of (B.2) the  $\mu \varepsilon \partial_t E$  only is relevant on relativistic scales. Therefore it can be neglected in the model and we obtain

$$\nabla \times B = \mu J. \tag{B.7}$$

### Derivation of the MHD equations

Then with (B.6) and (B.7) Maxwell's equations (B.2) change to the pre Maxwell equations

$$\nabla \cdot E = 0, \tag{B.8}$$

$$\nabla \cdot B = 0, \tag{B.9}$$

$$\nabla \times E = -\partial_t B,\tag{B.10}$$

$$\nabla \times B = \mu J. \tag{B.11}$$

In the following, we establish a closed formula for the magnetic and velocity field. We apply (B.10), (B.4) and (B.11) to infer

$$\begin{split} \partial_t B &= -\nabla \times E \\ &= -\nabla \times \left( \frac{1}{\sigma} J - V \times B \right) \\ &= -\frac{1}{\sigma \mu} \nabla \times \nabla \times B + \nabla \times (V \times B). \end{split}$$

Then we define the resistivity  $\kappa:=\frac{1}{\sigma\mu}$  and use  $\nabla\cdot V=\nabla\cdot B=0$  to infer

$$\begin{aligned} -\frac{1}{\sigma\mu} \nabla \times \nabla \times B &= \kappa \Delta B, \\ \nabla \times (V \times B) &= B \cdot \nabla V - V \cdot \nabla B. \end{aligned}$$

Therefore, for B we obtain the equation

$$\partial_t B + V \cdot \nabla B = \kappa \Delta B + B \cdot \nabla V$$
  
div(B) = 0. (B.12)

From (B.5), (B.6) and (B.11) we infer

$$F = J \times B = \operatorname{curl}(B) \times B = B \cdot \nabla B - \frac{1}{2} \nabla |B|^2$$

We define the pressure  $\Pi=\Pi_v+\frac{1}{2}\nabla|B|^2$  and thus we obtain for (B.1)

$$\partial_t V + V \cdot \nabla V + \nabla \Pi = \nu \Delta V + B \cdot \nabla B,$$
  
div(V) = 0. (B.13)

Combining the equations (B.12) and (B.13) we infer the MHD equations

$$\begin{aligned} \partial_t V + V \cdot \nabla V + \nabla \Pi &= \nu \Delta V + B \cdot \nabla B, \\ \partial_t B + V \cdot \nabla B &= \kappa \Delta B + B \cdot \nabla V, \\ \operatorname{div}(B) &= \operatorname{div}(V) = 0. \end{aligned}$$

### References

- [ABSPW22] D. Adhikari, O. Ben Said, U. R. Pandey, and J. Wu. Stability and large-time behavior for the 2d Boussineq system with horizontal dissipation and vertical thermal diffusion. Nonlinear Differential Equations and Applications NoDEA, 29(4):1–43, 2022.
- [Alf42] H. Alfvén. Existence of electromagnetic-hydrodynamic waves. Nature, 150(3805):405–406, 1942.
- [BBZD23] J. Bedrossian, R. Bianchini, M. C. Zelati, and M. Dolce. Nonlinear inviscid damping and shear-buoyancy instability in the twodimensional Boussinesq equations. *Communications on Pure and Applied Mathematics*, 76(12):3685–3768, 2023.
- [Bed20] J. Bedrossian. Nonlinear echoes and Landau damping with insufficient regularity. *Tunisian Journal of Mathematics*, 3:121–205, 2020.
- [BGM17] J. Bedrossian, P. Germain, and N. Masmoudi. On the stability threshold for the 3D Couette flow in Sobolev regularity. Annals of Mathematics, pages 541–608, 2017.
- [BGM19] J. Bedrossian, P. Germain, and N. Masmoudi. Stability of the Couette flow at high Reynolds numbers in two dimensions and three dimensions. Bulletin of the American Mathematical Society, 56(3):373–414, 2019.
- [BLW20] N. Boardman, H. Lin, and J. Wu. Stabilization of a background magnetic field on a 2 dimensional magnetohydrodynamic flow. SIAM Journal on Mathematical Analysis, 52(5):5001–5035, 2020.
- [BM14] J. Bedrossian and N. Masmoudi. Asymptotic stability for the Couette flow in the 2D Euler equations. *Applied Mathematics Research* eXpress, 2014(1):157–175, 2014.
- [BM15a] J. Bedrossian and N. Masmoudi. Inviscid damping and the asymptotic stability of planar shear flows in the 2D Euler equations. *Publ. Math. Inst. Hautes Études Sci.*, 122:195–300, 2015.

- [BM15b] J. Bedrossian and N. Masmoudi. Inviscid damping and the asymptotic stability of planar shear flows in the 2d Euler equations. *Publications mathématiques de l'IHÉS*, 122(1):195–300, 2015.
- [BMM16] J. Bedrossian, N. Masmoudi, and C. Mouhot. Landau damping: paraproducts and Gevrey regularity. *Annals of PDE*, 2(1):4, 2016.
- [BMV16] J. Bedrossian, N. Masmoudi, and V. Vicol. Enhanced dissipation and inviscid damping in the inviscid limit of the Navier-Stokes equations near the 2D Couette flow. Archive for Rational Mechanics and Analysis, 219:1087–1159, 2016.
- [BSS88] C. Bardos, C. Sulem, and P.-L. Sulem. Longtime dynamics of a conductive fluid in the presence of a strong magnetic field. *Transactions of the American Mathematical Society*, 305(1):175–191, 1988.
- [BVW18] J. Bedrossian, V. Vicol, and F. Wang. The Sobolev stability threshold for 2D shear flows near Couette. Journal of Nonlinear Science, 28(6):2051–2075, 2018.
- [BZD20] R. Bianchini, M. C. Zelati, and M. Dolce. Linear inviscid damping for shear flows near Couette in the 2D stably stratified regime. arXiv preprint arXiv:2005.09058, 2020.
- [CF23] D. Cobb and F. Fanelli. Elsässer formulation of the ideal MHD and improved lifespan in two space dimensions. Journal de Mathématiques Pures et Appliquées, 169:189–236, 2023.
- [CM91] X. L. Chen and P. J. Morrison. A sufficient condition for the ideal instability of shear flow with parallel magnetic field. *Physics of Fluids B: Plasma Physics*, 3(4):863–865, 1991.
- [CMRR16] J.-Y. Chemin, D. S. McCormick, J. C. Robinson, and J. L. Rodrigo. Local existence for the non-resistive MHD equations in Besov spaces. Advances in Mathematics, 286:1-31, 2016. URL https://www.sciencedirect.com/science/article/pii/ S0001870815003424.
- [CRW13] C. Cao, D. Regmi, and J. Wu. The 2D MHD equations with horizontal dissipation and horizontal magnetic diffusion. *Journal* of Differential Equations, 254(7):2661–2681, 2013.
- [CW13] C. Cao and J. Wu. Global regularity for the two-dimensional anisotropic Boussinesq equations with vertical dissipation. Archive for Rational Mechanics and Analysis, 208(3):985–1004, 2013.
- [CZ24] T. Chen and R. Zi. On the Sobolev stability threshold for shear flows near Couette in 2D MHD equations. Proceedings of the Royal Society of Edinburgh Section A: Mathematics, pages 1–51, 2024.

- [Dav16] P. A. Davidson. Introduction to Magnetohydrodynamics. Cambridge Texts in Applied Mathematics. Cambridge University Press, 2 edition, 2016.
- [DM23] Y. Deng and N. Masmoudi. Long-time instability of the Couette flow in low Gevrey spaces. Communications on Pure and Applied Mathematics, 2023.
- [Dol24] M. Dolce. Stability threshold of the 2D Couette flow in a homogeneous magnetic field using symmetric variables. Communications in Mathematical Physics, 405(4):94, 2024.
- [DWZ21] W. Deng, J. Wu, and P. Zhang. Stability of Couette flow for 2d Boussinesq system with vertical dissipation. Journal of Functional Analysis, 281(12):109255, 2021.
- [DWZZ18] C. R. Doering, J. Wu, K. Zhao, and X. Zheng. Long time behavior of the two-dimensional Boussinesq equations without buoyancy diffusion. *Physica D: Nonlinear Phenomena*, 376:144–159, 2018.
- [DYZ19] Y. Du, W. Yang, and Y. Zhou. On the exponential stability of a stratified flow to the 2d ideal MHD equations with damping. SIAM Journal on Mathematical Analysis, 51(6):5077–5102, 2019.
- [DZ21] Y. Deng and C. Zillinger. Echo chains as a linear mechanism: Norm inflation, modified exponents and asymptotics. Archive for rational mechanics and analysis, 242(1):643–700, 2021.
- [EW15] T. M. Elgindi and K. Widmayer. Sharp decay estimates for an anisotropic linear semigroup and applications to the surface quasigeostrophic and inviscid Boussinesq systems. SIAM Journal on Mathematical Analysis, 47(6):4672–4684, 2015.
- [FL19] E. Feireisl and Y. Li. On global-in-time weak solutions to the magnetohydrodynamic system of compressible inviscid fluids. Nonlinearity, 33(1):139, 2019.
- [FMRR17] C. L. Fefferman, D. S. McCormick, J. C. Robinson, and J. L. Rodrigo. Local existence for the non-resistive MHD equations in nearly optimal Sobolev spaces. Archive for Rational Mechanics and Analysis, 223:677–691, 2017.
- [Gra14] L. Grafakos. Classical Fourier Analysis. Springer-Verlag, New York, 2014.
- [HHKL18] Z. Hussain, S. Hussain, T. Kong, and Z. Liu. Instability of MHD Couette flow of an electrically conducting fluid. AIP Advances, 8(10), 2018.

- [HT01] D. Hughes and S. Tobias. On the instability of magnetohydrodynamic shear flows. Proceedings of the Royal Society of London. Series A: Mathematical, Physical and Engineering Sciences, 457(2010):1365–1384, 2001.
- [HTY05] M. Hirota, T. Tatsuno, and Z. Yoshida. Resonance between continuous spectra: Secular behavior of Alfvén waves in a flowing plasma. *Physics of plasmas*, 12(1), 2005.
- [HXY18] L.-B. He, L. Xu, and P. Yu. On global dynamics of three dimensional magnetohydrodynamics: nonlinear stability of Alfvén waves. Annals of PDE, 4(1):1–105, 2018.
- [IJ13] A. D. Ionescu and H. Jia. Nonlinear inviscid damping near monotonic shear flows. Acta Mathematica, 230(2):321 — 399, 2013.
- [JLWY19] R. Ji, H. Lin, J. Wu, and L. Yan. Stability for a system of the 2D magnetohydrodynamic equations with partial dissipation. Applied Mathematics Letters, 94:244–249, 2019.
- [JW22] F. Jiang and Y. Wang. Nonlinear stability of the inviscid magnetic bénard problem. Journal of Mathematical Fluid Mechanics, 24(4):1–18, 2022.
- [K] N. Knobel. Sobolev stability for the 2D MHD equations in the non-resistive limit. arXiv preprint arXiv:2401.12548, 2024.
- [Koz89] H. Kozono. Weak and classical solutions of the two-dimensional magnetohydrodynamic equations. Tohoku Mathematical Journal, Second Series, 41(3):471–488, 1989.
- [KZ1] N. Knobel and C. Zillinger. On echoes in magnetohydrodynamics with magnetic dissipation. Journal of Differential Equations, 367:625–688, 2023.
- [KZ2] N. Knobel and C. Zillinger. On the Sobolev stability threshold for the 2D MHD equations with horizontal magnetic dissipation. arXiv preprint arXiv:2309.00496, 2023.
- [LCZL18] Y. Liu, Z. Chen, H. Zhang, and Z. Lin. Physical effects of magnetic fields on the kelvin-helmholtz instability in a free shear layer. *Physics of Fluids*, 30(4):044102, 2018.
- [Lis20] K. Liss. On the Sobolev stability threshold of 3D Couette flow in a uniform magnetic field. Communications in Mathematical Physics, pages 1–50, 2020.
- [LL11] Y. C. Li and Z. Lin. A resolution of the Sommerfeld paradox. SIAM Journal on Mathematical Analysis, 43(4):1923–1954, 2011.

- [LMZ22] H. Li, N. Masmoudi, and W. Zhao. Asymptotic stability of two-dimensional Couette flow in a viscous fluid. arXiv preprint arXiv:2208.14898, 2022.
- [LWX<sup>+</sup>21] S. Lai, J. Wu, X. Xu, J. Zhang, and Y. Zhong. Optimal decay estimates for 2D Boussinesq equations with partial dissipation. *Jour*nal of Nonlinear Science, 31(1):1–33, 2021.
- [LXZ15] F. Lin, L. Xu, and P. Zhang. Global small solutions of 2-D incompressible MHD system. Journal of Differential Equations, 259(10):5440–5485, 2015.
- [MB01] A. J. Majda and A. L. Bertozzi. Vorticity and incompressible flow, volume 27. Cambridge Univ Pr, 2001.
- [MSHZ22] N. Masmoudi, B. Said-Houari, and W. Zhao. Stability of Couette flow for 2D Boussinesq system without thermal diffusivity. Archive for Rational Mechanics and Analysis, 245(2):645–752, 2022.
- [MV11] C. Mouhot and C. Villani. On Landau damping. Acta mathematica, 207(1):29–201, 2011.
- [MWGO68] J. H. Malmberg, C. B. Wharton, R. W. Gould, and T. M. O'Neil. Plasma wave echo experiment. *Physical Review Letters*, 20(3):95– 97, 1968.
- [MY06] C. Miao and B. Yuan. Well-posedness of the ideal MHD system in critical Besov spaces. Methods and Applications of Analysis, 13(1):89–106, 2006.
- [MZ22] N. Masmoudi and W. Zhao. Stability threshold of two-dimensional Couette flow in Sobolev spaces. Ann. Inst. H. Poincaré C Anal. Non Linéaire, 39(2):245–325, 2022.
- [MZZ23] N. Masmoudi, C. Zhai, and W. Zhao. Asymptotic stability for two-dimensional Boussinesq systems around the Couette flow in a finite channel. *Journal of Functional Analysis*, 284(1):109736, 2023.
- [Orr07] W. M. Orr. The stability or instability of the steady motions of a perfect liquid and of a viscous liquid. In Proceedings of the Royal Irish Academy. Section A: Mathematical and Physical Sciences, pages 69–138. JSTOR, 1907.
- [RWXZ14] X. Ren, J. Wu, Z. Xiang, and Z. Zhang. Global existence and decay of smooth solution for the 2-D MHD equations without magnetic diffusion. Journal of Functional Analysis, 267(2):503–541, 2014.
- [Sch88] P. G. Schmidt. On a magnetohydrodynamic problem of Euler type. Journal of differential equations, 74(2):318–335, 1988.

- [Sec93] P. Secchi. On the equations of ideal incompressible magnetohydrodynamics. Rendiconti del Seminario Matematico della Universita di Padova, 90:103–119, 1993.
- [ST83] M. Sermange and R. Temam. Some mathematical questions related to the MHD equations. HAL open science, RR-0185, 1983.
- [TWZZ20] L. Tao, J. Wu, K. Zhao, and X. Zheng. Stability near hydrostatic equilibrium to the 2D Boussinesq equations without thermal diffusion. Archive for Rational Mechanics and Analysis, 237(2):585– 630, 2020.
- [WZ17] D. Wei and Z. Zhang. Global well-posedness of the MHD equations in a homogeneous magnetic field. Analysis & PDE, 10(6):1361– 1406, 2017.
- [WZ21] J. Wu and Y. Zhu. Global solutions of 3D incompressible MHD system with mixed partial dissipation and magnetic diffusion near an equilibrium. Advances in Mathematics, 377:107466, 2021.
- [YOD05] J. H. Yu, T. M. O'Neil, and C. F. Driscoll. Fluid echoes in a pure electron plasma. *Physical review letters*, 94(2):025005, 2005.
- [Zil17] C. Zillinger. Linear inviscid damping for monotone shear flows. Trans. Amer. Math. Soc., 369(12):8799–8855, 2017.
- [Zil21a] C. Zillinger. On echo chains in Landau damping: Traveling wavelike solutions and Gevrey 3 as a linear stability threshold. Annals of PDE, 7(1):1–29, 2021.
- [Zil21b] C. Zillinger. On the Boussinesq equations with non-monotone temperature profiles. *Journal of Nonlinear Science*, 31(4):64, 2021.
- [Zil23] C. Zillinger. On echo chains in the linearized Boussinesq equations around traveling waves. SIAM Journal on Mathematical Analysis, 55(5):5127–5188, 2023.
- [ZZ24] W. Zhao and R. Zi. Asymptotic stability of Couette flow in a strong uniform magnetic field for the Euler-MHD system. Archive for Rational Mechanics and Analysis, 248(3):47, 2024.
- [ZZZ21] C. Zhai, Z. Zhang, and W. Zhao. Long-time behavior of Alfvén waves in a flowing plasma: Generation of the magnetic island. Archive for Rational Mechanics and Analysis, 242:1317–1394, 2021.